Port capacity investments under uncertainty: The use of real options models

Matteo Balliauw

Cover picture: Jasper Léonard Cover design and printing: Universitas, Antwerp

ISBN: 978-90-5728-638-4 Depot number: D/2019/12.293/24 Copyright © 2020 Matteo Balliauw

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or by an information storage and retrieval system without permission in writing from the author.

Members of the doctoral jury

Supervisors:	Prof. dr. Eddy Van de Voorde, Unversity of Antwerp
	Prof. dr. Hilde Meersman, University of Antwerp
Chair:	Prof. dr. Thierry Vanelslander, University of Antwerp
Examiners:	Prof. dr. Seraphim Kapros, University of the Aegean, Chios
	Prof. dr. Siri Pettersen Strandenes, NHH, Bergen
	Prof. dr. Peter Kort, University of Tilburg
	Prof. dr. Danny Cassimon, University of Antwerp

This research is funded by a PhD grant of the Research Foundation Flanders (FWO).

"The best way to predict the future is to create it." - Abraham Lincoln, 16th President of the United States

Word of thanks

First of all, I would like to thank my promoters Eddy Van de Voorde and Hilde Meersman, to give me the opportunity to write this PhD thesis. Thank you for the advice given to me, the challenging meetings and making me think beyond the models as well. Also, thank you for allowing me to alternate maritime and ports research with air transportation research, sports research and teaching activities. Not focusing on just one topic gave me valuable insights for each of my activities and a lot of energy to continue in each of them. Also, thank you other members of my commission: Thierry Vanelslander, for the useful meetings with suggestions and detailed feedback (even about commas!), and Siri Strandenes for your committed follow-up on my progress. I also would like to thank the other members of my jury, Peter Kort, Danny Cassimon and Seraphim Kapros, to take the time to evaluate this thesis and challenge me with insightful questions and suggestions to improve this work.

Subsequently, I want to thank everybody who co-authored with me for the nice output and the opportunities to learn. Next to Eddy, Hilde and Thierry, also, thank you Ann Verhetsel, for the first article we wrote together, as well as my Master's thesis promoters Dorien Herremans, Daniel Palhazi Cuervo and Kenneth Sörensen for our paper on piano fingering. A special word of thanks in this light also goes to Peter Kort, for introducing me to and unconditionally supporting me in real options modelling. Also, thank you Christophe Smet, for your good advice and the introduction to and joint exploration of the beautiful world of advanced mathematics. Thank you as well Anming Zhang, for inviting me to the Sauder School of Business at UBC in Vancouver, the fruitful collaboration and the scientific insights I learned from you. Moreover, thank you Trevor Heaver and Michele Acciaro, for showing me around and for offering me nice other experiences in my spare time in Vancouver.

Of course, I also want to thank the other university colleagues who made me have five nice years at the University of Antwerp and from whom I could learn. In this light, a special word of thanks goes to you, Thomas Verlinden, for all the nice moments of *gezever*, which eventually always brought us to good ideas and even some nice papers, colloquia and football games! Thank you for your unconditional support, your time to listen to me and to talk, and for the friend you are. Also, thank you Tom Vermeire, for the intellectual (and also pleasant) time together, which resulted in some nice footprints we could leave! Thank you also Christophe and Evy for proofreading my work, and Toon Reyns for printing. E natural-

mente, grazie Luca Talarico, per portare un po' d'Italia nella mia vita da PhD student.

After having finished this PhD trajectory, I decided to move into football, where I met new colleagues at the Royal Belgian Football Association. I want to thank them, and especially Peter Bossaert, Manu Leroy, Bart Cobbaert, Michaël Schockaert, Stephan Desmit, Amaia Lasa and Koen Landsheere for giving me the opportunity to continue working in the domain of my passion, but also for their great reception and support in the last bits of my PhD process. Also thank you everybody in sports whom I have met, some through the students who made nice theses with me, for making me learn a lot about this industry and for enabling me in this way to take this next step in my career. It is impossible to mention everybody, because I would definitely forget some people. But Chris Van Puyvelde, Tomas Van Den Spiegel, Dirk Geuens, Jesse De Preter, Gérard Linard, Jean Kindermans, Mila Djelosevic, Ariël Jacobs, Leander Monbaliu, David Pauwels, Laurenz Van Looveren and Stijn Francis, you deserve a special word of thanks for your help along the way.

A special word of thanks is also in place for my parents, who always support me in any decision I take. Thank you for supporting me to study Business Engineering at the University of Antwerp, where I could make some nice friends for life, such as Joris Van Geet, with whom I spent a beautiful time during nice trips. Thank you as well my friends Maxim Verstraeten and Cedric De Boom, for letting me learn from the way to a PhD in other domains.

I also want to thank Rita, as well as my parents, to take care of our children when Evy and I were too busy working. Thank you Flore and Lieze, the girls of Evy, for always letting me explain my work in easy terms, but even more for the cool moments in our spare time, with games, trips, sports and much more. Thank you, also, dear parents, grandparents and great grandparents for taking care of me since I was a kid, as well as my (former) (music) teachers for showing me the values of life, educating me and making me who I am.

Of course, a final paragraph is for you, my dear dr. Evy. It could have been a section, a chapter, a book or even a library, but I will try to keep it as concise as possible. Thank you for being the beautiful person you are. You are the guide in my life. First, you were at the university, where you taught me, where we made nice work together, where you also proofread my work, but where we especially had LOTS OF FUN! Thank you for your beautiful smiles. And then I was so lucky that you also wanted to share your life and love with me, making it the best life ever possible. But the most grateful I am for giving me the nicest gift on earth, our beautiful little son, Lucas, who makes us smile and love him all the time. Thank you for being the best mum for him, and for enabling me to be a good dad. And thank you little Lucas, for making me smile, with your funny expressions and reactions, as well as your unconditional love! I am looking forward to having many more happy moments together, where we can enjoy ourselves, for example with a car or in a sports stadium!

Matteo Balliauw, Sint-Niklaas, 24 Februari 2020.

Abstract

Port operations are important for worldwide and regional trade and for the development of regions. Investments in port capacity are required to perform these activities. Port infrastructure and some superstructure investments involve large sums of money, are irreversible and involve a lot of uncertainty. In the literature, real options (RO) have been identified as a methodology to improve investment decisions with a flexible size and timing under uncertainty. Elements of existing RO models from other sectors, suited for port capacity investment analyses, are combined in a framework, together with specific port-economic characteristics. Based on this, new RO port models are constructed to meet this thesis' objective, which is to study how optimal port capacity investment decisions are influenced by different port- and project-related economic characteristics under uncertainty.

In the developed models, throughput level, timing and size of the investment are flexible. As an addition to the literature, the users' congestion costs are added to the RO models for port capacity investments. Next to a base case benchmark model, a second model adds the possibility of a partially or fully publicly owned port authority (PA), as well as the division of cash flows and activities in a landlord port model between the two investing actors: the PA and the port operator. A third model adds inter-port competition to the base case model: two new ports, competing in quantities, are constructed according to a Stackelberg leader-follower model. A final model considers port expansion of one service port, as well as the construction lead time.

If port customers are on average more waiting-time averse, new port investment projects need to be developed later, and their size needs to be larger as well. In the case of port expansion, such an investment should be made earlier, whereas the impact on size is limited. Uncertainty leads to later and larger investments. Increased public ownership leads to earlier and larger investments in new ports, whereas expansions projects are even more anticipated if the public share is larger. In landlord ports, the two investing actors can agree to follow the investment strategy that would be optimal under a service port configuration. Otherwise, the PA can use the concession fee to force the terminal operator to invest in the PA's optimum. Inter-port competition reduces the option value of waiting. This leads to earlier and smaller leader investment, compared with its unrestricted strategy. The follower however will invest later and more.

Dutch executive summary

De activiteiten die in een haven plaatsvinden, spelen een belangrijke rol om de wereldhandel en de regionale groei te ondersteunen. Om deze activiteiten uit te voeren, moet een haven investeren. Elementen zoals maritieme en hinterlandtoegang, infrastructuur en superstructuur moeten allemaal aanwezig zijn om de goederen te kunnen overslaan tussen maritiem en achterlandvervoer. De hoeveelheid goederen die in een bepaalde tijdsperiode behandeld kunnen worden, bepaalt de capaciteit van een haven. In deze thesis wordt er gefocust op containerhavens, en meer bepaald op de investeringen in containerinfrastructuur en delen van de superstructuur. Daarom wordt capaciteit in deze thesis uitgedrukt in (miljoenen) TEU per jaar.

De beschouwde investeringen in havencapaciteit zijn grootschalig, duren lang om te bouwen, kosten enorm veel geld en zijn onderhevig aan heel wat onzekerheden. Daartegenover staat dat de beslissingsnemers een aantal flexibele opties hebben om de investeringsbeslissing aan te passen, zoals het vrij beslissen over de timing en de grootte van de investering, een flexibel outputniveau en gefaseerde investeringen. De literatuur heeft uitgewezen dat de reële optie (RO) benadering zulke beslissingen verder kan optimaliseren onder onzekerheid. RO zijn daarom een ideale uitbreiding van de huidige benadering voor het berekenen van kosten en baten. Aangezien deze RO methode nog niet uitgebreid toegepast wordt binnen haveninvesteringsbeslissingen, worden in deze thesis RO modellen uitgewerkt voor het beslissen omtrent haveninvesteringen. Het doel is om hiermee te bestuderen hoe optimale investeringen in havencapaciteit onder onzekerheid beïnvloed worden door project- en havengerelateerde economische omstandigheden.

Het tweede hoofdstuk start met een overzicht van de verschillende bronnen van onzekerheid in een haven. Deze omvatten de interacties van actoren in de maritieme logistieke keten, de onzekere evolutie van de wereldhandel en de technologische veranderingen als gevolg van verschuivingen in ecologische en politieke prioriteiten. Vervolgens worden voorbeelden van RO toepassingen binnen de transportsector besproken. Ook niettransportgerelateerde RO modellen met relevante elementen voor investeringen in havencapaciteit worden besproken en gebundeld in een raamwerk. Dit raamwerk bestaat uit verschillende elementen die in een RO model kunnen voorkomen, alsook specifieke haveneconomische elementen, zoals het type vraagfunctie, kostenfunctie, congestiekosten, constructietijd, gefaseerd investeren, flexibel outputniveau, havenconcurrentie, investeringskosten, andere doelstellingen van het havenmanagement en de te optimaliseren variabelen in de investeringsbeslissing. Het raamwerk kan gebruikt worden om bestaande RO modellen te beoordelen en te plaatsen in de literatuur, alsook om nieuwe modellen mee op te stellen. De RO modellen zijn continue tijd en continue status modellen, in lijn met de theorie van Dixit & Pindyck (1994). Zij maken gebruik van stochastische calculus (het lemma van Itō en de Bellman vergelijking) om de investeringsbeslissing onder onzekerheid te optimaliseren.

Het eerste uitgewerkte model in deze thesis is een basismodel waarmee de volgende nieuwe modellen vergeleken kunnen worden. Deze benchmark behandelt de flexibele investering in de infrastructuur en superstructuur voor een nieuwe private haven, die door de eigenaar ook wordt uitgebaat naar het service havenmodel en waarbij havenconcurrentie buiten beschouwing gelaten wordt. In dit model zijn output, timing en grootte van de investering flexibel. De grootste toevoeging aan de literatuur bestaat erin dat in dit model de congestiekosten meegenomen zijn in een continue tijd en status model op basis van een geometrische Brownse beweging. Deze congestiekosten zijn een onontbeerlijk element in haveneconomie. De resultaten tonen aan dat investeringsprojecten in nieuwe havens voor havengebruikers die meer avers zijn voor wachttijd, beter later gebouwd worden en aldus groter in omvang zijn. Hetzelfde geldt bij stijgende onzekerheid. In groeimarkten leidt langer wachten tot de mogelijkheid om meer inkomsten te realiseren. Deze dienen ook als buffer tegen mogelijke negatieve schommelingen ten gevolge van deze onzekerheid. Grotere markten vereisen over het algemeen ook grotere investeringen, om aan de vraag te kunnen voldoen. Andere veranderingen die havenprojecten interessanter maken, zoals lagere kosten, hogere groeivoeten of lagere discontovoeten, leiden ook tot grotere of eerdere investeringen, of de combinatie van beide.

In Hoofdstuk 4 wordt het eerste model uitgebreid met twee elementen. Enerzijds is er de mogelijkheid dat publieke instanties de haven gedeeltelijk of volledig in eigendom hebben. Anderzijds wordt er een opdeling gemaakt tussen de haveneigenaar en de havenuitbater, naar het landlord havenmodel. De resultaten tonen aan dat de nieuwe haven sneller en groter gebouwd moet worden naarmate het publieke aandeelhouderschap toeneemt. Bovendien kunnen de havenuitbater en de haveneigenaar negotiëren om samen te investeren in een gelijke capaciteit. Zij kunnen ervoor kiezen om de optimale investeringsstrategie van een equivalente service haven te volgen. Door middel van de hoogte van de concessievergoeding kan de havenautoriteit de terminaluitbater ook dwingen om de optimale strategie van de havenautoriteit te volgen.

Hoofdstuk 5 breidt het basismodel uit met concurrentie tussen twee havenbesturen die kunnen investeren in een nieuwe haven volgens een Stackelberg leider-volger model. Na de investeringen vindt er Cournot hoeveelheidsconcurrentie plaats tussen de havens. De haven met investeringskostenvoordelen zal als eerste investeren. Afhankelijk van de grootte van de investering van deze leider en de grootte van de markt op dat moment zal de andere havenautoriteit, de volger, wachten om toe te treden met een investering, of meteen investeren. Wanneer de kostenverschillen tussen de havens klein zijn, zal de leider eerder dan het eigen optimum moeten investeren. Hoe groter de concurrentie, bijvoorbeeld door een nabijere ligging, hoe eerder deze leider dient te investeren en hoe kleiner de investering bijgevolg zal zijn. Concurrentie holt de optiewaarde van wachten namelijk uit. Het gevolg is wel dat de volger later zal investeren en in meer capaciteit.

Het laatste model vertrekt opnieuw vanuit het basismodel, maar houdt rekening met de constructietijd en de reeds aanwezige capaciteit voor het bestuderen van havenuitbreidingen. De constructietijd verhoogt enerzijds onzekerheid, omdat er tussen de beslissing en het effectieve moment van ingebruikname schommelingen in de vraag kunnen plaatsvinden. Bovendien is dit ook een periode waarin nog geen inkomsten gegenereerd worden uit het nieuwe project, wat de attractiviteit hiervan negatief beïnvloedt. De impact van de constructietijd op de optimale investeringsbeslissing is echter niet eenduidig. Het is daarentegen wel duidelijk dat de expansiebeslissing niet identiek is aan de investering in een nieuwe haven. Havenuitbreidingen voor klanten die een grotere aversie hebben voor wachttijden, zullen in dit geval niet later maar eerder plaatsvinden, omdat er een grotere incentive is om de reeds bestaande congestie te verminderen. De impact op de grootte is echter beperkt. Ook een toename van publieke eigendom zal de havenuitbreiding relatief gezien veel vroeger doen plaatsvinden.

Hoewel de theoretische modellen belangrijke inzichten verschaffen in de invloed van verschillende factoren op de optimale investeringsbeslissing, zijn RO modellen complex om toe te passen in de realiteit. Bovendien is de kost voor het schatten van parameters of het maken van kleine aanpassingen in het model om de realiteit beter weer te geven, vaak zeer hoog. De reden is dat hiervoor vaak externe econometristen ingehuurd moeten worden. Omdat de potentiële baten van het toepassen van RO mogelijk nog hoger zijn, wordt in deze thesis een blauwdruk van een algoritme uitgewerkt om RO in een vereenvoudigde versie toe te passen in havens. Uiteraard dienen de gemaakte veronderstellingen steeds goed voor ogen gehouden te worden. Verder onderzoek kan op basis van de ontwikkelde modellen en inzichten in deze thesis leiden tot nieuwe modellen om de assumpties realistischer en de modellen toepasbaarder te maken. Een voorbeeld hiervan is een model met gefaseerd investeren, waarbij expansie, constructietijd, congestiekosten en concurrentie nog steeds worden meegenomen.

English executive summary

Port activities are important to support world trade and regional development. To perform these activities, ports need to invest. Elements such as maritime and hinterland access, infrastructure and superstructure all need to be present to transfer goods between maritime and inland transportation. The quantity of goods that can be handled in a given time period determines a port's capacity. This thesis focuses on container ports, and more specifically investments in container infrastructure and parts of the superstructure. Therefore, capacity is expressed in (million) TEU per year.

The considered port capacity investments involve large sums of money, long construction lead times, are large-scaled and are subject to a lot of uncertainty. Decision makers however have a number of flexible options to adapt their decisions, such as flexible investment timing and size, flexible throughput levels over time and phased investments. Literature showed that a real options (RO) approach can further optimise such decisions under uncertainty. Hence, this approach is an ideal extension for cost-benefit analyses. Since the RO methodology has not been widely used for port investment decisions, new port RO models are developed in this thesis. The objective is to use them to study how optimal capacity investment decisions in ports are influenced by different port- and project-related economic characteristics under uncertainty.

The second chapter starts with an overview of different sources of uncertainty in a port. These include interactions between actors in the maritime logistics chain, the uncertain evolution of world trade and technological changes due to changing ecological and political priorities. Subsequently, examples of RO applications in the transport sector are discussed. Also non-transportation RO models with relevant elements for port capacity investments are discussed and combined in a framework. This framework consists of different potential RO models as well as specific port-economic elements, such as the type of demand function, cost function, congestion costs, time to build, phased investment, flexible throughput level, port competition, investment costs, other port management objectives and the decision variables that are to be optimised in the investment decision. The framework can be used to discuss existing RO models in the literature, as well as to develop new models. The RO models in this thesis are continuous-time continuous-state models, in line with the theory of Dixit & Pindyck (1994). They use stochastic calculus (Itô's Lemma and Bellman

equation) to optimise investment decisions under uncertainty.

The first model developed in this thesis is a base case to which subsequent, new models can be compared. This benchmark deals with a flexible investment in infrastructure and superstructure for a new private port operated by the owner, i.e., a service port. Port competition is left beyond this model's scope. Throughput, timing and size of the investment are flexible. This model's main addition to the literature is the inclusion of congestion costs in a continuous-time continuous-state RO model involving a geometric Brownian motion. These congestion costs are an indispensable element in port economics. The results show that investment projects in new ports for more waiting-time-averse customers are to be larger and built later. The same holds for increased uncertainty. In growth markets, waiting longer to invest yields larger revenue opportunities, which serve as a buffer against potential negative deviations due to the uncertainty. Larger markets in general also require larger investments, to accommodate the demand. Other changes in the economic environment making the port project more attractive, such as lower costs, higher growth rates and lower discount rates, also result in larger or sooner investments, or a combination of both.

In Chapter 4, the first model is extended with two elements. On the one hand, public entities can fully or partially own the port. On the other hand, the distinction between port owner and port operator is made, i.e., a landlord port. The results here show that an increase in public ownership optimally leads to sooner and larger investments. Moreover, the port operator and owner can negotiate to invest at the same time in the same amount of capacity. They can choose to implement the optimal investment decision of an equivalent service port. Through the size of the concession fee, the port authority can also force the terminal operator to follow the optimal investment strategy of the port authority.

Chapter 5 extends the base case model with port competition between two port authorities that can invest in a new port according to the Stackelberg leader follower model. After investment, Cournot quantity competition takes place between the ports. The costadvantaged port will invest first. Depending on the investment size of the leader and the size of the market at that time, the other port authority, i.e., the follower, will wait to enter or invest right away. These situations are respectively known as entry deterrence or entry accommodation. When cost differences between the ports are small, preemption will force the leader to invest before the own unrestricted, optimal timing. The larger the port competition, e.g., due to the ports being situated closer to one another, the sooner the leader will invest, and the smaller the investment will be. The reason is that competition drives the option value of waiting down. As a consequence, the follower will invest later and in more capacity.

The final model adds time to build and the already existing capacity to the first model. Hence, it involves port expansions. Construction lead time increases uncertainty, as the market can change in the time between the investment decision and the generation of the first revenues. Moreover, revenues are foregone during this lead time, which reduces the project's attractiveness. The impact of time to build on the investment decision however is ambiguous. Oppositely, the results do show that optimal port expansion decisions differ from the decision to build a new port. Expansions for customers who are more waiting-time averse will not take place later, but earlier, due to the larger incentive to reduce the already existing congestion. The impact on size is however limited. Also an increase in public port ownership will lead to a much earlier investment in the case of expansion decisions.

Notwithstanding the ability of theoretical models to provide important insights into the impact of different factors on the optimal investment decisions, the complexity of RO models reduces its applicability in reality. Moreover, the cost of estimating parameters or changing model elements to increase realism can become very high, because it often involves hiring external econometricians. Since the potential gains of applying RO can be even higher, a blueprint of an algorithm to implement a simplified RO model in ports is developed in this thesis. Of course, the underlying assumptions always need to be taken into account. Further research can build on the models and insights developed in this thesis to develop new models that further relax the underlying assumptions, to increase applicability even more. As an example, a model including phased investment would be an interesting extension of this research.

Contents

Ał	ostrac	ct	iii
Dı	utch	executive summary	v
Er	nglish	n executive summary	ix
Li	st of f	figures	xix
Li	st of t	tables	xxi
Li	st of a	abbreviations x	αv
1	Intr	oduction	1
	1.1	A background on ports, port actors and port organisation	1
	1.2	The economic importance of ports	4
	1.3	Investing in ports is indispensable	6
	1.4	Throughput capacity of a port	9
		1.4.1 The determinants of capacity	10
		1.4.2 Port actors influence capacity as well	11
		1.4.3 The capacity investment decision making process in the port	12
	1.5	The cost of port investments	14

	1.6	Chara	cteristics of port capacity investments	15
	1.7	Resea	rch objective and scope	16
	1.8	Resea decisi	rch approach: advantages and limitations of investment analysis and on making with real options models	18
	1.9	Struct	ure of the thesis and research sub-questions	20
2	Tow	ards in	nproved port capacity investment decisions under uncertainty	27
	2.1	Source	es of uncertainty and risk in the port context	28
		2.1.1	Competitive position of the port: global logistics chains	30
		2.1.2	Global evolutions in international trade	32
		2.1.3	Technological change: environmental and legal uncertainties	32
	2.2	Real o	ptions models for better decisions under uncertainty	33
	2.3	Real o	ptions in the transportation sector	34
	2.4	Real o	ptions models from a port perspective	36
		2.4.1	The basic real options model of Dangl and its extensions	37
		2.4.2	Inclusion of port expansion in the model	39
		2.4.3	The impact of time to build	40
		2.4.4	Installing the project in stages	41
		2.4.5	Game-theoretic approaches	42
		2.4.6	Alternative uncertainty processes to model cyclicality and changes in legislation	44
	2.5	Towar	ds a real options application in ports	45
		2.5.1	A framework to enhance port capacity investment decisions	45
		2.5.2	Applying the framework to improve port capacity investment decisions	47

	2.6	Conclusion of this chapter	50
3	The mer	impact of congestion and uncertainty on a new private port's capacity invest- nt	53
	3.1	Economic setting and methodology	55
	3.2	Parameter calibration of a hypothetical numerical example	64
	3.3	Value of the investment project	65
	3.4	Investment decision: optimal size and timing	72
	3.5	Influence of the economic and operational context on the optimal invest- ment strategy of the port: sensitivity analysis	78
	3.6	Conclusion of this chapter	82
4	Port	t capacity investments with two actors and public money involved	85
	4.1	Economic setting and methodology	86
		4.1.1 Economic assumptions	87
		4.1.2 Differentiating between TOC and PA	88
		4.1.3 Allowing for both public and private ownership of the PA	93
		4.1.4 Summary of the model adaptations and assumptions	94
	4.2	Calibration of the parameters related to public ownership and the distribu- tion of income and costs	95
	4.3	Solving for the optimal investment strategy in a landlord port	99
	4.4	Results and discussion	101
		4.4.1 Division between PA and TOC	102
		4.4.2 Government involvement	107
		4.4.3 Landlord port with public ownership	108
	4.5	Investment decision sensitivity to an altered economic situation	110

	4.6	Conclusion of this chapter 1	112
5	The	capacity investment decisions of two competing ports 1	15
	5.1	Economic setting and methodology	117
		5.1.1 Individual port situations 1	118
		5.1.2 Game-theoretic modelling of competition	121
	5.2	Optimal investment strategies	125
		5.2.1 A port's throughput decision	127
		5.2.2 The follower's investment decision 1	127
		5.2.3 The leader's investment decision 1	129
		5.2.4 A synthesis of the leader's and follower's strategies 1	133
	5.3	The impact of competition and port ownership on the investment decisions . 1	134
	5.4	The impact of other parameters on the investment decision: sensitivity analysis	139
	5.5	Conclusion of this chapter 1	142
6	The	impact of time to build and uncertainty on port expansion projects 1	.45
	6.1	Recapitulation of the basic model for determining the project value 1	147
	6.2	Economic setting and methodology: model extensions 1	148
		6.2.1 Accounting for existing capacity in the value of expansion projects 1	148
		6.2.2 Accounting for the impact of time to build on the project value 1	148
	6.3	Determining the optimal project investment size and timing 1	149
	6.4	Setting the additional model parameters 1	150
	6.5	Results and discussion 1	150
		6.5.1 The impact of time to build	150

		6.5.2	The impact of existing port capacity	157
	6.6	Sensit	ivity analyses	158
		6.6.1	Sensitivity of the results to congestion costs and waiting-time aversion	ı 158
		6.6.2	Sensitivity of the results to public ownership	159
		6.6.3	Sensitivity of the results to growth and uncertainty	159
	6.7	Concl	usion of this chapter	161
7	Mov quir	ing fro ed step	om theoretical real options models to decision making in practice: re os, opportunities and threats	- 163
	7.1	A real	options tool blueprint for port capacity investment decisions	165
		7.1.1	Input required to calculate real options model parameters	165
		7.1.2	An application of the investment decision tool	168
	7.2	A criti capac	cal discussion of the advantages and challenges of real options for port ity investment decisions	173
		7.2.1	Potential gains of using real options	173
		7.2.2	Additional complexities and costs of using real options	174
	7.3	Concl	usion of this chapter	176
8	Con	clusior	ns, implications and future research	179
	8.1	Theor	etical findings and results	179
	8.2	Practi	cal implications and limitations	181
	8.3	Sugge	stions for future port-related real options research	183
Bi	bliog	raphy		187

A	Solution to differential equation (3.19), determining a new port's project value	207

xviii

List of figures

1.1	Revenue sources and investment outlays of the actors in a port	3
1.2	Maritime chain actors influencing capacity.	12
1.3	Overview of the relationships between the different RO models in the different chapters of this thesis.	24
2.1	Actors, uncertainties and risks in the maritime logistics chain.	29
2.2	Comprehensive framework of RO models and their components.	46
2.3	Delay curve illustrating the positive relationship between the port occupancy rate and the amount of delay.	49
3.1	Revenue sources and investment outlays of the single actor (PA) in the con- sidered private service port.	56
3.2	Sample evolution of demand shift parameter <i>X</i> following a GBM, with $X(t = 0) = 30$ euro per TEU, $\mu = 0.015$ and $\sigma = 0.1$	57
3.3	Realised total congestion cost function with installed capacity $K = 10$ million TEU per year	60
3.4	Total investment cost functions with and without a maximum boundary	62
3.5	Relationship between the optimal investment timing and the threshold for demand shift parameter <i>X</i> following a GBM, with $X(t = 0) = 30$, $\mu = 0.015$ and $\sigma = 0.1$.	63
3.6	Value of the project as a function of <i>X</i> at the moment of investment	71

3.7	Marginal project value of capacity as a function of <i>X</i>	72
3.8	Optimal capacity K^* as a function of $X \in R_2$	73
3.9	Optimal capacity K^* as a function of $X \in R_2$, with $I = I_{\text{Dangl}} = 220K^{0.6}$	74
3.10	Optimal threshold $X_T^* \in R_2$ as a function of <i>K</i>	75
3.11	Comparison between the option value $F(X)$ and the value of the project minus the investment cost $V(X, 11.17) - I(11.17)$ for $A = 5$ and $\sigma = 0.1$ in a private port.	76
3.12	Evolution of the (X_T^{**}, K^{**}) -equilibrium following a parameter change	80
4.1	Revenue sources and investment outlays of the PA and the TOC	87
4.2	Summary of the cooperative RO game methodology.	101
5.1	Total investment cost functions for the two projects.	121
5.2	Sample evolution of demand shift parameter X , following a GBM over time, showing the relationship between the optimal investment timing (T^{L} and T^{F}) and the threshold for X (X_{T}^{L} and X_{T}^{F}) of the leader and the follower port	124
5.3	Summary of the competitive RO Stackelberg investment game methodology.	126
5.4	Number line with the leader's critical investment thresholds.	135
7.1	Summary of the RO tool blueprint.	172

List of tables

1.1	Comparison between the characteristics of infrastructure and superstructure.	18
1.2	Overview of the next chapters' model assumptions.	21
1.3	Overview of this thesis' research question and sub-questions	25
3.1	Overview of Chapter 3's model and the selected parameters	66
3.2	Overview of Chapter 3's model assumptions.	67
3.3	Optimal private port investment decision (X_T^{**} , K^{**}) under different values of A and σ	78
3.4	Reaction of the optimal $X_T^*(K)$, $K^*(X)$ and (X_T^{**}, K^{**}) to a 10% increase in the other parameters.	79
3.5	Optimal investment decision (X_T^{**}, K^{**}) using a different discount rate	81
3.6	Optimal private port investment decision (X_T^{**} , K^{**}) under different values of r and σ	82
4.1	Overview of Chapter 4's model and the selected parameters	96
4.2	Overview of Chapter 4's model assumptions	97
4.3	Optimal $X_{T,a}^*(K)$, $K_a^*(X)$ and $(X_{T,a}^{**}, K_a^{**})$ under different α_5 in a private land- lord port.	102
4.4	V-I of the PA and TOC under different concession fees, if both actors invest at the optimum of a private service port.	104

xxii

4.5	$V - I$ of the PA and TOC with $\alpha_5 = 0.59475$ under possible investment strate- gies, equally followed by both actors
4.6	Illustration of the PA's concession fee strategy forcing the TOC to take the same investment decision
4.7	Discounted $V - I$ under under possible investment strategies equally fol- lowed by both actors in a private service port where the PA can force the TOC to invest in the PA optimum at the moment $t: X(t) = 35$ euro per TEU 106
4.8	Optimal $X_T^*(K)$, $K^*(X)$ and (X_T^{**}, K^{**}) under different s_G and s_{CS} in a service port, partly or fully publicly owned
4.9	Optimal $X_{T,a}^*(K)$, $K_a^*(X)$ and $(X_{T,a}^{**}, K_a^{**})$ under different α_5 in a landlord port with public ownership
4.10	Changes of the PA and TOC's optimal investment decisions $(X_{T,a}^{**}, K_a^{**})$ at the respective α_5^K 's and α_5^X 's as a result of different parameter changes in a land-lord port with public ownership
5.1	Overview of Chapter 5's model and the selected parameters
5.2	Overview of Chapter 5's model assumptions 123
5.3	Investment strategies of the leader (port 1: $X_{T,1}^{L}$, $K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})$) and follower (port 2: $(X_{T,2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})), K_{2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L}))))$ for each characteristic X_{T} in Figure 5.4
5.4	Investment strategies of the leader (port 1: $X_{T,1}^{L}$, $K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})$) and follower (port 2: $(X_{T,2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})), K_{2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})))$) for each characteristic X_{T} in Figure 5.4 under higher product diversification
5.5	Investment strategies of the leader (port 1: $X_{T,1}^{L}$, $K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})$) and follower (port 2: $(X_{T,2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})), K_{2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L}))))$ for each characteristic X_{T} in Figure 5.4 under lower product diversification
5.6	Investment strategies of the leader (port 1: $X_{T,1}^{L}$, $K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})$) and follower (port 2: $(X_{T,2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L})), K_{2}^{**,F}(K_{1}^{*,L_{det/acc}}(X_{T,1}^{L}))))$ for each characteristic X_{T} in Figure 5.4 under different parameter changes
6.1	Overview of Chapter 6's model and the selected parameters

6.2	Overview of Chapter 6's model assumptions
6.3	Optimal investment strategies $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ under different project lead times and initial capacities in a private port
6.4	Optimal investment strategies $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ under different project lead times and initial capacities in a port that is 50% publicly owned by a gov- ernment taking 50% of the consumer surplus into account
6.5	Optimal investment strategies $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ under different project lead times and initial capacities in a public port
6.6	Optimal investment strategies $(X_T^{**}(K_0 = 10, \theta = 4), \Delta K^{**}(K_0 = 10, \theta = 4))$ under different values of r and σ in a private port
7.1	Step 1 of the algorithm: input

xxiv

List of abbreviations

CS	Consumer surplus
FC	Fixed cost
FTE	Full-time equivalent
GBM	Geometric Brownian motion
GDP	Gross domestic product
GT	Gross tonnage
ICT	Information and communications technology
IRR	Internal rate of return
NPV	Net present value
NT	Net tonnage
PA	Port authority
РРР	Public private partnership
RMSE	Root-mean-square error
RO	Real options
RQ	Research question
TEU	Twenty-foot equivalent unit
тос	Terminal operating company

Chapter 1

Introduction

The different activities performed in a port are important for society. In order to transfer cargo between maritime and inland transportation, port investments are required. These investments provide the throughput capacity of a port. However, such investments are very costly for the port and for society, and need to be as good and efficient as possible. The maximum benefits from the investment are sought after at minimal cost. Since such investments are complex, lumpy and subject to a lot of uncertainty, and because some options for flexible investment are available for the decision makers, current investment appraisal and decision making methods fall short. More accurate models and more profound insights into port capacity investment decisions are required, both in theory and in practice.

1.1 A background on ports, port actors and port organisation

Ports are the interface between maritime and hinterland transportation. The main port business, cargo handling, involves activities to transfer cargo between maritime and inland transportation, such as mooring, cargo moving, handling (in a strict sense) and storage, piloting, tugging, customs, etc. In this way, ports form an important link in maritime logistics chains between origins and destinations everywhere in the world (Haralambides, 2002; Meersman & Van de Voorde, 2014a). Ports, port actors and decisions made in a port should be studied accordingly in a logistics chain approach (Coppens et al., 2007).

Ports are composed of many different intertwined actors influencing port decisions. Commonly, the most important actors are the port authority (PA) and the terminal operating company (TOC), but also the shipping companies and some smaller actors have an influence on the decisions taken in a port. Verhoeven (2015) identifies three important functions of a PA: landlord, regulator and in some cases operator of the port. Under the landlord function is comprised the daily management of the port; the maintenance of and investment in port infrastructure; and strategy, policy making and marketing. Additionally, the PA can act as a regulator, to ensure correct operations in the port. Finally, the PA may also operate the port and provide cargo throughput services itself. In that case, a port is classified as a *service port*.

In general, specific services such as tugging and piloting often remain under the responsibility of the PA (Meersman et al., 2015). Nevertheless, these special services have a relatively small impact on capacity. In some cases however, specialised providers operate these services under a concession agreement with the port, and earn a profit from this operation.

More frequently, the PA also grants a concession to provide the cargo throughput service to an external party, the TOC. The TOC obtains the right to exploit an area in the port, determined by a concession agreement, to service the ships and handle the cargo. The PA in this way outsources terminal operations to the TOC and in return earns a substantial revenue, the concession fee (Talley, 2009; G. W. Y. Wang & Pallis, 2014). Such a port is classified as a *landlord port*.¹ The landlord and service port models are historically the two main organisation typologies of a port.

The economic objectives of the PA and the TOC on both the operational and strategic level are strongly affected by their ownership structure. The PA can be owned publicly, privately or by a combination of both. The Anglo-Saxon Doctrine, stating that ports should be financially self-reliant, can be linked to private port ownership and striving for profit maximisation (Bennathan & Waters, 1979; Lee & Flynn, 2011). Public money is however very often involved in port projects throughout the world (e.g., public-private partnerships), because investments in port capacity involve large sums of money (Vanelslander, 2014). Meersman (2005) showed that in Europe the PA, whether or not corporatised, is frequently partly or fully publicly owned. In the Hamburg - Le Havre range for example, the ports of Antwerp and Rotterdam have this structure (Zachcial et al., 2006). The public PA invests and owns the infrastructure, but the superstructure and the equipment are both installed and exploited by TOCs. Under this Continental (European) Doctrine, ports are considered to be part of the social infrastructure. This is similar to the Asian Doctrine, where the central government involvement is however very high and where multiple ports are often developed centrally, possibly encompassing cross-subsidisation (Lee & Flynn, 2011). Oppositely, a TOC is most often a private party (Tsamboulas & Ballis, 2014; Estache & Trujillo, 2009), but it could in theory also be publicly owned (Lacoste & Douet, 2013). According to Heaver et al. (2001), competing private operators invigorate efficient terminal performance.

¹This is in some literature distinguished from a tool port, where the superstructure and equipment are operated but not owned by the TOC (See e.g., Estache & Trujillo (2009)). Tool ports however are not frequently observed in reality and are not further discussed in this thesis. In financial terms, they could be modelled using a landlord port model, but replacing the TOC's investment cost by an additional rent paid to the PA in exchange for using the superstructure and equipment, which are both financed by the PA.



Figure 1.1: Revenue sources and investment outlays of the actors in a port.

For a PA, Suykens & Van de Voorde (1998), Tsamboulas & Ballis (2014) and Xiao et al. (2012) discern potentially conflicting objectives: profit maximisation, throughput maximisation as a driver of economic development, social welfare maximisation and minimisation of the generalised cost for the customers. When a PA is privately owned, it focuses on profit maximisation, whereas a public party has more attention for social welfare and throughput maximisation. In the case of mixed ownership, it could be a weighted combination of these different objectives. Additionally, a port is well aware of it being part of a maritime supply chain wherein the customers are waiting-time averse. When waiting time and the extra cost to shipping companies increase, the latter are more likely to call a competing port, in order to guarantee a supply chain with a lower generalised cost. The port takes this consideration into account and tries to reduce the waiting time and generalised cost for its customers. Hence, profit, throughput and welfare maximisation are not the only objectives of the PA. Oppositely, the TOC's only objective as a private company is maximisation of the individual company's profit (Tsamboulas & Ballis, 2014). These considerations are reflected in the capacity investment decision and at the daily operational level of both the PA and the TOC. They also have an impact on the concession fee strategy of the PA. The PA will set a different size and type of concession fee depending on the relative importance of the profit and throughput maximisation objectives in its strategy (Heaver et al., 2001).

Figure 1.1 shows the sources of income a TOC and a (partly) publicly owned PA receive in exchange for the specific services they offer to the different actors. Alderton (2008) and Meersman et al. (2015) state that different determinants make up the price of the port service. The price structure is not uniform over different ports and is often lacking transparency. For example, port dues, paid by the shipping companies to the PA, are very complex. They may depend on gross tonnage (GT), net tonnage (NT), destination in the port, number of locks passed, length of the ship, amount of cargo handled, time staying in the port, liner service or not, quantity reductions, etc. (Degroote, 2015; Port of Antwerp, 2017a). In exchange, the PA takes care of facility investment, management and maintenance in the port (Zachcial et al., 2006). The terminal tariff is set by the TOC and paid by the shipping companies for the handling of their transported goods. It may depend on the number of movements or the number of containers handled. In some cases, extra payments are received, e.g., for storing goods. Depending on the trade and time period considered, the relative importance of these extra payments in the TOC's total income is however rather limited, as they consist at most of 15% of total TOC revenue. Moreover, these extra services are not applied to all goods handled (Jenné, 2017). Also, the discussed concession fee paid by the TOC to the PA is included in Figure 1.1.

1.2 The economic importance of ports

Port activities facilitate international transportation of goods, international and regional trade, of which more than 80% is seaborne, and economic development of a region (Munim & Schramm, 2018). The economic impact of ports is hence broader than profit generation alone. The economic importance of the port sector is typically measured and quantified through its direct, indirect, induced and catalytic effects in terms of value added and employment. Value added is defined as the value that a port adds through its activities to the value of the used inputs. In this way, it is a quantification of a port's contribution to the gross domestic product (GDP) of a country or region. Employment expresses the average number of full-time equivalents (FTE) that are working in the sector during a financial year (van Gastel, 2016). Cooper & Smith (2005) and Oxford Economics (2015) state that direct effects, which constitute the primary impact, are generated by the firms situated within the port. According to Davis (1983), the secondary impact of ports consists of indirect and induced effects. Indirect effects are generated by the actors down the supply chains of the port firms. Induced effects are generated by the organisations that earn from the the employees of the firms generating direct and indirect effects spending their income. Finally, economic catalytic effects encompass other spillover benefits and consumer surplus resulting from the port activities, as they contribute to trade and long term productivity (Musso et al., 2006). The latter however are the most difficult to quantify. Next to the economic impact, port activities also make an important contribution to the tax income of a country (Benacchio et al., 2000; Oxford Economics, 2015). Finally, ports are also an engine allowing other national industries to grow, such as manufacturing, logistics and transportation (Benacchio et al., 2000).

A meta-analysis by Olaf Merk for OECD shows that, on average, 100 dollar of direct and indirect added value is created per tonne handled in the port. For containers, the average is 90 dollar per tonne, the minimum is 40 and the maximum is 150. As far as employment is concerned, one million tonnes lead on average to about 800 jobs (OECD, 2014). Notwithstanding the positive impact of ports on the economy, they also cause a number of negative

effects and external costs. OECD (2014) mentions air, water and soil pollution, noise, waste production, an unpleasant visual impact, large land occupations and the creation of traffic congestion in the surroundings of the port.

A number of studies concentrating on individual countries and ports exist as well. In Belgium, the port of Antwerp is the major port in terms of value added. It generated between 10 (in 2012) and 11.5 billion euro (in 2017) of direct value added per year. It is followed by Ghent, Liège (an inland port), Zeebrugge, Brussels (a second inland port) and finally Ostend. In total, the direct effects generated by the Belgian ports ranged from 16.5 (in 2012) to 19.4 billion euro (in 2017), of which 4.4 to 4.8 billion euro were generated by the maritime cluster. This maritime cluster involves among other things cargo handling activities, shipping companies, shipping agents and forwarders. The cargo handling activities generated between 2 and 2.3 billion euro in the same years. Other, non-maritime port activities represented the majority of the direct value added: between 12 and 14.6 billion euro per year. The indirect effects ranged from 13.5 to 16 billion euro, again between 2012 and 2017. In total, Belgian ports generated between 30 and 35.3 billion euro per year in terms of direct and indirect value added. In 2016, the four Flemish seaports accounted for over 7% of the Belgian GDP, and more than 12% of the Flemish GDP. (Gueli et al., 2019; Merckx, 2018)

As far as employment is considered, Antwerp is still the largest port in Belgium, employing between 61 000 and 62 000 FTEs between 2012 and 2017. In total, the direct employment in the Belgian ports is slightly below 120 000 FTEs. Total employment is about 255 000 FTEs. The maritime cluster accounts for about 40 000 FTEs, of which about 20 000 FTEs perform activities related to cargo handling.

For the Netherlands, van der Lugt et al. (2018) estimated the direct added value of sea port activities at about 28 billion euro and the indirect added value at about 14.5 billion euro in 2017. As far as employment is considered, direct employment amounted to 184 000 FTEs and indirect employment to 166 000 FTEs. The port of Rotterdam accounts for almost half of the direct effects. When focusing on cargo activities, 5.9 billion out of a total of 7.2 billion euro added value is created within the Dutch port areas. 36 200 out of a total of 47 700 cargo related FTEs are working within the Dutch port areas.

In the UK, the direct value added of ports in 2015 was estimated by Cebr (2017) at 10.6 billion euro (1 GBP = 1.4 euro). The indirect value added for 2015 was 11.6 billion euro. Cebr (2017) also estimated the induced value added, estimated at 11.2 billion euro in 2015. In terms of employment, the direct, indirect and induced effect for the UK ports in 2015 amounted to 101 000, 319 000 and 276 000 jobs respectively (Cebr, 2017). These figures are in line with those of Oxford Economics (2015) for 2013.

In the USA, coastal seaport cargo activities are crucial for the economy. The direct business revenues generated by the port sector amounted to about 125 billion U.S. dollar in 2014. In total, when also importers' and exporters' activities are considered, seaport cargo activities accounted for 26% of the U.S. economy, as they generated about 4.6 trillion dollar in 2014. In terms of taxes, direct port sector activities resulted in about 8.2 billion dollar of income for the American government. The indirect and induced taxes amounted to about 33 billion dollar. As far as jobs are considered, about 542 000 direct, 372 000 indirect and 823 000 induced jobs are created within the port sector. (Martin Associates, 2015)

1.3 Investing in ports is indispensable

There are two main characteristics of good logistics performance of ports, namely low logistics costs and high reliability of the operations (Munim & Schramm, 2018). In order to perform port operations effectively and efficiently and generate added value in a port, investing to build new or expand existing ports is crucial. A port's ability to provide cargo throughput services is quantified in terms of its capacity. It is often expressed in twentyfoot equivalent units (TEU), tonnes or number of ships that can be handled per year. Next to building new capacity-increasing elements, capacity can also be increased through the optimisation of existing terminals (ESCAP, 2002; Kauppila et al., 2016). Each of these investment types will allow generating more throughput at better service levels (Musso et al., 2006). Since port capacity installed by investors increases the service level of the operators, the generalised cost in the logistics chain will fall, and also shippers and consumers will experience benefits from these investments. Musso et al. (2006) refer to this concept as a port investment chain.

Disposing of insufficient service capacity to handle cargo could involve congestion, leading to high logistics costs (Novaes et al., 2012). In a port, a service environment, and in transportation in general, it is even more important than in a production environment to dispose of the right amount of capacity, since the transport service is not storable (de Weille & Ray, 1974). Capacity that is not used at the actual time period cannot be stored and used in the next, as opposed to warehoused goods. Undercapacity cannot be covered either by unused outputs from a previous period. Since the demand for cargo throughput is uncertain and variability may become high, it might occur that moments of empty berths are followed by moments in which ships are waiting to be serviced at a berth that is currently occupied. In this way, congestion and waiting time might start to build up in the port from occupancy rates of about 50% of the theoretical design capacity, the maximum amount of throughput that can be handled by a port in a certain time period. Above 75% to 80%, waiting times increase more than linearly (Blauwens et al., 2016; Kauppila et al., 2016).

A mismatch between demand and capacity can result in periods of capacity scarcity or overcapacity. Without sufficient capacity, i.e., undercapacity, the port risks losing customers, throughput and profit due to the congestion building up. This results in a loss of social welfare in the form of lost consumer surplus. However, installing capacity comes at an investment cost, positively related to the amount of installed capacity. Hence in-
stalling too much capacity, i.e., overcapacity, poses a problem as well, as money is invested in capacity that is not used and that hence does not generate revenues. This gives rise to a trade-off between investing too much money in unused capacity and investing in too little capacity, which leads to higher user waiting costs and which might eventually lead to losing customers and profit (Meersman & Van de Voorde, 2014a). This capacity investment trade-off is even more complicated by the uncertainty in the port (Musso et al., 2006). As a result, installing the right amount of capacity at the right moment is crucial, as additional port capacity is required everywhere in the world (Vanelslander, 2014). Therefore, this thesis explores how ports can deal with this crucial trade-off between investment costs and internalised user waiting costs in port capacity investments under uncertainty.

In reality, overcapacity is often encountered. As a result of large investment sizes, high costs and long completion times of port capacity investment projects, it is more beneficial to initially install more capacity than needed in the first periods of use. If overcapacity is present, prices may be lowered to attain a desired capacity utilisation rate while awaiting demand increases for port services (Meersman et al., 2010; Xiao et al., 2013). Overcapacity may also exist in a port to reduce the attractiveness of a competitor's investment option, known in the literature as entry deterrence. Finally, (especially large) ships require overcapacity to reduce the probability of waiting at the dock (Huberts et al., 2015; Musso et al., 2006). The latter consideration demonstrates the importance of a joint analysis of the amount of available capacity, the amount and the timing of capacity expansions, the level of output and the output price, paid by the shipping company to the PA and the TOC (Strandenes, 2014). In case of undercapacity, congestion pricing could be applied to reduce utilisation rates and increase revenues, but the theoretical design capacity remains an upper limit to the port throughput (Zhang, 2007).

The previous discussion demonstrates that investing in port capacity is crucial, both in the long and in the short term. As Musso et al. (2006) argue, investing in port capacity allows increasing throughput at a moment that the demand for it becomes present. Due to increased port activities and resulting factor demand increases, different effects can be realised. Two effects are especially important in the short term: a direct profit for the owners of the investment and microeconomic benefits due to a reduction of generalised costs. Positive and negative macroeconomic effects in the long term involve economic growth and increased employment on the one hand and external costs on the other hand.

Long term

In the long term, consequences of port capacity shortages may involve freight delays and increased trade distances, leading to a growth slowdown of a region's GDP and worldwide trade. Maritime transportation on average is expected to more than quadruple between 2010 and 2050, although individual growth rates per port may differ and even become negative for some ports over a certain period of time. Consequently, the demand for cargo

throughput services is growing, especially in the container segment, although this demand is also uncertain (De Langen et al., 2018). By 2030, container traffic related to international trade is expected to more than double, whereas by 2050, it is expected to become the threefold of current container traffic. As a result, additional port capacity is a prerequisite to accommodate the expected increases in worldwide maritime trade volumes. (Musso et al., 2006)

In reality, ports are aware of this necessity. Port capacity expansions that are planned until 2030 are expected to be sufficient to handle the additional trade volumes, except in South and Southeast Asia. If such capacity expansions are not installed however, freight delays are expected to increase with 48% by 2030 and with 84% by 2050. Moreover, due to the congestion that would arise as a result, trade distances could rise by 43% by 2030 and by 65% by 2050. (Kauppila et al., 2016)

Although maritime trade on average is expected to grow, some future evolutions in socio-economic factors may slow down or even reverse this growth. One important factor is the level of economic activity, as it has a strong, positive impact on the demand for transportation. In turn, this is influenced by economic cycles, investment decisions, demographics, political decisions, trade wars, geopolitical conflicts and technological evolutions (both in transportation and in other sectors) (Preston, 2001). Transportation of goods by drones and 3D-printing might both alter demand for maritime transportation, as transportation of raw materials might substitute (half) finished products, in turn having an impact on the transportation mode used. For ports, it is important to monitor these long-term evolutions, as they will impact the need for port capacity.

Short term

As a result of bottlenecks in the port and subsequent delays, shippers, forwarders and shipping lines incur a cost, i.e., the congestion cost, because for example goods may deteriorate or services might have to be rescheduled. Given the fact that cost competitiveness drives the shipping industry, the shipping companies will avoid such waiting costs as much as reasonably possible (Kauppila et al., 2016). As shown by Blauwens et al. (2016) and De Borger et al. (2008), the value of time for a ship waiting in the port can be high and makes up an important part of the generalised cost of the maritime logistics chain. Aversion to waiting moreover differs among users and among goods categories. The value of time can be estimated using stated or revealed preference experiments (De Jong, 2007; OECD, 2014). For example, liners shipping (white) reefer containers with food are more waiting-time averse than liners transporting containers with construction materials.

Sufficient capacity is needed in the short term. Congestion and its costs, resulting from capacity shortages, add an extra aspect to inter-port competition. Moreover, because the overlap of port hinterlands has increased over the last decades, inter-port competition has

increased too (Musso et al., 2006). Such competition might lead to free, also called unused or idle, capacity in one port attracting throughput from a nearby congested port with high waiting times. This proves that the cost of congestion for the port customers needs to be considered by the port when studying the investment decision in additional capacity to decrease the probability of waiting and the resulting congestion costs (Macário, 2014).

Technological requirements

Small and fast capacity increases can be realised through the optimisation of terminal efficiency, e.g., through automation or a more efficient organisation of activities (Kauppila et al., 2016). Also extended opening hours can increase capacity. The environmental requirements for the maritime and port industry are expanding as well, in order to increase environmental sustainability through a reduced impact on the environment. To this end, ports may be obliged to invest in new noise and pollution reducing technologies (Kauppila et al., 2016).

Investing in safety measures to protect the port from adverse effects of climate change, such as rising sea levels, might be appropriate (K. Wang & Zhang, 2018). Due to spillover effects, investing early in climate change adaptations increases social welfare under uncertainty (Randrianarisoa & Zhang, 2019). In addition, investing in new, advanced information systems could enhance the exchange of information in port communities. In turn, operations in the logistics chain could be managed and coordinated in a better way. This is important, because disposing of correct information is at least as important as the cargo delivery itself for customers and transportation companies (ESCAP, 2002).

Finally, increasing average and maximum ship size as a result of liners further exploiting economies of scale and specific newly developed engines for large ships also have an impact on the need for port investments (ESCAP, 2002; Kauppila et al., 2016; Musso et al., 2006). Ports that want to accommodate these larger ships have different capacity requirements and need to adapt their operations. For example, access channels and berths need to be deeper, quays need to be longer and stronger, cargo storage areas need to be larger, cranes need to be larger (Kauppila et al., 2016), tugs need to be stronger and ship passages (such as channels, locks, turning points, etc.) need to be wider (Sys et al., 2008). Hence, it is important for port capacity investment decision makers to also take the investment decisions of ship owners into account.

1.4 Throughput capacity of a port

In order to discuss port capacity and the investment herein, it is important to first study in detail the different definitions of capacity and its constituting elements. The theoretical design capacity measures the maximum amount of cargo a port can handle (i.e., throughput) on a sustained basis over a predetermined time period, e.g., one year (Dekker, 2005). It is opposed to the commercial capacity of a port, which expresses the maximum amount of throughput that can be realised on a sustained basis at a predetermined service level (Kauppila et al., 2016; Rashed, 2016). This service level is often defined by the TOC. Hence in practice, commercial capacity does not exceed 75% of the theoretical design capacity, since waiting times start to increase sharply from such occupancy rates (Kauppila et al., 2016). Both capacity definitions are often expressed in TEU, tonnes or number of ships per year.

In this thesis, the focus is on the theoretical design capacity for two reasons. First, this definition is independent of the service level that a specific TOC wants to achieve. In that way, consistent comparisons of capacity between different ports and terminals are possible. Second, this approach does not necessarily exclude high occupancy rates and resulting waiting times in ports. This is a prerequisite to be able to analyse the economic impact of congestion on port investment decisions.

Capacity is determined by a complex interaction of different elements like infrastructure and superstructure, and is often approached from an operational perspective (Novaes et al., 2012). The capacity in a port in a certain time period is influenced by different constituting parts. These components are discussed in the following subsection. Subsequently, it is pointed out that all these parts need to be well adjusted to one another. A lack of alignment of only one component may already reduce capacity significantly, as capacity is very sensitive to bottlenecks (De Langen et al., 2018).

1.4.1 The determinants of capacity

In port capacity literature, different overviews of the constituting elements of port capacity are available. The overview here is composed from a capacity investment perspective (Alderton, 2008; Dekker, 2005; De Langen et al., 2018; Kia et al., 2000; Maloni & Jackson, 2005; Meersman & Van de Voorde, 2014a; Musso et al., 2006; Vanelslander, 2014; Zachcial et al., 2006):

- Maritime access: Locks, lights, buoys, channels and breakwaters;
- **Port infrastructure:** Berths, docks, basins, internal connections, storage areas and quay walls;
- Port superstructure and equipment: Paving,² lighting, gates, fences, sheds, build-

²Some sources classify paving under infrastructure (plus), others classify it under superstructure. This also depends on which actor invests in it. Here it is decided to classify it under superstructure, as for example in the port of Antwerp, paving is to be provided by the terminal operator.

ings, offices, cranes, pipes, warehouses, straddle or gantry carriers and ICT. These elements are strongly influenced by available technology;

- Labour and operations: Port workers, administrative staff, customs, working hours, labour organisation and division, stacking process, division between empty and full containers;³
- Hinterland access: Road, rail and inland navigation connections.

Not only the presence and amount of these different elements, but also their maintenance define the theoretical capacity (Alderton, 2008). Without maintenance, many of these elements may deteriorate, reducing their productivity and capacity. Additionally, a port's design capacity is limited by the element or elements with the lowest capacity, which can be seen as the bottleneck(s). The importance of this holistic approach is highlighted by Kauppila et al. (2016).⁴

1.4.2 Port actors influence capacity as well

The different, intertwined actors in a port influence the aforementioned elements of port capacity to handle the goods (De Langen et al., 2018). Figure 1.2 shows where the stake-holders from the maritime chain have an impact on capacity. First, there are the customers of the port: the shipping companies. They bring the goods to the port, through the maritime access provided by the port authority and governments. The port's throughput is not only limited by its own capacity, but also by the total capacity of the shipping companies calling at this port (Maloni & Jackson, 2005). Ports will not handle more goods than what the ships can carry.

Subsequently in the port, the PA and the TOC have the decision power over port infrastructure, superstructure and equipment investment. This is explained in detail in Section 1.4.3. Moreover, they organise labour by hiring port workers (Maloni & Jackson, 2005). This represents the majority of the variable costs of throughput. Additionally, both the number of pilots and the number of tugboat operators influence capacity.

³The amount of labour is considered as a flexible input. Increasing working time or the number of workers has a positive impact on port capacity. Also the way labour is organised, operations are coordinated and equipment and available technology are managed, determine the productivity and capacity of a port.

⁴An example clarifies this statement. Suppose that annually 10 million TEU could enter the port (maritime access capacity) and could also be handled (port infrastructure, superstructure and equipment capacity), but that only 8 million TEU could leave the port, because there is not sufficient hinterland access capacity (roads, railways and inland waterways). In this case, the design capacity of the port is only 8 million TEU and is limited by the hinterland access, which is the bottleneck. The capacity of the port can be expanded to 10 million by increasing hinterland accessibility in order to be able to ship the additional 2 million TEU into the hinterland. In that case, the bottleneck disappears. If the port subsequently wants to increase its capacity to 15 million TEU, this can be achieved by increasing all the aforementioned constituting elements of capacity by 5 million TEU.



Figure 1.2: Maritime chain actors influencing capacity.

The last actors in Figure 1.2 are the hinterland companies, using road, rail and inland waterways connecting the port with the hinterland. These connections are often partly provided by the port authority and partly by the government. Hinterland companies or connections being unable to move the goods out of the port before new goods arrive, or move the goods into the port before the ship leaves, may cause congestion within the port, leading to a capacity reduction as well (De Borger et al., 2008).

1.4.3 The capacity investment decision making process in the port

Different decision makers interact in port capacity investment decisions. This results in a complex decision making process, which varies with the type of port studied (Meersman, 2005). When the PA of an active port experiences congestion building up at the current infrastructure, when operational efficiency cannot be further increased and when the congestion pricing mechanism cannot be (further) exploited in a profitable way, the PA considers the infrastructure to be a bottleneck. Hence, the responsible government(s), possibly in collaboration with the PA, start(s) to develop plans for the investment in additional infrastructure (e.g., a new dock). If there is no port in place yet in a region, plans to build a new port are developed when sufficient maritime traffic to or from the region under consideration is expected to justify the construction of this new port (Novaes et al., 2012; Ward, 2015). If the port is a private service port, the decision making process is less complicated than for a public landlord port.

In the case of a private service port, there is one single actor which both installs and operates the infrastructure, superstructure and equipment. Hence all revenues and costs are on its account. Since profit maximisation is the objective, such a port needs to compare all expected revenues and costs. When sufficient profit, compared to the weighted average cost of capital, is expected so that it is better to invest rather than to wait longer, the port will decide to go ahead with the project and install the new capacity. Subsequently, it will receive all revenues, which include port dues and terminal handling charges.

However, due to the large scale of port infrastructure projects, a part of or the entire investment is very often funded with public money. In many cases, such as the port of Napier in New Zealand, service ports are partly or fully publicly owned. In such cases, not only profit will matter, but all welfare effects will be considered. As opposed to the private owner, a public owner is interested in the total throughput and employment as well. Not only direct economic effects are considered in the cost benefit analyses or economic impact studies. Also indirect and induced effects require attention. However, some other effects, such as the catalytic effects, are considered as well, although they are difficult to quantify in monetary terms. For a government, it matters that total benefits outweigh the costs (Musso et al., 2006).

Before, when governments decided on port investment projects, a lot of projects were funded on a first come, first served basis, as governments had sufficient money. Large amounts of money were invested in the development of ports by these governments (Haralambides, 2002). Nowadays, this is no longer the case. The scarce resources are allocated to projects with the best business cases, which optimise the objectives of the funder. Different business cases enter in competition with one another for the (public) money. In 2015, for example, the Flemish government decided to expand the container capacity of the Port of Antwerp, to avoid liners losing time due to congestion in the port (RebelGroup Advisory Belgium, 2015). This decision is only the first of four steps in the process. Different alternatives need to be considered and studied in detail in the second phase. The addition of a new dock, Saeftinghedok, is only one of the nine possible scenarios discussed in 2019. The outcome of this second phase is a preferred scenario, which needs to be developed in full in the third planning phase. With the project decision made in that third phase, the execution can be realised in the final, fourth phase. The time-consuming construction works however need to be as efficient as possible (Flemish Government, 2019).

In a landlord port, such as the port of Antwerp, the investment process is complicated by the involvement of multiple actors. In this case, the (often) partly or fully publicly owned PA is not the only actor deciding about future capacity. Also the TOC has a role to play. Since the PA and the TOC have different revenues and costs (see Figure 1.1), their optimal investment decision will differ as well. Additionally, in the case the PA is publicly owned and the TOC privately, the PA and the TOC would even have different and in some cases opposing objectives, i.e., welfare versus profit maximisation (Musso et al., 2006). The decision making process would comprise two decisions. The PA would first have to decide on and install the infrastructure. When deciding about the investment however, it could be beneficial to previously consult potential TOCs about their development plans (Port of Vancouver, 2015). After completion of the infrastructure, the PA can grant the TOC with the most attractive project the concession agreement. This TOC subsequently installs the superstructure and equipment. Hence, the timing of the TOC's investment cannot be be-

fore the PA's investment, nor can the capacity exceed the capacity installed by the PA. After completion of the terminal, the TOC can operate the terminal and handle the cargo. Nevertheless, the PA should already take into account the available information about the expected revenues from the future concession agreement, i.e., the concession fee paid by the TOC, when deciding about the infrastructure in the first step. This information is available to a port's PA to the extent that it disposes of the necessary information about the terminal operators already active in the port. The concession fee is an important revenue for the PA that allows recovering investment costs (Haralambides, 2002).

1.5 The cost of port investments

Investing in ports involves large sums of money. Not only the development of the land is costly, also the land itself has a high value. Land used for port activities is situated close to the water, and hence has a high opportunity cost due to a wide variety of valuable other potential destinations for the land (Haralambides, 2002). This section discusses the required investment budgets and the effectively made port investment outlays in different countries.

According to Merckx (2018), the Flemish region invested 415 million euro in its ports in 2017, of which slightly more than 60% was needed for maritime access works. Of the 69 million euro invested in Antwerp, the majority involved maintenance and renovation works. Gueli et al. (2019) show that in the Belgian ports, the direct investments amounted to 4.8 billion euro in 2017. Almost half of it was invested in Antwerp. The total maritime investments amounted to 1.9 billion euro, of which 0.9 accounted for cargo handling related investments, 0.3 for port construction activities and the remainder for shipping companies and other maritime activities. In 2003, when the new Deurganckdok project was nearing its completion, the investment of the Flemish region in Antwerp was at its highest point at 170 million euro. In 2004, 160 million euro were invested (Merckx, 2018). The total cost of the Deurganckdok, constructed between 1995 (initial study work) and 2006, was 670 million euro (Vanelslander, 2014). Quay wall construction and dredging accounted for the majority of the costs (almost 75%). At the same time, these were also the cost drivers causing the largest cost overruns. In total, cost overruns amounted to about 230 million euro. According to Freddy Aerts, head of the Maritime Access Division of the Flemish government, construction of the new Saeftinghedok is expected to cost between 0.9 and 1.5 billion euro.

In Rotterdam in the Netherlands as well, an important expansion took place between 2004 and 2013. The new facilities at Maasvlakte 2 have been installed in phases, at a total cost of 2.9 billion euro (Port Consultants Rotterdam, 2018). According to Vanelslander (2014), the list of future port infrastructure investments in Europe is small. However, according to a survey of De Langen et al. (2018) and their subsequent extrapolations, investments of about 50 billion euro are needed in the European ports in the period 2018-2027. As far as port investments are considered, there are important differences among geographical regions. Based on estimated trade volumes using a four step model approach, Kauppila et al. (2016) state that the need for additional capacity is the largest in Asia and Africa. In Africa, economic growth is supported by an increase in port projects, especially in the northern and western regions (Vanelslander, 2014). Large investments are taking place in Asia and the Pacific region. According to ESCAP (2002), between 2001 and 2011, 400 additional berths were needed, corresponding to an investment need of about 27 billion U.S. dollar. In India, the government planned to invest 12 billion U.S. dollar in its ports over the period 2012-2017, mainly in new terminals, expansions and navigation channels. Between 2010 and 2020, a total of 66 billion U.S. dollar is anticipated in India, of which the majority is to be funded privately. Also in China, where many ports are financed by the central government (Lee & Flynn, 2011), and in Singapore, large sums of money are invested in the ports (Vanelslander, 2014).

American ports are constantly seeking to expand their capacity as well (OECD, 2019). According to Vanelslander (2014), the major American ports had planned to invest about 8.5 billion dollar between 2012 and 2017, mainly in container terminals and dredging. However, investments of more than 30 billion dollar would be needed by 2020 to remain competitive. Therefore, the U.S. government provides, as part of the FAST programme, 4.5 billion dollar of funding over five years for a number of port projects (ASCE, 2019). Moreover, U.S. ports are planning to invest about 155 billion dollar between 2016 and 2020, mainly for expansion, maintenance and innovations.

In Latin America, the majority of port investments are taking place in Brazil. Between 2010 and 2013, the government invested about 6.4 billion U.S. dollar, especially in new ports, port expansions and port upgrades. Next to Brazil, also projects in Mexico, Chile and Peru are developed. Finally, also the Middle East, a small region disposing of a lot of capital, further develops its port infrastructure. (Vanelslander, 2014)

1.6 Characteristics of port capacity investments

Each port worldwide is organised, funded and built in a different way. Even every port investment project is different (Wiegmans & Behdani, 2018). Such investments can take the shape of a greenfield project, if no previous port exists in that place; a port expansion project, if e.g., a new dock and terminals are installed in an existing port; or an upgrade of existing facilities, e.g., by installing more efficient cranes based on new technologies.⁵

Port capacity investments have some peculiar characteristics. First of all, they are sub-

⁵In the literature, the following synonyms are sometimes respectively encountered for these three type of projects: greenfield port (see Chapters 3-5), greenfield expansion project (within an existing port: see Chapter 6) and brownfield (expansion) project. The latter type is beyond the scope of this thesis.

ject to a lot of uncertainty, resulting from many sources. Moreover, investing in the large elements of capacity involves large sums of money and sunk costs. These projects also take a lot of time to be completed, the so-called lead time. This lead time can easily exceed five years. This results in large gaps between investment outlays and revenues. Very often, this lead time increases due to slow planning and (political) decision making (Kauppila et al., 2016; Musso et al., 2006). On the other hand, the lifetime of these elements is extremely long (Musso et al., 2006). Examples of such large capacity elements include maritime access, hinterland access, port infrastructure and to a lesser extent some port superstructure elements and equipment, such as paving, buildings, warehouses, cranes and pipelines. Some of these capacity investments are also irreversible, including maritime access, hinterland access, port infrastructure and again some superstructure elements and equipment, such as paving, lighting, specific buildings and specific equipment that cannot be used at other terminals or for other activities (Musso et al., 2006). Moreover, infrastructure projects have a long expected economic life during which they generate revenues. In this light, maintenance is crucial to guarantee this extended economic life. Finally, due to the large scale of infrastructure investments, they are often funded with public money. All of these specific characteristics have an important impact on the decision making process and are taken into account in this thesis. (Vanelslander, 2014; Wiegmans & Behdani, 2018)

1.7 Research objective and scope

It has been discussed that ports have an important economic impact and that well-thoughtout investments in port capacity are indispensable. However, analysing and deciding about these investments is complex, since many actors and objectives are involved in an uncertain environment. The available literature does not provide clear theoretical insights into the factors that influence port capacity investment decisions under uncertainty. The literature does not provide practical guidelines for applications of such insights and developed models accounting for uncertainty either. As a result, the objective of this thesis is to study how optimal capacity investment decisions of ports under uncertainty are influenced by different port- and project-related economic characteristics. This objective is reflected in this thesis' main research question: How are optimal port capacity investment decisions influenced by changes in economic, port- and project-related characteristics, both theoretically and in practice?

In this thesis, the focus is on port investment decisions providing throughput capacity for cargo handling activities at berths, including loading and unloading ships. Other activities, such as ship facilitation (e.g., bunkering, ship building and repair), passenger cruises and industrial activity, are left beyond the scope of this thesis.⁶ To this end, the assumption

⁶The other activities could be included in an extension of the presented models by considering their relevant income, costs and actors.

is made that the maritime and hinterland access capacity elements are sufficiently present in the port and do not present a capacity bottleneck. They do not influence the considered investment decision. Such a focus on a single functional area has also been adopted by De Borger et al. (2008), who studied the investment in port hinterland capacity. It is also followed by de Weille & Ray (1974), who assume that berths are the bottleneck in order to study berth investments. In this way, this bottleneck selection methodology allows increasing or decreasing the scope of cargo handling capacity investment decisions to the necessary elements with their specific characteristics, revenues and costs.

The models developed in this thesis focus on the large-scale, irreversible infrastructure investments made by the port authority. However, also the subsequent superstructure investment decision of the terminal operating company is identified as a crucial influencing element of a port's throughput capacity, as the TOC handles the goods. Accordingly, the superstructure is also considered in the models of this thesis.⁷ Although the degree of irreversibility is considered to be lower, costly transportation of cranes and their specific design, the fixed character of paving, lightning, offices and buildings for storage, next to the large scale and long construction lead times, make the use of RO models also relevant for this type of investments. As a result, smaller and reversible capacity investments such as hiring additional pilots to compensate for a deficit, are left beyond the scope of this thesis.

Both infrastructure and superstructure investments are subject to a lot of uncertainty. In this thesis, the focus is on the macro level of uncertainty: demand uncertainty. The different types of uncertainty are further elaborated in Chapter 2. The implications of focusing on demand uncertainty in this thesis' models are clarified in Chapter 3. There are also some important differences between infrastructure and superstructure projects. Table 1.1 provides a synthesising overview of the characteristics of port infrastructure and superstructure. It is apparent that superstructure consists of a typical private investment of which all cash in- and outflows relate to the deciding actor (TOC). The investment is justified by an expected profit, generated by users paying tariffs to the TOC. Superstructure has a relatively short lifetime. This is opposed to the infrastructure, which has to be built before the superstructure can be developed. Infrastructure has a longer lifetime, but has no specific users. It is decided upon and funded by the PA receiving port dues and rents such as concession fees, but the government is often involved as well. For the latter, next to profit, also market share and welfare matter. These differences need to be taken into account in the investment analyses.

Since containers represent an important segment of seaborne trade, as they accounted for 45% of worldwide traffic in 2010 measured in tonnes (Kauppila et al., 2016), the developed models in this thesis focus on the investment in container ports. The advantages are

⁷Hence, in this thesis, the infrastructure, superstructure and equipment elements of capacity are assumed to be bottlenecks, all with an equal initial level of capacity (zero for a new port or the current capacity when expansion is studied). This approach is deemed realistic for lumpy investments (e.g., adding an entire new dock).

	Infrastructure	Superstructure
Users	Non user-specific	User-specific
Financing	Government, PA	Enterprise (i.e., TOC)
Revenues	Port dues, rents	Terminal tariff
Lifetime	Long (25-50 years)	Short (5-25 years)
Deciding actors	Government, PA	Enterprise (i.e., TOC)
Decision variables	Profit, market share, welfare	Profit

Table 1.1: Comparison between the characteristics of infrastructure and superstructure.

Source: own composition, based on Vanelslander (2014) and National Port Council (2001).

that a lot of container data is available, and that the uniformity of the containers leads to similar handling activities and infrastructure and superstructure requirements all over the world, making the developed models more generally applicable. Oppositely, it is more difficult to estimate average aversion to waiting for the different containers shipped. Since containers may carry diverse cargo types, the time sensitivity of containers may differ. Hence, the aversion to waiting is less clear-cut. Only general distinctions are easier to made, for example between reefer containers, which in general are highly time-dependent and other, regular containers. In this light, aversion to waiting for containers needs to be considered as an average value, as opposed to bulk cargo, where the type of cargo and hence waiting time aversion is clearer.

Since the focus is on containers, TEU is chosen as the unit of measurement of the goods handled in the port. According to Kauppila et al. (2016), the current container capacity in the major ports worldwide exceeds 500 M TEU per year. Although the analyses focus on container ports, the presented approach and models can be translated to ports handling other goods. To this end, another unit of throughput needs to be selected, and if applicable, specific characteristics of the cargo handling activities, infrastructure and superstructure need to be taken into account as well. The theoretical findings moreover remain qualitatively valid, independent of the unit of throughput used.

1.8 Research approach: advantages and limitations of investment analysis and decision making with real options models

The amount of capacity in which a port invests, is often determined by a net present value (NPV) or internal rate of return (IRR) approach in a cost benefit analysis that considers all relevant welfare effects, represented in monetary terms (Centraal Planbureau, 2001). However, port infrastructure and superstructure investment decisions consist of a number of flexible options to deal with different sources of uncertainty (e.g., the option to postpone the investment or to develop a new terminal in phases). These options bear an option value, which is overlooked by these traditional methods when applied to large-scale, irreversible investments under uncertainty. As a result, such traditional approaches can be very wrong when deciding about port infrastructure and superstructure projects (Dixit & Pindyck, 1994).

Real options (RO) have been identified as a more accurate approach than singular NPV or IRR to support the specific type of investment decisions under consideration in this thesis (Dixit & Pindyck, 1994). In Chapter 2, this is discussed in more detail. An important advantage of this methodology when modelled in continuous time and continuous state is that the investment decision's resulting profits and other benefits can be optimised with respect to the investment size and the crucial timing decision, given the large uncertainty present (Kauppila et al., 2016). However, currently available RO models are not able to fully incorporate the most important specific characteristics of ports and their capacity investment decision, such as the impact of (increasing) congestion costs, the different actors involved in the decision making process, inter-port competition, time to build and the amount of existing capacity (Benacchio et al., 2000). As a result, detailed insights into the impact of these characteristics on ports' optimal investment decisions are not available in the literature.

Since the main objective of this thesis is to examine and formalise the impact of portand project-related economic characteristics on the optimal capacity investment decision of a port, port-specific RO models are developed in this thesis. These models take some of the specific port and project characteristics into account. The complex economic reality is translated into mathematical investment decision making models. The data to calibrate the economic model parameters are based on existing port projects. The findings will allow port managers to better understand required reactions to changes in the economic environment within their investment strategy, an important added value of this thesis. Subsequently, econometrics and additional conversions are required to translate the theoretical RO models into specific, applicable port investment decision variables such as the optimal timing and size of the investment. This analysis moreover allows explicitly quantifying the added value of using RO for port investment decisions.

Modelling the correct patterns for the different uncertain variables in continuous-time continuous-state models is however the most difficult, and hence most costly aspect of real options modelling. A model always represents reality in a simplified way. As a model is always used for a specific purpose, simplifying assumptions on less important elements for this purpose, or regarding tractability, complexity and solubility are always adopted when translating reality into mathematical models (Provost & Fawcett, 2013). These assumptions may relate to the values of parameters, the shape of processes or even the number of uncertain variables that are considered.

In practice, both the demand (price) and supply (cost) side are uncertain. However, in RO models, often one side is modelled as and hence assumed to be deterministic. With-

out this, additional assumptions are needed about the interaction between both uncertain variables and simulations might be needed to solve the models. Moreover, it proves difficult to account for behavioural reactions to new decisions and policies in forecasts (Meersman & Van de Voorde, 2019). Finally, reality encompasses important unknown unknowns, uncertainty nobody has ever thought about (Thanopoulou & Strandenes, 2017). They also have an impact on the correctness of forecasts and parameter estimations and hence the applicability of the developed models. Notwithstanding these challenges and limitations, with acceptable parameters values and reasonable patterns for the uncertain variables at hand, investors and their externally hired econometricians should be able to implement the scientific insights from this thesis in their investment strategy. Nevertheless, caution is necessary when interpreting and using this thesis' findings in practice. To this end, the assumptions underlying each of the models, limited as much as possible to make the models as realistic as possible, are summarised in Table 1.2. The assumptions are discussed in greater detail and extended with some chapter-specific assumptions in the relevant chapters.

1.9 Structure of the thesis and research sub-questions

The thesis is divided into eight chapters in order to deal with a number of sub-questions derived from the main research question: How are optimal port capacity investment decisions influenced by changes in economic, port- and project-related characteristics, both theoretically and in practice?

The next chapter, Chapter 2, further builds on and extends the available literature on port uncertainties and RO models by formulating an answer to the sub-question: Why and how are RO suited to evaluate and decide on port capacity investments? Answering this question leads to uncovering the issues present in currently used methods and the sources which cause them. Subsequently, RO are explored in detail as an appropriate methodology to study flexible port capacity investment decisions under uncertainty. This chapter adds to the body of knowledge by developing an original framework to analyse port investments under uncertainty with practical implications, as advocated by Musso et al. (2006).

In Chapter 3, the first port RO model of this thesis is developed to deal with a one-shot or once-and-for-all investment in a new port, fully operated and owned by one actor: the privately owned port authority. This base case model, the benchmark for the subsequent chapters, allows answering two sub-questions: What is the impact of the degree of waitingtime aversion of port customers on a port's optimal capacity investment decision in a new private service port under uncertainty without competition? and What is the impact of the amount of uncertainty on a port's optimal capacity investment decision in a new private service port without competition? From a theoretical point of view, the answer to the first question extends the current state-of-the-art. In practice, only a few ports in the United

	3 Data 2000	4 Multiple actors	E Tatha sout	
	3. Base case	4. Multiple actors	5. Inter-port competition	6. Time to build in port expansion
Project type? Canacity constraint?	New port Hard constraint	New port Hard constraint	New ports Soft constraint	Port expansion Soft constraint
Port type?	Private service port*	Public and/or private,	Public and/or private	Public and/or private
Dort compatition?	No.(1 nort in model)	service* or landlord port**	Service ports* Duonolar Ctachelhora and	Service port*
LOUT COULDENTION:			Cournot competition	
Time to build? Dhasad innestment?	Not considered Droiact installed at once:	Not considered Droiact installed at once	Not considered Droigert installed at once	Included in the model Droigert installed at once.
Theory of the second second second second	available after comple-	available after comple-	available after comple-	available after comple-
	tion	tion	tion	tion
General assumptions	Price (sum of port dues and	d terminal tariff) transparenc	y → Linear additive inverse c	demand function
	Demand uncertainty, mod Growth market with $r > \mu$	lelled using a GBM with const > 0, to ensure RO model conv	tant, independent growth (μ) ^r ergence***) and uncertainty (σ) parameters
	Poisson-distributed ship an	rrivals at the port: delays star	t at about 50% and increase s	sharply beyond 75-80% occupancy
	User congestion costs dept	end on dock users' average va	alue of time, which evolve wit	th the market (i.c., increasing)
	Deterministic linear control	ig fourth-order polynomial wi	ith economies of scale and su	ze boundary
	Deterministic, miear opere Flavihility of invæstment fir	auonal and capacity nouning (custs sethone the investment	
	Once-and-for-all investme	and accision with flexible through	ughput levels	
	Infinite project life time (m	naintenance executed)		
		*: Owned and operated by one s	single actor.	
***: Condition r	$> \mu$ leads to a finite timing decision	a by the FA, operated by one FUC (at n, instead of infinite investment pos	stponement. Sensitivity tested with	n negative growth rates as well.
SCOPE: Theoretical design	n capacity for container throughpu	tt services. Other port services disre	garded (sufficiently present or reali	ised by actors other than the PA or TOC).
	Focus on lumpy invest	tment in port infrastructure and irre	eversible elements of superstructur	.e.

Table 1.2: Overview of the next chapters' model assumptions.

Kingdom and in New Zealand can to some extent be classified under the private service port model (Turpin, 2013). Moreover, this model is especially useful for the development of new private service ports in developing regions, such as Africa, where port competition is moreover limited. Port competition is also limited on relatively small islands. In Sardinia for example, the port of Cagliari is the only substantial container port. Nevertheless, since in reality the majority of ports are landlord ports, in which public money is involved, the model of Chapter 3 requires some extensions in the subsequent chapters.

Chapter 4 studies the investment in one new port that can also be partly or fully publicly owned and where two investing actors are considered (TOC and PA). The majority of newly developed ports adheres to this structure (Turpin, 2013), see e.g., the port of Gioia Tauro in Italy during its development phase (Genco et al., 2012, p. 10) and the port of Luanda in Angola (Porto de Luanda, 2018), with one PA and one TOC and without competition from nearby ports. The developed game allows answering the sub-questions: How are the investment decisions of the PA and the TOC in new port capacity influenced by each other's decisions under uncertainty without competition? and How are the investment decisions influenced by public PA ownership under uncertainty without competition? This innovative way of modelling different port actors under uncertainty adds to the body of knowledge, as it allows discerning the impact of different concession fee strategies on port investment decisions.

Competition from nearby ports has an impact on a port's investment decision too. If a nearby port invests earlier in additional capacity to alleviate congestion, the port that did not yet invest in new capacity may remain congested. Due to the waiting-time aversion of the customers, they might leave the congested port and move cargo to another port. As a result, expanding the port might no longer be required or profitable. A consequence might be that the congested port foregoes a lot of throughput and revenue. This is an important reason why competition between ports pushes them to anticipate their infrastructure investment when compared with a monopolistic case. These considerations have been disregarded in the first two models. For this reason, those strategies of preemption and entry deterrence are studied in Chapter 5. A second port, which will enter the market by installing a new port too, is added to the first model. Since ports are heterogeneous and for instance encounter different investment costs, one of the two can have an investment cost advantage, which leads to it becoming the leader in the investment timing game, unless the leader's capacity decision allows the follower to invest simultaneously (Pawlina & Kort, 2006). The sub-question answered in this chapter is: What impact does inter-port competition have on the investment decision of both the leader's and the follower's new port under uncertainty? Here, the scope is limited to service ports again, as the aggregate optimum from a landlord port can be obtained by optimising a service port model. However, the PA in this model can be owned privately, publicly or by a combination of both. Hence, this model is especially suited to study the investment decision in two new competing service ports, e.g., in New Zealand. From a theoretical point of view, Chapter 5 adds theoretical insights to the port investment literature considering congestion, competition and uncertainty in a

more realistic model that allows jointly analysing the timing and size of the port investment decisions.

Port expansion and time to build have not been considered in the previous chapters. However, the majority of port capacity investment projects involve expansion projects, and take considerable time to be completed. Examples include the extensions of the port of Antwerp or the port of Rotterdam with new docks. In Chapter 6, the first service port model is extended with these two important characteristics of port capacity investment projects. In this already active service port, where competition is again left beyond the scope to focus on the unrestricted investment decision, the following sub-questions are answered: What is the impact of time to build on the optimal capacity investment decision under uncertainty without competition? and What is the impact of existing capacity on the optimal capacity investment decision, with new docks added to an existing port, is considered.⁸ The results of this chapter show that the previous findings do not necessarily hold for expansion projects. Since time to build plays an important role in the considered port investment projects, these new insights are very relevant for practical applications too.

Another element that has not been considered in these models, is phased investment. The investment in Maasvlakte 2 in the port of Rotterdam involved such a phased investment option. Although port expansion is already one step closer to reality than a one-shot investment, considering phased investment in the RO model would involve another increase in realism. Nevertheless, as is explained in Section 2.4.4, this involves a considerably different modelling approach, for which this is left for follow-up research.

In each of the model chapters, sensitivity and robustness of the previous findings with respect to the added characteristics are included. Moreover, extensive numerical simulations are also provided to guarantee the robustness of the results. The interrelatedness of these chapters is displayed graphically in Figure 1.3, as a guide for the reader to the next chapters. Each of the developed models is based on some underlying assumptions, which are relaxed in subsequent chapters to add realism to the models. Port managers may gain insights and benefit from the model with the assumptions closest to their own situation. However, when their situation requires the combination of multiple models, insights of the different chapters need to be combined.⁹ Also mathematical elements of these models can be combined in a new RO model, as is shown in Chapter 2.

In order to be able to implement the theoretical findings of these model chapters, a blueprint of a practical tool is developed in Chapter 7. This provides a first step to implement the findings of this thesis in real-life port capacity investment decisions. Moreover, a

⁸This is opposed to brownfield (expansion) projects.

⁹For example, the derivation of the PA and TOC's individual investment optima and the impact of a competing port are derived from the different models in Chapters 4 and 5. These insights can be applied to an optimal port expansion decision of one service port, derived from Chapter 6's model.



Figure 1.3: Overview of the relationships between the different RO models in the different chapters of this thesis.

critical reflection on the potential gains and costs of implementing RO is provided. In this chapter the sub-questions answered are: (RQ 7.1) How can the developed RO models be implemented in reality? and (RQ 7.2) What are the potential gains and costs of this approach, compared with currently used approaches? An overview of all research sub-questions, related to this thesis' research question (RQ) is shown in Table 1.3.

The final Chapter 8 brings the answers to the sub-questions together in order to answer the main research question. Moreover, the added value, implications and limitations of this research are linked to these answers. This finally allows conceiving potential avenues for future research.
 Table 1.3: Overview of this thesis' research question and sub-questions.

RESEARCH QUESTION: How are optimal port capacity investment decisions influenced by changes in economic, port- and project-related characteristics, both theoretically and in practice?

Literature	Chapter 2 sub	r 2 sub-questions:	
Review	RQ 2.1	Why and how are RO suited to evaluate and decide on	
		port capacity investments?	
New	Chapter 3 sub	b-questions:	
Ports	ts <i>RQ 3.1</i> What is the impact of the degree of waiting-		
		sion of port customers on a port's optimal capacity in-	
		vestment decision in a new private service port under	
		uncertainty without competition?	
	RQ 3.2	What is the impact of the amount of uncertainty on a	
		port's optimal capacity investment decision in a new	
		private service port without competition?	
	Chapter 4 sub-questions:		
	RQ 4.1	How are the investment decisions of the PA and the TOC	
		in new port capacity influenced by each other's deci-	
		sions under uncertainty without competition?	
	RQ 4.2	How are the investment decisions influenced by public	
		PA ownership under uncertainty without competition?	
	Chapter 5 sub	o-questions:	
	RQ 5.1	What impact does inter-port competition have on the	
		investment decision of both the leader's and the fol-	
		lower's new port under uncertainty?	
Port	Chapter 6 sub	b-questions:	
Expansion	RQ 6.1	What is the impact of time to build on the optimal ca-	
		pacity investment decision under uncertainty without	
		competition?	
	RQ 6.2	What is the impact of existing capacity on the optimal	
		capacity investment decision under uncertainty with-	
		out competition?	
Implemen-	Chapter 7 sub	b-questions:	
tation	RQ 7.1	How can the developed RO models be implemented in	
		reality?	
	RQ 7.2	What are the potential gains and costs of this approach,	
		compared with currently used approaches?	

Chapter 2

Towards improved port capacity investment decisions under uncertainty

Port capacity investment decisions involve considerable uncertainty for the investing port actors, such as port authorities (PA), terminal operating companies (TOC) and governments (Kakimoto & Seneviratne, 2000b; Musso et al., 2006). They arise from many sources: different economic decisions taken by many logistics actors in competitive logistics chains; unpredictable income from trade, uncertainty about the amount of seaborne traffic and possible resulting congestion; uncertainty about new technological, environmental, political and legal developments (Meersman, 2005; Giuliano, 2007). Variability, uncertainty and managerial flexibility are not fully taken into account in traditional valuation methods such as net present value (NPV) and internal rate of return (IRR) analyses, which are often used in cost-benefit analyses (Trigeorgis, 1996; Anda et al., 2009). As Dixit & Pindyck (1994) noted, even if uncertainty is low or absent, the NPV rule can be (very) wrong, due to its underlying assumption that investing in an irreversible project is a now or never decision. In reality, managers have the flexibility to postpone investment to a better moment, e.g., in order to gain more information about the uncertain environment. This underlying option has a value, that is overlooked by the NPV rule.

As opposed to the traditional valuation methods, real options (RO) models offer a more appropriate method to evaluate flexible investments under uncertainty and make better investment decisions. RO models monetarily quantify the value of managerial flexibility to react to uncertainty in the best possible way. T. Wang & De Neufville (2005) discern two different types of real options. RO 'on' projects relate to the correct valuation of flexible investment opportunities. RO 'in' projects and the related models take the benefits of options of flexibility embedded in the specific project into account. Only when all relevant options of flexibility are included in project appraisal, the investment decision can be valued correctly. To this end, RO models use stochastic calculus and dynamic programming to optimise an objective function (such as the value of the firm), including a random term related to an investment decision variable (Dixit & Pindyck, 1994). It is however important to point out that RO models should not necessarily be seen as a replacement for NPV with discounted cash flows. It is complementary to it, because the value of a project can be calculated as the static NPV plus the value of the real options present (Hu & Zhang, 2015; Van Putten & MacMillan, 2004). The disadvantages of the RO approach are however its complexity, the need for even more data and parameter and function estimations. Its current applications to ports are limited.

The objective of this chapter is to discern a methodology that is able to correctly value and to enhance port capacity investment decisions under uncertainty. This is translated into the following research sub-question: (RQ 2.1) Why and how are RO suited to evaluate and decide on port capacity investments? The answers to this question are brought together by constructing a framework to further develop the RO methodology in a port context.

This chapter consists of six sections. In the next section, the relevant literature on uncertainties and risk factors is reviewed, since all of these elements have an impact on port capacity investment decisions. Some of the existing RO applications in a transportation context are discussed in Section 2.3. In order to allow building RO models to decide on port capacity investment decisions, building blocks and characteristics of general existing RO models relevant to the port context, are reviewed in Section 2.4. The insights from the reviews of both the port uncertainty and the RO modelling domain are combined in Section 2.5.1 in a port capacity investment RO modelling framework. This framework offers an overview that will support the RO model building for port capacity investment decisions in the remainder of this thesis. Section 2.5.2 uses the framework to elaborate a port capacity investment decision model under uncertainty. The final section concludes with the major findings of this chapter and makes the link to the next chapters, which contain the actual models. These RO models are based on the framework. The underlying assumptions are based on the information from this chapter and the previous one.

2.1 Sources of uncertainty and risk in the port context

Many sources of uncertainty and risk are present in the port context. Although the concepts uncertainty and risk are sometimes used interchangeably, they are different. Many scientists have debated on the difference, often starting from the seminal work of Knight (1921). Knight (1921) argues that uncertainty is related to an unknown probability of outcome, whereas risk is linked to a known distribution of outcomes. Moreover, discussions about risk are often related to the potential loss and its insurability (Langlois & Cosgel, 1993).



Source: own composition, based on Meersman & Van de Voorde (2014a); Meersman et al. (2010); Vanelslander (2014); Vilko & Hallikas (2012); Cucchiella & Gastaldi (2006); Koetse & Rietveld (2012).

Figure 2.1: Actors, uncertainties and risks in the maritime logistics chain.

Meyer & Reniers (2016) use the risk definition of ISO 31000:2009, which points out that uncertainty has a positive or negative effect on one's objectives. In this way, the amount of and exposure to negative uncertainty can lead to not achieving objectives and hence incurring losses. This is defined as negative risk, which is the focus of this chapter. However, there are also positive uncertainties which may lead to (better) realising objectives and gains. In this chapter, when the outcome can be positive or negative, this is referred to as an uncertainty.

The sources of uncertainty and risk in a port can be classified in three categories: (i) the competitive position of a port in a global logistics chain, (ii) global evolutions in international trade and (iii) technological evolutions, increased environmental considerations and resulting changes in legislation. In order to link the sources of uncertainty and risk to the actors in the entire logistics chain, Figure 2.1 represents the logistics chain. The different uncertainties and risks are grouped per activity in the logistics chain. Additionally, the actors per activity are displayed on top, to allow linking the uncertainties and risks at the bottom to the respective actors. The interaction of actors induces additional uncertainty and risks. Nonetheless, forwarders often intervene in this maritime logistics chain by coordinating different steps of the shipping process, resulting in the elimination of many direct contacts between shippers and different shipping lines (Meersman & Van de Voorde, 2014a). This might at times lead to a reduction of uncertainty. Moreover, it should be noted that many other activities bearing risks are performed in the port at the operational level and are not included in Figure 2.1, such as mooring, tugging and piloting (Blomme, 2014). Finally, general and unforeseen uncertainties that can have an influence on investment projects, such as tsunamis, earthquakes or terrorist attacks, are not discussed in further detail in this chapter either.

2.1.1 Competitive position of the port: global logistics chains

Ports are part of global logistics chains with a number of actors involved. The chain can be divided in three sections: maritime transportation, transferring cargo from sea to inland transportation in the port and hinterland transportation (Kapros, 2014). Maritime logistics chains are well-oiled machines (Van de Voorde & Vanelslander, 2014). Hence, for a port to be competitive, its connectivity in the transportation chain is vital (Jia et al., 2017). Such chains are multimodal, as it is necessary to switch between different transport modes, e.g., road, rail and maritime transport. This induces internal variability and risk in the demand for different transportation modes (Vilko & Hallikas, 2012). Decisions of one actor in the chain can have major implications for the other actors (Heaver, 2011; Slack & Frémont, 2005; Verhoeven, 2015). At first sight, they may seem insignificant but subsequently, they might cause a chain reaction creating bottlenecks that affect the competitive position of a port negatively (Panayides, 2006; Rodrigue & Notteboom, 2009; Van Der Horst & De Langen, 2008). Variability moreover complicates the decisions taken by the different owners in the chain (Cucchiella & Gastaldi, 2006). Finding out the most vulnerable links in the chain can help to efficiently manage these chains wherein goods are to be transported and information is to be shared. In order to steer the chain, cooperation and communication are vital (Vilko & Hallikas, 2012).

In the port, many different intertwined actors come together and influence the decisions taken there (Verhoeven, 2015). Coppens et al. (2007) identify port authorities, the government, shipping companies, terminal operating companies, shippers, integrators, forwarders, other logistics and transportation firms, brokers, customers at destination, production firms, project teams, regions and banks. The relationships between these actors can be rather complex (Estache, 2001), as they might have different, conflicting objectives. For example, a shipper wants to minimise the generalised transportation cost or the entire production cost,¹ including the port price and waiting time in a port, whereas a shipping company wants to maximise profits or its market share and a terminal operator wants to maximise its profit by reducing concession prices and increasing handling prices. The port authority in turn wants to ascertain the competitive position of the port. In addition, the port authority requires a sufficient business and/or welfare ROI to be generated in the port (Meersman & Van de Voorde, 2014a; Y.-T. Chang et al., 2012).

The complexity of the relationships between these actors, which hampers a correct specification of the objective functions of the port capacity investment decision makers (Xiao et al., 2012), is reinforced by the unequal distribution of power between them. As Van de Voorde & Vanelslander (2009) state for example, the concession policy is for the port authority one of the few instruments to express their negotiation power over TOCs. The negotiated concession fee needs to be taken into account in the objective functions. Moreover, shipping companies have a lot of decision power in the maritime chain too, since they

¹Depending on the shipper's specific activities and cost function.

decide on the ports of call for large throughput volumes. Their organisational structure is changing towards more concentration (Kauppila et al., 2016). In addition, shipping lines may invest in dedicated terminals, in order to reduce waiting times by preventing being served on a first come, first served basis (Heaver et al., 2001). This reduction of port and hinterland congestion is an objective of the shipping company, as well as the PA and the TOC. As a result, the impact on port capacity needs to be considered (Strandenes, 2014; Meersman et al., 2010).

The question could be raised whether port authorities aspire cost minimisation or profit maximisation, or even a different objective (Y.-T. Chang et al., 2012). The objectives of port authorities are influenced and complicated by their mixed ownership structures and competition inside and between ports (Cullinane et al., 2005). In some cases, they could be captured by a multiple objective function. As Suykens & Van de Voorde (1998) indicate, not only private, but also public money plays an important role in port ownership. This leads to the consideration of social welfare. Xiao et al. (2012) define such a multiple objective function to deal with the different actors in the port. Different weighted objectives are combined for the private and public actors involved. Together, they can realise the objectives of both parties, which are profitability for the private actors and social welfare for the public actor (Musso et al., 2006). A private investor aims at maximising profit:

$$\pi = (p-c)q - c_h K, \tag{2.1}$$

with *p* the unit price, *c* the unit cost, *q* the output quantity, c_h the capacity (or capital) holding cost of one unit of capacity (to hold capacity in place, e.g., through maintenance) and *K* the total available capacity, which forms the upper bound of *q*. A local government considers the sum of this profit and the spillover effects for the local economy, λq , with λ a constant scale factor. Finally, a central government additionally includes total consumer surplus in its value function. The objectives of the different actors are then weighted and summed in a global objective function, which is to be maximised. The delay *D*, caused by congestion, is defined by a function of the capacity utilisation rate q/K:

$$D = f(q/K). \tag{2.2}$$

Delay is included in the analysis of Xiao et al. (2012) as a user cost. Xiao et al. (2012) find that a higher share of ownership for the private investor results in lower capacity investments and higher prices. Indeed, when positive spillover effects of port business are not taken into account, capacity investment will be smaller (Asteris et al., 2012).

The weights of the different objectives in the approach of Xiao et al. (2012) are defined by the share of ownership of the different parties in the port. This is a good first approximation when voting power is directly related to the shares of ownership of the different parties. In some ownership structures (e.g., a majority shareholder or public-private partnerships) or when dealing with dominating parties however, this approach might not be a correct representation of the decision making process. In such a case, only the objectives of the decision makers with influence need to be retained. A second limitation of the approach of Xiao et al. (2012) encompasses the omission of demand variability.

Competition has an impact on the chain as well (Notteboom & Yap, 2012). Competition between ports increased over the last decades, because captive market areas became less incontestable, increasing the overlap of hinterlands of the ports (Musso et al., 2006). Competition does not only exist within one chain, but also between different chains. First, the different actors in the chain might compete internally to have the power to influence the decisions made in the chain (Meersman & Van de Voorde, 2014b). Second, different TOCs and port authorities might compete to be part of a specific logistics chain. It is important to be part of the chain with the lowest generalised cost, because the goods are more likely to flow through this chain (Meersman & Van de Voorde, 2014a). This in turn leads to competition between different chains to actually have the lowest generalised cost and attract the highest possible throughput volumes. This competition induces additional uncertainty and risk in the port. To account for this competition between ports, many models use a competitive game with two or more ports (Zhuang et al., 2014; Xiao et al., 2012; Saeed & Larsen, 2010; Zhang, 2008).

2.1.2 Global evolutions in international trade

The amount of throughput handled in a port depends on international trade in three ways. First, the total amount of trade plays an important role, and is influenced by the macro environment (see Figure 2.1). Future levels of world trade flows are very uncertain and volatile, although they are expected to grow, leading to an increase in goods shipped over seas (UNC-TAD, 2015). Second, the geographical distribution of these trade flows can change, resulting in different routes for the ships. Third, the composition of trade and the selected transport modes can change too, influencing among other things containerisation rates. This uncertainty in global trade has a negative impact on forecast accuracy. Accurate forecasts estimating future port demand are required to obtain valuable outcomes of NPV analyses, since the latter cannot correctly account for uncertainty (Kauppila et al., 2016; Meersman & Van de Voorde, 2014b; Blonigen & Wilson, 2008; Jacks & Pendakur, 2010). Oppositely, RO models valuing flexibility have the advantage of being able to account for the uncertainty (de Neufville, 2016). This is because RO explicitly model uncertainty and account for it in the calculations based on stochastic calculus. Nevertheless, the more exact the forecast of demand, the more accurate the outcome of the RO analysis will be too.

2.1.3 Technological change: environmental and legal uncertainties

Technological and environmental developments are another prominent macro-environmental factor influencing the port. Asteris et al. (2012) describe ports as an industry facing numerous technological developments, internally as well as externally. Vessel size increases, new routes become available (e.g., through widening the Panama Canal, the North Sea Route to Asia, etc.) and the environmental impact of technology and the investment herein under uncertainty (e.g., LNG fuel) gain increasing attention (Kauppila et al., 2016; Verbruggen, 2008; Acciaro, 2014a,b). Koetse & Rietveld (2012) prove that climate change can add to the uncertainty and risk encompassed in transport infrastructure investments. Next to transport inducing climate change, there also exists an influence of climate change on transportation. More extreme weather conditions can pose a problem to different transportation modes, even more than a change in the average condition. An interesting example in this context is the change in the sea level, complicating certain port activities. This in turn could eventually result in modal shifts for regional and global goods transportation, altering the demand for some types of transport infrastructure and increasing demand variability. All of these technological innovations and environmental impacts, in combination with the aforementioned variable economic demand, add to the already present port uncertainty.

Since technological and environmental impacts, but also safety and security gain more interest, new legislations are constituted to protect the goods traded, the workers and the port environment (Harrald et al., 2004). Also the environmental impact of ports on climate change can be addressed by governments imposing additional legislation. Furthermore, changes in labour regulations may induce additional uncertainty (Barton & Turnbull, 2002). An example of typical port labour regulations are laws organising which workers may perform which type of work in a port area.

2.2 Real options models for better decisions under uncertainty

Port capacity investment analysis is a field where RO can be a very powerful methodology, as was already suggested by Juan et al. (2002) and Herder et al. (2011). RO show great potential to improve project valuation and decision making under uncertainty in large-scale, complex, irreversible projects involving long planning and construction periods, which are influenced by technological shifts and opposing objectives of different stakeholders involved (de Neufville, 2003). A lot of port infrastructure investment projects have these characteristics, as was discussed in Section 1.6. Moreover, the port environment itself is complex, as it is characterised by competition and uncertainty (see Section 2.1), especially because ports form a link in the maritime logistics chain, involving intertwined actors with possibly conflicting objectives (Heaver et al., 2001; Meersman & Van de Voorde, 2014a).

As a result, the value of port projects can be increased by means of flexible size and timing of the investments and through flexible options embedded in the project elaboration (Hull, 2012). Ports that want to excel in a dynamic world need to incorporate options to allow reacting flexibly to changes. RO models attribute value to such flexibility (Taneja, 2013; Van Putten & MacMillan, 2004). Moreover, uncertainty is considered as an opportunity, not a threat, because managers can adapt their strategies to internal and external changes. Literature even indicates that higher demand uncertainty results in larger capacity investments (Dangl, 1999; Guthrie, 2012; Xiao et al., 2013).

An advantage of RO beyond its included flexibility and decision making realism, is the possibility of a stepwise elaboration of the mathematical model. Trigeorgis (2005) shows that RO can be approached as building blocks, where different options could be interrelated and interacting with each other. Some important options to be considered are: the option to expand the project, possibly in stages; the option to defer (timing flexibility); the option to switch to other activities or goods categories handled; the option to adapt output levels; and the option to fully shut down operations. Such RO models are often expanded. In an initial stage, only one firm might be considered. In a following stage of the research, the model might be extended for example to more firms operating under competition (Xiao et al., 2013).

In order to explore in more detail how RO can improve port capacity decision making, some previously elaborated RO applications in a transportation context are first identified in the next section. Afterwards, the focus is narrowed down to a port context.

2.3 Real options in the transportation sector

Real options have been applied to many transportation investments, because they can deal very well with managerial flexibility under different sources of uncertainty. The models discussed in this section relate to the maritime sector, to other modes and to environmental projects. When transportation infrastructure is funded by the public government, the social welfare function has to be maximised, complicating the analysis, because now more variables have an impact on the value of the project (Szymanski, 1991). Chow & Regan (2010) exemplify this when applying RO to transportation network investments. Tibben-Lembke & Rogers (2006) show how financial options could be adopted as transportation options to hedge logistical contracts.

Real options have amongst others been used in toll road investment analyses (Ashuri et al., 2011). In aviation too, RO have been applied. Hu & Zhang (2015) for example have analysed airline investment decisions. A lot of interesting and relevant work about airport infrastructure investments under uncertainty is available (Smit, 2003; Zhang & Zhang, 2003, 2006; Xiao et al., 2013, 2017). As there are some similarities between ports and airports, these papers allow drawing useful conclusions to be considered in the port context. Xiao et al. (2017) for example find that for an airport it is important to buy enough reserve capacity to expand at a later moment. When uncertainty increases, the option value is higher, explaining this need to invest in reserve capacity. This finding may prove important for ports

as well. Nevertheless, there are some important differences between the described context for ports and airports. These differences should be correctly taken into account before transferring results from one sector to another.

A first important similarity is that airport capacity expansion is lumpy and involves a long lead time as well. Moreover, a lack of sufficient infrastructure, compared to demand, leads to capacity shortages in both ports and airports. When there are not enough runways, terminals or gates in an airport, delays will occur as well. Third, also in the airport, multiple actors interact. However, their interactions are different, especially when maritime transportation of goods is compared to passenger air transportation. This leads to a first distinction. As far as maritime transportation of goods is considered, the shipping line effectuates the transportation and decides through which ports the goods are shipped. In practice, these will be the ports with the lowest generalised cost. In passenger air transportation, the airline effectuates the transportation, but it are the passengers who choose the airline and decide on the origin and destination airport. This important transportation decision making difference should also be expressed in the specific models for ports and airports. Moreover, price transparency is higher in passenger transportation than in goods transportation (Onghena, 2013). A second difference is linked to the investment decision making process. Notwithstanding the more complex transportation decision making process, the investment decision making process in the airport involves fewer actors. The airport holding company is often the main actor deciding about the investment in new capacity. In a landlord port however, both the TOC and the PA have an impact on this investment decision. A final important distinction is related to the concessions in ports and airports. While concessions in ports are often related to the provision of products or services linked to the transported goods, concessions in airports are also related to the generation of nonaeronautical or commercial revenues from e.g., retail and real estate. As shown by Gillen (2011), airports are two-sided platforms, providing both aeronautical and non-aeronautical services to airlines and passengers.

There have been many applications of RO to maritime investments as well. Bendall & Stent (2003, 2005, 2007) and Dikos (2008) show how investing in a new ship or maritime technology can incorporate an option value. Bjerksund & Ekern (1995) discuss the value of mean-reverting cash flows through contingent claim analysis, applied to time charters in shipping. Sødal (2006) and Balliauw (2017) use RO to study entry and exit decisions in shipping markets. Rau & Spinler (2017), as an extension of Rau & Spinler (2016), discuss the performance of RO modelling in a cooperative container shipping game concerning alliances. Moreover, Sødal et al. (2008) use RO to value the option to switch operations in combination carriers. These examples confirm that RO models offer a good approach to valuing volatile and capital-intensive projects as in the maritime business. This is because projects in the maritime business are characterised by financial risks, as revenues are uncertain and situated in the distant future (Kakimoto & Seneviratne, 2000a). As a result and because the market environment changes continuously towards more integration (Bergantino & Veenstra, 2002), accurate forecasts are also indispensable in maritime economics (Meersman et

al., 1997).

RO analysis has been used to evaluate alternative technologies dealing with environmental issues as well (Kim et al., 2009; Koetse & Rietveld, 2012). RO are well suited to deal with the uncertainty associated with climate change and the effectiveness of technological innovations (Acciaro, 2014a). RO demonstrate the option value in postponing the technological investments. However, adapting to climate change during regular maintenance or retrofitting could be beneficial pro-active actions (Cullinane & Bergqvist, 2014).

2.4 Real options models from a port perspective

Herder et al. (2011) already showed that real options can be used in ports to decide on the optimal timing and size of a capacity project, including flexibility in terms of investment in and use of the infrastructure and to open the option of postponement and abandonment. This section contains a critical overview of existing RO models to analyse decisions with a number of similar characteristics. The advantages and drawbacks of the models and their constituting building blocks are discussed in light of inclusion in a new port investment RO model. This model has to take the specific, complex port context into account. An overview of models to find the optimal size and timing of investment projects is supported by the work of Huberts et al. (2015) and also includes game-theoretic approaches. This is followed by the discussion of a model incorporating cyclicality, because demand in maritime markets is cyclical. In ports, such growth cycles are estimated to have a duration of 8 to 10 quarters (Paflioti et al., 2015; Stopford, 2009).

Novaes et al. (2012) analysed the development of terminals and berths under demand uncertainty and economies of scale in a developing country. In that paper, capacity is considered as a hard constraint and relevant insights from queuing theory are taken into account. Although engineering approaches are relevant for port capacity investment decisions, they should always be supported by economic considerations, such as revenues and costs generated by the project. This is lacking for example in Kia et al. (2002) and Noritake & Kimura (1983). Chen & Liu (2016) and Chen et al. (2017) in their analyses of port infrastructure investments prove that uncertainty alters the investment decision a lot and needs to be included in the port capacity investment analysis, as well as congestion costs. Although the latter two papers consider uncertainty, as modelled by a uniform distribution, it is possible to go one step further by taking an uncertain process that allows modelling expected growth (drift) and uncertainty independently. K. Wang & Zhang (2018) apply RO to the investment in abatement technology to reduce the impact of natural disasters, resulting from climate change, in a port with inter and intra-port competition. To this end, they assume the probability of the event of a disaster following some ambiguous probability density function and hence being unknown ex-ante, but known after the investment decision (Knight, 1921). This Knightian uncertainty is compared to a Poisson jump to model

the uncertain occurrence of the event (see Section 2.4.6 for a discussion of Poisson or jump processes).

This section's review of RO models in the literature is the basis for developing new models suitable to the specific, complex port context with its considerable uncertainties and demand characteristics. In order to model investment decisions as realistically as possible, without relying on too many simplifying assumptions, the relevant decision variables need to be included in the port model. Nevertheless, some variables need to be kept exogenous in order to have a solvable model with sufficient explanatory power. The included decision variables alter the outcomes of the model and the impact of uncertainty. For example, considering timing and size of an investment together could be more realistic in some cases and would lead to results that differ from models with only one single decision variable.

2.4.1 The basic real options model of Dangl and its extensions

An initial RO model to decide on the size and timing of capacity investments is available in Dangl (1999). His continuous-time continuous-state model starts from the linear (in Q) additive (in X) inverse demand function

$$p(t) = X(t) - Bq(t),$$
 (2.3)

with the following variables defined at time t: p(t) the price, X(t) the demand shift parameter, intercept or reservation price that follows a geometric Brownian motion (GBM), B the (negative) slope and q(t) the output quantity. The increment of a GBM is defined by

$$dX(t) = \mu X(t)dt + \sigma X(t)dZ(t), \qquad (2.4)$$

with μ the drift (a measurement of growth) parameter, σ the volatility (a measurement of uncertainty) and *Z* a standard Wiener process. For convergence of the model, it is required that the risk-free discount rate $r > \mu$ (Dixit & Pindyck, 1994). Although the rate of uncertainty, σ , can vary in practice, it is assumed fixed in RO models based on a GBM. Instead of imposing uncertainty on the uncertainty parameter, decision makers are advised to reevaluate RO models as soon as changes in one of the parameters are suspected. As far as the GBM is concerned, Marathe & Ryan (2005) already empirically showed that the GBM is a good process to model demand for established services such as the number of passengers in airline transportation. Additionally, Lindsey & De Palma (2014) highlight the frequent use of a GBM in a transportation context.

In ports there are two possible ways to deal with capacity shortage: investing in additional capacity or increasing the output price (Xiao et al., 2012). As a result, it is important in a port context to jointly analyse the output price and quantity through a demand function when studying capacity investments (Aguerrevere, 2003; Vanelslander, 2014). Of course, the use of a demand function in the models assumes sufficient price transparency for the customers in the market for the considered port activities, which is not necessarily true in practice, as the price is set by a number of actors (Meersman et al., 2015). For the sake of clarity, time dependencies are omitted in the remaining formulae in this chapter, e.g., p instead of p(t).

Dangl (1999) models a flexible firm, since the firm can choose its output level, as long as it satisfies

$$q = \min\left\{q^{opt}, K\right\},\tag{2.5}$$

as opposed to a non-flexible firm, with $q \in \{0, K\}$. Ports are typical examples of flexible firms, as operating at full capacity is in general an exception (Rashed, 2016). According to Hagspiel et al. (2016), a flexible firm will invest more than a non-flexible firm, proving that flexibility adds value to the firm.

The cost of investment is given by

$$I = \gamma K^{\varepsilon}, \tag{2.6}$$

with γ a constant scale factor, K the total capacity and $\varepsilon < 1$ to indicate economies of scale in investment outlays and to exclude infinitesimal expansion, which would have a relatively high cost under the current specification. Profit is equal to

$$\pi = (p - c(K))q, \qquad (2.7)$$

where c(K) is the average operational cost, being dependent on the installed capacity K. In the numerical calculations of Dangl (1999) however, c is assumed constant. Equation (2.7) is a reduced version of the profit function used by Xiao et al. (2012) in Equation (2.1) for a port environment. Maximisation of this profit will lead to the optimal output q^{opt} at each point in time, given X and K. Subsequently, Dangl (1999) maximises the value of the firm, given by the integral of the discounted expected future profit streams minus the discounted investment cost. This forms the input of the option value maximisation. By using dynamic programming, an optimal investment strategy (X_T, K) is found. The first element indicates the optimal timing, expressed as an investment threshold for the demand shift parameter. As soon as X(t) reaches this threshold for the first time from below, or if X is initially above this threshold, investment takes place. The second element indicates the optimal amount of capacity of the investment.

The main conclusion of Dangl (1999) is that an increase in uncertainty will lead to investing in more capacity, but later. Bar-Ilan & Strange (1999) analyse the difference between lumpy and incremental investments. They confirm the findings of Dangl (1999), but only for lumpy investments, namely that uncertainty delays the investment and increases its size. Lumpy investments are defined by two properties: combinable investments and the presence of adjustment costs. Investments in port capacity are combinable, since adding a dock increases capacity. Moreover, there are adjustment costs in public infrastructure, such as planning costs (Szymanski, 1991). Exactly those costs lead to the presence of economies

of scale in the size of investment. So indeed, investing in port capacity is considered lumpy, as adding a dock means increasing capacity by a large leap (Kauppila et al., 2016). It is not possible to extend a dock's capacity just by a few TEU per year through infrastructure investments (Musso et al., 2006).

An extension to the model of Dangl (1999) is provided by Hagspiel et al. (2016), who add a capacity holding cost $c_h K$ to the cost function of Dangl (1999). The resulting profit function is similar to the one of Xiao et al. (2012) in Equation (2.1). However, the typical (port) congestion and delays are still omitted. The new profit function is given by

$$\pi = (p - c(K))q - c_h K.$$
(2.8)

2.4.2 Inclusion of port expansion in the model

A slightly different continuous-time continuous-state model to analyse the impact of scale economies on the investment decision is presented by Guthrie (2012). It takes into account a firm that can undertake subsequent capacity expansion investments. For a port, this could be achieved by means of adding multiple docks, one after another. The inverse demand curve used in this model is not linear additive, but has a constant price elasticity of demand (-1/a). It is given by

$$p(t) = X(t)q(t)^{-a},$$
 (2.9)

with X(t) again following a GBM, that could be adapted for risk-aversion by subtracting a risk premium.

The investment cost function *I* is given by

$$I = \left(\Delta q\right)^{\varepsilon},\tag{2.10}$$

with $\varepsilon < 1$, leading to economies of scale and preventing infinitesimal capital expansions. This investment cost function is similar to the one of Dangl (1999), although Guthrie (2012) sets γ to 1 and substitutes K by the amount of capacity expansion Δq , which is the difference between the new and previously installed capacity. Output in this case is always set at maximum capacity, i.e., output inflexibility, because a < 1 makes this the optimal decision in every circumstance. This is related to the non-flexible firm of Hagspiel et al. (2016), where output was also equal to full capacity, or zero. Another element that is included in the model, is depreciation:

$$\mathrm{d}q = -\zeta q \,\mathrm{d}t,\tag{2.11}$$

with ζ the depreciation rate. The objective function is the market value of the firm and is to be maximised. It is calculated as the integral of the discounted net operating incomes minus the sum of the discounted capital investment costs. In addition, the expansion decision is analysed in terms of scaling up the existing capacity by a scale factor κ . This is possible because of the homogeneity of the objective function.

The outcome of maximising the firm's market value is a set of three decisions $(\hat{y}_0, \kappa, \hat{y}_b)$. \hat{y}_0 defines the initial capacity investment. κ and \hat{y}_b define the subsequent capacity expansion strategy. κ is the scale of each expansion (related to the current capacity level) and \hat{y}_b is the return on assets threshold indicating the investment timing. The return on assets is calculated as

$$y = pq/q^{\varepsilon} = q^{1-a-\varepsilon}X, \qquad (2.12)$$

where q^{ε} is the replacement cost of the assets in place. It is noteworthy to indicate that investing will decrease the actual value for \hat{y}_b again through an increase in q.

The main conclusion of Guthrie (2012) is in line with the findings of Dangl (1999): increased uncertainty will lead to larger investment outlays that are in turn less frequent. Next to output inflexibility, another limitation of the model is that capacity expansion is assumed to be realised instantaneously, which is not in line with the long lead times incurred in port infrastructure construction.

2.4.3 The impact of time to build

Aguerrevere (2003) includes time to build in his RO model of capacity expansions, a project characteristic that was not included in the previously discussed models. He uses the same demand curve as Dangl (1999). Aguerrevere (2003) uses the demand function to explicitly study the resulting pricing decision for a chosen output level of a non-storable product. This could be useful in a port to shed some additional light on the complicated port pricing decision.

Aguerrevere (2003) uses an operational cost function that differs from the one of Dangl (1999):

$$c(q) = c_1 q + \frac{1}{2} c_2 q^2, \qquad (2.13)$$

with *q* being flexible, but limited by the available operational capacity. As a result of flexible output, this model considers overinvestment in capacity less harmful than underinvestment, which is a direct application of the inequality of Jensen (1906). Indeed, there is an imbalance between the two sides of uncertainty, as the capacity investment decision in combination with flexibility rather has to be considered as a determination of the maximum output threshold.

The investment cost in this model is given by Equation (2.6), with ε set to 1. Time to build is modelled through three functions: O(t) represents the amount of capacity present at time t, N(t) is the amount of capacity that is under construction with lead time θ and K(t), the committed capacity, is the sum of the previous two. K(t) is non-decreasing over time, because of investment irreversibility and the omission of depreciation in the model.

An important insight of Aguerrevere (2003) is that capacity investments increase with

uncertainty when time to build is included in the model. The same logic applies when the time lag increases. The results explain why expansion takes place even when current capacity is not fully used. It also turns out that when capacity is fully used, the pricing behaviour as a result of increased demand will be entirely different. The price will rise much more following a demand increase under full capacity utilisation than in a case where capacity is not fully used. This is a consequence of the limit that capacity places on throughput levels. As a result, a model like that of Aguerrevere (2003) could be used to quantify congestion pricing, which could be an important contribution to understanding the complex pricing strategy of a port.

2.4.4 Installing the project in stages

Building a new dock can be realised in different steps and should not necessarily be completed at once. This approach allows reducing the lumpy character of port investments, since the capacity can be installed in phases, and then also be brought into use gradually (Kauppila et al., 2016). A well-known example of the implementation of such a real option is Maasvlakte 2 in Rotterdam, where the water basin and part of the terminals were realised in the first stage. In the second stage, the rest of the terminals can be built, but only when the additional capacity is required. Also the port of Antwerp is considering expansion. Therefore, it considers digging a first phase of the Saeftinghedok and developing the terminals around it (Timperman, 2017). When additional capacity is needed, the second phase can be installed, by further expanding this dock and its terminals.

An important advantage of the phased investment approach can be a shorter lead time to add extra capacity, because part of the project has already been realised. The downside of such a project however is that in general, a premium has to be paid in terms of a higher total investment cost. Stage 1 has an investment cost I_{s_1} , whereas Stage 2 involves I_{s_2} and realising the project at once costs I_l . Hence in reality, $I_{s_1} + I_{s_2} > I_l$ will hold.

A model that confronts lumpy and stepwise investment is provided by Chronopoulos et al. (2017). They confirm the findings of Kort et al. (2010), who work with exogenous capacities and who calculate a maximum (relative) cost premium $(I_{s_1}+I_{s_2})/I_l > 1$ for which phased investment is more profitable. Chronopoulos et al. (2017) state that if the capacity of the project is determined ex-ante, lumpy investment is better under a high price uncertainty. However, if the final capacity is not fixed, stepwise investment leads to a higher value and more capacity will be installed.

2.4.5 Game-theoretic approaches

Most of the previous models only consider the investment decision of one monopoly entity. In reality however, there is competition between ports and even within ports to attract cargo flows (Meersman et al., 2010), since the logistics chain with the lowest generalised cost will be selected. The preference of shipping goods to Western Europe for instance through the port of Rotterdam or the port of Antwerp will be determined by the cost and the service provided by the port (i.e., transit time, the frequency of service for container lines, the amount of congestion, etc.), but also by switching costs (determined by existing contracts, the presence of liner-owned terminals and the speed of new information availability) (Veldman & Bückmann, 2003). The ports in the Hamburg - Le Havre range should hence not be considered as monopoly ports. They are in competition for the same market or hinterland, like for example in De Borger et al. (2008). In this regard, Pallis et al. (2008), Saeed & Larsen (2010) and Van de Voorde & Vanelslander (2014) describe port negotiations as a game. Sufficient transparency can be assumed, because entities in the public sector can also be included in the analysis. As a result, information asymmetry should not necessarily be included in the game.

When insights from industrial organisation are added to the models and a game-theoretic approach is chosen, individual value functions have to be derived and the type of game has to be chosen. Dynamic programming can then be used to find the optimal decision (Dixit & Pindyck, 1994; Azevedo & Paxson, 2014; Chevalier-Roignant et al., 2011; Huberts et al., 2015). Huisman & Kort (2015) discuss a model that analyses the initial investment decision of a leader and a follower firm. In a duopoly setting, they use a linear (in *Q*), multiplicative (in *X*), inverse demand function given by

$$p = X(1 - BQ),$$
 (2.14)

where *X* again follows a GBM and *Q* represents the total market output. This function is a special case of the one introduced by Dixit & Pindyck (1994). The authors find that in this case, overinvestment of the leader forces the follower to invest later and in less capacity. However, this model is not directly applicable to a competitive port setting. An important implication of this model is that the maximum output is limited to 1/B in order to prevent negative prices. This demand function could serve as a good first approximation. A linear (in *Q*), additive (in *X*) inverse demand function would however be better suited to analyse a market wherein potential demand growth is possible. Nevertheless, more sophisticated functions and their applicability could also be considered when developing port investment models.

Another important limitation of the model of Huisman & Kort (2015) lies in the number of competitors involved. As shown by Veldman & Bückmann (2003), in the Hamburg - Le Havre range, five ports are in competition. For this reason, modelling ports in an oligopoly, rather than in a duopoly could be a better alternative. Some examples are given by Azevedo
& Paxson (2014). Aguerrevere (2009) finds in his model with an exogenously determined number of firms in the industry that with more companies in the market, the option to expand is exercised earlier, as its value is eroded by the presence of competitors. In order to analyse an n-firm game, Grenadier (2002) uses a constant elasticity demand function, equivalent to the one used by Guthrie (2012) in Equation (2.9). It is shown that the value of waiting is negatively correlated to the amount of competition, as a result of preemptive strategies. As stated before, competition between ports is not the only competitive element in the analysis of port capacity investments. Potentially conflicting interests in combination with a number of aligned objectives of the actors involved in the maritime logistics chain require a much more complex analysis in a new port investment model too.

Since ports are heterogeneous, this should be translated into asymmetric firms in an RO game model. Firms or ports could differ in terms of investment costs, revenues or operational costs (Azevedo & Paxson, 2014). In addition, ports offer products or services that are not identical (e.g., different service levels or hinterland distances). Kamoto & Okawa (2014) present an RO model that allows for product differentiation, more specifically in an innovative environment with a leader and follower entrant. Their inverse demand function of firm i is related to the one used by Huisman & Kort (2015). A correction is applied for the output of the other firm j, weighted by the amount of product similarity. The inverse demand function is given by

$$p_i = X \left(1 - q_i - \delta q_j \right), \tag{2.15}$$

with $\delta \in [0, 1]$ the differentiation parameter, taking the value of 1 in the case of perfect substitution and 0 in the case of perfect complementarity. If more than two ports are competing in the market, Equation (2.15) could be extended to

$$p_i = X \left(1 - q_i - \delta \sum_{j \neq i} q_j \right), \tag{2.16}$$

with an equal product differentiation between the considered ports or to a more complex situation

$$p_i = X \left(1 - q_i - \sum_{j \neq i} \delta_j q_j \right), \tag{2.17}$$

with different diversifications between the services of the considered ports. Adding the latter asymmetry between the ports might however complicate the calculations too much, so that more complex solution techniques such as agent based modelling might be needed.

The models discussed above involve firms that were not yet active in the market and have to decide on their first investment size and timing. However, practically every port is already active and has made one or more capacity investments in the past. For this reason, a realistic port model should allow for ports to be not only a new market entrant, but also an incumbent with previously installed capacity. Also in such a case, it would be interesting to analyse which port would invest first (preemption), and when they would invest again (Huberts et al., 2015).

2.4.6 Alternative uncertainty processes to model cyclicality and changes in legislation

RO models frequently use a GBM as the stochastic process expressing uncertainty (Li & Cai, 2017). This non-negative process allows uncertainty and drift to be set by two independent parameters, and its mathematical complexity is relatively limited (Schöne, 2014). Moreover, Marathe & Ryan (2005) and Lindsey & De Palma (2014) already proved the suitability of this process in a transportation context. However, as Dixit & Pindyck (1994) also indicate, alternatives exist to model different market patterns. Maritime markets typically follow a cyclical pattern, with cycles lasting about eight to ten quarters (Stopford, 2009; Paflioti et al., 2015).

Ruiz-Aliseda & Wu (2012) built an RO model analysing the entry and exit decisions in such cyclical markets, like maritime markets are. Balliauw (2017) applied this model to study entry and exit decisions in shipping markets. To model cyclical markets, the traditionally used GBM is replaced by a discrete-time Markov process, defined by

$$\mathrm{d}\pi = \alpha(t)\pi\mathrm{d}t,\tag{2.18}$$

with π the instantaneous profit at time *t*, and *a* equalling the growth rate (α_g) when the industry is in a growth phase, or the decline rate (α_d) when the industry is in a decline phase. The probabilities of going from the initial phase to the second phase and of the inverse transition are given by

$$\begin{cases} P(\alpha(t+dt) = \alpha_2 | \alpha(t) = \alpha_1) = \psi_1 dt + o(dt), \\ P(\alpha(t+dt) = \alpha_1 | \alpha(t) = \alpha_2) = \psi_2 dt + o(dt). \end{cases}$$
(2.19)

Whereas a GBM is used to model a process showing a positive, negative or zero trend with some random variations, the Markov process can model different states in one model (e.g., growth and recession phases). As a result, the presented Markov model can take into account periods of decline, as opposed to many economic models assuming only a growth trend. Because of this, it is expected that capacity investments will be lower under this approach, since overcapacity in times of recession has a higher cost. Substituting the GBM by a cyclical Markov process offers a worthwhile approach to study the impact of this alternative stochastic market demand process on the decisions made in the port.

Alternatively, Poisson or jump processes are often used in models in order to represent the possibility of sudden changes or events, for example following a newly available technology in R&D or a new legislation (K. Wang & Zhang, 2018; Weeds, 2002; Dixit & Pindyck, 1994). In this case, the sudden change, called the event, results in a jump. Such a stochastic process is written as a Poisson differential equation:

$$dx = f(x, t)dt + g(x, t)dJ, \qquad (2.20)$$

with f dt the non-random evolution over time, g also a known function and dJ the increment of the Poisson process:

$$dJ = \begin{cases} 0, & \text{probability } 1 - \nu dt, \\ u, & \text{probability } \nu dt. \end{cases}$$
(2.21)

Here, *u* is the size of the jump caused by the event and *v* the mean arrival rate of the event.

2.5 Towards a real options application in ports

Elements of RO models that could be used in a port environment were examined in the previous section, identifying some of their important characteristics. In this section, the different models and their characteristics are linked to each other within the port environment, with the objective of effectively improving port capacity investment decisions. This section first presents a framework that will lead to enhanced port capacity investment decisions. The framework is then applied in Section 2.5.2.

2.5.1 A framework to enhance port capacity investment decisions

The framework in Figure 2.2 combines the insights from the port uncertainty and RO models reviews. This framework visualises the different choices of components to be made when building a new model to decide on port capacity investments. In this figure, the dashed blocks together constitute the building block 'objective'. First of all, decisions are to be made about the demand function for the output (i.c., port throughput). The mathematical specification of the function and the way of including uncertainty in this function need to be considered. Possible alternatives include a GBM, a Markov process, a Poisson process, a binomial process (Smit, 2003), a uniform distribution (Xiao et al., 2013), a mean reverting process (Dixit & Pindyck, 1994) or a vector autoregressive (VAR) process (Huisman et al., 2013). The inverse demand function should also account for the cost of congestion (Xiao et al., 2012). When more than one port needs to be considered, the inverse demand function should reflect competition between ports, with or without differentiated products.

A cost function for producing the output has to be selected as well. This function could be a linear, quadratic or even higher-order function of q. Besides a capacity-dependent variable cost, a capacity holding cost could be added to the function. When a capacity holding cost is included, investments to maintain the level of capacity are already accounted for. In that case, depreciation should not be included in the model. Next to the profit, also other benefits matter and could be included in the analysis (e.g., spillover effects or consumer surplus). Different objectives can be combined by means of a weighted sum of these objectives. The sum of discounted profits and other benefits in each period of time minus the



CHAPTER 2. TOWARDS IMPROVED PORT CAPACITY INVESTMENT DECISIONS

46

cost of investment together determine the value of the project. Other than the investment cost function of Dangl (1999), also a polynomial specification could be used, for example a higher-order term to account for a limit to the project size (e.g., because of limited space and the subsequent need of expropriation). The value of the port authority's project is the objective that needs to be maximised through the flexible investment. A slightly different approach is maximising the market value of the port authority, as a function of the return on assets (Guthrie, 2012).

Other components that could be included are time to build, depreciation (when there is no capacity holding cost) and output flexibility. Finally, single-firm models can be expanded to game-theoretic models with more than one firm. The most appropriate specification will depend on a number of factors, such as the port characteristics, uncertainties and the research objectives. The presented framework should support the selection of the best building blocks when developing new port capacity investment models (Trigeorgis, 2005). The framework will also help better understanding, categorising and comparing existing RO models.

2.5.2 Applying the framework to improve port capacity investment decisions

In order to illustrate a potential application of the framework, the investment decision of a port in new capacity is analysed. A port that wants to maximise the revenues consisting of the sum of port dues and terminal handling prices is considered. In such a port, the value of the firm comprises the combined objectives of the port authority and the operators of the port (Meersman et al., 2015; Strandenes & Marlow, 2000). Those two actors may coincide in one enterprise, like in the private port of Felixstowe, owned by the port operator Hutchison Port Holdings (Juhel et al., 2007). This single entity wants to maximise the stream of discounted profits minus the investment cost with respect to the investment timing (T) and the capacity (*K*). The stream of profits, which only become available after investment, is composed of the revenue (pq) minus the operational cost (a linear function in this initial setting) and the capacity holding cost (see Figure 2.2). In the profit function, the concession payments are cancelled out, because this is an internal transfer from the TOC to the port authority and should not be considered here. The opportunity cost of waiting, namely foregone profits before the investment takes place, is also reflected in the objective function through the discounting of the profit stream to the moment of investment. These considerations result in the following system of equations, which can be optimised through dynamic programming (Dixit & Pindyck, 1994):

$$\begin{cases} \max_{T,K} \left\{ \mathbf{E} \int_{0}^{+\infty} e^{-r(T+t)} \pi(T+t) dt - e^{-rT} I(K) \right\}, \\ \pi(t) = 0, & t < T, \\ \pi(t) = (p(t) - c)q(t) - c_h K, & t \ge T, \\ p(t) = p(X(t), q(t), D(t)), \\ + & - & - \\ dX(t) = \mu X(t) dt + \sigma X(t) dZ(t), \\ D(t) = D(q(t)/K), \\ 0 \le q(t) \le K, \\ I(K) = \gamma K^{\varepsilon}. \end{cases}$$
(2.22)

System (2.22) indicates that the price p from the inverse demand function is negatively related to the throughput quantity q and delay D and that it is positively related to a random shift parameter X. This parameter follows a GBM and is related to the size of the market. Since congestion can result in a loss of potential demand (Jansson & Shneerson, 1982), the demand and congestion level are endogenised in the model. When congestion and delay increase, the customers' willingness to pay decreases, because they will incur the additional costs caused by congestion and delays. A possible shape of the positive relationship between the occupancy or utilisation rate q/K and the amount of delay D is visualised in the delay curve in Figure 2.3. According to Kauppila et al. (2016), such a delay curve is a good approach when modelling at port level. To derive a good equation for the delay curve, insights from queuing theory can be used (Novaes et al., 2012). Also computer simulations (e.g., enterprise dynamics) can be used to further estimate the impact of operations on capacity (Kia et al., 2002). Additionally, System (2.22) specifies that the upper limit to throughput is defined by the theoretical design capacity, K. Producing no output (q = 0) is also possible. The cost of investment I is initially based on the specification of Dangl (1999). With $\varepsilon < 1$, economies of scale in investment costs can be included in order to prevent infinitesimal investment. When capacity expansions are studied, K has to be replaced by ΔK in the investment cost function.

In infrastructure investment projects, demand and revenues are often overestimated, whereas costs are frequently underestimated. Both could reduce the value of a project significantly. For this reason, the inclusion of uncertainty both on the demand and cost side of the model is suggested (Van Putten & MacMillan, 2004). The latter could result in the inclusion of an additional state variable, with its own uncertainty. This would however complicate the analysis considerably, because the interaction between both distributions has to be included. This will in turn reduce the mathematical tractability of the model (Huisman et al., 2013). As a result, Monte-Carlo simulations might be needed. Tsamboulas & Kapros (2003) for example apply Monte-Carlo simulations to the investment decision in freight villages under uncertainty.



Figure 2.3: Delay curve illustrating the positive relationship between the port occupancy rate and the amount of delay.

The presented model from System (2.22) can be used to calculate an optimal investment threshold for X (expected to be reached at time T), the amount of capacity K, the optimal throughput quantity q and price p for a port. When building new RO models however, one always has to keep in mind the solution technique as well. A frequently used technique to solve RO models is dynamic programming. This technique is explained by Dixit & Pindyck (1994). Often however, no analytical expression can be derived for the optimal values of some decision variables. Nevertheless, numerical solutions can be calculated and they allow drawing conclusions on the dynamics of the model (Aguerrevere, 2003; Guthrie, 2012). In some cases, the optimum might also be approached by a good heuristic algorithm, but it does not guarantee reaching the actual global optimum. Azevedo & Paxson (2014) provide such an example in RO games. Cucchiella & Gastaldi (2006) also use a software package to solve their RO model. Furthermore, the RO analysis might be simplified by a binomial tree approach (Hu & Zhang, 2015) or a Monte-Carlo simulation to estimate some parameters involving processes characterised by uncertainty (Anda et al., 2009). Even simplifying spreadsheet approaches are available (de Neufville, 2006). This shows that the trade-off between the complexity and realism of the model on the one hand and the feasibility of the solution technique on the other hand, should be well considered when building RO models.

2.6 Conclusion of this chapter

It is very important for a port to install the right amount of capacity when making lumpy investments. On the one hand, undercapacity leads to congestion and waiting costs for the port users. On the other hand, persistent overcapacity means that unnecessary investments and capital costs are incurred not only by the port authorities and private investors, but often also by the society. However, it is not easy to determine the optimal size of investments in port capacity, because a substantial amount of uncertainty has to be taken into account, especially because it often concerns large-scale, irreversible investment projects.

As ports are nodes in often long and complex logistics chains, they are subject to a wide range of uncertainties. It is necessary to account for the complex relationships between the maritime logistics chain actors, as well as uncertainty resulting from changes in global trade, technology, environmental factors and the legal framework. Applying the traditional NPV method in such a complex situation will in general lead to the wrong investment decision. In this case, RO models are a better alternative, since they specifically take into account the value of managerial flexibility under uncertainty. However, introducing all the different sources of uncertainty ports are facing in an RO model is not straightforward.

Starting from a review of these uncertainties and of the available RO models, a framework has been set up in order to help categorising and applying RO models for port capacity investments decisions, taking into account the complex environment ports are operating in. New RO port models should include the relevant characteristics of the port or ports studied. This could for instance be achieved by selecting the most suitable components from the presented framework. The port RO models can be expanded even further, with major or minor additions, in order to include even more of the complex relationships between the actors in the maritime logistics chain. This however requires an increasing number of assumptions about the specifications of the underlying stochastic processes and their interactions; and the estimates of their variances. Moreover, RO models are only applicable in a limited range of parameter settings, because of convergence conditions. Due to the complex mathematics of RO, the main challenge will be to find the right balance between on the one hand having an acceptable and reliable representation of the port environment and on the other hand ending up with a solvable RO model.

In order to improve the performance of investment decision making models in transportation, ex-post analyses of these models are advised (Mackie & Preston, 1998), with the careful attention for correct parameter estimations (Azevedo & Paxson, 2014). For this purpose, case studies are an excellent approach in the subsequent stage of this research in order to verify and fine-tune the developed models. Already completed projects might be useful, as data confidentiality might be less critical. It is however necessary to possess information about the entire decision process, including all the alternatives and final decisions. When these data are lacking, other finished projects might allow making valid estimates of missing data and obtaining information about the different parties involved in the project.

This chapter is only the first step in evolving towards better port capacity investment decisions. In the following chapters, new port RO models are developed in order to study the impact of different typical port characteristics on the investment decision and take investment decisions with embedded options of flexibility. These different models contain components from the developed framework, taking into account the relevant characteristics of the port(s) studied. The next chapter deals with a private port owner that wants to invest in and operate a new port. To this end, a linear additive inverse demand curve with a GBM is selected. For the cost function, the traditional first-order function of throughput is retained, to which a capacity holding cost is added. This leads to the private port profit function. Combined with a polynomial investment cost function, the project value is to be maximised. To this model, output flexibility is added. In subsequent chapters, the model is expanded gradually. From Chapter 4 on, other public benefits are also considered by publicly owned ports. Moreover, in that chapter, game theoretic considerations within the port are explored, since the port authority and terminal operating company need to take the port capacity investment decision together in a landlord port. In Chapter 5, competition between two ports is considered in a duopoly. Finally, in Chapter 6, time to build is added to the model and port expansions are considered. A synthesising overview of how the models of each chapter relate to one another has been provided in Figure 1.3. An overview of their underlying assumptions has been shown in Table 1.2. These overviews can be interpreted in combination with the framework in Figure 2.2 to enhance insight into the differences and the applicability to different situations. They moreover offer the reader a guide to the next chapters. For port managers, they help to select the most appropriate model or combination of models to support the investment decision in the own port. These are the models with the most reliable assumptions, compared to the case at hand.

Chapter 3

The impact of congestion and uncertainty on a new private port's capacity investment

As port infrastructure capacity investments are irreversible, subject to many sources of uncertainty and involve large sums of money, the real options (RO) approach to investment was chosen in the previous chapter to help determine the optimal amount of capacity for this type of investment projects and the ideal moment to invest. The basic principle is that potential investors monitor information about costs and returns as it becomes available, and make the investment as soon as the cost of deferring the project is greater than the expected value of the information gained by postponing investment (Bernanke, 1983). Decisions about the timing of this type of port investment are related to decisions about the optimal capacity of infrastructure with flexible throughput. Pindyck (1988) examined this issue in the context of incremental investment decisions, while Dangl (1999) considered investments in which there was only one opportunity to choose the maximum capacity. The latter approach is more suitable for analysing port infrastructure investments, as large-scale port infrastructure projects are often lumpy once-and-for-all decisions (Bar-Ilan & Strange, 1999). Hence, this latter approach is also used in this thesis.

As discussed in the previous chapters, ports are affected by environmental issues on the one hand, and cause a number of negative external effects themselves on the other hand (Benacchio et al., 2000). In this chapter, the focus is on the impact of the aversion to waiting time of the shipping companies using the port's capacity on the investment decision. This waiting time generates delays involving considerable costs (Novaes et al., 2012). As pointed out in Chapter 1, aversion to waiting time and resulting congestion costs differ among users and goods categories. From queuing theory, it is known that waiting time in ports generally begins to increase as infrastructure occupancy rates pass 50% and that it is the result of

uncertainty regarding the distribution of ship arrivals at a terminal with a given theoretical capacity (Blauwens et al., 2016). Installing a larger port would lead to less waiting time and less congestion when the same amount of throughput is handled. As a result, travel times would be shorter, since capacity serves as a buffer against waiting time that is caused by future demand growth. This is of measurable value to shippers. Yet, previous studies have shown that investment in more capacity attracts additional traffic and throughput, offsetting part of this capacity investment benefit (Zhang, 2007; Wan et al., 2013). What remains unclear however is how the investment decision in a new port is influenced by port users' average aversion to waiting time.

In this chapter, the investment in a new container port in a growth market under uncertainty is studied. The considered port is operated by one single actor which also owns the port. In the considered region, only one significant port is assumed, leaving port competition beyond the scope of this chapter. Two once-and-for-all decisions have to be made: the size of the port and when to build it. The analysis takes into account that next to the capacity decision in the first (investment) stage, the port can flexibly adapt its price level in the second (operational) stage to influence throughput and port occupancy. The approach omitting competition moreover allows focusing on the impact of congestion and uncertainty on an individual port's investment decision. Such investment decisions are relevant in growing developing countries with a lot of uncertainty. Examples include the greenfield container ports in Manaus and Porto Central (both in Brazil) and north of Izmir (in Turkey), as well as port development in African countries (Vanelslander, 2014; Ward, 2015).

This chapter sheds light on how the cost of congestion for port users affects the optimal investment in new port infrastructure under demand uncertainty. To this end, a real options approach is used. The proposed approach to congestion costs allows differentiating between projects involving goods categories and shippers with different waiting-time aversions. The RO methodology in this chapter represents a more accurate continuous-time continuous-state model for ports to decide on their optimal investment strategies, compared with models omitting congestion and the costs it involves (Dangl, 1999; Hagspiel et al., 2016; Novaes et al., 2012). Moreover, uncertainty is modelled through a geometric Brownian motion, which allows analysing the impact of economic growth and uncertainty independently of one another.

As was shown in the previous chapter, many sources of uncertainty impact demand, such as trade uncertainty and the different maritime logistics chain actors' uncertain decisions, which are exogenous for a port. Although here the scope of uncertainty is limited to the macro level, also a number of micro-level uncertainties, such as disasters, technological, legal and political uncertainty exist. They are however left beyond the scope of this thesis, since they involve a different modelling approach. Nevertheless, other research is available that studies the impact of these sources of uncertainty on port investment decisions. Randrianarisoa & Zhang (2019) study the impact of climate change on port investment decisions under uncertainty. Also K. Wang & Zhang (2018) study the impact of natural disaster

risk on port adaptation investments to reduce potential damage. They use Knightian uncertainty and Poisson jumps to model uncertainty. Moreover, other port investment types are concerned. As a result, ports' policies need to be supported by a combination of both types of research. It could be interesting to study in future research how different sources of uncertainty and investment types can be combined in one decision making model.

The structure of this chapter is as follows. The next section deals with the economic context considered in this chapter and explains the methodology used to identify the optimal size and timing of the (new) port capacity investment decision under uncertainty. Section 3.2 explains the selection of the parameters used for the numerical application of the model to a new port, since analytical solutions are not obtainable. In Section 3.3, the value calculation of an investment project under uncertainty is elaborated. In Section 3.4 the optimal size and timing of the investment decision are derived and the impact of congestion and uncertainty on the decision is quantified. A sensitivity analysis in Section 3.5 shows how the investment decision depends on the various economic parameter values. The final section presents the conclusion of this chapter.

3.1 Economic setting and methodology

In this chapter, a profit-maximising private port with flexible throughput, operated and owned by one single actor, is considered. Figure 3.1 illustrates the revenue sources and investment outlays of the considered single port actor. This port authority (PA) has the option to invest in a new port. The PA must take a lumpy once-and-for-all decision regarding the maximum capacity of the new port and the timing of the investment. However, the port's customers are averse to waiting, which implies additional costs for them. Hence, the first sub-question to be answered in this chapter is: (RQ 3.1) What is the impact of the degree of waiting-time aversion of port customers on a port's optimal capacity investment decision in a new private service port under uncertainty without competition? Since the port experiences a random but growing demand, the second sub-question is: (RQ 3.2) What is the impact of the amount of uncertainty on a port's optimal capacity investment decision in a new private service port without competition?

The assumed price transparency gives rise to the following linear additive inverse demand function for the port at time *t*:

$$\rho(t) = X(t) - Bq(t), \qquad (3.1)$$

with $\rho(t)$ the gross willingness to pay,¹ q(t) the throughput, X(t) the value of an exogenous demand shift parameter at time t, sometimes also called the reservation price, and B the (negative) slope of the inverse demand curve. Throughput is measured in terms of

1

¹Synonyms used in the literature are generalised cost or full price.



Figure 3.1: Revenue sources and investment outlays of the single actor (PA) in the considered private service port.

container TEU. It is assumed here that the theoretical design capacity of the new port, K, puts an upper limit to the throughput that can be handled. As such it is assumed that $0 \le q(t) \le K$ and as a consequence the capacity utilisation rate q(t)/K is constrained between zero and one. Hence, exceptional cases where realised throughput q(t) temporarily exceeds K at peak moments are disregarded in this chapter (Luo et al., 2010).²

Demand uncertainty is introduced in the model through random shifts of the demand function. These are introduced by a random process for X(t) which is assumed to follow a geometric Brownian motion:

$$dX(t) = \mu X(t)dt + \sigma X(t)dZ(t), \qquad (3.2)$$

which is specified by parameters μ (expected drift in demand shift) and σ (variability) and with Z a standard Wiener process. A GBM is often used to model uncertainty, because both drift and variability are included in the random process as independent parameters. The implementation of this approach is however new in the context of port capacity investment decisions under congestion. Chen & Liu (2016) for example use a uniform distribution to model demand uncertainty in a port. Another advantage is that with an initial value of X(t = 0) > 0, the variable X(t) always remains positive, even if expected drift is negative (Dixit & Pindyck, 1994). Here, the focus is on a base case with a positive drift μ , reflecting economic growth and growing international trade (UNCTAD, 2015). In such a case, demand shifts at the rate of μ , on average, but over certain time intervals, growth may be higher, lower or even negative due to the uncertainty, as shown in Figure 3.2. This simulated figure displays a possible evolution of demand for throughput in a port in a growth market. This approach of a positive growth rate is common in real options models (see e.g., Dixit & Pindyck (1994), Dangl (1999) and Hagspiel et al. (2016)), but is not a necessary condition for finding a solution. For the sake of readability, the (t)-dependencies will be omitted in the following equations.

According to Strandenes & Marlow (2000) and Xiao et al. (2012), the gross willingness to pay of customers is the sum of the unit price of services at the port (*p*) and the unit cost of

²This assumption is relaxed in Chapters 5-6. In Chapter 6, a comparison between the two approaches shows that the impact of this assumption on the optimal investment decisions is negligible.



Figure 3.2: Sample evolution of demand shift parameter *X* following a GBM, with X(t = 0) = 30 euro per TEU, $\mu = 0.015$ and $\sigma = 0.1$.

congestion the users incur:

$$\rho = p + AX \frac{q}{K^2}.$$
(3.3)

Hence, *p* can be rewritten as:

$$p = X - Bq - AX \frac{q}{K^2},\tag{3.4}$$

with the final term the unit cost of congestion which reduces the customer's gross willingness to pay. Total congestion cost *TCC* equals $AX(q/K)^2$, an own specification based on queuing theory insights from other authors (De Borger & Van Dender, 2006; Yuen et al., 2008; Xiao et al., 2012; Zhang & Zhang, 2006; Chen & Liu, 2016).³ In the unit congestion cost term, the capacity utilisation rate is a proxy indicator for the amount of congestion and delays.

Xiao et al. (2012), De Borger & Van Dender (2006), Yuen et al. (2008) and Chen & Liu (2016) make delays *D* linearly dependent on capacity occupancy. However, as previously discussed, queuing theory learns that congestion only starts at occupancy rates of about 50%, and increases sharply beyond 75% to 80% (Blauwens et al., 2016; Kauppila et al., 2016). To this end, Zhang & Zhang (2006) use a specification where delay depends on a scale factor *times q* /[K(K-q)]. This specification is based on estimates from steady-state queuing theory, with the underlying assumption of Poisson distributed arrivals (Lave & DeSalvo, 1968). It satisfies four conditions (Zhang & Zhang, 2006):

$$\frac{\partial D}{\partial q} > 0, \frac{\partial D}{\partial K} < 0, \frac{\partial^2 D}{\partial q^2} > 0, \frac{\partial^2 D}{\partial q \partial K} < 0.$$
(3.5)

³Although congestion costs are modelled here on the demand side, a mathematically equivalent approach could have been to model congestion on the supply side (i.e., to include it in the port's costs), since congestion can impose costs on both suppliers and customers of port capacity.

These conditions imply that higher throughput and lower capacity lead to an increase in congestion and that these effects are stronger when congestion is already high.

The condition $\partial^2 D / \partial q^2 > 0$ however does not hold for the linear specification on the one hand. On the other hand, the disadvantage of the specification of Zhang & Zhang (2006) is that delays, and hence costs as well after a multiplication with a monetary scale factor, are infinitely high at full occupancy. This is not the case in reality either. A functional specification that highly resembles the specification of Zhang & Zhang (2006) would be a higherorder term of occupancy in the delay function, such as a fourth- or fifth-order term. This would satisfy all conditions of Zhang & Zhang (2006) in Equation (3.5), whereas full occupancy would not ex-ante be excluded. However, the exact, analytical optimisations performed in the remainder of this thesis would become too complicated under such a specification of the delay function, due to the consideration of quantity flexibility.⁴ A secondorder dependency would also already be more realistic than the first-order relationship, and would also satisfy the required conditions, but would still complicate the calculations too much. Therefore, a proxy for q is required, which can then replace the factor q^1 in q^2 . The uncertain reservation price, X, is positively correlated with the optimal throughput at any point in time. This explains why q^2 is replaced by Xq, using X as a proxy for q. The advantage of this approach where congestion costs are linked to the state of the market is discussed further on in this section, using Equation (3.8).

In the next step, delays are translated into costs through the monetary scale factor A (Xiao et al., 2012). The parameter A depends on the port users and goods categories involved (Yuen et al., 2008). The monetary scale factor (A) is an expression of the value of time (De Jong, 2007) and is lower for shipping companies that transport goods which are less urgently needed or less impacted by time (e.g., wood compared with perishables or critical machine components) and for shipping companies with less stringent sailing schemes. In such cases, congestion poses less of a problem for the port user. However, different individual liners that will use the new capacity might have different values of time. Therefore, A should be interpreted as the average value of time of the new capacity's users. Moreover, A is fixed over time in this RO model. Therefore, including X in the congestion cost function has the advantage that the excess growth of X compared to the growth of q can account for rising values of time in the future (De Jong, 2007). Moreover, the customer base of a port is relatively stable over time, because container liners became less footloose (Verhoeven, 2015; Heaver et al., 2001). Hence, the assumption of a fixed average scale factor A per project is reasonable. Nevertheless, the value for A may differ between projects with different users. It is exactly this approach that allows illustrating the effect of different waitingtime aversions of users on the timing and size of the investment in a new port, through the variation of A in the analysis.

⁴The maximisation of profit, which is the first optimisation, with respect to the optimal quantity would then involve a sixth order function to be maximised. Even a second-order dependency on occupancy would involve a third-order function in this optimisation.

It can be derived from the previous expressions that in order to reduce or avoid congestion, ports can choose two strategies. On the one hand, they can charge higher prices. Congestion, ceteris paribus, reduces the willingness to pay of users. When there is more waiting time for the same amount of throughput, the aversion to waiting of the customers will lead to the port having to charge a lower price if it wants to retain all of these customers. As a result, handling the same amount of throughput for customers with the same willingness to pay, leads to lower revenues if congestion is higher (i.c., due to less capacity), because the users' additional waiting costs are part of their generalised costs. Equivalently, customers have a willingness to pay in order to avoid congestion. Hence, charging a higher price will lead to a lower throughput level, as a result of the negative slope of the demand curve, and less congestion, which in turn tempers the throughput reduction and might increase port revenues. This mechanism is known in the literature as congestion pricing. However, to implement this in a good way, the port needs to accurately measure instantaneous port occupancy rates at every moment in time. Moreover, the congestion pricing approach is especially useful in the short term. On the other hand, ports can accommodate the additional demand by investing in more capacity. Since this latter approach takes more time, this is only feasible in the long term. The investment approach will be preferred when the effect of the throughput increase will outweigh the previously discussed price effect following from the specific price elasticities of the port users.

Optimising both instantaneous profit and these discounted profits over time, in relation to investment outlays, includes the potential strategy of congestion pricing in the model. Prices will be set at the level that is most profitable for the port under consideration and will depend amongst others on the willingness to pay of the users and the actual value for the aversion to waiting, *A*, leading to an instantaneous optimal level of congestion. At a certain point in time however, investing becomes more profitable than further increasing prices and reducing throughput, and will then be the preferred option.

The total operational cost function of the port is given by

$$TC_O(q) = c q, \tag{3.6}$$

where *c* is the constant marginal operational cost, like it was used by Haralambides (2002). In the literature, no consensus exists about the presence of economies or diseconomies of scale in port operations. S. Chang (1978) for example encounters constant scale economies, whereas Reker et al. (1990) find diseconomies of scale, as opposed to the economies of scale of Tongzon (1993).⁵ Due to the ambiguity in the literature, a simple linear production function is reasonable in this model.⁶ The instantaneous port profit, π , can now be calculated

⁵For a comprehensive overview, see the literature review of Tovar et al. (2007, p. 210).

⁶A practical example justifies this specification. The available employees can handle one or more extra units of throughput, a typical case of economies of scale. However if even more additional throughput is to be handled, a new employee needs to be hired, leading to a sudden cost leap (diseconomies of scale). Moreover, the linear cost function approach is in line with other real options models such as Dangl (1999). Altering the specification of TC_0 would however not substantially complicate the remainder of the analysis. For example, a second-order term as in Chapter 2 could be added if it increases realism for a specific case.



Figure 3.3: Realised total congestion cost function with installed capacity K = 10 million TEU per year.

as

$$\pi(X, K, q) = p(q)q - cq - c_h K, \qquad (3.7)$$

total revenue minus operational costs (TC_O) and $c_h K$, the total cost to hold the capacity in place, for example through maintenance.

The optimal throughput $q^{opt}(X, K)$ for given X and capacity level K is the result of maximising profit π with respect to throughput q. This leads to the following total congestion cost *TCC* dependent on given X and K:

$$TCC(X,K) = AX \left(\frac{q^{opt}(X,K)}{K}\right)^2.$$
(3.8)

It is represented by the black curve in Figure 3.3 for the fixed port capacity level *K* of 10 million TEU per year.

Figure 3.3 hence demonstrates that including *X* in the congestion cost allows for an exponential realisation of the total congestion cost function, as observed in Figure 3.3. It moreover satisfies the conditions of Zhang & Zhang (2006) and it leads to a good approximation of the discussed queuing theory insights and reality, in which congestion and delays begin to emerge once an occupancy rate of approximately 50% is exceeded and increase substantially beyond 75% to 80% (Kauppila et al., 2016; Memos, 2004; Leachman & Jula, 2011; Novaes et al., 2012; Blauwens et al., 2016). The grey lines in Figure 3.3 are the quadratic total congestion cost functions that correspond to given values for the factor *X* and the variable throughput (*q*). Subsequently, a $q^{opt}(X, K)$ corresponds to each factor *X* and the fixed capacity level *K*. This q^{opt} divided by *K* results in the optimal occupancy rate.

These are marked on their corresponding congestion cost curves by black dots. Connecting those dots results in the realised total congestion cost function, shown in black, which has an exponential shape similar to the one of Leachman & Jula (2011). This is the case because the impact of X on the total congestion cost is twofold. On the one hand, an increase in X shifts the demand function upwards.⁷ On the other hand, an increase in X leads to a higher occupancy rate and also to more delays.⁸ As a consequence, the missed fraction of the potential demand grows more than linearly (Blauwens et al., 2016).

The investment cost I(K) of installing a new port of size K is independent of the investment timing and is given by the following function:⁹

$$I(K) = FC_{I} + \gamma_{1}K - \gamma_{2}K^{2} + \gamma_{3}K^{3} + \gamma_{4}K^{4}, \qquad (3.9)$$

where FC_I is the fixed investment cost, resulting among other things from a preliminary feasibility study. The cost of the investment grows with size K. Installing a larger dock involves more dredging activities and installing more quay walls and paving. This is reflected by a positive first-order term. General RO literature (Dangl, 1999; Hagspiel et al., 2016) and port literature (Haralambides, 2002) indicate the existence of economies of scale in the investment size. Investment costs increase less than proportionally with an increase in project size. This is reflected by the negative second-order term. Additionally, waterfront land is very costly (Haralambides, 2002). The available land for port infrastructure is limited. A larger investment than the available land would, for instance, imply expropriating houses, which is very costly. Therefore, a fourth-order term is added to the investment cost, reflecting an investment size boundary with a very sharp cost increase. Although a third-order term could also be used to model this boundary, the advantage of a fourth-order polynomial (with the third-order term omitted, $\gamma_3 = 0$) is that the function is not symmetric about the inflection point. Hence, the proposed mathematical specification does not impose that the economies of scale need to follow the same functional shape as the investment size boundary.

The previous considerations are taken into account by selecting appropriate values for the γ 's in Equation (3.9), see Section 3.2. An example is already given in Figure 3.4. This investment cost function is compared with the dashed line representing the investment cost function of Dangl (1999):

$$I_{\text{Dangl}} = \gamma K^{\varepsilon}, \qquad (3.10)$$

with the $\varepsilon < 1$ expressing the economies of scale in the investment size and without a size boundary. It is similar over the first part of its domain, but only the specification of this

⁷In the example, this leads to an upward shift in the grey curve. The impact on the congestion cost can easily be verified by considering a case in which *q* already equals *K*. With an increase in *X*, the occupancy rate would remain constant at 1(= K/K), but more potential demand would be foregone. The model accounts for this mathematically by multiplying the occupancy rate with *X*.

⁸The black dots move to the right.

⁹As Haralambides (2002) and Musso et al. (2006) highlighted, port investments are lumpy. This leads to discontinuous jumps in expansion paths of ports. Nevertheless, the cost of one individual investment project as a function of its size can reasonably be modelled as a continuous function.



Figure 3.4: Total investment cost functions with and without a maximum boundary.

chapter's model then displays the required sharp cost increase reflecting the investment size boundary. This required shape is obtained when the γ 's adhere the condition: $|\gamma_1| > |\gamma_2| > |\gamma_4|$. Although it could have been a possible alternative to just add a fourth-order term to I_{Dangl} , the advantage of the proposed function is that the shape is more flexible (especially over the first part of its domain) and that it is easier to estimate empirically using linear estimation techniques.

Additionally, it is important to point out that in practice, the shape and height of the investment cost function depend to a large extent on the characteristics of the project, such as the geographical situation, the production technology to be used, etc. The mathematical specification of the investment cost function needs to be considered as a translation of the reality into a model. In practice, it is necessary for the deciding actor to have a good insight into the different possible sizes of the project and the respective investment costs. The two most important aspects are the correct cost estimation for each size and the first-order derivative with respect to size, to express correct cost increases between different project sizes. To meet these requirements, for each case, different more or less equivalent functional specifications can be used. The exact functional shape will barely impact the investment decision, as long as the height and first-order derivative are sufficiently reflecting reality.

Given the discussed economic setting, the decision is to be made when and how much to invest. The former means to decide on the critical value X_T of the demand shift parameter for which it becomes optimal to invest, instead of waiting. This implies that investment will take place as soon as X reaches this threshold for the first time from below, as illustrated in Figure 3.5 at T = 4.65. Hence, a higher threshold value corresponds to investing



Figure 3.5: Relationship between the optimal investment timing and the threshold for demand shift parameter *X* following a GBM, with X(t = 0) = 30, $\mu = 0.015$ and $\sigma = 0.1$.

later. If X(t = 0) however exceeds X_T , investment takes place right away. At the moment of investing, also the optimal size of the capacity K has to be determined. As long as the demand shift parameter X remains below X_T , it is better to wait and keep the option open. Therefore, this region $X < X_T$ is called the *waiting region*. In this region, for t < T such that $X(t) < X_T$, the value of the option is given by

$$F(X|X < X_T) = e^{-rdt} \mathbf{E}[F(X) + dF(X)].$$
(3.11)

At the moment (*T*) the demand shift parameters *X* equals X_T for the first time, the demand is sufficiently high to make it optimal to invest. In this *investment region* $X \ge X_T$, the value of the option becomes equal to the return of the investment. This return will depend on the demand at the moment of investment (as determined by *X*) and the installed capacity *K*. This capacity will be chosen such that the return of the investment is maximised. This implies that for $t \ge T$ where $X(t) \ge X_T$, the value of the option is given by

$$F(X|X \ge X_T) = \max_{K} [V(X, K) - I(K)],$$
(3.12)

with *V* the project value. This project value equals the present value of the annual profits (or losses) resulting from the project after it has been installed.

The model can be summarised as follows. The objective function for the investment maximises the expected future discounted profit stream minus the investment outlay of the project with respect to the timing *T*, where $X(t = T) = X_T$, and the capacity *K* (Huberts et al., 2015). If $X(t = 0) < X_T$ so that it is not optimal to invest from the beginning, this investment problem objective function is given by

$$\max_{T \ge 0, K \ge 0} \mathbf{E} \left\{ [V(X_T, K) - I(K)] e^{-rT} | X(t=0) = X \right\},$$
(3.13)

with the value of the investment project, which is assumed to have an infinite life time due to maintenance outlays: ∞

$$V = \mathbf{E} \int_{0}^{\infty} \max_{q} \{\pi(T+\tau)\} \mathrm{e}^{-r\tau} \mathrm{d}\tau, \qquad (3.14)$$

the (inverse) demand function where *X* is following a GBM determined by drift μ and variability σ , specified as follows:

$$p = X - Bq - AX \frac{q}{K^2},$$

the total cost function

$$TC = c q + c_h K$$

and the investment cost function

$$I = FC + \gamma_1 K - \gamma_2 K^2 + \gamma_3 K^3 + \gamma_4 K^4.$$

The next section determines numerical values for the model parameters, which are needed to calculate numerical solutions, since analytical solutions are not attainable.¹⁰

3.2 Parameter calibration of a hypothetical numerical example

In order to illustrate the working of the model and calculate solutions for the remainder of this thesis, the theoretical model is applied to the investment decision about a dock of about 8 to 14 million TEU. In this chapter, this dock is to be installed in a region where no port is active yet. The parameters are calibrated using data of projects such as Deurganck-dok or Saeftinghedok in the port of Antwerp (Port of Antwerp, 2016; Vanelslander, 2014) and Maasvlakte 2 in the port of Rotterdam (Zuidgeest, 2009) in the Hamburg - Le Havre range in Western Europe. The derived parameters in this chapter will be retained in the next chapters.¹¹ Through this numerical approach, the impact of these parameters on the optimal investment decision can be derived. Moreover, outcomes of the different models in the different chapters are better comparable.

Given the investment costs discussed in Section 1.5, the cost of the investment in a new port, I(K), is expected to be between one and three billion euro, depending on the size of the investment. In the example, throughput (*q*) and capacity (*K*) are expressed in million TEU per year, while price *p* and costs c(= 1) are in euro per TEU and $c_h(= 0.5)$ in euro per TEU per year (Vergauwen, 2010; Meersman & Van de Voorde, 2014b; Vanelslander, 2014).

¹⁰This is due to the factor X not appearing in each term of the profit function. Hence, the solution for the optimal throughput, the resulting expression for the optimal profit and the expression for V become so complicated that the optimal solution for the option value cannot be expressed analytically.

¹¹Additional parameters for the following models in the next chapters will be discussed when they are introduced for the first time.

Since sufficient data to accurately estimate parameter *B* are not available for this thesis, this parameter is normalised to 1. This is a common approach in RO models, see e.g., Dangl (1999). The impact of a change in *B* is analysed in the sensitivity analysis in Section 3.5. Profit is calculated in million euro. The investment cost I(K) is also expressed in million euro. The values for drift ($\mu = 0.015$) and drift variability ($\sigma = 0.1$ to 0.2) can be validated empirically in the port context. Regressing an exponential growth model on annual container throughput of significant existing ports in Antwerp and Rotterdam between 2010 and 2015 (Vlaamse Havencommissie, 2016) resulted in a growth rate of between 1.5% and 2% [p < 0.05] for this port area. The root of the squared error of the regression led to a standard deviation of 15% to 17%, justifying the 10% to 20% interval for σ . Both parameters are however varied in the analysis. In transport infrastructure studies, a discount rate in the range of 4% (Blauwens, 1988) to 8% (Centraal Planbureau, 2001) is often used. The discount rate r = 0.06 is selected here in the base case, as did Aguerrevere (2003). Since the choice of the discount rate is often subject to much discussion, this value is varied as well in the sensitivity analysis in Section 3.5.

One of the most difficult aspects is determining suitable values for the monetary scale factor *A*. Ultimately, several different values for parameter *A* were selected from a range which produces realistic port investment strategies and occupancy rates.¹² It should be kept in mind that the objective is to show the impact of differing customer aversions to delays on the investment decision, rather than estimating the project's exact *A*. The model of this chapter with its selected parameter values is summarised in Table 3.1. The assumptions behind this model are summarised in Table 3.2.

3.3 Value of the investment project

The first step of the dynamic programming methodology determines the value V(X, K) of the new port project of capacity K, once it is built at the moment the demand shift parameter equals X,¹³ at a cost I(K). For a private port, the project value V(X, K) is the sum of the discounted maximum realised profits (or losses), specified in Equation (3.14), in which the discount rate $r > \mu$ in order to guarantee convergence of the model (Dixit & Pindyck, 1994).¹⁴ Once the investment is made and capacity K is determined and becomes fixed,

¹²For A = 5, X = 37 euro per TEU, K = 11 M TEU p.a., c = 1 euro per TEU and B = 1, the throughput q would equal 7.11 M TEU p.a. This yields an occupancy rate of 68%. The congestion cost per TEU would on average equal about 11 euro per TEU (AXq/K^2). This seems plausible, compared to the value of about 40 euro per container (not per TEU) per hour (De Jong, 2007) and given that congestion will not yet pose too large a problem in this example. Moreover, in the more highly congested ports of Antwerp and Rotterdam, a congestion charge of about 20 euro per container has been imposed temporarily (CONTARGO, 2018). This validates the plausibility of the selected values for A.

¹³Note that as a result of this definition, an X used in V(X, K) is considered de facto as a threshold value (X_T) .

¹⁴Without this convergence, discounted future cash flows would grow infinitely over time, resulting in eternal postponement of the investment. This behaviour is similar to the reaction of customers in periods of deflation.

Variables								
p	=	price						
q	=	throughput						
Κ	=	capacity						
Inverse demand func	tion:	$p = X - Bq - AX \frac{q}{K^2}$						
B(=1)	=	slope						
$A (\in [4;5])$	=	monetary scale factor of congestion cost						
Demand shift parameter X: $dX(t) = \mu X(t) dt + \sigma X(t) dZ(t)$								
t (=annual)	=	time horizon						
Ζ	=	standard Wiener process						
$\mu(=0.015)$	=	drift of Z (i.e., growth)						
$\sigma(\in [0.1; 0.2])$	=	drift variability of Z (i.e., uncertainty)						
Total cost $TC = cq + c$	$c_h K$							
c(=1)	=	constant marginal operational cost						
$c_h(=0.5)$	=	cost to hold one unit of capacity in place						
Investment cost $I = FC_I + \gamma_1 K - \gamma_2 K^2 + \gamma_3 K^3 + \gamma_4 K^4$								
$FC_{I}(=80)$	=	fixed investment cost						
$\gamma_1(=180)$	=	first-order coefficient						
$\gamma_2(=19)$	=	coefficient reflecting economies of scale of investment cost						
$\gamma_3(=0)$	=	omitted third-order coefficient						
$\gamma_4 (= 0.12)$	=	coefficient posing boundary to maximum project size						

Table 3.1: Overview of Chapter 3's model and the selected parameters.

Table 3.2: Overview of Chapter 3's model assumptions.	Assumption(s)	New portint?Hard constraintPrivate service port*?No (1 port in model)Not considerednt?Project installed at once: available after completion	<i>ions</i> Price (sum of port dues and terminal tariff) transparency \rightarrow Linear additive inverse demand function Demand uncertainty, modelled using a GBM with constant, independent growth (μ) and uncertainty (σ) parameters Growth market with $r > \mu > 0$, to ensure RO model convergence** Poisson-distributed ship arrivals at the port: delays start at about 50% and increase sharply beyond 75-80% occupancy User congestion costs depend on dock users' average value of time, which evolve with the market (i.c., increasing) Investment cost: increasing fourth-order polynomial with economies of scale and size boundary Deterministic, linear operational and capacity holding costs Flexibility of investment timing and size \rightarrow Option to postpone the investment Once-and-for-all investment decision with flexible throughput levels Infinite project life time (maintenance executed)	*: Owned and operated by one single actor. ion $r > \mu$ leads to a finite timing decision, instead of infinite investment postponement. Sensitivity tested with negative growth rates as well. design capacity for container throughput services. Other port services disregarded (sufficiently present or realised by actors other than the PA or TOC). Focus on lumpy investment in port infrastructure and irreversible elements of superstructure.
		Project type? Capacity constrain Port type? Port competition? Time to build? Phased investment	General assumptic	**: Conditio SCOPE: Theoretical de

the instantaneous profit under different values for the demand shift parameter *X* can only be altered and maximised through the pricing mechanism, to yield an optimal throughput quantity q^{opt} . The optimal throughput quantity at each point in time is given by the first-order condition for maximising π in Equation (3.7) and satisfies the second-order condition. It depends on the current value of the demand shift parameter *X* and the installed capacity *K* and is given by

$$q^{opt}(X,K) = \begin{cases} 0, & 0 \le X < c, \\ \frac{(X-c)K^2}{2(XA+BK^2)}, & c \le X < \frac{K(2BK+c)}{K-2A}, \\ K, & X \ge \frac{K(2BK+c)}{K-2A}. \end{cases}$$
(3.15)

Through the boundary values for *X* ensuring $0 \le q^{opt} \le K$ in Equation (3.15), three regions can be identified for *X*:

$$\begin{cases} R_1 = [0, c), \\ R_2 = \left[c, \frac{K(2BK + c)}{K - 2A}\right], \\ R_3 = \left[\frac{K(2BK + c)}{K - 2A}, \infty\right]. \end{cases}$$
(3.16)

In R_1 , no throughput will be handled, since the operational, variable cost will always be higher than the return. In R_2 , q^{opt} increases in X and in K, which means that throughput increases along with demand and installed capacity.¹⁵ Increased throughput as a result of installing more capacity may be a consequence of the congestion cost reduction. In R_3 , optimal throughput is constrained by the theoretical design capacity K. Since X > 0 in a GBM, it can also be seen that R_3 only exists if K > 2A. Otherwise, a fully occupied port is always prevented by price increases, as port users are willing to pay to avoid very high levels of congestion, resulting from fully occupied port capacity. In that case, it always holds that $q^{opt} < K$.

Each region identified has a resulting optimal instantaneous profit (or loss):

$$\pi(X, K, q^{opt}(X, K)) = \overline{\pi}(X, K) = \begin{cases} \overline{\pi}_1(X, K) = -c_h K, & X \in R_1, \\ \overline{\pi}_2(X, K) = \frac{(X - c)^2 K^2}{4(XA + BK^2)} - c_h K, & X \in R_2, \\ \overline{\pi}_3(X, K) = (K - A)X - (BK + c + c_h)K, & X \in R_3. \end{cases}$$
(3.17)

Now, Equation (3.14) can be rewritten as a dynamic programming problem:

$$V(X,K) = \overline{\pi}(X,K)\mathrm{d}t + \mathrm{e}^{-r\mathrm{d}t} \mathbf{E}(V(X,K) + \mathrm{d}V(X,K)).$$
(3.18)

¹⁵As expected and required, $\partial q^{opt} / \partial X > 0$ in R_2 . A higher *X* leads to a higher potential demand. Economic theory states that, if available capacity allows, optimal realised throughput q^{opt} increases when demand shifts upwards and the supply curve has a positive inclination.

Here, *V* is a function of the value for *X* at the moment of investment and the installed capacity *K*. Using Itô's Lemma and the Bellman equation (Dixit & Pindyck, 1994), a differential equation for the project value V(X, K) as a function of the investment threshold and capacity installed is obtained, which must be solved in each region R_i :

$$\frac{\sigma^2}{2}X^2\frac{\partial^2 V}{\partial X^2}(X,K) + \mu X\frac{\partial V}{\partial X}(X,K) - rV(X,K) + \overline{\pi}(X,K) = 0.$$
(3.19)

The solution for j = 1, 2, 3 is given by

$$V(X,K)|_{X \in R_j} = V_j(X,K) = G_{j,1}(K)X^{\beta_1} + G_{j,2}(K)X^{\beta_2} + \overline{V}_j(X,K),$$
(3.20)

where $\overline{V}_j(X, K)$ is a particular solution of differential equation (3.19) with $\overline{\pi} = \overline{\pi}_j$. The factors $G_{j,i}$ are parameters that can be determined through the boundary conditions. The roots β_1 and β_2 satisfy the quadratic equation

$$\frac{1}{2}\sigma^{2}\beta(\beta-1) + \mu\beta - r = 0$$
(3.21)

(see Dixit & Pindyck (1994)), and are equal to

$$\beta_{1} = \frac{\frac{\sigma^{2}}{2} - \mu + \sqrt{\left(\frac{\sigma^{2}}{2} - \mu\right)^{2} + 2r\sigma^{2}}}{\sigma^{2}} > 1, \qquad \beta_{2} = \frac{\frac{\sigma^{2}}{2} - \mu - \sqrt{\left(\frac{\sigma^{2}}{2} - \mu\right)^{2} + 2r\sigma^{2}}}{\sigma^{2}} < 0.$$
(3.22)

All of these considerations lead to the following particular solution in each region R_i :

$$\overline{V}_{j}(X,K) = \begin{cases} -\frac{c_{h}K}{r}, & j = 1, \\ \frac{K^{2}}{4A(r-\mu)}X - \frac{1}{r} \left[\frac{K^{2}}{4A}\left(2c + \frac{BK^{2}}{A}\right) + c_{h}K\right] \\ -\frac{\left(c + \frac{BK^{2}}{A}\right)^{2}}{2\sigma^{2}B(\beta_{2} - \beta_{1})} \left[\frac{1}{\beta_{1}}{}_{2}F_{1}\left(1, -\beta_{1}; 1 - \beta_{1}; -\frac{AX}{BK^{2}}\right) \\ -\frac{1}{\beta_{2}}{}_{2}F_{1}\left(1, -\beta_{2}; 1 - \beta_{2}; -\frac{AX}{BK^{2}}\right) \\ -\frac{\Gamma(\beta_{1})\Gamma(1 - \beta_{1})(AX)^{\beta_{1}}}{(BK^{2})^{\beta_{1}}}\right], & j = 2, \\ \left\{\frac{K - A}{r - \mu}X - \frac{BK^{2} + cK + c_{h}K}{r}, & j = 3, \end{cases}$$

$$(3.23)$$

with $_2F_1$ () hypergeometrical functions. Appendix A contains more details on the calculation of the solution to differential equation (3.19).

Six¹⁶ boundary conditions are imposed (see Dixit & Pindyck (1994) and Dangl (1999))

¹⁶If R_3 does not exist, the final two conditions can be omitted, the second condition holds for $\overline{V}_2(X, K)$, and $G_{2,1}$ and $G_{3,2}$ both equal zero.

in order to calculate the factors $G_{i,i}$:

$$\begin{aligned}
V(0,K) &= \mathbf{E} \int_{0}^{\infty} -c_{h} K e^{-rt} dt = \frac{-c_{h} K}{r}, \\
\lim_{X \to +\infty} (V(X,K) - \overline{V}_{3}(X,K)) &= 0, \\
\lim_{X \to c} V(X,K) &= \lim_{X \to c} V(X,K), \\
\lim_{X \to c} \frac{\partial V}{\partial X}(X,K) &= \lim_{X \to c} \frac{\partial V}{\partial X}(X,K), \\
\lim_{X \to c} V(X,K) &= \lim_{X \to c} V(X,K), \\
\lim_{X \to \frac{K(2BK+c)}{K-2A}} V(X,K) &= \lim_{X \to \frac{K(2BK+c)}{K-2A}} V(X,K), \\
\lim_{X \to \frac{K(2BK+c)}{K-2A}} \frac{\partial V}{\partial X}(X,K) &= \lim_{X \to \frac{K(2BK+c)}{K-2A}} \frac{\partial V}{\partial X}(X,K).
\end{aligned}$$
(3.24)

The first condition implies that $G_{1,2} = 0$ and the second that $G_{3,1} = 0$. This leaves the final four (value matching and smooth pasting) conditions to define the values for $G_{1,1}$, $G_{2,1}$, $G_{2,2}$ and $G_{3,2}$ (see Hagspiel et al. (2016)). $G_{1,1}$ represents the option to start production, $G_{3,2}$ the option of flexible throughput levels below full capacity and $G_{2,1}$ and $G_{2,2}$ are two correction factors in R_2 . The $G_{j,i}$ are dependent only on K, since X is evaluated in c and $\frac{K(2BK+c)}{K-2A}$.

The port investment project value for a numerical example is displayed (partly) in Figure 3.6 as a function of the chosen threshold *X* with a fixed capacity *K*. This figure shows that in a situation with higher variability σ , the project value will be slightly higher, ceteris paribus. This is because the management of the port is able to take advantage of the higher upside potential and can avoid the larger downside potential by reducing or stopping operations when port price *p* falls too low. When congestion poses less of a problem to the customers, as expressed by a lower value of time, reflected by a lower *A*, the value of the project will also be higher. This is because the available capacity can be used at higher occupancy rates, without imposing a too high generalised cost ρ on the shipping lines. Moreover, the project value will be higher when investment is made at a higher threshold for *X*, i.e., a higher willingness to pay, and higher revenues can be generated from the outset. The negative value of the project at *X* = 0 is caused by this point being an absorbing barrier of the GBM process of *X*, resulting in a port revenue being equal to zero forever, in combination with a capacity holding cost $c_h K$.

Moreover, adding extra capacity increases the project value when capacity is low. This is because at high occupancy rates, due to high demand and low capacity on the left hand side of the graph, additional capacity would sharply reduce congestion. This has a high marginal value. When there is sufficient capacity and congestion is sufficiently low, adding more capacity only reduces value. Large sums of money are invested in capacity that is not required, as current port occupancy rates and congestion are sufficiently low for the customers. Hence in such a situation, additional congestion reductions as a result of additional capacity are much more costly than beneficial, because it is costly to hold this capacity at a cost c_h . Dangl (1999) shows in this light that the value of an additional unit of capacity K



Figure 3.6: Value of the project as a function of *X* at the moment of investment.

can be calculated in each region R_i as

$$\frac{dV(X,K)}{dK}\bigg|_{X\in R_{j}} = \frac{dG_{j,1}(K)}{dK}X^{\beta_{1}} + \frac{dG_{j,2}(K)}{dK}X^{\beta_{2}} + \frac{\partial\overline{V}_{j}(X,K)}{\partial K}.$$
(3.25)

This can also be written as

$$\nu(X,K)|_{X \in R_j} = \nu_j(X,K) = g_{j,1}(K)X^{\beta_1} + g_{j,2}(K)X^{\beta_2} + \overline{\nu}_j(X,K).$$
(3.26)

v(X, K) can subsequently be determined by the conditions derived from (3.24).

$$\begin{cases} \nu(0,K) = \frac{-c_h}{r}, \\ \lim_{X \to +\infty} (\nu(X,K) - \overline{\nu}_3(X,K)) = 0, \\ \lim_{X \to c} \nu(X,K) = \lim_{X \to c} \nu(X,K), \\ \lim_{X \to c} \frac{\partial \nu}{\partial X}(X,K) = \lim_{X \to c} \frac{\partial \nu}{\partial X}(X,K), \\ \lim_{X \to \frac{c}{K-2A}} \nu(X,K) = \lim_{X \to \frac{K(2BK+c)}{K-2A}} \nu(X,K), \\ \lim_{X \to \frac{K(2BK+c)}{K-2A}} \frac{\partial \nu}{\partial X}(X,K) = \lim_{X \to \frac{K(2BK+c)}{K-2A}} \frac{\partial \nu}{\partial X}(X,K). \end{cases}$$
(3.27)

The resulting graph for v(X, K) is presented in Figure 3.7, showing that the marginal value of the project with respect to capacity is higher when shipping lines have a lower value of time (as expressed by *A*). This difference grows with the capacity utilisation rates.¹⁷ In

¹⁷When the maximum willingness to pay (*X*) is higher and *K* is fixed, capacity utilisation will be higher, as q^{opt} will increase.



Figure 3.7: Marginal project value of capacity as a function of *X*.

the case of a lower value of time, the additional capacity K can relatively be more exploited and thus generates more additional revenue. Moreover, the figure shows that the marginal project value with respect to K is increasing in X, because the willingness to pay of the port customers increases with X. The marginal value is also higher when uncertainty is higher, because extra capacity is required to serve the potentially higher upward deviations in the demand (see Equation (3.2) and Figure 3.5), contributing to a higher option value under higher uncertainty.

3.4 Investment decision: optimal size and timing

In this section, the numerical values for the parameters and the calculated value of a new port project at the time of investment are used as inputs. Based on this, the optimal capacity and timing threshold are determined in order to maximise the expected discounted profit stream minus the investment cost. This allows identifying the ideal point in the trade-off between gaining additional information and foregoing income by delaying investment in the new port. When the investment is postponed, the port cannot earn a profit from handling cargo. However, when the port knows the demand is sufficiently high, there is a wedge to protect the port from negative demand fluctuations and guarantee sufficient profitability.

The option value included in the option to postpone the port capacity investment can



Other parameter values: B = 1, c = 1, $c_h = 0.5$, $\mu = 0.015$, r = 0.06, $FC_I = 80$, $\gamma_1 = 180$, $\gamma_2 = 19$, $\gamma_3 = 0$, $\gamma_4 = 0.12$.



be written as

$$F(X) = \max\left\{ e^{-rdt} \mathbf{E}[F(X) + dF(X]), \max_{K}[V(X, K) - I(K)] \right\},$$
(3.28)

where the outer maximisation indicates whether the investment should be made or not yet (Dangl, 1999). In this way, the option F(X) can be regarded as a trade-off between waiting longer to achieve a higher project value V, which however comes at the expense of higher discounting.

First, to obtain the new port's optimal capacity K^* for a given level of X at which investment takes place, a now or never (NPV-type) investment decision is examined, which is the inner maximisation of Equation (3.28). This is equivalent to obtaining the value $K = K^*(X)$ such that

$$\nu(X,K) - \frac{\mathrm{d}I(K)}{\mathrm{d}K} = 0.$$
 (3.29)

The resulting curves $K^*(X)$ for various values of *A* and σ are plotted in Figure 3.8.

This analysis is relevant for ports to know their optimal design capacity K^* , should they wish to invest at an exogenously determined moment. Numerical calculations show that optimal capacity K^* is higher when the investment takes place at a higher X – in other words, later. Postponing port investment ultimately leads to a larger investment, due to the once-and-for-all nature of the investment decision. If demand is higher, a larger capacity investment is required. It is also best to invest in a larger port when uncertainty is higher, because more spare capacity should be made available for two reasons. First, more capacity allows the port to handle higher upward deviations from the expected market demand for throughput. Second, it reduces the likelihood that these extra port users will face waiting time caused by congestion (Meersman & Van de Voorde, 2014a). If the project is



Figure 3.9: Optimal capacity K^* as a function of $X \in R_2$, with $I = I_{\text{Dangl}} = 220K^{0.6}$.

characterised by a higher average aversion to waiting, as expressed by a higher A, the port invests in less capacity. This is the result of two opposing impacts. On the one hand, with a higher A, capacity is needed more to reduce the occupancy rate and the costly congestion. On the other hand, however, the project value appears less attractive because of the reduced willingness to pay of customers and the required lower occupancy rates of the installed capacity. The latter implies an incentive for the port to invest in less capacity. The outcome of the trade-off is dependent on the specific investment cost function. Here, there is an inverse relationship between A and $K^*(X)$. Under the investment cost function I_{Dangl} of Dangl (1999) for a production plant however, the outcome of this trade-off would be the opposite, as is illustrated in Figure 3.9. Moreover, due to the sharp port investment cost increase beyond the boundary to the investment size, the increase in optimal capacity $K^*(X)$ is decreasing in X. Indeed, the characteristics of the port area with its size limit temper the positive relationship between the size of the project and the demand for port services. This is not the case with the investment cost function I_{Dangl} , where there was no boundary to the investment size considered.

In addition to inner maximisation, outer maximisation of Equation (3.28) must also be solved. The option to delay investment with flexible throughput from Equation (3.28) satisfies a second-order differential equation (Dangl, 1999):

$$\frac{\sigma^2}{2}X^2\frac{\partial^2 F}{\partial X^2}(X,K) + \mu X\frac{\partial F}{\partial X}(X,K) - rF(X,K) = 0.$$
(3.30)

The solution is given by

$$F(X) = H_1 X^{\beta_1} + H_2 X^{\beta_2}.$$
(3.31)

No particular solution is required here, because there is no profit from the port project before investment takes place, and the β 's equal those in Equation (3.22). The conditions



Figure 3.10: Optimal threshold $X_T^* \in R_2$ as a function of *K*.

required to find factors H_1 , H_2 and the solution for state variable X (i.e., $X_T^*(K)$) are:

$$\begin{cases} F(0) = 0 \to H_2 = 0, \\ H_1 X_T^{*\beta_1} = V(X_T^*, K(X_T^*)) - I(K(X_T^*)), \\ \beta_1 H_1 X^{\beta_1 - 1} = \frac{\partial}{\partial X} (V(X, K(X)) - I(K(X))) \Big|_{X = X_T^*}, \end{cases}$$
(3.32)

in which the last two conditions, known as value matching and smooth pasting, guarantee a continuous option value function (Dixit & Pindyck, 1994). The solution of this system is the optimal investment threshold X_T^* as a function of K, which can be useful for determining the optimal investment timing of a new port with a predetermined capacity.

A graph with numerical solutions is presented in Figure 3.10. The graph shows that larger investment projects are rolled out later, when demand is high enough to operate the port capacity at a sufficiently high price. It also shows that higher variability leads to later investments, as obtaining more information about the evolution of demand bears an option value. Both observations are consistent with S. Zheng & Negenborn (2017). Additionally, it can be seen that the higher the cost of delays, as expressed by *A*, the later the investment. This is because, in order to provide more of the costly spare capacity required in this case, more profit is needed. This is reflected by a higher threshold value for *X*.

In the next step, combining the optimal timing and size functions in the system

$$\begin{cases} X_T^{**} = X_T^*(K^{**}), \\ K^{**} = K^*(X_T^{**}), \end{cases}$$
(3.33)



Figure 3.11: Comparison between the option value F(X) and the value of the project minus the investment cost V(X, 11.17) - I(11.17) for A = 5 and $\sigma = 0.1$ in a private port.

yields the optimal flexible investment strategy (X_T^{**}, K^{**}) wherein both the timing and size of the investment are optimised. Using this solution, the option value (F(X)) and value of the project $V(X, K^{**}) - I(K^{**})$ can be plotted as a function of X, with K fixed at K^{**} . Figure 3.11 gives an example for A = 5 and $\sigma = 0.1$. In the example, an optimum can only be found in R_2 , where capacity is not fully utilised. However, given the once-and-for-all nature of the investment, this will not prevent q^{opt} from eventually equalling K when X becomes large enough to reach R_3 , according to Equation (3.15).

Figure 3.11 shows that in ports too, the general real options theory holds. The option to postpone investment bears a value and leads to investment postponement until $V \ge I + F$, as opposed to the traditional NPV rule V > I (Dixit & Pindyck, 1994).¹⁸ The new port in the example will dispose of capacity K^{**} (= 11.17 million TEU per year), which will be installed at threshold X_T^{**} (= 37.63 euro per TEU), leading to an optimal throughput at the moment of investment of 7.3 million TEU per year. The initial occupancy rate will hence be 65%, which is indeed below full occupancy. The initial price per TEU after investment, according to Equation (3.4), would be 19.31 euro, which leads to obtaining a discounted project value of 3.35 billion euro at the time of investment and compensating for the project's investment costs of 1.59 billion euro. It is interesting here to note that through the inverse demand function and given the optimal capacity, X_T^{**} can be converted into an optimal throughput quantity (in million TEU per year) that needs to be demanded at a certain price (in euro per TEU) before investment takes place. This is a more practical decision variable for port

¹⁸The RO rule leads to postponing the investment to $X_T = 37.63$ euro per TEU, as opposed to $X_T = 22.49$ euro per TEU under NPV.

managers.

In order to further analyse the impact of congestion costs and uncertainty on the investment decision, a number of other investment pairs (X_T^{**} , K^{**}) are calculated numerically for altered values of A and σ , as shown in Table 3.3.

A crucial question to be answered is how the port investment decision is influenced by congestion costs under uncertainty. If congestion costs were not considered (A = 0), the new port would be too small and it would be installed at a threshold X_T^{**} that is too low. Table 3.3 demonstrates this major result.

Result 3.1. Under the assumptions given in Table 3.2, ports with customers that are on average more waiting-time averse, should invest in more capacity, but later.

This observation is the result of three effects with conflicting outcomes. First, an increase in the monetary scale factor of congestion, A, leads to a lower profit. This makes the investment less attractive, resulting in a lower capacity K^* . Second, the port may also wish to wait longer before investing to increase the project's value, resulting in a higher value for threshold X_T^* . Due to the positive inclination of the $K^*(X)$ -function, this results at the same time in an increase in the optimal capacity K^* . The final effect is a direct increase in capacity K^* , since congestion now poses a greater problem for port users and lower occupancy rates q/K are required. The final two effects dominate, since capacity has a higher marginal value when it is not fully occupied. Moreover, increasing the size of the project implies taking more advantage of the investment size scale economies.

Regions that want to invest in a new port, should take this result into account. If the future customers are very waiting-time averse, it is better to wait longer before building the new port. In this way, more (spare) capacity can be provided. At the same time, the investment size scale economies are exploited more by the investors. However, if the customers' value of time is lower, the investors optimise their expected present value by investing sooner in less capacity, which will however be occupied more.

The second result from the table is consistent with a well-known RO finding and is here proven to hold in a port environment as well (Dangl, 1999; Huisman & Kort, 2015):

Result 3.2. Under the assumptions given in Table 3.2, higher demand uncertainty leads to later but more extensive investments in port capacity.

When uncertainty is lower (or zero), the option value of waiting will be lower.¹⁹ Consequently, the investment in less port capacity will take place at a lower threshold value.

¹⁹It should be noted that even in the deterministic case where $\sigma = 0$, there is an option value inherent to waiting when the project value and investment costs differ little, as shown by Dixit & Pindyck (1994, p. 138-139). In real terms, the investment cost is reduced by e^{-rt} , whereas the returns over time decrease by only a factor $e^{-(r-\mu)t}$.

		Uncertainty		
		$\sigma = 0.10$	$\sigma = 0.15$	$\sigma = 0.20$
Congestion	A = 0	(18.61, 8.15)	(22.82, 9.20)	(24.40, 10.31)
monetary	A = 4	(33.67, 10.83)	(44.74, 12.21)	(63.27, 13.98)
scale	A = 4.5	(35.96, 11.02)	(48.44, 12.55)	(70.60, 14.53)
factor	A = 5	(37.63, 11.17)	(52.32, 12.88)	(78.97, 15.10)

Table 3.3: Optimal private port investment decision (X_T^*, K^{**}) under different values of A and σ .

Other parameter values: $B = 1, c = 1, c_h = 0.5, \mu = 0.015, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12.$

Less capacity is required to handle smaller upward deviations from the expected market demand for throughput, leading to congestion. Hence, when demand uncertainty can be reduced, the new port installed at optimal conditions will be sooner available.

The sensitivity of the investment decision to the other parameters determining the economic and operational context of a port requires additional attention and analysis. This is the focus of the next section, which considers operational costs c, investment costs I, economic growth μ and the steepness of the demand curve as influenced by B.

3.5 Influence of the economic and operational context on the optimal investment strategy of the port: sensitivity analysis

This section discusses the sensitivity tests of the results to changes in the other parameters included in the model, which are operational costs, investment costs, economic growth and the steepness of the demand curve. It is explored how the optimal capacity (K^*) and timing (X_T^*) for the new port, displayed in Figure 3.11, alter individually following a limited increase in a single parameter and what the new optimum (X_T^{**} , K^{**}) will be.

Table 3.4 shows some important results. First of all, the direct impact of the increase on the optimal level of capacity $K^*(X)$ at a given moment is as expected for all parameters: higher operating, capital and investment costs lead to investing in a smaller port, since they reduce the project's attractiveness. One exception is the fixed investment cost parameter FC_I . Since it is a constant in the investment cost function I(K), it has no influence on the optimal capacity K, which is calculated through a derivation of I(K) with respect to K. A steeper demand curve caused by a higher value for B also leads to a lower investment size K, as the market is less profitable and the port can then alter its price more freely without attracting or losing many customers. If B is lower, the port stands to gain by reducing the price slightly and attracting additional customers. For this to work, however, sufficient capacity is required. Investment size is also lower for a higher discount rate, since future revenues will be of lower value, especially compared to the immediate investment cost. In
Parameter change	Timing $X_T^*(K=11.17)$	Size <i>K</i> *(<i>X</i> = 37.63)	Decision (X_T^{**}, K^{**})			
Base case	37.63	11.17	(37.63, 11.17)			
c +10%	37.79 (+)	11.16 (—)	(37.87 (+), 11.19 (+))			
c_h +10%	37.77 (+)	11.16 (—)	(37.85 (+), 11.19 (+))			
$FC_I + 10\%$	37.75 (+)	11.17 (=)	(37.86 (+), 11.20 (+))			
γ_1 +10%	40.59 (+)	11.04 (—)	(42.39 (+), 11.62 (+))			
γ_2 +10%	41.11 (+)	10.88 (—)	(42.29 (+), 11.45 (+))			
$\gamma_4 + 10\%$	40.38 (+)	10.74 (–)	(39.52 (+), 10.97 (-))			
r +10%	39.74 (+)	10.74 (–)	(38.43 (+), 10.83 (-))			
μ +10%	37.37 (–)	11.34 (+)	(38.50 (+), 11.45 (+))			
B +10%	38.70 (+)	11.01 (—)	(38.51 (+), 11.12 (-))			
Base case parameter values:						

Table 3.4: Reaction of the optimal $X_T^*(K)$, $K^*(X)$ and (X_T^{**}, K^{**}) to a 10% increase in the other parameters.

 $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12.$

contrast, a higher expected average economic growth (μ) leads to investment in a larger port. This finding explains why the growing countries such as China have been investing in mega-ports.

Moreover, a parameter change which has a positive direct impact on optimal capacity $K^*(X)$ has a negative direct impact on optimal timing threshold X_T^* , and vice versa. This is plausible, because the response to a positive economic effect might be either to increase the size of a project at a given moment (to realise more of the feasible profit, growing with the economy), or to implement a project sooner (X_T^* can be lower to obtain the same profitability). Nonetheless, there is one exception. Although an increase in FC_I has no impact on $K^*(X)$, it does have a positive impact on $X_T^*(K)$. If the fixed expenditure for making an investment increases, a port may wish to postpone investment, so that higher revenues can compensate for this increased fixed cost.

As opposed to a negative direct effect, the indirect effect of an increase in capacity (K^*) on the optimal timing is positive, because $K^*(X)$ and $X_T^*(K)$ are both increasing functions. Hence, if size and timing are both considered jointly, the net effect on the (X_T^{**}, K^{**})-equilibrium is ambiguous. One of the variables always moves in the direction of the individual direct impact. The evolution of the other variable depends on the relative sizes of the indirect and direct effects. This is determined by the relative positions and slopes of the $X_T^*(K)$ and $K^*(X)$ functions when a specific parameter is altered. This is illustrated in the example of Figure 3.12, which shows an economic change that drives $K^*(X_T)$, capacity as a function of a chosen threshold, down (direct effect on capacity). This moves the equilibrium down left along $X_T^*(K)$ (indirect effect). But also because of this change, $X_T^*(K)$ increases and the



Figure 3.12: Evolution of the (X_T^{**}, K^{**}) -equilibrium following a parameter change.

curve moves to the right as a result of the direct effect on the threshold value for the demand shift parameter. According to the new position of $X_T^*(K)$, the new equilibrium is in quadrant I, III or IV. Only a solution $X_T^*(K)$ in quadrant II would be impossible to reach, as this would move both X_T^* and K^* in a direction opposite to their expected direct effect.

Table 3.4 contains a number of results concerning the sensitivity of the model, based on the numerical simulations. After numerically simulating with a lot of different parameter settings, the reaction of the new optimum to a change of one parameter in Results 3.1, 3.2 and 3.3 turns out to be independent of the other parameter choices. The changes of the new optimum in the other results however do depend on the specific parameter settings.

Result 3.3. Under the assumptions given in Table 3.2, higher port costs result in later investments in more capacity.

Result 3.4. If the discount rate r increases, smaller and/or later port capacity investment will be made. The impact on the new optimum depends on the specific parameter values and holds under the assumptions given in Table 3.2.

Result 3.5. If the growth rate increases, it is much more attractive to invest in a larger port and/or at an earlier moment. The impact on the new optimum depends on the specific parameter values and holds under the assumptions given in Table 3.2.

Result 3.6. A steeper demand curve leads to smaller and/or later investment in port capacity. The impact on the new optimum depends on the specific parameter values and holds under the assumptions given in Table 3.2.

To further explore Result 3.5, the impact of a negative growth rate is examined in a scenario with $\mu = -0.005$. Here, the optimal investment decision would equal (44.30 euro per TEU, 10.32 million TEU per year). The port would be much smaller than in the case of positive growth, since the project would be less attractive due to expected decreasing profits

Discount rate	Decision (X_T^{**}, K^{**})	
r = 0.03	(87.81, 21.88)	
r = 0.04	(43.96, 14.27)	
r = 0.06	(37.63, 11.17)	
<i>r</i> = 0.08	(41.24, 10.36)	
Other parameter values:		

Table 3.5: Optimal investment decision (X_T^{**}, K^{**}) using a different discount rate.

 $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12.$

over time. In such a situation, less capacity would be needed too, as throughput levels will fall. Additionally, in order to compensate for future decreases in demand, the project would only be implemented at a much higher threshold, which is less likely to be reached. This explains why port investments are unlikely to take place in a decreasing market.

As the discount rate is subject to a lot of discussion in the literature, it is varied in Table 3.5 between the usual 4% and 8%. Recently, however, interest rates have been falling. Therefore, also the investment decision at 3% is calculated.

A higher discount rate r leads to a decreased net present value of future profits, hence reducing the value of the project and making the investment less attractive. It can be seen from the results that increased discount rates lead to smaller investment sizes. If the discount rate is small, an increase in the discount rate leads to lower a threshold X_T^{**} . For higher discount rates, the threshold will rise with an increased discount rate (see from r = 0.06 to r = 0.08).

If the discount rate is very low, it becomes very profitable to invest in more capacity, even to the extent that more capacity is installed than space allows. This is only feasible for the port if additional land and property is purchased. Hence, costly expropriation to extend the port area will take place to install the large amount of capacity. Since a high net growth rate is a consequence of the low discount rate, it is a good strategy to make the investment only at a large threshold for X so that the option value in the project can be exploited fully. Yet this observation should be interpreted with caution, for two reasons. First, at low discount and growth rates, the investment decision is also very sensitive to the growth rate. A growth rate of 1% in combination with a discount rate of 3% results in an investment in a capacity of 13.64 million TEU per year at threshold $X_T^{**} = 34.26$ euro per TEU, which is much more similar to the base case decision. Second, at historically low interest and hence discount rates, these rates can be expected to rise again in the future. Such volatility has a significant negative impact on investment size (Cassimon et al., 2002) and moderates the increase in investment triggered by low discount rates. Such discount rate volatility was however not included in this chapter's real options model.

$\sigma = 0.10$ $\sigma = 0.11$
) 12 10) (40 75 12 5
2, 12.10) (40.75, 12.5
3, 11.17) (39.87, 11.4
4, 10.67) (41.01, 10.9
1

Table 3.6: Optimal private port investment decision (X_{T}^{**} , K^{**}) under different values of r and σ .

Another macroeconomic phenomenon that requires attention is the relationship between interest rates and uncertainty. When uncertainty rises, households increase precautionary savings, which in turn has a negative effect on interest rates (Hartzmark, 2016). The impact of these combined effects can be seen in Table 3.6. As discussed before, both an increase in uncertainty and a decrease in the interest rate, used as the discount factor, lead to an increase in the size of the port investment. As a result, the outcome of the combined effect is a considerable increase in the project size. Although increased uncertainty leads to investment postponement, the impact of the discount rate on timing was already shown to be ambiguous and dependent on the size of the discount rate. Therefore, the net impact of the combined effects on timing depends on the size of the interest rate as well.

3.6 Conclusion of this chapter

In order to account for the uncertainty, irreversibility and lumpiness of port capacity investments, a real options model has been built in this chapter to derive a first decision rule to find the optimal investment size and timing of the investment in a new port, fully operated by one private actor. As an addition to the available literature, the presented model includes the variable aversion to waiting time of the port's users in the investment analysis. Moreover, a continuum of potential investment decisions is considered to determine the optimal decision. To this end, a variable monetary scale factor allows differentiating the impact of waiting between different goods categories and port users involved. Moreover, the impact of uncertainty and other project and economic variables on the optimal investment decision has been analysed.

The main conclusion of this chapter is that considering the congestion cost in the analysis under uncertainty alters the investment decision considerably. If the congestion cost were not considered (A = 0), a new port that would be too small would be installed too early. As the examples here show too, the port will invest in more capacity at a later time through a higher threshold if the port customers are more waiting-time averse. When the uncertainty or port costs are higher, the port authority is advised to postpone investment and invest in more capacity as well, as this allows awaiting sufficient market information to become available to justify the investment. Moreover, waiting longer for a larger market provides the required wedge to protect against the downside of this uncertainty. Finally, increased net economic growth leads to a larger, later investment, as the investment project will be much more attractive.

The presented first real options model of this thesis serves as the base case for the next chapters. However, some of the assumptions of this chapter need relaxing in the following chapters, in order to deal with different situations. These outcomes can then be compared with the findings of this chapter and related to the impact of the different assumptions. For example the complicating impact of the port's ownership structure on the investment decision requires attention. When governments are involved, local benefits and consumer surplus as part of social welfare are frequently considered in the investment decision (Xiao et al., 2012). For governments, investment postponement not only involves foregoing profit, but also other economic effects are foregone and economic growth may be hampered. The specific distribution of the different revenues and costs between the port authority and terminal operating companies might alter the final investment decision as well. These impacts of public ownership and of the different actors involved in the capacity investment decision are studied in the next chapter.

Chapter 4

Port capacity investments with two actors and public money involved

In the previous chapter, the port capacity investment decision has been studied for a privately owned port, entirely operated by one single actor. In such a service port, the port operator owns the infrastructure and superstructure and is responsible for providing cargo handling services (Trujillo & Nombela, 2000; Slack & Frémont, 2005). In the majority of ports worldwide, the port product is realised by a combination of actors under the landlord model (Suykens & Van de Voorde, 1998). From Chapter 1 it was concluded that the capacity investment decision in such ports is influenced by different actors. This complicates the investment decision in two ways. First, the infrastructure and land are owned by the port authority (PA), while the terminal operating company (TOC) owns the superstructure and handles the cargo under a concession agreement with the PA. A second aspect typical for large and costly infrastructure projects such as port infrastructure is the involvement of public money, because ports create value beyond the border of a port (Brooks & Cullinane, 2007). In this light, Xiao et al. (2012) studied the influence of multiple port owners on capacity investment, showing that private ports tend to invest less in capacity than publicly owned ports. A port with public involvement exhibits a faster expansion path, since profit maximisation is not its only objective (Asteris et al., 2012). They also want to maximise social welfare, value added, and/or employment, which is often linked to the amount of throughput. This offers an explanation for a large number of ports trying to maximise their throughput (Tsamboulas & Ballis, 2014; Jiang et al., 2017). A similar study of Zhang & Zhang (2003) for airports unveiled that publicly owned airports invest sooner in capacity than private airports as well.

The previously discussed papers allowing for different types of ownership do not consider uncertainty. As Chen & Liu (2016), Xiao et al. (2012) and the previous chapter already demonstrated in a port operated by a single actor, uncertainty as well as congestion alter the investment decision a lot and need to be considered in the port capacity investment analysis. Therefore, the objective of this chapter is to analyse how a new port's optimal capacity investment decision under congestion and uncertainty is influenced by the interaction between the PA and the TOC and the involvement of public money. This allows analysing investment cases such as the port of Luanda and the port of Gioia Tauro in its development phase. To this end, the previous chapter's real options model is extended, which allows answering the following research sub-questions in this chapter: (RQ 4.1) How are the investment decisions of the PA and the TOC in new port capacity influenced by each other's decisions under uncertainty without competition? and (RQ 4.2) How are the investment decisions influenced by public PA ownership under uncertainty without competition? The results of this chapter are compared with the optima resulting from the previous chapter's set-up.

The structure of this chapter is as follows. The next section introduces the adaptations to the economic model to incorporate the distinction between the PA and the TOC in a landlord port and the involvement of public money in the ownership of a PA. Section 4.2 explains how the values for the additional model parameters are selected. Section 4.3 shows how the RO model and calculations are influenced by introducing a cooperative game between the PA and the TOC. Section 4.4 discusses the results, followed by a sensitivity analysis in Section 4.5. The conclusion of this chapter is given in Section 4.6.

4.1 Economic setting and methodology

In the type of ports considered here, the PA and the TOC invest in complementary elements of port capacity to realise their individual activities and earn in return different revenues. The sources of these revenues and the investment outlays of both actors are displayed in Figure 4.1. Other TOC income sources than the terminal tariff (e.g., storage) are ignored here, as they do not apply for every unit of throughput handled or they are negligible. Moreover, other activities such as tugging and piloting are assumed to be provided by other actors and left beyond the scope of this analysis. Demand originates from a receiver buying goods from the shipper, who ships them through a shipping line (Coppens et al., 2007). Containers, the pricing base of the TOC that handles them, are carried by shipping lines on their ships, the pricing base of the PA. As the focus is on the supply side with the PA and the TOC, the complexity of the demand side needs simplification. The number of ships can therefore be expressed in terms of the amount of throughput, or vice versa, through a conversion factor. As throughput generates welfare (Xiao et al., 2012), the number of ships is converted to the number of containers in Section 4.2. In this way, demand depends on one single variable, which reduces mathematical complexity.

The objectives of the PA and the TOC often diverge, because of their different activities and type of ownership (Heaver et al., 2000; Meersman et al., 2015; Xiao et al., 2012). As a result, this chapter's model presents two extensions compared with the previous chapter and



Figure 4.1: Revenue sources and investment outlays of the PA and the TOC.

the available literature on port capacity investments and real options. First, the distribution of revenues and costs over the different actors is included in the model. Second, public ownership is accounted for through an extension of the PA's operational objective function with social welfare. Before discussing and quantifying both elements, the assumptions regarding the economic environment and the behaviour of the port are discussed.

4.1.1 Economic assumptions

Like the previous chapter, this chapter deals with the development of a new container port. The dock and terminal are in this chapter assumed to be installed at once and without time to build, by two actors, respectively one PA and one TOC. Since again the port is assumed to be situated in a location where little or no competition is present, the investment decision in only one port is studied. Moreover, the PA and the TOC have the option to make a once-and-for-all investment decision, of which the size and timing are flexible.

Next to these assumptions related to the economic environment and the investment decision, some behavioural assumptions are made in relation to landlord ports studied in this chapter. The assumption is made that the PA has full information about the TOC's price and cost decisions. In reality, the PA often only has full information about the realised throughput of the TOC. The PA could however make decent predictions of prices and costs of an efficient TOC, based on the limited available information about terminal operators already active in other ports and the expected demand. These predictions allow the PA to ex-ante calculate its expected part of the income from terminal operations. This income is realised by the PA through a concession fee, charged to the TOC. This income constitutes a considerable part of the port authority's profit and its correct inclusion in the calculations is required to accurately decide on the optimal size and timing of the infrastructure investment. The specific assumptions regarding the concession agreement and the mathematical model are included in the next subsections.

4.1.2 Differentiating between TOC and PA

This subsection describes how the distinction between the PA and the TOC is made in the economic model. The division of revenues and costs is described first. Next, the modelling of the concession fee and the related assumption of a renewal of the concession agreement are discussed. Finally, it is explained why the omission of time to build leads to assuming the PA and the TOC to install infrastructure and superstructure of the same capacity and at the same time.

The distribution of revenues and costs between both actors

The port customer faces the same demand function as in Equation (3.4) from the previous chapter. Since the price p is the sum of the terminal tariff

$$p_{\rm TOC} = \alpha_1 \cdot p, \tag{4.1}$$

paid by the shipping company to the TOC, and port dues

$$p_{\mathrm{PA}} = (1 - \alpha_1) \cdot p, \qquad (4.2)$$

paid by the shipping company to the PA; it holds that

$$p = p_{\rm TOC} + p_{\rm PA}.\tag{4.3}$$

In this chapter, the TOC and the PA independently set their respective prices for servicing ships and handling cargo, before the capacity is operated. The chosen relative prices determine α_1 , the average share of the terminal tariff in the total price. The calculation of α_1 in practice, using real data and average port occupancy, is illustrated in Section 4.2. This share then remains fixed over time, as both the port dues and terminal tariff are assumed to subsequently follow the evolution of the demand shift parameter *X*, which determines the evolution of the market.¹ A growth of *X* leads to a higher total price *p*. This market growth, with its uncertainty as expressed by a GBM, is distributed accordingly over the TOC and the PA by the shares α_1 and $1 - \alpha_1$. As a result, the customers' reduced willingness to pay due to congestion is also distributed by the same shares α_1 and $1 - \alpha_1$ respectively.

A similar reasoning holds for the operational costs of the TOC and the PA. The shares of the TOC and the PA in the total operating cost cq, with c the constant marginal operational cost, are α_2 and $1 - \alpha_2$ respectively. These shares, like the other division parameters, are constant over time. The operational cost for the TOC encompasses for example labour and electricity, whereas the cost of the PA encompasses among other things administration (Lacoste & Douet, 2013). Also the total investment cost and the capacity holding

¹In reality however, it might happen that the PA and the TOC alter their prices separately over time, e.g., when an actor applies peak-load pricing.

cost $c_h K$ are divided between the TOC and the PA. The investment cost is the same fourthorder function of capacity as in Equation (3.9), which involves fixed investment costs (FC_I), economies of scale in investment size (negative second-order term) and a boundary to the maximum investment size (positive fourth-order term). For the division of the investment cost, the shares α_3 and $1 - \alpha_3$ are used, whereas for the capacity holding cost, they are α_4 and $1 - \alpha_4$ respectively. These express the cost structure differences between superstructure and infrastructure investments.²

Modelling the concession fee

An additional element that needs to be modelled to account for the different actors involved, is the concession fee. The PA grants a TOC the right to exploit a certain area of the port to handle the cargo, in exchange for a fee. The concession agreement stipulates for how long the TOC is allowed to exploit this area, the size of the concession fee, what specific activities will be performed, the minimum amount of activity required (often expressed in terms of throughput, the port's output) and the penalty linked to not meeting these requirements, e.g., the immediate end of the agreement or a penalty fee (Aronietis et al., 2010; Pallis et al., 2008; G. W. Y. Wang & Pallis, 2014).

According to Aronietis et al. (2010) and Meersman et al. (2009), the concession policy of a PA is the most important, or even unique, instrument to keep control over the terminal operators, which have become more powerful, and related shipping companies. Hence, the time length and contractual requirements of the concession agreement should be carefully considered and decided upon by the PA. In this way, the PA both retains sufficient control over the TOC and at the same time allows the TOC to have realised a profit at the end of the concession's life span. Concessions of 30 years including targets that match the common interests of both the PA and the TOC well, are a commonly observed practice. Longer concessions would imply the PA losing a significant part of its (already limited) power over the TOC, whereas shorter concessions would hamper the TOC to fully recover the investment cost during the project lifespan.

In this chapter, one TOC handles all the containers in the port. In return, the PA receives a concession fee from the TOC. Many different concession fee schemes exist, such as a lump sum, an annual fee, a quantity-dependent fee, a percentage of the revenue or a combination of these elements (Saeed & Larsen, 2010). The impact of the different forms of concession agreements is beyond the scope of this chapter. Hence and in order to avoid setting an arbitrary value for the concession fee, it is modelled as the TOC paying a share α_5 of its annual operational profit to the PA.³ In this way, the modelled concession fee values are

²Note that α_1 , α_2 , α_3 and α_4 do not need to be equal, as they encompass different elements of the project cash flows (i.e., revenues and costs). The meaning of these α 's is summarised in Table 4.1.

³A consequence is that the capacity holding cost does not influence the concession fee payment, as it is a fixed cost related to capacity and not to throughput.

endogenous and more realistic. This would not be the case if the concession fee had been based on revenue, since the PA creaming off a too high percentage of the TOC's revenue would leave the TOC with losses and being discouraged to invest.

Setting an arbitrary percentage of the revenue can only be avoided by ex-ante calculating the concession fee as a share of the TOC's operational profit, taking in this way both revenues and costs into account. If operating is profitable for the TOC, it will remain so after paying the concession fee, since only a share of this profit is to be paid. In case the TOC cannot be profitable from operations, it could decide to suspend operations, especially if losses are persistent in the long term. Hence, operational profit will always remain greater than or equal to zero after deducting the concession fee. This is a prerequisite for a meaningful calculation of the concession fee.⁴

In reality, different ways to grant a concession and close the contract exist. According to Farrell (2012), who studied 781 international terminals, different procedures are used in different countries. Examples of recent awarding schemes include open tenders often involving auctions or one- or multiple-round negotiations, direct talks with terminal operators and joint ventures (Farrell, 2012; Meersman, 2005; Notteboom et al., 2012; G. W. Y. Wang & Pallis, 2014). Additionally, new concession contract types are constantly being developed (Tsamboulas & Ballis, 2014; Cruz & Marques, 2012). An important aspect of the concession agreement is that the port passes some of its demand risk, i.e., potential losses caused by the difference between forecasts and the actual realisation of the uncertain demand level, on to the concessionaire. For this, models have been developed to assess concessions under uncertainty (Niu & Zhang, 2013). To obtain a concession, the TOC has to present a business plan, including the expected NPV of the project. This serves as a good basis for the PA to determine the required minimum activity, a correct concession fee and a possible penalty for not meeting the predetermined objectives (G. W. Y. Wang & Pallis, 2014).

In this chapter, it is assumed that the PA has full information about the TOC and that negotiations about the final concession agreement between the PA and the TOC take place before the PA's investment decision.⁵ As a result, the PA can *a priori* estimate the discounted profit of the TOC over time. With this information, the PA can decide on the concession fee, modelled by parameter α_5 , the percentage of the TOC's operational profit to be creamed off, resulting in a discounted income stream. This can subsequently be converted into one of the in reality commonly applied concession fee systems with an equal net present value,

⁴This explains why it is crucial to focus on operational profit and leave the capacity holding cost out. The capacity holding cost is a sunk cost once the investment is made and has no influence on the optimal level of output. As a result, the TOC could decide to operate if operational profit is positive, but total profit is negative, as long as a part of the sunk cost $c_h K$ is recovered. Since total profit could become negative, calculating the concession fee as a percentage of this could lead to the unrealistic case of a negative concession fee. This would be equivalent to the port subsidising a TOC.

⁵In practice, concession agreements are negotiated at a later point. Although this implicit assumption is needed for modelling purposes, the impact on the outcomes is limited when the PA has a good ex-ante insight into the TOC's demand for concessions.

e.g., a lump sum, annual, throughput or revenue based fee (Pallis et al., 2008). This conversion is required to be able to implement the outcome of the ex-ante calculations and outcomes from the model in reality.

Since the concession fee has an impact on the returns of both the TOC and the PA and is set by the PA, the latter can influence its own profit and optimal investment decision as well as those of the TOC. This makes the concession fee an important decision variable for the PA to obtain desired behaviour of the TOC, without resorting to penalties that are in reality difficult to enforce.⁶ In fact, such penalties involve negative consequences for both parties. When the concession agreement terminates prematurely, the port foregoes future throughput and income from this site. Due to the limited number of TOCs available following horizontal and vertical concentration in the market, it is complicated to replace a TOC. For the TOC, suspending the concession agreement involves the loss of the residual value (future profits) from the investment in the irreversible elements of the superstructure, such as paving, warehouses or specific equipment (e.g., cranes) that is very costly to transport. Moreover, not meeting the concession requirements might sometimes be caused by an adverse economic period, which is beyond the control of the TOC's management. A penalty might then induce even higher losses for the TOC than already incurred. As a consequence of assuming the TOC fulfilling the negotiated concession agreement conditions to the extent that the economic situation and demand allow it, future renewals of the agreement are assumed (G. W. Y. Wang & Pallis, 2014). This and the capacity holding expenses to maintain the capacity ($c_h K$ from the previous chapter) lead to assuming an infinite project lifetime in the calculations.

Possible investment and concession strategies for the PA

In the described port, the optimal timing (expressed as threshold X_T) and size (K) of the investment decision of the PA and the TOC may differ, because they have different operational objective functions to optimise. In the framework of Xiao et al. (2015) studying port infrastructure investments preventing disasters, different investments of the PA and the TOC are possible. In this chapter however, the chosen investment size and timing of both actors need to be the same, as they are considered the outcome of a cooperative game. Also Xiao et al. (2015) take the contract of cooperation between the PA and the TOC into account. Since a public party considers aggregated social welfare rather than individual profits, cooperation can be justified. An additional advantage of this approach is that it allows elaborating on how the optimum from the previous chapter can be reached in a landlord port. However, results are compared with a non-cooperative outcome as well, where the PA as a Stackelberg leader has all the power to set the concession fee. Although a PA prefers higher

⁶See for example the case where PSA only had to pay a fraction of the total penalty for handling less cargo between 2009 and 2012 than agreed upon in the concession agreement with the Port of Antwerp. In this case, the reason was a significant exogenous fall in demand due to the financial crisis of 2008.

concession fees, it should set the concession fee in a way that the incentive for the TOC is such that it would not invest too little nor too late.

In a landlord port, the PA and the TOC need each other's efforts to maximise their objectives through throughput generation. The PA's infrastructure investment facilitates the TOC servicing the ships and handling the goods. In the described cooperative game considering a lumpy once-and-for-all investment without time to build or phased investment, it would be disadvantageous for both parties or even impossible to invest at a different moment or in a different size. Infrastructure needs to be installed before superstructure can be installed and the capacity of the infrastructure poses a limit to the capacity of the superstructure. In this way, the decision of the PA can limit the TOC's investment options. If phased investment is not an option and construction lead times are omitted from the analysis, it does not make sense for a PA to invest in more capacity than the TOC's capacity investment under the concession agreement, as it would only cause a loss of money because of unused capacity. The same holds for the PA investing before the TOC, which would result in a period without profits, as the infrastructure would not yet be operated. As a result, the PA would not invest in the infrastructure before the moment the TOC is willing to install the superstructure. Hence, if optimal timing and size differ for both individual actors, their optima constitute an interval from which the unique final investment decision needs to be determined. (Meersman & Van de Voorde, 2014a)

Two possible investment strategies are discerned in this setting, depending on the negotiation power the PA possesses during the concession negotiations. Both are numerically illustrated in Section 4.4. A first investment strategy involves both the TOC and the PA giving in from their individual optima to invest at the aggregate optimum of a service port, which is often comprised in the decision interval. If the PA invests in a size at a threshold deviating from its optimum, it can urge the TOC to deviate as well through the concession negotiation or auction following the concession tender. If the TOC is not willing to deviate, it could be denied the concession. In that case, the PA of course needs to search for a new concessionaire. Under the described cooperative strategy, the concession fee can be interpreted as a redistribution of the project value after the deduction of investment costs (V-I)from one party to another. The concession fee should always be below a critical value that ascertains that the TOC's V - I is sufficient to be willing to invest. It should be positive and exceed the opportunity cost of investing in a different port with a higher V - I. In this way, the concession fee size is market- and competition-dependent. These conditions set, a port wondering what the best concession fee is, could pursue different objectives when determining its concession fee:

- reaching an equal distribution of the entire project's V I over both actors;
- reaching a distribution of V-I according to each actor's share in the operational cost, investment cost or a weighted sum of both;
- reaching a distribution according to other objectives based on the port's strategy, e.g.,

the relative effort made for marketing and attracting port customers;

• equalling the amount of discounted V - I given up by each actor to deviate from the individual optimum to the agreed decision.

An important consideration when following this first investment strategy is that incentives for the TOC to deviate from the concession agreement should be avoided or at least minimised. Such incentives are present if the optimal size and/or threshold of the TOC are respectively below and/or above the optima of the PA.

A second investment strategy for the PA to deal with deviating optima in the described scenario could be to urge the TOC to decide at the same time to invest in the same amount of capacity through economic incentives.⁷ If the PA invests later or in a smaller amount than the TOC's optimum, the TOC is urged to adapt its strategy by taking this limiting investment decision variable as given for its own investment decision. The remaining decision variable can then be optimised conditionally on the already fixed variable. With a carefully selected concession fee, the TOC's conditional optimal value for this remaining decision variable will equal the PA's optimum. This competitive strategy hence also leads to the TOC and the PA investing in the same amount of capacity at the same market threshold. The downside of this strategy however is that less profit is realised at port level, aggregated over TOC and PA, as it differs from the global optimum in a service port.

4.1.3 Allowing for both public and private ownership of the PA

The analysis in Chapter 3 considered a private port. In reality however, very few ports are privately owned. Very often, public money is involved in a landlord port with a (partly) publicly owned PA (Suykens & Van de Voorde, 1998). If a government is involved, the PA will not maximise profit, but social welfare (*SW*), as infrastructure involves a benefit for society as a whole (Jenné, 2017). Social welfare is the sum of profit, spillover benefits for the local economy generated by the throughput handled in the port as discussed in Chapter 1, and consumer surplus (*CS*). Spillover benefits are included in the operational objective function as λq , with λ the spillover benefits per unit q. According to Xiao et al. (2012), consumer surplus is in this case calculated as

$$CS = \int_{0}^{q} \rho(Q) dQ - \rho(q)q = Bq^{2}/2, \qquad (4.4)$$

with ρ the gross willingness to pay. Some governments however only tend to consider the *CS* relevant for the region they govern. The Belgian government might for example only consider the *CS* of Belgian customers, and might disregard the *CS* of other shipping companies. To account for this in the operational objective function, s_{CS} is the share of total

⁷In this light, it is crucial to recall that time to build and phased investment are disregarded in the RO model.

consumer surplus considered by the government. This adaptation of the framework of Xiao et al. (2012) better models the European port doctrine.

The previous reasoning results in the following expression for social welfare:

$$SW(X, K, q) = \pi_{PA}(X, K, q) + \lambda q + s_{CS} \cdot Bq^2/2,$$
 (4.5)

with $\pi_{PA}(X, K, q)$ the instantaneous annual profit of the PA, given the instantaneous value for the demand shift parameter X, the theoretical design capacity of the port K and the instantaneous annual throughput q. As discussed in Section 2.1.1, it might also be the case that the port is owned by a combination of public and private entities. The aggregated operational objective function (II) of the PA is assumed to become the weighted sum of the individual owners' operational objective functions. The shares of ownership are used as the weights:

$$\Pi_{\text{PA}}(X, K, q) = (1 - s_G) \cdot \pi_{\text{PA}}(X, K, q) + s_G \cdot SW(X, K, q)$$

= $\pi_{\text{PA}}(X, K, q) + s_G \cdot \lambda q + s_G s_{CS} \cdot Bq^2/2,$ (4.6)

with s_G the relative number of PA shares owned by the government. As the sum of the shares needs to equal 1, the private parties together own a share of $1 - s_G$ of the PA.⁸

4.1.4 Summary of the model adaptations and assumptions

To summarise, the profit and investment cost functions resulting from the division of revenues and costs between the two actors are given. For the TOC, this results in:

$$\pi_{\text{TOC}}(X, K, q) = (1 - \alpha_5) \cdot \left\{ \alpha_1 \cdot \left[p(q)q - AX\left(\frac{q}{K}\right)^2 \right] - \alpha_2 \cdot c q \right\} - \alpha_4 \cdot c_h K, \quad (4.7)$$

$$I_{\text{TOC}}(K) = \alpha_3 \cdot \left(FC_I + \gamma_1 K - \gamma_2 K^2 + \gamma_4 K^4 \right), \tag{4.8}$$

whereas for the PA, it results in:

$$\pi_{\text{PA}}(X, K, q) = [(1 - \alpha_1) + (\alpha_1 \alpha_5)] \cdot \left[p(q)q - AX \left(\frac{q}{K}\right)^2 \right] -[(1 - \alpha_2) + (\alpha_2 \alpha_5)] \cdot c q - (1 - \alpha_4) \cdot c_h K,$$
(4.9)

$$I_{\rm PA}(K) = (1 - \alpha_3) \cdot \left(FC_I + \gamma_1 K - \gamma_2 K^2 + \gamma_4 K^4 \right). \tag{4.10}$$

The profits and costs of both actors are determined by the same drivers. However, these drivers are distributed differently over the TOC and the PA. It is hereby interesting to note that the sum of the profit functions on the one hand and the investment functions on the other hand both result in the profit and investment function of a private service port with

⁸As noted in Chapter 2, in some cases with majority ownership, a winner-take-all approach is better suited to model the PA's operational objective function. This comes down to setting s_G to 1 if the public owner has the majority share, and to 0 if the private owner has the majority share.

one single actor. In this way, the presented model can be used to calculate the private service port optimum as well. Since a (partly) publicly owned PA also considers social welfare, the operational objective function of the PA has been extended in Equation (4.6) with spillover benefits and consumer surplus, weighted by the share of public ownership.

Table 4.1 provides an overview of the entire economic model. It contains a short explanation of all the variables, equations and parameters, together with the values for the numerical examples as determined in the previous chapter and the next section. The assumptions of this chapter are summarised in Table 4.2.

4.2 Calibration of the parameters related to public ownership and the distribution of income and costs

In Section 3.2 in the previous chapter, a hypothetical example of a new container terminal of about 8 to 14 million TEU was used to illustrate the theoretical analysis. The parameters that were already calibrated are retained in Table 4.1. Since the model in this chapter accounts for public ownership and for a distinction between the PA and the TOC, additional parameters are introduced. In this section, the reasoning behind the chosen numerical values is explained.

Spillover effects within the port perimeter are estimated at 20% to 30% of the cost c to process one TEU (Coppens et al., 2007). However, depending on the method used, a wide variety is observed (Benacchio & Musso, 2001). Depending on the level of aggregation of the local government's jurisdiction, the spillover effects could even amount to 60% of c. This would leave λ in the range of 0.2 to 0.6 euro per TEU. Here it is set to 0.4 euro per TEU, to account for a port's spillover effects in an entire country. Two other parameters that result from allowing multiple owners are the share of ownership of the government and the share of the CS taken into account by this government, respectively s_G and $s_{CS} \in [0; 1]$. These parameters are not fixed in advance, as they will be varied to study different types of port ownership.

As a result of differentiating between the TOC and the PA, five alphas are introduced. $\alpha_5 (\in [0; 1])$ is a variable that can be set by the PA to determine the size of the concession fee. It represents the share of the TOC's operational profit that is paid to the PA. The share of the superstructure cost in the total investment cost is expressed by α_3 . Based on Vanelslander (2005), Jacob (2013) and Jenné (2017), it is set to 0.35.⁹ The share of superstructure holding cost in the total capacity holding cost $c_h K$, i.e., α_4 , varies a lot depending on the specific

⁹For example, for a project with a capacity of 10 million TEU, the infrastructure would cost about 1 billion euro and the superstructure about 550 million euro, including cranes, straddle carriers, warehouses and gates. Note that this is an average of large and small terminals. For the investment analysis of a specific project, it is crucial to recalculate the share of the superstructure cost in this specific project's total cost.

Variables	6		
1	Ø	=	price
C	9	=	throughput
i	K	=	capacity
C	α_5	=	concession fee parameter:
			share of TOC's annual operational profit paid to PA
Aggregat	ed inverse d	lema	and function: $p = X - Bq - AX \frac{q}{K^2}$
j	B(=1)	=	slope
1	A(=5)	=	monetary scale factor of congestion cost
Demand	shift param	eter	$X: dX(t) = \mu X(t) dt + \sigma X(t) dZ(t)$
1	t(=annual)	=	time horizon
	Ζ	=	standard Wiener process
ļ	$\mu(=0.015)$	=	drift of Z (i.e., growth)
($\sigma(=0.1)$	=	drift variability of Z (i.e., uncertainty)
Aggregat	ed total cost	t TC	$= c q + c_h K$
(c(=1)	=	constant marginal operational cost
($c_h(=0.5)$	=	cost to hold one unit of capacity in place
TOC inve	estment cost	t I _{TO}	$C = \alpha_3 \cdot \left(FC_I + \gamma_1 K - \gamma_2 K^2 + \gamma_3 K^3 + \gamma_4 K^4 \right)$
PA invest	ment cost I	$_{\rm PA} =$	$(1-\alpha_3) \cdot \left(FC_I + \gamma_1 K - \gamma_2 K^2 + \gamma_3 K^3 + \gamma_4 K^4 \right)$
C	$\alpha_3 (= 0.35)$	=	share of total investment cost I incurred by TOC
i	$FC_I(=80)$	=	fixed investment cost
)	$r_1(=180)$	=	first-order coefficient
)	$r_2(=19)$	=	coefficient reflecting investment economies of scale
)	$r_3(=0)$	=	omitted third-order coefficient
)	$r_4 (= 0.12)$	=	coefficient posing boundary to maximum project size
TOC prof	fit = operati	onal	objective function $\pi_{ m TOC}$
		=	$(1-\alpha_5) \cdot \{\alpha_1 \cdot [p(q)q] - \alpha_2 \cdot cq\} - \alpha_4 \cdot c_h K$
PA profit	$\pi_{\rm PA} = [(1 - a)$	$(t_1) + t_2$	$(\alpha_1\alpha_5)]\cdot [p(q)q] - [(1-\alpha_2) + (\alpha_2\alpha_5)] \cdot c q - (1-\alpha_4) \cdot c_h K$
($\alpha_1 (= 0.9)$	=	share of terminal tariff in total price p
C	$\alpha_2(=0.95)$	=	share of <i>c</i> incurred by TOC
C	$\alpha_4 (= 0.5)$	=	share of capacity holding cost incurred by TOC
PA opera	tional objec	tive	function $\Pi_{\text{PA}} = \pi_{\text{PA}} + s_G \cdot \lambda q + s_G s_{CS} \cdot CS$
7	$\lambda(=0.4)$	=	spillover benefits per unit q
(CS	=	consumer surplus, i.c., $Bq^2/2$
5	$s_G(\in [0;1])$	=	share of PA owned by the government
5	$s_{CS}(\in [0;1])$	=	share of total CS taken into account by the government

 Table 4.1: Overview of Chapter 4's model and the selected parameters.

	Assumption(s)
Project type? Capacity constraint? Port type? Port competition? Time to build? Phased investment?	New port Hard constraint Public and/or private, service* or landlord port** No (1 port in model) Not considered Project installed at once: available after completion
Specific assumptions	PA can derive full information about TOC prices and costs Distribution of port income and costs between PA and TOC are fixed over time PA invests in infrastructure, TOC invests in irreversible superstructure elements Investment timing and size of PA and TOC need to be equal: negotiated before construction, or forced by PA Concession fee is <i>modelled</i> as a percentage of the TOC's operational profit and is paid by the TOC to the PA Concession agreements are renewed over time \rightarrow infinite project lifetime TOC handles the goods \rightarrow TOC sets the throughput level (q^{opt})
General assumptions	Price (sum of port dues and terminal tariff) transparency \rightarrow Linear additive inverse demand function Demand uncertainty, modelled using a GBM with constant, independent growth (μ) and uncertainty (σ) parameters Growth market with $r > \mu > 0$, to ensure RO model convergence ^{***} Poisson-distributed ship arrivals at the port: delays start at about 50% and increase sharply beyond 75-80% occupancy User congestion costs depend on dock users' average value of time, which evolve with the market (i.c., increasing) Investment cost: increasing fourth-order polynomial with economies of scale and size boundary Deterministic, linear operational and capacity holding costs Flexibility of investment timing and size \rightarrow Option to postpone the investment Once-and-for-all investment decision with flexible throughput levels Infinite project life time (maintenance executed)
***: Condition <i>r</i> > SCOPE: Theoretical design	*: Owned and operated by one single actor. *: Owned by the PA, operated by one TOC (actor differs from the PA). ** Dwned by the PA, operated by one TOC (actor differs from the PA). µ leads to a finite timing decision, instead of infinite investment postponement. Sensitivity tested with negative growth rates as well. Capacity for container throughput services. Other port services disregarded (sufficiently present or realised by actors other than the PA or TOC). Focus on lumpy investment in port infrastructure and irreversible elements of superstructure.

Table 4.2: Overview of Chapter 4's model assumptions.

project considered and the related dredging contract. As an average, it is set here to 0.5, but it is varied in the sensitivity analysis (T. Zheng, 2015; Jan De Nul, 2012; Keskinen et al., 2017; Luck, 2017). Next, α_2 expresses the relative share of the TOC's operational cost in the total operational cost of the cargo handling services. To load and unload ships, the TOC encounters the cost of labour and electricity, which is the major part. The PA's marginal operational cost is negligible and mainly incurred by administration, since other services such as piloting and tugging are left beyond the scope of this analysis. By consequence, α_2 is set to 0.95.

Finally, α_1 , the share of the terminal tariff in the total price, needs to be set. The share of the port dues is then $1 - \alpha_1$. This is the most difficult parameter to determine. Port dues are not only throughput dependent. They depend on many different factors like ship characteristics, such as ship size, and the location in the port, which has an influence on the number of locks that need to be passed (Meersman et al., 2015). Hence, the quantity of goods loaded and unloaded only partly influence the size of these port dues. As a result, a conversion is required to find an expression for the average amount of port dues per TEU. Conversions have an influence on α_1 , which is in reality not fixed, but is the result of the demand function taking into account the ship size, location in the port. The difference in price calculation method per port and the limited transparency complicate the calculation of α_1 even more. As a result, in this chapter, α_1 is calculated as an average ratio over a full year and based on data from the Port of Antwerp (2017a,b) to come to a realistic value.

In 2016, 4500 container ships called the Port of Antwerp (2017b), with an average GT of 55 000. Moreover, about 10 million TEU was handled in the same year, which results in an average of 2222 TEU per container ship. The prices of Port of Antwerp (2017a) show that the port dues per GT are 0.2 euro, if the ship is operated by a container line, without reductions included. The container supplement is 0.2 euro per ton and the Port of Antwerp (2017b) assumes 12 ton per TEU on average, so that the additional port dues for handling one TEU are 2.4 euro. As a result, the average total port dues equal 7.35 euro per TEU. The terminal tariff is 69 euro per TEU, as handling a container, involving two moves, costs about 110 euro and the average container is 1.59 TEU (Port of Antwerp, 2017b; Saeed & Larsen, 2010; Wiegmans & Behdani, 2018). Based on these numbers, the relative share of terminal tariffs in the total port income is calculated as:

$$\alpha_1 = \frac{69 \cdot 10 \times 10^6}{(69 \cdot 10 \times 10^6) + (0.2 \cdot 4500 \cdot 55000 + 0.2 \cdot 12 \cdot 10 \times 10^6)} = 0.90.$$
(4.11)

Altering the terminal tariff, port dues and average weight of one TEU within reasonable boundaries, indicates that for many ports, α_1 could be in a range of 0.8 to 0.95.

To study the impact of the determined values and changes in these values, they will be altered in the sensitivity analysis in Section 4.5.

4.3 Solving for the optimal investment strategy in a landlord port

With the operational objective functions of the port actors deciding on capacity and the numerical values for the parameters at hand, the optimal investment decision in a landlord port can be calculated. In reality, the PA first decides at which moment it would invest in infrastructure and how much capacity it would provide. Once this investment is made, it poses two boundaries to the superstructure investment of the TOC that obtained a concession agreement through negotiations or an auction following a tender. Since the TOC is responsible for the operation of the terminal, it sets the optimal throughput maximising its own profit: q_{TOC}^{opt} . This needs to be below *K*, because in this setting throughput cannot exceed the total design capacity *K*. Moreover, the assumption of full ex-ante information for the PA and the interactions with the TOC preceding a concession agreement lead to the consequence that the PA and the TOC select the same size and timing of the investment. The decision will often be somewhere between the extrema of both actors' optima, which constitute the decision interval as described in Section 4.1.2. If the TOC deviates from the PA's decision, the PA will not grant the concession and the TOC cannot invest in and operate the terminal.

In this chapter, the same dynamic programming methodology of the previous chapter is used to backwards solve the real options problem including congestion and multiple actors (a: TOC and PA) in a port. The way congestion is modelled gives again rise to hypergeometric functions ($_2F_1$) in the expressions of the project value functions (after investment at time T_a) of the TOC:

$$V_{\text{TOC}} = \mathbf{E} \int_{0}^{\infty} \max_{q_{\text{TOC}}} \{ \pi_{\text{TOC}} (T_{\text{TOC}} + \tau) \} e^{-r\tau} d\tau$$
(4.12)

and the PA:

$$V_{\rm PA} = \mathbf{E} \int_{0}^{\infty} \Pi_{\rm PA} (q_{\rm TOC}, T_{\rm PA} + \tau) \mathrm{e}^{-r\tau} \mathrm{d}\tau.$$
(4.13)

These equations highlight the important difference compared with a service port. The optimisation is to be executed for the two actors separately, but only one actor handles the goods. In order to find the optimal investment decision of each actor, each actor maximises the expected future discounted profit stream minus the investment outlay of its project, with respect to timing T_a and capacity K_a (Huisman & Kort, 2015). If $X(t = 0) < X_{T,a}$ so that it is not optimal to invest from the beginning, this gives rise to each actor's investment problem objective function:

$$\max_{T_a \ge 0, K_a \ge 0} \mathbf{E}\left\{ \left[V_a(X_{T,a}, K_a) - I_a(K_a) \right] e^{-r T_a} | X(t=0) = X \right\},$$
(4.14)

which is the same maximisation as for a service port, but now defined for each actor a.¹⁰

¹⁰Although this unrestricted optimisation problem for each actor can be solved independently, the final in-

Since each actor has the option to postpone the investment, each investment decision contains an option value $F_a(X)$, based on the economic equations in Section 4.1. These option values are similar to the one derived in Section 3.1. Analogously, the optimisation with respect to timing again comes down to finding the critical threshold value $X_{T,a}$ for which it is better to invest rather than to wait.

In order to solve this RO problem and optimise the investment decisions of both actors, similar calculations as in the previous chapter are required. However, some extensions and differences are present too. Here, the first step of the methodology determines the TOC's optimal throughput $q_{TOC}^{opt}(X, K)$ through the first and second-order conditions for the $\pi_{TOC}(q, X, K)$ of Equation (4.7). This results in

$$q_{\text{TOC}}^{opt}(X,K) = \begin{cases} 0, & X < \frac{\alpha_2}{\alpha_1}c, \\ \frac{(\alpha_1 X - \alpha_2 c)K^2}{2\alpha_1 (AX + BK^2)}, & \frac{\alpha_2}{\alpha_1}c \leq X < \frac{(2\alpha_1 BK + \alpha_2 c)K}{\alpha_1 (K - 2A)}, \\ K, & X \ge \frac{(2\alpha_1 BK + \alpha_2 c)K}{\alpha_1 (K - 2A)}, \end{cases}$$
(4.15)

given the size of the port *K* and the instantaneous value for *X*. As it should hold that $0 \le q_{\text{TOC}}^{opt} \le K$, q_{TOC}^{opt} is again divided into three mathematical regions, defined by boundaries for *X*, as was also done in Chapter 3. Plugging q_{TOC}^{opt} into $\pi_{\text{TOC}}(q, X, K)$ leads to $\overline{\pi}_{\text{TOC}}(X, K)$, defined in the same three regions.

Second, through differential equation

$$\frac{\sigma^2}{2} X^2 \frac{\partial^2 V_{\text{TOC}}}{\partial X^2}(X, K) + \mu X \frac{\partial V_{\text{TOC}}}{\partial X}(X, K) - r V_{\text{TOC}}(X, K) + \overline{\pi}_{\text{TOC}}(X, K) = 0, \qquad (4.16)$$

 $V_{\text{TOC}}(X, K)$, the value of the TOC's project of size K, installed at the moment that the demand shift parameter equals X and for which the TOC pays $I_{\text{TOC}}(K)$, is derived. Here again, the same three mathematical regions apply. Third,

$$F_{\text{TOC}}(X) = \max\left\{ e^{-r dt} \mathbf{E}(F_{\text{TOC}}(X) + dF_{\text{TOC}}(X)), \max_{K_{\text{TOC}}}[V_{\text{TOC}}(X, K_{\text{TOC}}) - I_{\text{TOC}}(K_{\text{TOC}})] \right\}$$
(4.17)

gives the option value of investment postponement, which is maximised in order to find the optimal investment timing threshold and size of the TOC's investment ($X_{T,TOC}^{**}, K_{TOC}^{**}$). Fourth, the resulting operational objective function of the PA, $\overline{\Pi}_{PA}(X, K)$, is determined by plugging q_{TOC}^{opt} into $\Pi_{PA}(q, X, K)$ from Equation (4.6), as it is the TOC that sets the throughput quantity. Using q_{TOC}^{opt} results in the same three regions for the PA's operational objective function as found for the TOC's profit. Fifth, the differential equation

$$\frac{\sigma^2}{2} X^2 \frac{\partial^2 V_{\text{PA}}}{\partial X^2}(X, K) + \mu X \frac{\partial V_{\text{PA}}}{\partial X}(X, K) - r V_{\text{PA}}(X, K) + \overline{\Pi}_{\text{PA}}(X, K) = 0$$
(4.18)

vestment decision made in a landlord port is the result of multiple interactions between the interdependent actors. The modelling of these interactions has been discussed in Section 4.1.2 and relates to the final step in Figure 4.2.



Figure 4.2: Summary of the cooperative RO game methodology.

allows deriving $V_{PA}(X, K)$, the value of the PA's project of size K, installed at the moment that the demand shift parameter equals X and for which the PA pays $I_{PA}(K)$. Finally, the option value

$$F_{\rm PA}(X) = \max\left\{ e^{-r \,\mathrm{d}\,t} \,\mathbf{E}(F_{\rm PA}(X) + \mathrm{d}F_{\rm PA}(X)), \max_{K_{\rm PA}}[V_{\rm PA}(X, K_{\rm PA}) - I_{\rm PA}(K_{\rm PA})]\right\}$$
(4.19)

allows calculating the optimal investment timing threshold and size for the PA $(X_{T,PA}^{**}, K_{PA}^{**})$. Both $(X_{T,TOC}^{**}, K_{TOC}^{**})$ and $(X_{T,PA}^{**}, K_{PA}^{**})$ together constitute the decision interval from which the final investment decision is to be selected. The flowchart in Figure 4.2 summarises the steps of the cooperative RO game methodology of this chapter.

4.4 Results and discussion

In this section, numerical solutions for different port types are calculated using the previously described methodology. The investment decisions are compared with the optimal decision for a private service port with the same values for the common parameters. This optimum has been calculated as $X_T^{**} = 37.63$ euro per TEU and $K^{**} = 11.17$ million TEU

<i>a</i> ₅	Actor a	Timing $X_{T,a}^*(K_a = 11.17)$	Size $K_a^*(X = 37.63)$	Decision (X ^{**} _{T,a} , K ^{**} _a)
0.4	PA	47.12	10.21	(47.30, 11.21)
	TOC	28.97	12.45	(28.85, 11.14)
0.469	PA	43.18	10.56	(43.18, 11.17)
	TOC	31.30	12.07	(31.30, 11.17)
0.5	PA	41.69	10.71	(41.63, 11.16)
	TOC	32.53	11.89	(32.59, 11.19)
0.59475	PA	37.86	11.12	(37.63 , 11.12)
	TOC	37.31	11.26	(37.63 , 11.26)
0.6	PA	37.67	11.14	(37.44, 11.11)
	TOC	37.62	11.22	(37.97, 11.26)

Table 4.3: Optimal $X_{T,a}^*(K)$, $K_a^*(X)$ and $(X_{T,a}^{**}, K_a^{**})$ under different α_5 in a private landlord port.

Parameter values: $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12, \alpha_1 = 0.9, \alpha_2 = 0.95, \alpha_3 = 0.35, \alpha_4 = 0.5.$

per year in Chapter 3. In this section, the impact on the investment decision of the division between the PA and the TOC and the size of the concession fee in a private landlord port is first studied separately from the impact of government involvement as a PA shareholder in a service port, partly or fully owned by the government. Afterwards, both are combined in a landlord port setting with public money involvement.

4.4.1 Division between PA and TOC

If the distinction between the PA and the TOC in a private landlord port (no government involvement, or s_G set to zero) is made according to the model in Table 4.1, an individual optimal investment threshold can be calculated for both the PA and the TOC. Indeed, given their individual revenues and costs, both the PA and the TOC might have a different individual investment optimum. Numerical simulations show that an α_5 leading to the same optimal investment size (α_5^K) and an α_5 leading to the same optimal timing for both actors (α_5^X) exist for different divisions of the revenues and costs between the PA and the TOC (see e.g., Table 4.10). Moreover, this optimal value common for both actors is almost equal to the optimum of the private service port where the port is directed and operated by one single actor. The other decision variable determining the optimal investment strategy then differs per actor, giving rise to a decision interval. This interval often also comprises the optimum value from the private service port.

The previous reasoning is illustrated with the numerical example in Table 4.3. If the TOC needs to pay 59.48% of its operational profit as a concession fee to the PA, the optimal investment threshold $X_{T,TOC}^{**} = X_{T,PA}^{**} = 37.63$ euro per TEU will be the same for both actors,

which is nearly the same investment threshold as found in a private service port.¹¹ In that case, the optimal capacity for the PA will equal 11.12 million TEU per year, whereas for the TOC it will be optimal to invest in 11.26 million TEU per year. Indeed, the optimal capacity from a similar private service port (11.17 M TEU p.a.) is comprised in this decision interval. With a lower concession fee (α_5 set to 0.469) however, the optimal size of the investment is equal for both the PA and the TOC, almost equalling the optimal capacity $K^{**} = 11.17$ M TEU p.a. from a private service port. The optimal thresholds are then $X_{T,PA}^{**} = 43.18$ euro per TEU and $X_{T,TOC}^{**} = 31.30$ euro per TEU. Through mutual concessions, the same optimal threshold from a private service port, $X_T^{**} = 37.63$ euro per TEU, is attainable. It is also possible for other concession fees to select the global optimum from the decision interval.

Table 4.3 moreover illustrates the influence of the concession fee on the optimal investment decisions of both the PA and the TOC. If the TOC is required to pay a higher share of its operational profit, the TOC will invest later, or in less capacity ceteris paribus, because the project will be less attractive. This is opposed to the investment becoming more attractive for the PA, leading to an earlier investment or a larger investment size. In the previous chapter, it was observed that every parameter change increasing X_T^* , reduces K^* and vice versa. When observing the final optimal investment decision combining timing and size however, X_T^{**} will always be close to X_T^* , which does not hold for K^{**} and K^* . A later optimal timing X_T^{**} coincides with a higher optimal design capacity K^{**} , because the effect of the increasing $K^*(X)$ and $X_T^*(K)$ -functions dominate the opposite shifts of these functions following a change in α_5 . More specifically, due to the specific shape of the investment function with its economies and diseconomies of scale, the net impact of changes in the economic environment on the timing is larger than the net impact on the size of the investment.

Additional numerical calculations further illustrate the strategy of selecting the private service port optimum, which has a total project value minus investment costs V - I of 1.76 billion euro. If the PA and the TOC in a private landlord port invest both at this optimum, their aggregated V - I also equals 1.76 billion euro, independent of the size of the concession fee. Hence, this concession fee only has an impact on the distribution of the revenues and costs among both actors. The different concession fee strategies from Section 4.1.2 are illustrated in Table 4.4. The table shows that in this numerical example, $\alpha_5 = 0.5204$ leads to an equal V - I for both actors. Setting α_5 to 0.6058 equals the share each actor has in both V - I and I, which is 65% for the PA and 35% for the TOC.

The case involving a TOC paying 59.475% of its operational profit to the PA as a concession fee allows easily calculating each actor's impact of diversion from its own optimum. Since it is optimal with this concession fee for both actors to invest at the same time, V - I of both can be compared at the moment of investment. No additional discounting is required. This scenario is quantified in Table 4.5 for different investment strategies included

¹¹Only the rounded numbers are slightly different. As $X_T >> c$ and the chosen α_1 and α_2 do not differ by much in this numerical example, their impact on q_{TOC}^{opt} in Equation (4.15) and the optimal investment strategy is limited.

<i>a</i> ₅	$(V-I)_{\rm PA}$	$(V-I)_{\text{TOC}}$	$\Sigma_i (V-I)_i$
0.4	508	1252	1760
0.469	721	1039	1760
0.5	817	943	1760
0.5204	880	880	1760
0.59475	1110	650	1760
0.6	1126	634	1760
0.6058	1144	616	1760

Table 4.4: V - I of the PA and TOC under different concession fees, if both actors invest at the optimum of a private service port.

Parameter values: A = 5, $\overline{B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12, \alpha_1 = 0.9, \alpha_2 = 0.95, \alpha_3 = 0.35, \alpha_4 = 0.5, X_T = 37.63, K = 11.17.$

Table 4.5: V - I of the PA and TOC with $\alpha_5 = 0.59475$ under possible investment strategies, equally followed by both actors.

$(V-I)_{\rm PA}$	$(V-I)_{\text{TOC}}$	$\Sigma_i (V-I)_i$
1110.0	649.8	1759.8
1109.8	650.1	1759.9
1109.0	650.3	1759.3
	(V-I) _{PA} 1110.0 1109.8 1109.0	(V-I) _{PA} (V-I) _{TOC} 1110.0 649.8 1109.8 650.1 1109.0 650.3

Parameter values: $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12, \alpha_1 = 0.9, \alpha_2 = 0.95, \alpha_3 = 0.35, \alpha_4 = 0.5, \alpha_5 = 0.59475.$

in Table 4.3.

The results show that the PA investing in the service port optimum, which entails more capacity than its own optimum, leads to a decrease in V - I for itself of 0.2 million euro. This however allows the TOC to make a larger investment, which is already closer to its own optimum. As the TOC can now invest in K = 11.17 M TEU p.a. instead of 11.12, the TOC's V - I increases with 0.3 million euro. This leads to an aggregated gain of 0.1 million euro. In reality, such a situation would hardly be observed if the PA were privately owned, since only the own welfare would be maximised. Therefore, a public PA owner, caring about aggregated social welfare, is introduced in the next subsections. Deviating from the aggregate optimum would lead to a destruction of welfare, compared with situation b). Both actors investing at the TOC's optimum would make the TOC win an additional 0.2 million euro, but the PA would lose 0.8 million euro, destroying 0.6 million euro of aggregated profit. As argued before, the additional aggregated profit in situation b) can be distributed among the PA and the TOC through the adaptation of the concession fee. The size of the concession fee set by the PA is part of its strategy as discussed in Section 4.1.2 and is an important element in concession negotiations. Additionally, it is noteworthy that the deviations in project V - I are limited in absolute terms for both actors. This favours the negotiations between the actors.

Actor	Optimal investment strategy		Conditional optimal investment strategy			
PA	(37.43 , 11.11)					
TOC	(38.00 , 11.26)	\rightarrow	$(X_{T,\text{TOC}}^*(11.11), 11.11) = (37.43, 11.11)$			
Parameter values: $A = 5$, $B = 1$, $c = 1$, $c_h = 0.5$, $\mu = 0.015$, $\sigma = 0.1$, $r = 0.06$, $FC_I = 80$, $\gamma_1 = 180$, $\gamma_2 = 19$, $\gamma_3 = 0$, $\gamma_4 = 100$, $\gamma_2 = 10$, $\gamma_3 = 0$, $\gamma_4 = 100$, $\gamma_4 = 100$, $\gamma_5 = 1000$, $\gamma_5 = 1000$, $\gamma_5 = 1000$, $\gamma_5 = 1000$, $\gamma_$						
$0.12, \alpha_1 = 0.9, \alpha_2 = 0.95, \alpha_3 = 0.35, \alpha_4 = 0.5, \alpha_5 = 0.60035.$						

Table 4.6: Illustration of the PA's concession fee strategy forcing the TOC to take the same investment decision.

The results in Table 4.3 also bear worthy information for a PA searching for a good concession fee. For any value of α_5 between 46.9% and 59.475%, the PA poses a limit to both the size and timing of the TOC's optimal investment decision. In this range for the concession fee, the optimal timing of the TOC is earlier than the timing of the PA and the optimal capacity of the TOC would exceed that of the PA. In such a case, the PA knows that as soon as it invests in the negotiated capacity, the TOC will be willing to invest too in the same amount of capacity, as long as this still is a profitable strategy. This is important from a game-theoretic point of view, as the TOC's incentives to deviate from the contract should be minimised. In this light, the PA needs to have sufficient power to enforce the concession contract (G. W. Y. Wang & Pallis, 2014).

However, if α_5 is too low (i.c., below 0.469), the TOC may be willing to invest in less capacity than what has been decided on, whereas a too high α_5 (i.c., above 0.59475) could lead to the TOC investing later than the moment agreed upon. Although these latter two cases bear an incentive for the TOC to deviate from the PA's optimum, the PA still has some negotiation power that could turn out to be sufficient. In the first case, the PA could install the project at its own optimal threshold, which is higher than what is optimal for the TOC. As was explained before, this is a limiting factor and the TOC will internalise this higher threshold. Subsequently the TOC's optimal size is determined as $K^*_{\text{TOC}}(X^{**}_{T,\text{PA}})$, which might come closer to or even equal or exceed the optimal size of the PA. In this way, the PA retains a strong position. In the second case, a smaller project than what is optimal for the TOC could be installed. The TOC then takes the size as given and determines its remaining investment decision degree of freedom, its optimal threshold, conditionally on the size of the PA. This $X^*_{T,\text{TOC}}(K^{**}_{\text{PA}})$ might come closer to or even equal or be below the optimal threshold of the PA. An illustration is given in Table 4.6 for the unique α_5 , given the other parameters, allowing the PA to force the TOC to invest exactly at the PA's optimum.

If $\alpha_5 = 0.60035$, the optimal decision for the TOC is to invest later and in more capacity than the PA. Hence, the TOC knows that it has to reduce its investment size accordingly, to the 11.11 million TEU per year of the PA. Taking this into account, the TOC calculates its conditional optimal threshold $X^*_{T,TOC}(11.11) = 37.43$ euro per TEU, which is equal to the threshold of the PA. The impact on the individual and aggregated discounted V - I is shown in Table 4.7. It contains the discounted V - I for both the PA and the TOC under different possible strategies at the moment where X(t) = 35 euro per TEU. These strategies are equally followed by both actors.¹² It is shown that the PA forcing the TOC to invest at its own optimum does in this case only cause small deviations in individual and aggregated discounted V - I from the private service port optimum, or even from the optimal TOC's investment strategy.

Table 4.7: Discounted V - I under under possible investment strategies equally followed by both actors in a private service port where the PA can force the TOC to invest in the PA optimum at the moment t: X(t) = 35 euro per TEU.

Common investment strategy:	Discounted	Discounted	Discounted		
(X_T, K)	$(V-I)_{\rm PA}$	$(V-I)_{\text{TOC}}$	$\Sigma_i (V-I)_i$		
PA individual optimum: (37.43, 11.11)	933.3	523.7	1457.0		
Private service port optimum: (37.63, 11.17)	933.2	523.9	1457.1		
TOC individual optimum: (38.00, 11.26)	932.9	524.0	1456.9		
Parameter values: $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.015, \sigma = 0.1, r = 0.06, FC_I = 0.015, \sigma = 0.1, r = 0.06, FC_I = 0.015, \sigma = 0.0$					
$0.12, \alpha_1 = 0.9, \alpha_2 = 0.95, \alpha_3 = 0.35, \alpha_4 = 0.5, \alpha_5 = 0.60035, X = 35.$					

For any other α_5 than 0.60035, $X_{T,TOC}^*(K_{PA}^{**})$ will be either higher than the $X_{T,PA}^{**}$, meaning that the TOC is even more forced to deviate from its conditional optimum, or below $X_{T,PA}^{**}$, still implying an incentive for the TOC to invest below the PA's optimal capacity. For the given parameter values, in the other situation where the optimal timing of the PA is later than the optimal timing of the TOC, the resulting optimal size of the TOC will still exceed the PA's size. So there the TOC has to deviate even more from its optimum. There is no concession fee leading to $X_{T,PA}^{**} > X_{T,TOC}^{**}$, coinciding at the same time with $K_{TOC}^*(X_{T,PA}^{**}) = K_{PA}^{**}$.

The discussion of the numerical results of this subsection is summarised in the following result:

Result 4.1. Under the assumptions given in Table 4.2, there are two ways to decide on the PA and TOC's common final investment decision. First, a well-thought-out concession fee and subsequent negotiations may allow choosing the aggregated investment optimum, which at the same time avoids incentives to deviate from this decision. Alternatively, the PA may set a concession fee and invest in its individual optimum which forces the TOC to invest at the same time in exactly the same amount of capacity. This strategy is the TOC's conditionally optimal strategy.

This result shows that in the port that is to be built, the selection of the concession fee is

¹²This discounted
$$V - I$$
 is calculated as $(X/X_T)^{\beta_1} \cdot (V - I)$, with $(X/X_T)^{\beta_1}$ the stochastic discount factor at time t where $X(t) = X$ and with $\beta_1 = \frac{\frac{\sigma^2}{2} - \mu + \sqrt{\left(\frac{\sigma^2}{2} - \mu\right)^2 + 2r\sigma^2}}{\sigma^2}$ (Huisman & Kort, 2015).

		Timing	Size	Decision			
\$ _G	s _{cs}	$X_T^*(K = 11.17)$	$K^*(X = 37.63)$	(X_T^{**}, K^{**})			
0	N/A	37.63	11.17	(37.63, 11.17)			
1/2	1/2	35.89	11.40	(35.98, 11.19)			
1/2	1	34.39	11.64	(34.74, 11.27)			
1	1/2	34.05	11.66	(34.25, 11.22)			
1	1	30.70	12.22	(31.40, 11.38)			
Parameter values:							

Table 4.8: Optimal $X_T^*(K)$, $K^*(X)$ and (X_T^{**}, K^{**}) under different s_G and s_{CS} in a service port, partly or fully publicly owned.

 $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, \sigma = 0.1, r = 0.06, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12, \lambda = 0.4.$

important, and has implications for the investment decision, given the negotiation power of both actors. If it turns out difficult for the PA to negotiate with the TOC about the investment strategy and the concession fee, which will be strongly dependent on the specific port situation, the PA can still pursue the second strategy. Using a well-chosen concession fee, the PA can force the TOC to also invest in this optimum of the PA.

4.4.2 Government involvement

Next to a private owner maximising profit, also a social welfare-maximising government can be a PA shareholder. As was explained, the government in the model owns a share s_G of the considered service port. The private partner then owns the remaining share $1 - s_G$. The share of total consumer surplus taken into account by the government is given by s_{CS} . Some possible scenarios are presented in Table 4.8.

The first line in Table 4.8 reflects the situation from Chapter 3 with a privately owned single port actor in a private service port. If the government's share of ownership or the considered share of consumer surplus is higher, the considered project benefits will be larger as well, as the spillover benefits and consumer surplus are taken more into account in the PA's operational objective function $\overline{\Pi}_{PA}(X, K)$. This is translated into a lower threshold for $X_T^*(K)$, which goes again hand in hand with a higher investment size $K^*(X)$. The analysis confirms the finding that public entities tend to invest sooner or in more capacity than private entities (Asteris et al., 2012). The optimal investment strategy (X_T^{**}, K^{**}) changes accordingly. If the government's share of ownership or the considered share of *CS* increases, the project is valued a lot higher because social welfare is taken more into account. As a result, the individual effects of earlier and larger investment dominate the positive relationship between size and timing, to lead to the following result:

Result 4.2. Under the assumptions given in Table 4.2, increased government ownership or this government taking social welfare more into account leads to installing the new port ear-

lier. This new port will moreover have a larger design capacity.

This finding is remarkable, because it is opposite to the common real options finding, where more capacity leads to a later investment or vice versa due to the dominating effect of the increasing $K^*(X)$ and $X_T^*(K)$ -functions. When welfare effects are considered more, it proves beneficial to dispose of the new port capacity soon enough, and make it sufficiently large so that sufficient welfare effects are realised. Indeed, waiting implies foregoing consumer surplus and spillover benefits, as economic growth of the region is hampered by the current lacking of a port. Moreover, sufficient throughput to generate these effects is desired as well.

The optimal port capacity investment decision of the government is derived under the assumption that sufficient public budget is available and allocated to port investments. However, if the optimal investment size exceeded the available budget, the size of the actual investment would be limited by the available budget, and only the investment timing could be optimised, taking this maximum size as given: $X_T^*(K_{\text{max}})$. Moreover, this result is reconsidered for a port expansion decision in Chapter 6.

4.4.3 Landlord port with public ownership

The previous subsections illustrated separately the impact of the landlord port model and public ownership on the port capacity investment decision. In this subsection, both are combined. The analysis is made for a landlord port in which the PA's shares are equally divided among the private parties and the government, which in turn takes 50% of total *CS* into account.

The results are similar to the outcomes in Table 4.3 for a private landlord port. Hence, Result 4.1 still holds in this context. The table shows that the inclusion of mixed ownership with governments involved does not have an impact on the TOC's decision, as this remains a private party with the same profits, costs and operational objective function as before. For the same concession fee, their optimal decision remains unchanged. Public involvement in the model only has an impact on the PA's optimal decision, which will be earlier and larger. As a result, the α_5 's matching both actors' investment timing or size will alter. This result is summarised as follows:

Result 4.3. Under the assumptions given in Table 4.2, public ownership in a landlord port has no impact on the optimal investment decision of a private TOC. It however alters the optimum of the PA, and hence the final decision made.

From both Table 4.3 and Table 4.9, it can be verified that, independent of the amount of public money involved, the project will become less attractive for the TOC with a higher

<i>a</i> ₅	Actor a	Timing $X_{T,a}^*(K_a = 11.17)$	Size $K_a^*(X = 37.63)$	Decision $(X_{T,a}^{**}, K_{a}^{**})$
0.4	PA	43.19	10.61	(43.57, 11.26)
	TOC	28.97	12.45	(28.85, 11.14)
0.504	PA	38.45	11.09	(38.54, 11.20)
	TOC	32.70	11.86	(32.77, 11.20)
0.55	PA	36.77	11.28	(36.76, 11.17)
	TOC	34.82	11.57	(35.01, 11.23)
0.5693	PA	36.12	11.35	(36.08 , 11.16)
	TOC	35.84	11.44	(36.08 , 11.24)
0.6	PA	35.16	11.47	(35.06, 11.15)
	TOC	37.62	11.22	(37.97, 11.26)

Table 4.9: Optimal $X_{T,a}^*(K)$, $K_a^*(X)$ and $(X_{T,a}^{**}, K_a^{**})$ under different α_5 in a landlord port with public ownership.

Parameter values: A = 5, B = 1, c = 1, $c_h = 0.5$, $\mu = 0.015$, $\sigma = 0.1$, r = 0.06, $FC_I = 80$, $\gamma_1 = 180$, $\gamma_2 = 19$, $\gamma_3 = 0$, $\gamma_4 = 0.12$, $\alpha_1 = 0.9$, $\alpha_2 = 0.95$, $\alpha_3 = 0.35$, $\alpha_4 = 0.5$, $s_G = 1/2$, $s_{CS} = 1/2$, $\lambda = 0.4$.

concession fee. The reason is that they retain a lower share of their profit. Unexpectedly however, it is optimal for the TOC individually to invest in more capacity. This counter-intuitive result is explained by the fact that the TOC reacts by postponing the investment to a moment when the market has grown more, requiring more capacity. At the same time, the investment becomes more attractive for the PA. As a result, the PA wants to invest earlier, although its optimal capacity will then be lower.

The combination of mixed ownership of the PA and a landlord port model leads to two different results compared with the private landlord port setting. First, as was also apparent from Table 4.8, an increased share of public involvement leads to the PA investing earlier and in more capacity, because welfare effects other than profit are considered in the analysis too. Second, the optimum with the same $s_G = s_{CS} = 1/2$ as for the equivalent service port in Table 4.8 lies in some cases further outside the decision interval than in the case of a private port. At α_5^X (= 0.5693) for example, $X_{T,PA}^{**}$ equals $X_{T,TOC}^{**}$ = 36.08 euro per TEU, which is higher than the threshold X_T^{**} (= 35.98) in Table 4.8. This is caused by the fact that the PA's optimum is influenced by its public ownership, whereas the TOC's optimal investment decision and optimal throughput level are not altered, as the private TOC does not take social welfare into account. As a private party, it only considers profit in its operational objective function.¹³

In the present scenario, the two described concession strategies remain possible. First, the PA could aggregate the operational objective functions of itself and the TOC and invest

¹³Since a private TOC is a profit and not a social welfare maximiser, the optimal throughput set by the TOC in a public landlord port will differ from the optimal throughput set by the single public actor in a public service port. This in turn influences the operational objective function and investment decision of both actors.

at the optimum of a service port. Through the concession agreement, the PA could then urge the TOC to handle at least a certain minimal throughput. Negotiating a favourable (i.e., lower) concession fee could be an adequate incentive for this. Especially for publicly owned PAs, the approach of applying a concession fee that allows for the aggregate investment optimum is relevant. Second, if $\alpha_5 = 0.57252$, the optimal investment decision for the PA would be (35.96, 11.16), and the TOC would be forced to reduce its optimal investment of (36.27, 11.24) to a size of $K_{\text{TOC}} = 11.16$ M TEU p.a. The corresponding $X_{T,\text{TOC}}^*(11.16)$ would then be 35.96 euro per TEU, which equals $X_{T,\text{PA}}^{**}$.

4.5 Investment decision sensitivity to an altered economic situation

In this section, the sensitivity of the results to changes in other parameters is discussed. Table 4.10 shows how the investment decisions of the different actors alter with each parameter change. To this end, the decisions at respectively α_5^X and α_5^K are given for each altered situation, as this information allows understanding the direction of change of the optimal investment decision caused by different concession fees.

With the monetary scale factor of congestion A = 4 instead of 5, the equalled investment threshold and installed capacity for the PA and the TOC are lower, confirming Result 3.1. Also, α_5^K and α_5^X are lower than in the base case. With a lower *A*, congestion poses less of a problem to the port users, so that relatively more throughput is acceptable at the same infrastructure and that less capacity is required. If uncertainty (σ) is higher, the investment is made at a later moment, but the installed capacity will also be higher. Since these conclusions were also drawn in a private service port setting in the previous chapter, the robustness of this chapter's model is guaranteed. Additionally, the increase in uncertainty (σ) leads to another interesting observation. In this case, $\alpha_5^X (= 0.5624)$ is below $\alpha_5^K (= 0.829)$. This inversion of α_5 's has an important consequence for the negotiation power of the PA. Below $\alpha_5 = 0.5624$, the port still has negotiation power through timing the project at a higher threshold than what is optimal for the TOC. Above $\alpha_5 = 0.829$, the power of the PA also still lies in providing less capacity than what would be optimal for the TOC. However, between α_5^X and α_5^K , the TOC has a larger incentive to deviate from the PA's optimum. This is because it is optimal to later install less capacity than what has already been provided by the PA. This contains a reasonable incentive for the TOC to deviate by handling less cargo than agreed under the concession agreement. To avoid such a deviation, the PA could select the second investment strategy of urging the TOC to follow the PA's optimal strategy by reducing the TOC's investment decision degrees of freedom.

The sensitivity of the results to changes in the parameters discerning a public landlord port from a private service port are also included in Table 4.10. If the average local benefits

Parameter change	α_5	Actor a	Decision $(X_{T,a}^{**}, K_{a}^{**})$
Base case	0.504	PA	(38.54, 11.20)
		TOC	(32.77, 11.20)
	0.5693	PA	(36.08, 11.16)
		TOC	(36.08 , 11.24)
<i>A</i> = 4	0.472	PA	(35.09, 10.85)
		TOC	(28.40, 10.85)
	0.5643	PA	(32.12 , 10.81)
		TOC	(32.12 , 10.89)
$\sigma = 0.15$	0.5624	PA	(49.88 , 12.74)
		TOC	(49.88 , 13.02)
	0.829	PA	(41.04, 12.81)
		TOC	(91.83, 12.81)
$\lambda = 0.5$	0.491	PA	(38.89, 11.19)
		TOC	(32.19, 11.19)
	0.5673	PA	(35.96 , 11.15)
		TOC	(35.96 , 11.24)
<i>α</i> ₁ = 0.95	0.546	PA	(37.94, 11.19)
		TOC	(33.41, 11.19)
	0.594	PA	(36.07 , 11.17)
		TOC	(36.07 , 11.23)
<i>α</i> ₂ = 0.9	0.521	PA	(37.92, 11.19)
		TOC	(33.43, 11.19)
	0.5714	PA	(36.07 , 11.17)
		TOC	(36.07 , 11.23)
<i>α</i> ₃ = 0.3	0.5124	PA	(40.20, 11.20)
		TOC	(30.29, 11.20)
	0.6235	PA	(36.08 , 11.15)
		TOC	(36.08 , 11.28)
$\alpha_4 = 0.45$	0.5448	PA	(37.13, 11.19)
		TOC	(34.45, 11.19)
	0.57434	PA	(36.08 , 11.18)
		TOC	(36.08, 11.21)

Table 4.10: Changes of the PA and TOC's optimal investment decisions $(X_{T,a}^{**}, K_a^{**})$ at the respective α_5^K 's and α_5^X 's as a result of different parameter changes in a landlord port with public ownership.

Base case parameter values: A = 5, B = 1, c = 1, $c_h = 0.5$, $\mu = 0.015$, $\sigma = 0.1$, r = 0.06, $FC_I = 80$, $\gamma_1 = 180$, $\gamma_2 = 19$, $\gamma_3 = 0$, $\gamma_4 = 0.12$, $\alpha_1 = 0.9$, $\alpha_2 = 0.95$, $\alpha_3 = 0.35$, $\alpha_4 = 0.5$, $s_G = 1/2$, $s_{CS} = 1/2$, $\lambda = 0.4$.

per TEU were higher ($\lambda = 0.5$), the social welfare generated by the project would be higher too, making the project itself more attractive for the PA. As a result, the investment would be made slightly earlier, but it would also be smaller. Moreover, lower α_5 's are required for the port to equal the size or timing of both actors' investment decision. As was already explained, local benefits and consumer surplus are not included in the private TOC's operational objective function (π_{TOC}) and hence do not influence the TOC's optimal investment decision. This is opposed to the PA, whose project's attractiveness is now higher. Hence, the PA requires less income from the concession since already more welfare has been generated. The last four blocks of Table 4.10 show the impact of the PA receiving less of the total port revenue or incurring a higher share of the port costs (represented respectively by an increase in α_1 and a decrease in α_2 , α_3 or α_4). Qualitatively, the decision intervals remain similar. Additionally, almost identical optima as in the base case can be achieved, although through a higher value for α_5 . In each of the altered cases, the PA has a lower share of total port profit. Hence, the PA requires a higher concession fee, expressed as a share of the TOC's profit, to obtain the same level of welfare as in the base case.

4.6 Conclusion of this chapter

A private service port with one actor is the easiest setting to analyse new port capacity investment decisions, since this single, profit-maximising actor takes all decisions in the port. This chapter presented the case of a new landlord port with two actors: a private TOC handling the cargo under a concession agreement with the PA that manages the port and owns the land. This PA can be (partly or fully) publicly owned and is assumed to experience little competition from nearby ports. The case studied in this chapter leads to two extensions to real options modelling of capacity investment decisions in a port under congestion and uncertainty. Compared with the previous chapter, this chapter adds (i) the division of the port income and costs between the actors, as expressed by the α 's in the model, and (ii) the inclusion of social welfare in the PA's operational objective function to the model.

The results show that the PA and TOC's different objectives in a landlord port lead to a decision interval constituted by the different optimal investment strategies for both actors. From this interval, a common investment decision is to be made. Through the concession agreement, the PA could persuade the TOC to follow the strategy that is optimal for a service port in order to maximise the aggregate project value minus investment costs V - I. The concession fee can then be used as a redistribution mechanism of this V-I. Another possible strategy for the PA is to urge the TOC to invest in the PA's optimal strategy. This can be achieved by setting a concession fee that limits one of the two investment decision degrees of freedom of the TOC. The remaining investment decision variable is then conditionally optimised and could equal the optimal value of the PA. This discussion shows that although the PA would like to maximise and the TOC would like to minimise the concession fee to maximise their respective income, the optimal fee should also be considered from an investment analysis perspective. In this process, the negotiation power of the two actors has an impact on the optimal fee as well. In future research, it would be interesting to further elaborate the non-cooperative game approach with the PA acting as a Stackelberg leader, who moreover has all the power to set the concession fee.

The results also bear worthy capacity investment policy lessons to be learnt. Since the

benefits of port capacity are not only for current and future port users, but also for the economy of an entire region, the involvement of public money has a positive impact. First of all, more PA shares held by the government leads to larger and earlier investments in port capacity. More benefits and positive externalities are generated for the entire economy. This finding is opposite to the common RO finding, where more capacity leads to a later investment or vice versa. Another advantage of involving a public owner in the PA is that the concession fee can be set to trigger investment at the aggregated optimum. In such a case, the concession fee instrument will not be used to maximise PA profits, but can rather enable an equitable distribution of income and costs in the port. Of course, the PA should at the same time avoid incentives for the TOC to deviate from the concession agreement. Interesting to observe in light of reaching the aggregate optimum is that the optimum of a service port is not always reachable by a landlord port when the PA is (partly) publicly owned and the TOC privately. The TOC sets the optimal throughput without taking social welfare into account. This leads to a higher deviation from the aggregate optimal throughput and hence the investment strategy in a service port made by a single port actor, partly or fully owned publicly. As a result, the concession agreement is an important instrument to align the objectives of the TOC, the PA and the government. Finally, the model allows confirming two conclusions from the previous chapter. Both an increase in congestion costs and uncertainty lead to a new port investing in more capacity, but at a later moment in time.

The analyses made in this chapter and the previous chapter consider the port as a monopolist. In reality however, many ports do not operate as monopolists. They experience competition from nearby ports. The port that is part of the chain with the lowest generalised cost has the highest probability to attract and handle the cargo. In the Hamburg - Le Havre range for example, Antwerp and Rotterdam are fierce competitors in their attempts to attract important loops of global shipping lines. In such cases, the capacity investment decision of a neighbouring port influences the own decision. Both ports try to invest before the other, to offer free capacity that reduces waiting time in the own port. This is crucial to attract the shipping lines and their cargo for the overlapping hinterland, as these shipping lines are averse to waiting. As this competition was left out of the scope of this chapter, it forms the starting point of the next chapter. In that chapter, the capacity investment decisions of two competing service ports are considered, instead of the decision of one monopoly port. As was shown in this chapter, the PA and TOC can decide to invest in the aggregate optimum. By readopting the service port model in the next chapter, the focus will be on the impact of competition on the aggregate optimal port investment decision. Also the isolated functions of the PA could be used to focus on the PA's individual optimum, which could subsequently be imposed on the TOC. Moreover, both port authorities will be assumed to have the same ownership structure, which can be private, public or a combination of both, in line with those of this chapter.
Chapter 5

The capacity investment decisions of two competing ports

Ports experience competition from nearby ports with overlapping hinterlands to attract cargo (De Langen et al., 2012). Examples can be found all over the world: ports in the Hamburg - Le Havre range, the Spanish ports, East Asian ports and ports on the East and West Coast of the USA (Jacobs, 2007; Meersman & Van de Voorde, 2002; Yap et al., 2006; Ng, 2006; Cullinane et al., 2005). A number of reasons cause this competition (De Langen, 2007). A port should strive to be part of the logistics chain with the lowest generalised cost to serve the hinterland. Such a competitive advantage increases the probability of the port handling the goods (Heaver et al., 2000; De Langen et al., 2012; Talley et al., 2014). Location, service quality and efficiency play an important role herein (Meersman et al., 2010). One of the critical problems in the port is congestion (Novaes et al., 2012). Shipping lines are averse to the waiting time and logistics costs caused by delays. Since delay costs depend on the goods transported, the aversion to waiting is different for each shipping line (Blauwens et al., 2016; De Borger et al., 2008). As a result, ports tend to avoid congestion in order to be competitive with respect to non-congested ports nearby.

In order to avoid congestion, ports can charge a higher price, leading to a demand reduction (Xiao et al., 2013), or accommodate the demand by investing in more throughput capacity (Meersman & Van de Voorde, 2014a; Xiao et al., 2012). Y.-T. Chang et al. (2012) used an economic approach to balance all elements of capacity in a port, given the objectives of the port owners, explicitly taking waiting time costs into account. An important consideration is that a congestion-reducing capacity investment in one congested port increases its own demand, and that it reduces the demand in the other congested port, if the level of congestion in the investing port will become relatively lower than the level of congestion in the other port (Wan et al., 2013). When there is more competition, excess (free) capacity is required more. In this light, the port's investment decision also depends on the decision of the competitor which serves (a part of) the same hinterland. In this way, competition has an impact on the capacity investment decision of ports and vice versa (Huisman & Kort, 2015; Xiao et al., 2012; Haralambides, 2002). Therefore, the port needs to endogenise the other port's investment decision to avoid investing in capacity that is not used due to the other port's capacity serving part of the demand (Huberts et al., 2015).

The capacity decision is even more complicated by the fact that the demand faced by the port is very uncertain (Vilko & Hallikas, 2012; Huisman & Kort, 2015). Many different sources of this uncertainty have been identified in Chapter 2, including not only the impact of the financial crisis on global trade and the uncertain decisions of many actors in the logistics chain, but also technological, environmental and regulatory changes. To correctly account for the value of different sorts of managerial flexibility in capacity investment decisions under uncertainty in a competitive environment, real options models have been identified (Dixit & Pindyck, 1994; Bar-Ilan & Strange, 1996; Dangl, 1999; Aguerrevere, 2003; Hagspiel et al., 2016), in combination with game-theoretic modelling (Azevedo & Paxson, 2014). The options considered in this chapter are a one-shot investment in a new port with flexible size and timing and a flexible throughput level. The present chapter extends the previous models and the real options literature in general by determining the investment timing and capacity choice in a continuous-time framework that simultaneously deals with competition and volume flexibility.

The objective of this chapter is to analyse how the capacity investment in two new ports is influenced by inter-port competition under congestion and demand uncertainty. To this end, the following research sub-question is put forward: (RQ 5.1) What impact does interport competition have on the investment decision of both the leader's and the follower's new port under uncertainty? In order to accurately analyse the capacity investment decisions of the competing ports, a game-theoretic RO model including congestion and uncertainty is developed in this chapter. This model extends the state-of-the-art in the field of industrial organisation with a port-specific application, wherein congestion plays an important role (De Borger & Van Dender, 2006; De Borger et al., 2008).

As opposed to the majority of the port competition literature, the timing decision is included, which allows for heterogeneous ports in a game where the ports do not necessarily invest at the same time. In this way, the model accounts for entry deterrence and preemption as additional strategies to the often solely considered accommodation strategy. Luo et al. (2012) modelled the preemptive prices by the dominant port, but did not consider uncertainty and congestion.¹ In order to study a port context, the real options model of Huisman & Kort (2015) is extended in three ways, to account for the cost of port congestion, differentiated services offered by different ports (Bichou & Gray, 2005) and mixed ownership by both private and public actors (Xiao et al., 2012).

¹Ishii et al. (2013) also examined port competition and stochastic demand with the size and timing of capacity investment. However, they did not consider the endogenous timing issue and the issue of entry deterrence or preemption.

As a result of this chapter's methodological additions, it is possible to account for the frequently observed phenomenon that port investments in two competing ports do not necessarily take place at the same time, notwithstanding that simultaneous investment is also possible. However, some simplification with respect to the previous chapter's model is needed too, in order to guarantee the mathematical tractability of this model and without losing general applicability. The previous chapter indicated that the concession fee is a viable port authority (PA) instrument to steer the capacity investment of the terminal operating company (TOC) to the desired direction. The PA could force its own optimum, or the optimum from an aggregated viewpoint could be obtained through mutual compromises. Hence, to study the impact of competition between neighbouring ports on the capacity investment decision of both ports, it could be possible to resort to the individual PA's objective functions or the aggregated objective functions. Since the previous chapter showed that the differences in discounted cash flows and investment decisions are negligible between the individual PA's optimum or the aggregate optimum, it is assumed in this chapter that the aggregate optimal strategy is followed. Hence, the considered ports are modelled as public service ports. This argument especially holds in the case of a publicly owned port or if there is only one actor operating the port. Hence, the model from Chapter 3 forms the starting point here. The extension with public and private ownership from Chapter 4 is however retained.

This chapter is structured as follows. The next section describes the basic model for the two-port setting and each port's investment decision making objectives. Subsequently in Section 5.2, this chapter's specific methodology and analysis are described. Since no closed-form solutions can be derived here either, the numerical results with respect to the impact of competition combined with different government ownership structures are discussed in Section 5.3. Sensitivity analysis in Section 5.4 identifies the impact of other factors on the investment decision in a competitive setting. The final section presents this chapter's conclusion.

5.1 Economic setting and methodology

In this chapter, two private or public service ports that compete to handle a flexible amount of throughput, but offer differentiated services, are considered. The next subsection deals with the individual situation in, and the relevant information for each port. This is followed by a discussion of the type of competition between both ports and the related game-theoretic modelling.

5.1.1 Individual port situations

The fact that ports offer differentiated services is mainly the result of the different geographic locations of the ports, implying different distances to the hinterland. Moreover, each port being organised differently adds to this service differentiation. Also their service levels as expressed by their occupancy rates, with an impact on their prices, may differ (De Borger & Van Dender, 2006). The product market's heterogeneity is expressed through the differentiation parameter δ in the inverse demand function of the previous chapters, giving rise to the gross willingness to pay, ρ_i , for port *i* at time *t* (Vives, 1999; Xiao et al., 2012):

$$\rho_i(t) = X(t) - Bq_i(t) - \delta Bq_i(t), \qquad (5.1)$$

with *X* being the demand shift parameter, q_i the throughput of port *i* and q_j the throughput of port *j* ($j \neq i$). A similar function holds for port *j*, with subscripts *i* and *j* interchanged.²

Depending on the location and services of both ports, parameter δ can vary between zero and one. In the case of a port not experiencing competition from another port, e.g., with two distinct hinterlands, δ would equal 0 and the model would simplify to a monopoly model. For two identical, hypothetical ports at the same location and offering exactly the same services, δ would equal 1. In this way, δ also reflects the difference in hinterland and port access (tugging and piloting) costs. Initially, δ is set to 0.6 to take the different characteristics of two competing ports in a close range into account (e.g., Antwerp and Rotterdam in the Hamburg - Le Havre range). This value is altered in the sensitivity analysis to 0.9, to account for ports situated back to back but offering slightly different services, (e.g., Los Angeles (LA) and Long Beach in California, Seattle and Tacoma in Washington, Vancouver and Prince Rupert in British Columbia, or Gdynia and Gdansk in Poland), and to 0.3 to account for two ports that are situated further from each other, possibly at different coast lines and with only partly overlapping hinterlands (e.g., LA/Long Beach and New York/New Jersey in America, or Hamburg - Le Havre and Trieste/Koper in Europe (De Langen, 2007)).

In order to model the uncertain demand evolution, the intercept of the inverse demand function, X, again follows a GBM with an independent parameter for the drift (economic growth, μ) and the variability (uncertainty, σ): see Equation (2.4). Additionally, the slope of the demand function (B) is again normalised and set to 1. For the sake of readability, denoting the functional dependence on time will be omitted in the remainder of this chapter. Since the gross willingness to pay involves not only the price paid in port i, p_i , but also a cost incurred because of congestion and resulting waiting times at high occupancy rates (Zhang & Zhang, 2006; Xiao et al., 2012), according to the discussion in Chapter 3, price p_i can be written as

$$p_i = X - Bq_i - \delta Bq_j - AX \frac{q_i}{K_i^2},$$
(5.2)

²As a result of port services being differentiated products leading to different prices in different ports, a single market demand function cannot be used.

with the latter term the congestion unit cost term and K_i the theoretical design capacity of port *i*. Total congestion cost for port *i* then equals $AX(q_i/K_i)^2$.

An important difference from the economic models of Chapters 3 and 4 is that in this chapter, the design capacity K_i is not a hard constraint for the maximum throughput. This adaptation is necessary to avoid unnecessary mathematical complexity.³ Realism is not to be lost if A, the monetary scale factor in the congestion cost, is sufficiently high. In that case, the price would become too low to profitably operate if $q_i > K_i$. This will be avoided as much as possible, but in some emergency cases, the port has no other choice than to handle the additional cargo. If one of the main customers wants the port to handle more than its capacity, the port needs to find solutions such as temporarily stocking containers on the quay. These organisational complexities imply high extra costs, but losing the customer is even a lot worse. To have a realistic setting, A = 5 is retained from the previous chapters. This results in a congestion cost size that discourages full occupancy in most situations, but that does not prevent it de facto and that allows in some exceptional cases for exceeding design capacity, at a high cost. Since in most cases, ports like to avoid these high costs and because a nearby port with free capacity offers an alternative, such high levels of occupancy are hardly encountered in reality as well as in the numerical simulations of the described ports in this chapter.

Since not only situations with two private ports are considered, but also with two public ports, it is not sufficient to only consider annual profit maximisation,⁴ which is the objective of a private port. Governments also consider positive externalities or local spillover benefits per unit of throughput handled (e.g., employment and local industry growth), and consumer surplus in their social welfare (SW_i) maximisation (Xiao et al., 2012; Jiang et al., 2017). In this chapter, social welfare generated by port *i* is calculated as the sum of the profit of port *i*, the spillover benefits $\lambda \cdot q_i$ and a share s_{CS} of consumer surplus generated by port *i* (CS_i), since some governments only consider the part that is relevant for the region they govern.⁵ To account for the fact that the two ports are owned by a different, independent

³The impact of this change on the results in a monopoly setting is negligible, as is proven in Chapter 6. The disadvantage of the constraint $q_i \leq K_i$ in a two-port setting is that more regions for q_i^{opt} arise (e.g., one of the two ports at full capacity, both at full capacity, etc.) and that the relative order of these regions is endogenous and capacity dependent. It could be added to the model, but only complexity would increase and tractability would be reduced.

 $^{{}^{4}\}pi_{i} = (p_{i} - c) \cdot q_{i} - c_{h}K_{i}$, with c the marginal operational cost and c_{h} the capacity holding cost

⁵The Antwerp city council, shareholder of the Port of Antwerp's port authority, might only be interested in local spillover benefits next to port profit, which in turn differs from the objectives of Rotterdam's council. Similarly, the Canadian government, shareholder of the Port of Vancouver, might or might not consider the *CS* of Chinese shipping lines, having an influence on parameter s_{CS} . To compare with a full social planner in each country, state or city, instances with s_{CS} set to 1 are included as well. In Section 5.4, the case where one government owns shares of both ports, giving rise to a different CS_i calculation, is considered as well.

government, consumer surplus in port *i* is calculated as follows (Xiao et al., 2012):

$$CS_{i}(q_{i},q_{j}) = CS_{i}(q_{i}) = \int_{0}^{q_{i}} \rho_{i}(y,q_{j})dy - \rho_{i}(q_{i},q_{j})q_{i}$$

$$= \int_{0}^{q_{i}} (X - By - \delta Bq_{j})dy - (X - Bq_{i} - \delta Bq_{j})q_{i}$$

$$= \frac{Bq_{i}^{2}}{2}.$$
 (5.3)

This calculation reflects that the difference between the gross willingness to pay for each unit of throughput in port i and the actual generalised cost in port i is not affected by the throughput realised by port j. In the case of two independent governments owning only shares of port i and j respectively, q_i is considered as exogenous by port i.

The aggregated operational objective function Π_i of port *i* is then again the weighted sum of the individual owner's objectives, with the shares of ownership used as the weights. If the government owns a share s_G of the port, and the private party hence $1 - s_G$, the weighted operational objective function of port *i* is composed as follows:

$$\Pi_{i}(X, K_{i}, q_{i}, q_{j}) = \pi_{i}(X, K_{i}, q_{i}, q_{j}) + s_{G} \cdot \lambda q_{i} + s_{G} s_{CS} \cdot B q_{i}^{2}/2.$$
(5.4)

In this chapter, s_G is varied between 0 and 1 in the numerical simulations. Hence, the same range of ownership structures as in Chapter 4 is considered, but in every case the two governments own the same share of their respective ports, and attach the same weight to consumer surplus. Nonetheless, as higher s_G and s_{CS} lead to social welfare being taken more into account by the ports, two public ports fully considering consumer surplus will generate the highest social welfare.

In this chapter, the analysis is not restricted just to ports that are homogeneous in costs, as is often done in game-theoretic approaches. Here, the ports may differ in costs. The investment cost of the project, I_i , and more specifically the fixed investment cost $FC_{I,i}$ is chosen to reflect the cost difference of a similar project. This is illustrated by the example of Antwerp and Rotterdam. In Antwerp, the Deurganckdok has been dug, whereas the Maasvlakte 2 in Rotterdam has been constructed in open water through rainbowing sand, resulting in an $FC_{I,2}$ that is 20 million euro higher for this second project (Rotterdam) than the $FC_{I,1}$ of the project in Antwerp. The existence of the cost difference is assumed to be known, as both ports can ex-ante observe the type of construction technology used. This situation gives rise to a Stackelberg leader-follower game, with the possibility of simultaneous investment under entry accommodation (Huisman & Kort, 2015). According to Musso et al. (2006), the Stackelberg oligopoly structure with leaders and followers is a good framework to analyse port investments in a competitive environment. It is however impossible for the port with a cost disadvantage to become the leader. It will be the follower, unless it chooses to invest simultaneously (Pawlina & Kort, 2006).⁶ Hence in this chapter, the in-

⁶Only if the port with the cost disadvantage had a higher number of shares owned by the government, this result would not necessarily hold. Such cases are however not considered in this chapter.



Figure 5.1: Total investment cost functions for the two projects.

vestment timings of both ports are endogenous, allowing for simultaneous investment as well.

The investment cost functions of one single capacity project of respectively the port with a cost advantage (project 1) and with a cost disadvantage (project 2) are graphed in Figure 5.1. These functions are similar fourth-order polynomials as in Chapter 3, with the same parameters except for the two different intercepts $FC_{I,i}$ to express cost differences between the two ports. In Table 5.1, the full economic model is summarised. The parameters derived in Chapters 3-4 are retained here. The assumptions of this RO competitive game model are recapitulated in Table 5.2.

5.1.2 Game-theoretic modelling of competition

As opposed to the majority of the literature, the analysis in this chapter is not limited to both ports investing simultaneously. When the leader invests in sufficient capacity at the moment the market is not large enough to accommodate two ports, the leader is deterring the entry of the follower to a later moment. In this way, a four-stage game is discerned in the case of entry deterrence (Huberts et al., 2015). In the first stage, the leader installs capacity K^{L} as soon as its relevant investment threshold X_{T}^{L} is reached for the first time from below at time T^{L} ; see Figure 5.2.⁷ In the second stage, immediately after the first stage, the

⁷In line with the terminology of Huberts et al. (2015), the first port to enter the market, i.e., the leader, becomes the incumbent after investment. If this initial investment is sufficiently large, it will be entry deterring, to delay the follower's investment. If the leader's initial investment is relatively small, it will be entry accommodating to allow simultaneous follower investment. This is described later in a three-stage game.

Variabl	es		
	p_i	=	price in port <i>i</i>
	q_i	=	throughput in port <i>i</i>
	K_i	=	capacity in port <i>i</i>
Inverse	e demand function	n (∀ <i>i</i>	$i): p_i = X - Bq_i - \delta Bq_j - AX \frac{q_i}{K_i^2}$
	B(=1)	=	slope
	$\delta(=0.6)$	=	product differentiation parameter
	A(=5)	=	monetary scale factor of congestion cost
Deman	ld shift parameter	:X:	$dX(t) = \mu X(t)dt + \sigma X(t)dZ(t)$
	t (=annual)	=	time horizon
	Ζ	=	standard Wiener process
	$\mu(=0.015)$	=	drift of Z (i.e., growth)
	$\sigma(=0.1)$	=	drift variability of Z (i.e., uncertainty)
Total c	$ost (\forall i) \ TC_i = c \ q_i$	$+c_h$	K_i
	c(=1)	=	constant marginal operational cost
	$c_h(=0.5)$	=	cost to hold one unit of capacity in place
Investr	nent cost ($\forall i$) $I_i =$	FC_I	$\gamma_{i,i} + \gamma_1 K_i - \gamma_2 K_i^2 + \gamma_3 K_i^3 + \gamma_4 K_i^4$
	$FC_{I,i}(=80, 100)$	=	fixed investment cost of port 1 and 2 respectively
	$\gamma_1(=180)$	=	first-order coefficient
	$\gamma_2(=19)$	=	coefficient reflecting investment economies of scale
	$\gamma_3(=0)$	=	omitted third-order coefficient
	$\gamma_4 (= 0.12)$	=	coefficient posing boundary to maximum project size
Operat	ional objective fu	ncti	on $(\forall i)$ $\Pi_i = \pi_i + s_G \cdot \lambda q_i + s_G s_{CS} \cdot CS_i$
	π_i	=	annual profit of port <i>i</i> , i.e., $p_i q_i - TC_i$
	$\lambda(=0.4)$	=	spillover benefits per unit q_i
	CS_i	=	consumer surplus in port <i>i</i> , i.e., $Bq_i^2/2$
	$s_G (\in [0;1])$	=	share of the port owned by the government
	$s_{CS}(\in [0;1])$	=	share of total CS_i taken into account by the government

Table 5.1: Overview of Chapter 5's model and the selected parameters.

	Assumption(s)
Project type? Capacity constraint? Port type? Port competition? Time to build? Phased investment?	Two new ports Soft constraint Public and/or private service ports* Duopoly: Stackelberg (investment) and Cournot quantity (operations) competition Not considered Project installed at once: available after completion
Specific assumptions	Each port is owned by a different, independent government Each government has the same objectives and ownership structures of both ports are equal Ports differ in location and service quality, as well as in fixed investment costs Fixed investment cost difference between ports due to different construction technology: known by both ports Cost-advantaged port will be the investment leader. Follower invests simultaneously or later
General assumptions	Price (sum of port dues and terminal tariff) transparency \rightarrow Linear additive inverse demand function Demand uncertainty, modelled using a GBM with constant, independent growth (μ) and uncertainty (σ) parameters Growth market with $r > \mu > 0$, to ensure RO model convergence ^{**} Poisson-distributed ship arrivals at the ports: delays start at about 50% and increase sharply beyond 75-80% occupancy User congestion costs depend on dock users' average value of time, which evolve with the market (i.c., increasing) Investment costs: fourth-order polynomial with economies of scale and size boundary Deterministic, linear operational and capacity holding costs Flexibility of investment timings and sizes \rightarrow Options to postpone the investment Once-and-for-all investment decisions with flexible throughput levels Infinite project life times (maintenance executed)
**: Condition <i>r</i> > SCOPE: Theoretical design	*: Owned and operated by one single actor. μ leads to a finite timing decision, instead of infinite investment postponement. Sensitivity tested with negative growth rates as well. capacity for container throughput services. Other port services disregarded (sufficiently present or realised by actors other than the PA or TOC). Focus on lumpy investment in port infrastructure and irreversible elements of superstructure.

Table 5.2: Overview of Chapter 5's model assumptions.



Figure 5.2: Sample evolution of demand shift parameter *X*, following a GBM over time, showing the relationship between the optimal investment timing (T^{L} and T^{F}) and the threshold for $X(X_{T}^{L} \text{ and } X_{T}^{F})$ of the leader and the follower port.

incumbent starts to operate under a temporary monopoly at its optimal throughput level, $q(X, K^{L}, 0)$. This throughput level is flexible, and depends on the instantaneous value of X. In the third stage, when the market has grown sufficiently until X_T^F is reached at time T^F , the follower will invest in K^F . This size is not only dependent on the timing of the follower's investment, but also conditional on the investment size of the leader. Once the follower invests, the incumbent loses its monopoly. Both ports will now operate at their optimal throughput level, given the level of the other port, their capacity investment size and the current value for the demand shift parameter X. In this light, both ports compete in quantities, leading to a Cournot equilibrium.

Following Wan & Zhang (2013), ports competing on quantity are considered (i.e., Cournot competition). In general, which model of competition is applicable to a particular industry depends largely on its production technology. In Cournot competition, firms commit to quantities, and prices are then adjusted to clear the market (i.e., the committed quantities) implying the industry is flexible in price adjustments, even in the short term. On the other hand, in Bertrand (price) competition, capacity is unlimited or easily adjusted in the short term. There are some good reasons to believe that quantity competition may be more realistic than price competition in the case of ports. For instance, Quinet & Vickerman (2004, p. 263) remarked:

"The general idea which emerges from the theoretical analysis is that when transport capacities are high, or can be enlarged through the transfer of capacity from other locations, and the services provided are not differentiated, then competition is likely to be of a Bertrand type, based on price. [...] If, on the other hand, capacity is difficult to increase, then competition is likely to be of a Cournot type, based on quantities. This is the case found, for example, in rail, maritime or inland waterway transport."

The main reason why port capacity is difficult to change (relative to the ease and speed with which prices can be adjusted) is that port capacity investment is lumpy, time-consuming and irreversible, which is consistent with the set-up described above. Moreover, it involves high investment costs, including high fixed costs of designing, scheduling and implementing investments.⁸ Indeed, with capacity constraints, Van Reeven (2010) assumes quantity competition between TOCs, based on Kreps & Scheinkman's (1983) argument of capacity-constrained price competition yielding quantity competition. Furthermore, the market "conduct parameters" with respect to port charges of the three largest, competing Australian seaports were empirically estimated by Menezes et al. (2007). The calculation based on their results indicates that at the 10% level of statistical significance, the hypothesis of price competition among the ports is rejected.⁹ As a result of Cournot competition, Wan & Zhang (2013) find that investment increases the own port's profit, which is not necessarily true under Bertrand competition (De Borger et al., 2008). Additionally, the use of quantity as a decision variable has been observed for firms in the production of goods as well as for firms in the production of services (e.g., competition among airlines or airports; see Zhang & Czerny (2012) for a recent survey of the literature). A final advantage of using quantity competition is that it is found to increase competition between ports, whereas price competition leads to collusion (Musso et al., 2006). The latter would be a less realistic framework to describe the competing ports.

As a final remark to the described game, note that it is slightly, but not substantially, different in case the leader invests when the market is large enough to accommodate two ports. This would lead to a three-stage game, as stage 2 is omitted. In this situation, leader investment (stage 1) is immediately followed by the follower's investment (stage 3). Right after this investment, both ports again set their optimal throughput levels under Cournot competition in stage 4. The structure of the game is summarised in Figure 5.3.

5.2 Optimal investment strategies

With two heterogeneous competing ports making a decision to invest in new capacity, of which the size and the timing are not ex-ante defined, the optimisation problem is solved

⁸Here, abstraction is made of the fact that ports can increase capacity to a limited extent in the short term, e.g., by increasing productivity or extending working hours. Such an approach is plausible when studying the investment of market entrants.

⁹For studies using conduct parameters to assess the empirical relevance of certain oligopoly models to a particular market, see, e.g., Bresnahan (1989), and Brander & Zhang (1990).



Figure 5.3: Summary of the competitive RO Stackelberg investment game methodology.

backwards using dynamic programming. First, the decision of the follower is optimised, taking the decision of the leader as given. Subsequently, the optimal entry-deterring or entry-accommodating decision of the leader is explicated, taking the information about the resulting follower's optimal decision into account as expressed by its reaction function.

5.2.1 A port's throughput decision

First, port *i*'s optimal throughput quantity, once its investment is made, needs to be determined. Differentiating the operational objective function $\Pi_i(X, K_i, q_i, q_j)$ w.r.t. q_i leads to

$$q_{i}^{opt}(X, K_{i}, q_{j}) = \max\left\{0, \frac{X + s_{G}\lambda - c - \delta B q_{j}}{2\frac{AX}{K_{i}^{2}} + (2 - s_{G} s_{CS})B}\right\}.$$
(5.5)

By substituting the expression for $q_j^{opt}(X, K_j, q_i)$ into $q_i^{opt}(X, K_i, q_j)$, q_i^{opt} can be expressed in terms of X, K_i and K_j . The advantage of such an expression is that this value is dependent only on X once both investments are made, as opposed to also being dependent on the continuously changing q_j . This leads to

$$q_{i}^{opt}(X, K_{i}, K_{j}) = \begin{cases} 0, & X < c - s_{G}\lambda, \\ (X + s_{G}\lambda - c) \left(2\frac{AX}{K_{j}^{2}} + (2 - s_{G}s_{CS} - \delta)B\right) \\ \hline \left(2\frac{AX}{K_{i}^{2}} + (2 - s_{G}s_{CS})B\right) \left(2\frac{AX}{K_{j}^{2}} + (2 - s_{G}s_{CS})B\right) - \delta^{2}B^{2}, & X \ge c - s_{G}\lambda. \end{cases}$$
(5.6)

Through the boundary value ensuring $q_i^{opt} \ge 0$ in Equation (5.6), two regions for *X* are identified for both ports: $R_1 = [0, c - s_G \lambda)$ where throughput is zero and $R_2 = [c - s_G \lambda, \infty)$ where throughput equals the optimal throughput. In this R_2 , throughput could exceed capacity, but only at a substantial congestion cost. Hence, as opposed to Yang & Zhang (2012), not just the interior solutions are considered; rather also corner solutions are considered.

For each region R_k , with $k \in \{1, 2\}$, the resulting optimum of port *i*'s operational objective function can be calculated as $\Pi_i(X, K_i, q_i^{opt}(X, K_i, K_j), q_j^{opt}(X, K_j, K_i))$. This is rewritten as $\overline{\Pi}_i(X, K_i, K_j)$.

5.2.2 The follower's investment decision

To determine the follower's optimal investment decision, the approach of Chapter 3 can be followed, thereby taking the capacity installed by the leader additionally into account.

 $\overline{\Pi}_i(X_{T,i}^{\rm F}, K_i^{\rm F}, K_j^{\rm L})$ at the time port *i* invests after port *j*'s investment in $K_j^{\rm L}$, forms the input to calculate the value of the investment project for port *i*, $V_i^{\rm F}(X_{T,i}^{\rm F}, K_i^{\rm F}, K_j^{\rm L})$, for which it pays $I_i(K_i^{\rm F})$ when it invests at threshold $X_{T,i}^{\rm F}$ at time $T^{\rm F}$ in capacity $K_i^{\rm F}$, $n^{\rm C}$ given that the other port has invested in $K_j^{\rm L}$. The differential equation for $V_i^{\rm F}(X_{T,i}^{\rm F}, K_i^{\rm F}, K_j^{\rm L})$ in a two-port setting is found by applying Itô's Lemma and Bellman Equation to

$$V_i^{\mathrm{F}} = \mathbf{E} \int_{0}^{\infty} \max_{q_i} \{\Pi_i (T^{\mathrm{F}} + \tau)\} \mathrm{e}^{-r\tau} \mathrm{d}\tau = \mathbf{E} \int_{0}^{\infty} \overline{\Pi}_i (T^{\mathrm{F}} + \tau) \mathrm{e}^{-r\tau} \mathrm{d}\tau.$$
(5.7)

This results in

$$\frac{\sigma^{2}}{2} (X_{T,i}^{\mathrm{F}})^{2} \frac{\partial^{2} V_{i}^{\mathrm{F}}}{\partial (X_{T,i}^{\mathrm{F}})^{2}} (X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) + \mu X_{T,i}^{\mathrm{F}} \frac{\partial V_{i}^{\mathrm{F}}}{\partial X_{T,i}^{\mathrm{F}}} (X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) - r V_{i}^{\mathrm{F}} (X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) + \overline{\Pi}_{i} (X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) = 0,$$
(5.8)

with r the discount rate (Dixit & Pindyck, 1994).

The solution for each region R_k is given by

$$V_{i}^{F}(X_{T,i}^{F}, K_{i}^{F}, K_{j}^{L})|_{X_{T,i}^{F} \in R_{k}} = V_{i,k}^{F}(X_{T,i}^{F}, K_{i}^{F}, K_{j}^{L})$$

$$= G_{i,(k,1)}^{F}(K_{i}^{F}, K_{j}^{L}) \cdot (X_{T,i}^{F})^{\beta_{1}} + G_{i,(k,2)}^{F}(K_{i}^{F}, K_{j}^{L}) \cdot (X_{T,i}^{F})^{\beta_{2}}$$

$$+ \overline{V}_{i,k}^{F}(X_{T,i}^{F}, K_{i}^{F}, K_{j}^{L}), \qquad (5.9)$$

with $\overline{V}_{i,k}^{\text{F}}(X_{T,i}^{\text{F}}, K_i^{\text{F}}, K_j^{\text{L}})$ the particular solution of differential equation (5.8) and $\overline{\Pi}_i = \overline{\Pi}_{i,k}$ in each region R_k . The roots β_1 and β_2 are given by Equation (3.22), whereas the boundary conditions are given by

$$\begin{cases} V_{i}^{\mathrm{F}}(0, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) = \mathbf{E} \int_{0}^{\infty} -c_{h} K_{i}^{\mathrm{F}} \mathbf{e}^{-rt} \mathrm{d}t = \frac{-c_{h} K_{i}^{\mathrm{F}}}{r} \\ \lim_{X_{T,i}^{\mathrm{F}} \to +\infty} \left(V_{i}^{\mathrm{F}}(X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) - \overline{V}_{i,2}^{\mathrm{F}}(X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) \right) = 0 \\ \lim_{X_{T,i}^{\mathrm{F}} \stackrel{\leq}{\to} c - s_{G} \lambda} V_{i}^{\mathrm{F}}(X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) = \lim_{X_{T,i}^{\mathrm{F}} \stackrel{\geq}{\to} c - s_{G} \lambda} V_{i}^{\mathrm{F}}(X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{L}}) \\ \lim_{X_{T,i}^{\mathrm{F}} \stackrel{\leq}{\to} c - s_{G} \lambda} \frac{\partial V_{i}^{\mathrm{F}}}{\partial X_{T,i}^{\mathrm{F}}} (X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{F}}, K_{j}^{\mathrm{I}}) = \lim_{X_{T,i}^{\mathrm{F}} \stackrel{\geq}{\to} c - s_{G} \lambda} \frac{\partial V_{i}^{\mathrm{F}}}{\partial X_{T,i}^{\mathrm{F}}} (X_{T,i}^{\mathrm{F}}, K_{i}^{\mathrm{I}}). \end{cases}$$
(5.10)

The follower's investment problem objective function

$$\max_{T_i^{\rm F} \ge 0, K_i^{\rm F} \ge 0} \mathbf{E}\left\{ \left[V_i^{\rm F}(X_{T,i}^{\rm F}, K_i^{\rm F}, K_j^{\rm L}) - I_i(K_i^{\rm F}) \right] e^{-r T_i^{\rm F}} | X(t=0) = X \right\},\tag{5.11}$$

gives rise to the following option value

$$F_{i}^{\rm F}(X, K_{j}^{\rm L}) = \max\left\{ e^{-r \, \mathrm{d}t} \, \mathbf{E}(F_{i}^{\rm F}(X)) + \mathrm{d}F_{i}^{\rm F}(X)), \max_{K_{i}^{\rm F}} \left[V_{i}^{\rm F}(X, K_{i}^{\rm F}, K_{j}^{\rm L}) - I_{i}(K_{i}^{\rm F}) \right] \right\},$$
(5.12)

¹⁰Note that in contrast to the other chapters, the threshold notation is used here to clearly diversify between the different thresholds considered in this chapter.

which is to be maximised. This allows determining the optimal timing and size of the follower port's capacity investment decision, given the size of the leader's investment. This analysis is relevant for deciding on new port projects near existing ports. If the Italian government for instance wants to install a new port close to the French border and the port of Marseille, e.g., near Ventimiglia, this approach can be used. The inner maximisation of Equation (5.12) considers the project's net present value, $V_i^F(X_{T,i}^F, K_i^F, K_j^L)$ minus $I_i(K_i^F)$, in order to determine the optimal capacity of the follower $(K_i^{*,F})$ in terms of its timing threshold $(X_{T,i}^F)$ and the capacity already installed by the other port (K_j^L) . This optimal capacity needs to satisfy

$$\frac{\partial V_i^{\mathrm{F}}}{\partial K_i^{\mathrm{F}}} (X_{T,i}^{\mathrm{F}}, K_i^{*,\mathrm{F}}, K_j^{\mathrm{L}}) = \frac{\partial I_i}{\partial K_i^{\mathrm{F}}} (K_i^{*,\mathrm{F}}), \qquad (5.13)$$

stating that capacity is added up to the point where the marginal added value of extra capacity equals the marginal cost of extra capacity. Using the relevant boundary conditions from Section 3.4, including the smooth pasting and value matching conditions, on the outer maximisation of Equation (5.12), the follower's optimal timing threshold $(X_{T,i}^{*,F})$ can be determined in terms of its own installed capacity (K_i^F) and the capacity already installed by the other port (K_i^L) (Dangl, 1999; Huisman & Kort, 2015; Hagspiel et al., 2016).

Solving the system

$$\begin{cases} X_{T,i}^{**,F}(K_j^{\rm L}) = X_{T,i}^{*,F}(K_i^{**,F}(K_j^{\rm L}), K_j^{\rm L}), \\ K_i^{**,F}(K_i^{\rm L}) = K_i^{*,F}(X_{T,i}^{**,F}(K_i^{\rm L}), K_i^{\rm L}), \end{cases}$$
(5.14)

results in the investment decision optimal in both timing and size of the follower, only dependent on $K_j^{\rm L}$ of the leader. It is written as $(X_{T,i}^{**,F}(K_j^{\rm L}), K_i^{**,F}(K_j^{\rm L}))$. It is worth highlighting that the timing of the leader has no impact on the follower's decision, as opposed to the leader's investment size. Concerning timing, it only matters that the leader has invested (Huisman & Kort, 2015). As a result, it could be that the optimal timing of the follower is before the investment timing of the leader. This typically happens in large markets, so that $X_{T,i}^{**,F}(K_j^{\rm L}) < X_{T,j}^{\rm L}$ and the leader plays an entry-accommodating strategy (see the next section). The investment strategy of the follower will in such case be $X_{T,j}^{\rm L}, K_i^{*,F}(X_{T,j}^{\rm L}, K_j^{\rm L})$, as both ports will invest right away from the start.

5.2.3 The leader's investment decision

The leader port with the cost advantage needs to decide on its own optimal timing and capacity, taking the calculated reaction function of the follower into account. There are three investment alternatives for the leader. First, if the market is initially large enough, the leader could invest in an entry-accommodating capacity, which is low enough so that the follower can profitably invest at the same time and serve a part of the market as well (Huisman & Kort, 2015). If the market is initially smaller and the leader has a large cost advantage, it is also possible to invest in a capacity that is higher, so that the entire market is served at the moment of investment, the unrestricted entry deterrence threshold. In this case, entry of

the follower is deterred to a later moment, when the market will have grown and it will have become also for this port profitable to invest. If the cost differences are rather small however, both ports could strive to be the first to invest. The leader's investment then occurs at the preemption point of the port with the cost disadvantage, where the latter's follower value, $V_2^{\rm F} - I_2$, equals its leader value $V_2^{\rm L} - I_2$. This is called the preemption equilibrium. All of these strategies are calculated and discussed in greater detail in the following subsections.

Entry deterrence

To derive the optimal investment decision of the leader, first the value of this port after its investment, $V_i^{\rm L}(X_{T,i}^{\rm L}, K_i^{\rm L})$, is determined. If the leader port chooses the entry deterrence strategy by investing in such a high capacity that it (temporarily) serves the entire market, this value is made up of two parts. Since leader investment can only be optimal in region R_2 , the investment value is given by:

$$V_{i,2}^{\mathrm{L}_{\mathrm{det}}}(X_{T,i}^{\mathrm{L}}, K_{i}^{\mathrm{L}}) = V_{i,2}^{\mathrm{M}}(X_{T,i}^{\mathrm{L}}, K_{i}^{\mathrm{L}}) - \left(\frac{X_{T,i}^{\mathrm{L}}}{X_{T,j}^{**,\mathrm{F}}(K_{i}^{\mathrm{L}})}\right)^{\beta_{1}} \times \left(V_{i,2}^{\mathrm{M}}(X_{T,j}^{**,\mathrm{F}}(K_{i}^{\mathrm{L}}), K_{i}^{\mathrm{L}}) - V_{i,2}^{\mathrm{F}}(X_{T,j}^{**,\mathrm{F}}(K_{i}^{\mathrm{L}}), K_{i}^{\mathrm{L}}, K_{j}^{**,\mathrm{F}}(K_{i}^{\mathrm{L}}))\right), \quad (5.15)$$

for $X_{T,i}^{L} \in R_2$. As demonstrated by Huisman & Kort (2015), the monopoly value $V^{M}(X_{T,i}^{L}, K_{i}^{L})$ is corrected for the reduced profit stream once the follower invests. This follower investment will always take place in region R_2 as well, since this is the only region wherein it is optimal for the follower to invest. Equation (5.15) indicates that the leader has an incentive to increase its investment size, as it further delays the follower's moment of entry. In that way, the leader can take advantage, because it earns the higher monopoly profit for a longer time. The monopoly value $V_{i,2}^{\rm M}(X_{T,i}^{\rm L}, K_i^{\rm L})$ in region R_2 can be calculated like in Chapter 3, with the exception that here only two regions are retained and that the boundary conditions need to be adopted accordingly, such that $G_{2,1}$ is set to zero.

The correction to the monopoly value is calculated as the discounted future reductions of the profit flows due to the investment of the follower.¹¹ The same stochastic discount factor as used by Huisman & Kort (2015),

$$\left(\frac{X_{T,i}^{\mathrm{L}}}{X_{T,j}^{**,\mathrm{F}}(K_{i}^{\mathrm{L}})}\right)^{\beta_{1}}$$
(5.16)

is used here. This factor accounts for the time until the follower port's investment threshold is expected to be reached from the current market state. However, the discounted future

¹¹Alternatively, the correction term could be written as $+\left(\frac{X_{T,i}^L}{X_{T,j}^{**,F}(K_i^L)}\right)^{\beta_1} \times \left(V_{i,2}^F(X_{T,j}^{**,F}(K_i^L), K_i^L, K_j^{**,F}(K_i^L)) - V_{i,2}^M(X_{T,j}^{**,F}(K_i^L), K_i^L)\right)$, as $V_{i,2}^F$ replaces $V_{i,2}^M$ for the leader after the follower's invest-

reductions of the profit flows are more complicated to explicate than in the case of Huisman & Kort (2015), because of output flexibility and the two resulting regions for $\overline{\Pi}_i$. In that light, the correction can be considered as a reduction of the residual value of the project after the follower's investment. As a result, it is calculated as the difference in value V for the leader between being the sole active port and the duopoly situation at the moment the follower invests. The value at the moment that both ports have invested is calculated using Equation (5.9) with port *i* being the leader and port *j* the follower. The solution takes into account that the follower invests in its own optimal size based on the capacity installed by the leader.¹²

It should be noted that each V in Equation (5.15) contains a term of the form $G_{i,(k,l)}^{S} X^{\beta_{l}}$, with S the role (L or F) of port *i*, *k* the region and $l \in \{1, 2\}$ to account for the possibility of X crossing the boundary between R_1 and R_2 . The value matching and smooth pasting conditions automatically hold for $V_i^{L_{det}}$. This is because these conditions need to be imposed for $V_{i,1}^{\rm M}(X_{T,i}^{\rm L}, K_i^{\rm L})$ and $V_{i,2}^{\rm M}(X_{T,i}^{\rm L}, K_i^{\rm L})$ (see Hagspiel et al. (2016)) and because the correction terms remain the same for each region of $V_i^{L_{det}}$. The latter is because the follower will always invest only in the profitable region R_2 , giving rise to a sequential problem.¹³

Analogously to the previous section, $V_i^{L_{det}}(X_{T,i}^L, K_i^L)$ and $I_i(K_i^L)$ together allow calculating the optimal capacity $K_i^{*,L_{det}}$ for a given timing $(X_{T,i}^L)$, whereas the relevant value matching and smooth pasting conditions for $F_i^{L_{det}}(X)$ lead to the optimal unrestricted timing $X_{T,i}^{*,L_{det}}$ as a function of installed capacity (K_i^{L}) . Combining both leads to $(X_{T,i}^{**,\text{L}_{det}}, K_i^{**,\text{L}_{det}})$, the investment decision jointly optimal in both timing and size for the leader playing the entry deterrence strategy. According to Huisman & Kort (2015) however, the leader can only play the deterrence strategy when $X \in [X_{T,i}^{\det_1}, X_{T,i}^{\det_2}]$. This is the case when demand is small enough so that one port suffices to handle the cargo. $X_{T,i}^{\det_1}$ on the one hand is calculated as the lowest X for which an optimal $K_i^{*,L_{det}}$ exists, so that an entry-deterring investment becomes possible. Since infinitesimal investment is never profitable due to the fixed investment cost, a slightly different calculation is needed than the one of Huisman & Kort (2015). Here, the following condition is solved:

$$V_{i,2}^{\mathcal{L}_{det}}(X_{T,i}^{\det_{1}}, K_{i}^{*,\mathcal{L}_{det}}(X_{T,i}^{\det_{1}})) - I_{i}(K_{i}^{*,\mathcal{L}_{det}}(X_{T,i}^{\det_{1}})) = 0.$$
(5.17)

Any investment in another capacity level at this threshold would not be profitable, nor can a profitable investment be made at a lower threshold. On the other hand, $X_{T,i}^{\text{det}_2}$ is implicitly defined as

$$X_{T,j}^{*,F}(K_i^{*,L_{det}}(X_{T,i}^{det_2})) = X_{T,i}^{det_2}$$
(5.18)

(Huisman & Kort, 2015).

In this light, it should always be verified that $X_{T,i}^{**,L_{det}} \in [X_{T,i}^{det_1}, X_{T,i}^{det_2}]$ to make sure that this

¹²This reasoning is analogous to

 $V_1^{M}(X_T, K) = \overline{V}_1^{M}(X_T, K) + (X_T/X_{T, \text{boundary}})^{\beta_1} \times (V_2^{M}(X_{T, \text{boundary}}, K) - \overline{V}_1^{M}(X_{T, \text{boundary}}, K)),$ where the correction expresses the possibility of *X* crossing *X*_{T, \text{boundary} to enter a different region *R*_k. $^{13}X_{T,\text{boundary}} < X_{T,i}^{**,F}$.

optimal entry deterrence strategy is feasible and indeed deters the follower. In the case of sufficiently large cost differences, the leader will wait to invest until the optimal deterrence threshold is reached if $X(t = 0) < X_{T,i}^{**,L_{det}}$, whereas the port will invest right away if the market is large enough from the outset, so that $X(t = 0) > X_{T,i}^{**,L_{det}}$.

Entry accommodation

It may occur that entry deterrence is not the optimal strategy for the port to follow. This is the case when the market is initially large enough to accommodate two operating ports right away from the start. Then the leader does not have the incentive to overinvest in order to delay the follower's investment, since it would be too costly to do so. This is because installing a very large port could involve diseconomies of scale, or the required space could be lacking in the port area. Consequently, it is better for the leader port to invest in less capacity than what would be optimal under entry deterrence, so that the other port can be installed simultaneously to handle the large demand together with the first port.

Under entry accommodation, the value function of the leader is different than under entry deterrence, because the follower invests right away and there is no $V^{\rm M}$ -term. It is however similar to Equation (5.9), as the two ports invest at the same time (Huisman & Kort, 2015). The difference is that the capacity of the other port, i.e., the follower, is now endogenous, since it depends on the capacity of the leader and the given threshold, which is by definition equal to the timing of the leader. The value of the leader now becomes:

$$V_i^{\mathrm{L}_{\mathrm{acc}}}(X_{T,i}^{\mathrm{L}}, K_i^{\mathrm{L}}) = V_i^{\mathrm{F}}(X_{T,i}^{\mathrm{L}}, K_i^{\mathrm{L}}, K_j^{\mathrm{F}}(X_{T,i}^{\mathrm{L}}, K_i^{\mathrm{L}})).$$
(5.19)

The entry accommodation strategy can only be played when the market is large enough such that $X \in [X_{T,i}^{\text{acc}_1}, \infty)$, with $X_{T,i}^{\text{acc}_1} \leq X_{T,i}^{\text{det}_2}$ and $X_{T,i}^{\text{acc}_1}$ implicitly defined as

$$X_{T,j}^{*,F}(K_i^{*,L_{acc}}(X_{T,i}^{acc_1})) = X_{T,i}^{acc_1}$$
(5.20)

(Huisman & Kort, 2015). In a similar way as under entry deterrence, the optimal entry accommodation investment strategy ($X_{T,i}^{**,L_{acc}}, K_i^{**,L_{acc}}$) can be calculated. Again however, this optimum is only feasible if $X_{T,i}^{**,L_{acc}} > X_{T,i}^{acc_1}$.

It is moreover possible to determine whether it is optimal to play the entry deterrence or entry accommodation strategy if $X \in [X_{T,i}^{\text{acc}_1}, X_{T,i}^{\text{det}_2}]$. If $X_{T,i}^{\text{acc}_1} < X_{T,i}^{\text{det}_2}$, there exists an $\hat{X}_{T,i} \in [X_{T,i}^{\text{acc}_1}, X_{T,i}^{\text{det}_2}]$ for which it holds that for $X < \hat{X}_{T,i}$, it is optimal for the leader to play the entry deterrence strategy, while for $X > \hat{X}_{T,i}$ the entry accommodation strategy is optimal.¹⁴ $\hat{X}_{T,i}$ satisfies

$$V_i^{\mathrm{L}_{\mathrm{det}}}(\hat{X}_{T,i}, K_i^{*,\mathrm{L}_{\mathrm{det}}}(\hat{X}_{T,i})) - I(K_i^{*,\mathrm{L}_{\mathrm{det}}}(\hat{X}_{T,i})) = V_i^{\mathrm{L}_{\mathrm{acc}}}(\hat{X}_{T,i}, K_i^{*,\mathrm{L}_{\mathrm{acc}}}(\hat{X}_{T,i})) - I(K_i^{*,\mathrm{L}_{\mathrm{acc}}}(\hat{X}_{T,i})), \quad (5.21)$$

meaning that at this market state, the value of both strategies is equal for the leader port.

¹⁴ If
$$X_{T,i}^{\operatorname{acc}_1} = X_{T,i}^{\operatorname{det}_2}$$
, $\hat{X}_{T,i} = X_{T,i}^{\operatorname{acc}_1} = X_{T,i}^{\operatorname{det}_2}$.

Preemption

In the case of a small cost asymmetry between the ports, both ports have an incentive to be the first investor (Corchón & Marini, 2018). At the optimal investment threshold $X_{T,1}^{**,L_{det}}$ of the first port with the cost advantage, the project value $V_1^L - I_1$ of being the leader for this port is higher than the discounted value of being the follower $V_1^F - I_1$. If the same holds for the second port, this follower would then prefer to become the leader, as this port would achieve higher value if it invested first. The second port could achieve this by investing at an *X* that is infinitesimally smaller than $X_{T,1}^{**,L_{det}}$:

$$X_{T,1}^{**,\mathrm{L}_{\mathrm{det}}} - \varepsilon, \qquad (5.22)$$

with ε an infinitesimal small positive number. The first port would then invest at $X_{T,1}^{**,L_{det}}-2\varepsilon$ to remain the leader. This process of epsilon preemption as described by Huisman & Kort (2015) would continue until the preemption threshold X_T^P for which it holds that

$$V_{2}^{L}(X_{T}^{P}, K_{2}^{*,L}(X_{T}^{P})) - I_{2}(K_{2}^{*,L}(X_{T}^{P})) = \left(\frac{X_{T}^{P}}{X_{T,2}^{**,F}(K_{1}^{*,L}(X_{T}^{P}))}\right)^{\beta_{1}} \times \left[V_{2}^{F}(X_{T,2}^{**,F}(K_{1}^{*,L}(X_{T}^{P})), K_{2}^{**,F}(K_{1}^{*,L}(X_{T}^{P})), K_{1}^{*,L}(X_{T}^{P})) - I_{2}(K_{2}^{**,F}(K_{1}^{*,L}(X_{T}^{P})))\right].$$
(5.23)

At this point, the port with the cost disadvantage is indifferent between leader and the follower role. The cost advantage for the other port leads to its leader value, $V_1^M - I_1$, at this point still being higher than its discounted follower, $V_1^F - I_1$. Hence, at this point no further preemption will take place, and port 1 will act as the leader and invest at X_T^P in $K_1^{*,L}(X_T^P)$ to preempt port 2, whose investment will be deterred. However, if the market is currently small enough and the cost difference between the two ports is so large that the optimal investment timing is earlier than the preemption point so that $X < X_{T,1}^{**,L_{det}} < X_T^P$, the port with the cost advantage will invest at its deterrence optimum and the preemption point will be insignificant.

5.2.4 A synthesis of the leader's and follower's strategies

As the previous subsections illustrated, the leader port can choose from three different investment strategies. The choice it will make, depends on the initial state of the market as expressed by X. In this subsection, these possible strategies are discussed using a number line for X in Figure 5.4.

If *X* is initially very small, i.e., below $X_{T,i}^{\det_1}$, no profitable investment is possible for any port at the beginning. As soon as $X \ge X_{T,i}^{\det_1}$, the leader (i.c., port *i*) can profitably invest in an entry-deterring quantity. However, it will not decide to do so, since waiting is an even more valuable strategy. As soon as $X = X_T^P$ and assuming that $X_T^P < X_{T,i}^{**,L_{det}}$ due to relatively

small cost differences between the ports, the port with the cost advantage will decide to invest in the optimal entry-deterring quantity $K_1^{*,L_{det}}(X_T^P)$. If it does not do so, the other port will invest first to preempt the port with the cost advantage. The latter will then become the follower. This would imply a lower value for the cost-advantaged port than if it were the leader. However, if the preemption point were higher than the optimal investment threshold of the leader due to a larger cost advantage, the leader could wait longer until X reaches $X_{T,i}^{**,L_{det}} \in [X_{T,i}^{det_1}, X_{T,i}^{det_2}]$. At that point, the port with the cost advantage could invest at its unrestricted leader threshold, involving entry deterrence with $K_i^{**,L_{det}}$ being the amount of capacity.

If X initially exceeds $\min\{X_T^p, X_{T,i}^{**, L_{det}}\}$, the leader will invest right away. The investment strategy and related capacity are dependent on the actual value of X. As long as $X < X_{T,i}^{acc_1} \leq X_{T,i}^{det_2}$, the optimal instantaneous investment strategy for the leader is the deterrence strategy, which is at the same time the only feasible strategy. If $X > X_{T,i}^{det_2} \geq X_{T,i}^{acc_1}$, it is optimal for the leader to invest right away in an entry-accommodating strategy, which is in this situation, again, the only feasible strategy. If $X \in [X_{T,i}^{acc_1}, X_{T,i}^{det_2}]$, both the deterrence and accommodation strategies are feasible. If $X < \hat{X}_{T,i}$, the deterrence strategy to play. As a result, a higher $\hat{X}_{T,i}$ decreases the probability that the leader will invest in an entryaccommodating capacity level.

The decision of the follower (i.c., port *j*) is easier to determine. If the leader invests in an entry-deterring capacity, the follower will wait with investment until *X* reaches its optimal threshold, which is $X_{T,j}^{**,F}$. At that point, the follower will invest in the corresponding optimal capacity $K_j^{**,F}$. Both depend on the capacity of the leader, as described in Section 5.2.2. If the leader invests in an entry-accommodating quantity, the follower will invest at the same time in its corresponding optimal capacity $K_i^{*,F}(X_{T,i}^{\text{Lacc}})$ as calculated before.

5.3 The impact of competition and port ownership on the investment decisions

In this section, the previously described methodology is applied to a two-port setting, specified by the parameters derived in Chapter 3 and summarised in Table 5.1. The specific $X_{T,i}$ and K_i from Figure 5.4 are calculated in order to describe the full domain of possible decisions for both the leader and the follower port. In order to examine the impact of competition intensity, the product differentiation parameter δ is varied from 0.6 to 0.9 for intensified competition and to 0.3 for less competition with more diversified port services. For each δ , the analysis is carried out under three different ownership structures: a private port (such as the English ports, e.g., the port of Felixstowe), a public port taking full consumer surplus into account (such as Spanish ports, e.g., the port of Valencia) and a port that

Table 5.	3: Inve	sstment strategies of t X_T in Figure 5.4.	Figure 5.4 the leader (port 1:	$X_{T,i}^{\det_1}$: Number line wit $X_{T,1}^{*, \operatorname{L}_{\det(\operatorname{acc}}(X_T^{\operatorname{L}}))$	$X_T^P X_T^P X_T$ h the leader's crit	$\frac{ }{T,i}$ <i>T,i i</i> :al investment the ical investment the port 2: $(X_{T,2}^{**,F}(K_1^{*,1}$	$X_{T,i}^{\operatorname{acc}}$ resholds.	$\hat{X}_{T,i}$ $\hat{X}_{T,i}$ $X_{T,i}^{\det_2}$ $(K_1^{*,\operatorname{Ldet},\operatorname{acc}}(X_{T,1}^{\operatorname{L}}))))$	for each character-
sG	s _{CS}	Port role	$X_{T,i}^{\det_1}$	X_T^{P}	$X_{T,i}^{**,\mathrm{L}_{\mathrm{det}}}$	$X_{T,i}^{\mathrm{acc}_1}$	$\hat{X}_{T,i}$ (acc)	$\hat{X}_{T,i}$ (det)	$X_{T,i}^{{ m det}_2}$
0	0	Leader (1)	19.84, 8.39	32.75, 10.35	(36.24, 10.79)	48.72, 12.02	48.84, 12.04	48.84, 12.19	48.91, 12.20
		\longrightarrow Follower (2)	(44.67, 11.76)	(46.89, 11.91)	(47.38, 11.94)	(48.72, 12.02)	48.84, 12.04	(48.90, 12.03)	(48.91, 12.03)
1/2	1/2	Leader (1)	19.09, 8.45	30.97, 10.34	(34.68, 10.83)	47.89, 12.15	48.06, 12.17	48.06, 12.36	48.16, 12.38
		\longrightarrow Follower (2)	(43.35, 11.84)	(45.67, 12.00)	(46.28, 12.05)	(47.89, 12.15)	48.06, 12.17	(48.15, 12.17)	(48.16, 12.17)
I	Ι	Leader (1)	17.01, 8.73	25.93, 10.39	(30.49, 11.10)	47.75, 12.90	48.48, 12.99	48.48, 13.46	48.96, 13.52
		\longrightarrow Follower (2)	(40.25, 12.29)	(43.06, 12.53)	(44.34, 12.64)	(47.75, 12.90)	48.48, 12.99	(48.85, 12.99)	(48.96, 13.00)
		Parameter values: $A = 5$	$\overline{\delta}, B = 1, \delta = 0.6, c = 1$	l, $c_h = 0.5, \mu = 0.015$,	$\sigma = 0.1, r = 0.06, F_0$	$C_{I,1} = 80, FC_{I,2} = 100$	$0, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 10, \gamma_4 = 10, \gamma_5 = 10, $	$\gamma_3 = 0, \gamma_4 = 0.12, \lambda =$	= 0.4.

.De	s _{CS}		$X_{T,i}^{\det_1}$	X_T^{P}	$X_{T,i}^{**,\mathrm{L}_{\mathrm{det}}}$	$X_{T,i}^{\mathrm{acc}_1}$	() 1'I	- 1 -	$X_{T,i}^{{ m det}_2}$
0	0	Leader (1)	19.50, 8.36	35.12, 10.71	(36.57, 10.89)	43.19, 11.61	43.20, 11.61	43.20, 11.65	43.20, 11.65
		\longrightarrow Follower (2)	(41.52, 11.51)	(42.75, 11.59)	(42.84, 11.59)	(43.19, 11.61)	43.20, 11.61	(43.20, 11.61)	(43.20, 11.61)
1/2	1/2	Leader (1)	18.75, 8.42	33.31, 10.69	(34.93, 10.90)	41.88, 11.68	41.89, 11.69	41.89, 11.73	41.90, 11.74
		\longrightarrow Follower (2)	(40.04, 11.56)	(41.34, 11.65)	(41.46, 11.66)	(41.88, 11.68)	41.89, 11.69	(41.90, 11.69)	(41.90, 11.69)
Ι	Ι	Leader (1)	16.67, 8.68	28.12, 10.74	(30.45, 11.08)	39.04, 12.13	39.11, 12.13	39.11, 12.25	39.15, 12.26
		\longrightarrow Follower (2)	(36.27, 11.90)	(37.91, 12.04)	(38.20, 12.06)	(39.04, 12.13)	39.11, 12.13	(39.15, 12.13)	(39.15, 12.13)
s_G	SCS	Port role	$X_{T,i}^{\det_1}$	$X_T^{ m P}$	$X_{T,i}^{**,\mathrm{L}_{\mathrm{det}}}$	$X_{T,i}^{\mathrm{acc}_1}$	$\hat{X}_{T,i}$ (acc)	$\hat{X}_{T,i}$ (det)	$X_{T,i}^{\mathrm{det}_2}$
0	0	Leader (1)	20.10, 8.43	31.15, 10.14	(36.28, 10.80)	55.27, 12.52	55.76, 12.57	55.76, 12.93	56.08, 12.96
		\longrightarrow Follower (2)	(47.76, 12.02)	(50.87, 12.23)	(52.09, 12.32)	(55.27, 12.52)	55.76, 12.57	(56.02, 12.56)	(56.08, 12.57)
1/2	1/2	Leader (1)	19.35, 8.50	29.43, 10.13	(34.79, 10.86)	55.20, 12.73	55.94, 12.80	55.94, 13.26	56.43, 13.31
		\longrightarrow Follower (2)	(46.60, 12.12)	(49.81, 12.36)	(51.29, 12.46)	(55.20, 12.73)	55.94, 12.80	(56.33, 12.80)	(56.43, 12.80)
Ι	Ι	Leader (1)	17.24, 8.79	24.64, 10.22	(30.90, 11.23)	60.33, 14.02	64.82, 14.46	64.82, 15.64	68.30, 16.03
		\longrightarrow Follower (2)	(44 06, 12, 69)	(47.88, 13.02)	(50.89, 13.27)	(60.33, 14.02)	64.82, 14.46	(66.66, 14.49)	(68.30, 14.61)

is 50% owned publicly by a government taking 50% of port *i*'s generated consumer surplus into account (such as the port of Tauranga in New Zealand).¹⁵ This allows discerning the impact of different government ownership structures on the investment decision. The resulting investment decisions for port 1 (the leader) and port 2 (the follower) are shown in Tables 5.3 - 5.5.

 $X_{T,i}^{\text{det}_1}$ is the first critical threshold that requires analysis. If the share of the public owner is larger, investment will be considered beneficial sooner already, when the market might still be smaller. This threshold decreases with the involvement of public money in the port. In this case, the project becomes more attractive, since more welfare effects are taken into account. Because of this, the threshold for the inverse demand function intercept from which investment for the leader becomes profitable, decreases. Moreover, the leader's installed capacity will be larger. The follower's decision is impacted in the same way. The increase in project attractiveness leads to an earlier and larger investment. In the case of a lower δ , product diversification is higher, for which competition between the two ports is lower. As a result, the ports experience a lower impact from each other's operations on the own price. Hence, the price will be higher, ceteris paribus, and the project will be more attractive, so that the leader will be willing to invest earlier, whereas the size will be slightly smaller. Nevertheless, the follower will invest earlier too, since its project will also be more attractive because the competition from the leader is lower. This leads to the leader having a monopoly position for a shorter time, but this is largely made up for by the higher price because of less competition, even after the monopoly period has ended.

The second threshold to be discussed is the preemption point $X_T^{\rm P}$. In the numerical simulations, with a sufficiently low cost advantage for port 1, the preemption point is always below the unrestricted investment threshold for the leader. This implies that when X is initially below the preemption point, the port with the cost advantage will wait until X reaches X_T^P to invest in an entry-deterring capacity, in order to preempt the follower. If government involvement is higher in both ports, the preemption threshold will be lower too, since more accounting for social welfare increases the projects' attractiveness. The willingness to invest of each port increases. As a result, epsilon preemption will continue further to lead to a lower preemption threshold. In order to preempt port 2, port 1 needs to invest earlier. The impact of intensified competition (through a lower product diversification) on the timing of the preemption point is as expected, namely a lower X_T^P -threshold. Since port 2's incentive to become the leader is larger, following a relative increase in the leader value as compared with the follower value, port 1 needs to advance its investment further to preempt port 2. The impact on the size of the investment is however remarkable. The earlier leader investment in order to preempt the follower implies investing at the moment that the market has a smaller size, which results in the leader installing less capacity. The optimal leader capacity investment size as a function of the timing increases however with

¹⁵The port of Tauranga is 55% state owned and the rest is listed on the New Zealand Stock Exchange (Cairns, 2013).

the amount of competition. As a result of increased competition, the leader would invest in more capacity than what would be optimal under a monopoly setting in order to delay follower entry for a longer time, leading to a prolonged monopoly position (Huberts et al., 2019). This explains why follower investment is later and hence larger when there is more competition, notwithstanding the leader's earlier investment timing under the preemption strategy. Hence, if for example the Italian government wanted to install a new port closer to Marseille, it would be advised to wait until the market is large enough for two ports to handle the cargo. The findings are summarised in the following results, which are derived numerically, since closed-form analytical results could not be obtained.

Result 5.1. Under the assumptions given in Table 5.2, intensified competition and resulting preemption not only lead to earlier investment of the leader, but also to less capacity installed. The follower however invests later and in more capacity.

If there are large cost differences, the leader is able to invest at the unrestricted threshold $X_{T,i}^{**,L_{det}}$. The impact of competition on this optimal timing threshold is ambiguous and limited. The impact of an increase in public money involvement however is less ambiguous and the result from the previous chapter can be confirmed:

Result 5.2. Under the assumptions given in Table 5.2, an increase in public money involvement leads to an earlier unrestricted investment in more capacity by both the leader and the follower.

Result 5.2 confirms the finding of the previous chapter and is in line with Asteris et al. (2012), namely that public companies invest sooner and in more capacity than private companies.

If the initial X is high enough (above $\hat{X}_{T,i}$), the market is so profitable that the costdisadvantaged port will also invest immediately. This implies that entry accommodation becomes the most profitable strategy for the leader. The leader will invest in less capacity than under entry deterrence, because then there is no incentive to prolong the monopoly period. The thresholds indicating the beginning of the accommodation region and the end of the deterrence region increase with a higher δ . As previously noted, if the competitive impact of the other port increases, the leader has a larger incentive to deter the follower's investment. The impact of public involvement on the two boundaries is ambiguous. Additionally, (increased) public money involvement and product diversification both widen the interval wherein both strategies (entry deterrence and accommodation) are possible. This is caused by a combination of effects. Due to public money involvement, investment becomes more beneficial for both ports. Since it is more profitable for the leader, deterrence is feasible over a longer time span, but because investment is also more profitable for the follower, at the same time it is more difficult to deter entry. The follower's decision does not depend on the initial value of *X* and the different leader thresholds. It only depends on the capacity choice of the leader. Given the size of this capacity, the follower can decide to invest right away, or delay the investment until the market is large enough for two ports to be operating. In the latter case, the follower will calculate its optimal size and timing of the investment. If the leader follows an entry-accommodating strategy, the follower will de facto find it optimal to invest at the same timing, and it calculates its optimal size accordingly. If the leader has invested in more capacity, the follower will invest later, independent of the timing of the leader. The reason is that the leader has been able to capture a larger share of the market. Hence the follower needs to await more market growth in order to be able to invest profitably. Given the positive relationship between timing and size, the follower will not only invest later, but also invest in more capacity.

If relatively more public money is involved, the follower will ceteris paribus invest earlier in more capacity. This observation is analogous to the impact on the leader's decision. If the port services are more homogeneous due to being situated closer to one another, in turn leading to increased competition, the negative impact of the leader's activity on the follower's price is higher. As a result, the follower needs to wait longer in order to be able to invest profitably and the installed capacity will be larger. The reason is that under more intense competition, investment by the leader has more effect on the follower's profitability. By consequence, it requires less effort of the leader port to reduce profitability of the follower, thus making entry deterrence a relatively easier strategy to follow. This effect can however be offset by public government involvement, since social welfare being considered more, can facilitate the construction of a new port closer to an existing port.

5.4 The impact of other parameters on the investment decision: sensitivity analysis

In order to analyse the impact of the size of congestion costs (*A*), uncertainty (σ), growth (μ) and the investment cost difference between the two ports (as expressed by $FC_{I,i}$) on the investment decision, the respective parameter is each time altered and compared with the base case (where $\delta = 0.6$ and $s_G = s_{CS} = 0.5$). In this light, the outcomes in Table 5.6 allow verifying the sensitivity of the results with respect to the altered parameters and the robustness of some of the previous chapters' findings with respect to adding a second port to the model.

If the port customers are more averse to waiting, the willingness to pay is lower for the same utilisation rate. As a reaction, both the leader and the follower port will invest in more capacity, but at a later moment. This result confirms the finding of Chapter 3 in a monopoly port.

Parameter								
change	Port role	$X_{T,i}^{\det_1}$	X_T^{P}	$X_{T,i}^{**,\mathrm{L}\mathrm{det}}$	$X_{T,i}^{\mathrm{acc}_1}$	$\hat{X}_{T,i}$ (acc)	$\hat{X}_{T,i}$ (det)	$X_{T,i}^{\mathrm{det}_2}$
Base case	Leader (1)	19.09, 8.45	30.97, 10.34	(34.68, 10.83)	47.89, 12.15	48.06, 12.17	48.06, 12.36	48.16, 12.38
	\longrightarrow Follower (2)	(43.35, 11.84)	(45.67, 12.00)	(46.28, 12.05)	(47.89, 12.15)	48.06, 12.17	(48.15, 12.17)	(48.16, 12.17)
A=4	Leader (1)	17.23, 8.25	27.56, 10.07	(31.29, 10.61)	44.57, 12.05	44.75, 12.07	44.75, 12.30	44.86, 12.31
	\longrightarrow Follower (2)	(40.04, 11.72)	(42.25, 11.89)	(42.90, 11.94)	(44.57, 12.05)	444.75, 12.07	(44.85, 12.07)	(44.86, 12.07)
$\sigma = 0.15$	Leader (1)	18.63, 8.51	41.32, 11.86	(49.36, 12.78)	N/A	N/A	N/A	N/A
	\longrightarrow Follower (2)	(69.38, 14.79)	(80.77, 15.67)	(84.30, 15.93)	N/A	N/A	N/A	N/A
$\sigma = 0.14$	Leader (1)	18.75, 8.50	38.62, 11.48	(44.84, 12.21)	78.10, 15.18	78.70, 15.23	78.70, 15.51	79.12, 15.55
	\longrightarrow Follower (2)	(60.79, 13.90)	(68.11, 14.47)	(70.02, 14.61)	(78.10, 15.18)	78.70, 15.23	(79.02, 15.25)	(79.12, 15.25)
$\mu = 0.02$	Leader (1)	16.82, 8.47	33.08, 11.23	(38.09, 11.90)	62.51, 14.42	62.89, 14.45	62.89, 14.70	63.15, 14.73
	\longrightarrow Follower (2)	(50.83, 13.45)	(56.10, 13.91)	(57.44, 14.02)	(62.51, 14.42)	62.89, 14.45	(63.10, 14.46)	(63.15, 14.46)
$\mu = -0.01$	Leader (1)	29.51, 8.44	33.29, 8.89	(36.07, 9.20)	43.35, 9.60	43.54, 9.61	43.54, 9.93	43.69, 9.95
	\longrightarrow Follower (2)	(44.24, 9.56)	(42.67, 9.57)	(42.97, 9.58)	(43.35, 9.60)	43.54, 9.61	(43.68, 9.61)	(43.69, 9.61)
$FC_{I,2} = 196.3$	Leader (1)	19.02, 8.44	(34.74, 10.84)	(34.74, 10.84)	51.33, 12.51	51.51, 12.53	51.51, 12.74	51.63, 12.75
	\longrightarrow Follower (2)	(46.12, 12.17)	(49.20, 12.38)	(49.20, 12.38)	(51.33, 12.51)	51.51, 12.53	(51.61, 12.53)	(51.63, 12.53)
Single	Leader (1)	19.02, 8.44	31.46, 10.41	(34.70, 10.83)	46.29, 12.03	46.40, 12.04	46.40, 12.19	46.46, 12.19
government	\longrightarrow Follower (2)	(42.53, 11.77)	(44.61, 11.92)	(45.06, 11.95)	(46.29, 12.03)	46.40, 12.04	(46.46, 12.04)	(46.46, 12.04)
Base case paran	neter values: $A = 5, B =$	$1, \delta = 0.6, c = 1, c_h =$	$= 0.5, \mu = 0.015, \sigma =$	$0.1, r = 0.06, FC_{I,1} =$	80, $FC_{I,2} = 100, \gamma_1 =$	$180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 $	$\gamma_4 = 0.12, s_G = 0.5, s_G$	$s_{CS} = 0.5, \lambda = 0.4.$

CHAPTER 5. CAPACITY INVESTMENT DECISIONS OF COMPETING PORTS

Table 5.6: Investment strategies of the leader (port 1: $X_{T,1}^{\text{L}}$, $K_{T,1}^{\text{l}}$, $K_{T,1}^{\text{l}}$, $K_{T,1}^{\text{l}}$, $K_{T,1}^{\text{l}}$, $K_{T,2}^{\text{l}}$, $(K_{T,1}^{\text{st,F}}(K_{T,1}^{\text{l}}))$, $K_{2}^{\text{st,F}}(K_{T,1}^{\text{l}})$)), $K_{2}^{\text{st,F}}(K_{T,1}^{\text{l}})$)), for each character-

If the economic environment is more uncertain, both ports benefit from waiting longer to invest, in order to gain more information. Increased uncertainty not only leads to waiting longer before investment due to a higher option value of waiting, but also to more capacity installed. This confirms the frequent real options observation, also found in the previous chapters. However, this effect is lower under competition, since competition has a negative impact on the option value of waiting. Next to the common real options result, uncertainty also has another effect. It might even be that due to high uncertainty, entry accommodation is no longer a feasible strategy. If uncertainty increases, the value of waiting for the follower rises. As a result, for sufficiently high values of uncertainty, the follower always waits whenever the leader invests. This implies the absence of an investment accommodation region. Since higher uncertainty implies that the follower invests later, the leader is a monopolist for a longer time in the deterrence region. This makes deterrence more attractive for the leader, implying that $X_{T,i}^{det_1}$ can be lower.

If average growth is higher, both ports need to install substantially more capacity, in order to be able to accommodate the future demand for throughput without too much congestion. However, a larger investment requires a larger market, implying later investment. This result was also found in Chapter 3. Additionally, the $X_{T,i}^{\det_1}$ threshold will be lower as well, since the project is more attractive, and the port will be willing to install the project earlier. However, also the impact of negative economic growth needs some attention in this setting. If growth is negative, the port will need to wait for a higher threshold to be reached, in order to be able to profitably install the project. Due to the negative growth rate, reaching this threshold is much less probable. Moreover, the project will be much smaller, since future demand is expected to decrease. This finding is also in line with the results found in Chapter 3. Since the leader's project will be much smaller, the follower speeds up investing. Hence, the follower's investment size will be smaller too, due to the declining market and the earlier investment.

The impact of a cost advantage increase for the leader, such that $FC_{I,2}$ increases, is that the leader can wait longer to invest in order to still deter or preempt the follower. The reason is that the follower needs to wait longer in order to be able to profitably invest in capacity, due to its higher investment cost. This leads to both ports not only investing later, but also in more capacity. As a consequence, when the cost advantage becomes large enough, preemption no longer takes place. In such a case, the role of each port is ex-ante determined. The first port (leader) can invest at its unrestricted leader threshold, while at this point the follower value of the second port is at least as high as its leader value. Additionally, it is observed that $X_{T,i}^{det_1}$ slightly decreases. The leader knows that it will benefit for a longer time of its monopoly position due to the *FC*-cost increase for the other port. As a result, investing in capacity becomes profitable earlier. This result is summarised as follows:

Result 5.3. Under the assumptions given in Table 5.2, a larger fixed investment cost difference decreases the competing port's impact on the cost-advantaged port's optimal investment decision.

Throughout this chapter, the analyses were made under the assumption that the public owner involved in port *i* differs from the public owner involved in port *j*. However, it is worthwhile to check the robustness of the analysis with respect to the same government owning a share of both ports *i* and *j*. This could be the case if those two ports are located in the same country or province. In such a case, the government would consider the consumer surplus of both ports, CS_{i+j} , which under product differentiation equals

$$CS_{i+j}(q_i, q_j) = B/2 \cdot (q_i^2 + 2\delta q_i q_j + q_j^2),$$
(5.24)

due to the calculation through a line integral (Singh & Vives, 1984). As a result, CS_i in Equation (5.3) would no longer be valid, as it needs to be replaced by

$$CS_i = B/2 \cdot (q_i^2 + \delta q_i q_j). \tag{5.25}$$

The consumer surplus of the other port would equal

$$CS_j = B/2 \cdot (q_j^2 + \delta q_i q_j), \tag{5.26}$$

so that the sum of both equals $CS_{i+j}(q_i, q_j)$. Implementing this in the model yields the investment strategies in the last two lines of Table 5.6, indicated by 'single government'. The effect on the calculated investment decisions is limited and is qualitatively similar to the effect of a decreased impact of the other port's throughput quantity q_j on the own port's price p_i , mathematically expressed as a decrease in δ .

5.5 Conclusion of this chapter

Many examples of competing ports, operating in an uncertain environment, exist. Depending on the geographical situation and services offered, the amount of competition may however differ. This chapter has explored the impact of inter-port competition and other typical port characteristics under uncertainty on port capacity investment decisions. Simultaneous investment when the leader invests in an entry-accommodating capacity was allowed for, next to the possibility for the leader to deter entry in a leader-follower timing game.

When competition is more intense, the option value of waiting is reduced, because each port has a larger incentive to invest before the other port. As a result of this earlier investment, the leader's investment will be smaller as well. The follower will however invest later and in more capacity. The same effect on the timing is observed when government involvement increases. The consideration of social welfare in the operational objective function of publicly owned ports leads to a more attractive project. As a result, the ports will invest earlier, but in more capacity. If growth expectations are higher in this competitive market, it is beneficial to wait longer and benefit from the larger market and higher price. As a result, more capacity will be installed at this later moment too. The same holds for higher uncertainty, since the option value of waiting and gaining more information increases with uncertainty. Subsequently, it can be confirmed that if the customers are less waiting-time averse, less capacity is needed, and because the project is more attractive due to higher generalised costs, the port will want to invest earlier. Finally, if the cost advantage of the leader is small enough, both ports compete for the leader role, as they are seeking to preempt each other. If the cost difference is large enough however, the cost-advantaged port is guaranteed the leader role, so that it can invest at its unrestricted leader threshold. Additionally, it will be easier for the leader to secure the monopoly position for a longer time. Hence, policy makers that want to benefit as much as possible from the advantages of inter-port competition, are advised to reduce investment cost differences between different ports. This could for instance be done through taxes or subsidies.

Although not considered in this and the previous chapters, it takes time to build the infrastructure, during which the market evolves in an uncertain way. As a result, the exposure to uncertainty is larger in reality than accounted for in the presented model. Additionally, a difference in time to build between two ports, e.g., due to a different political and legal environment, may also impact competition and the determination of the leader and follower port. Moreover, this and previous chapters' models neglect the option to expand. When the port is already active, it might want to mitigate sooner the already present congestion, in order to increase the port operations' profitability in the future. The impact of time to build on the port expansion investment decision of one single port is studied in the next chapter.

Chapter 6

The impact of time to build and uncertainty on port expansion projects

The majority of the theoretical literature on port capacity investments considers greenfield projects, i.e., ports that are built from scratch (Xiao et al., 2012).¹ The previous chapters all dealt with greenfield projects too. One of the counter-intuitive findings was that higher congestion costs and waiting-time aversion lead to delaying the investment in a new port, which would be larger though. In reality however, the majority of port capacity investment projects encompass expansion projects. Chen & Liu (2016) are one of the few authors studying port expansion investments in a theoretical way. However, they do not account for the current design capacity level of the port. In the Hamburg - Le Havre range, many examples of expansion projects in ports of different sizes are present, e.g., Maasvlakte 2 in Rotterdam and the planned Saeftinghedok in Antwerp. The question can be asked whether the same conclusions hold for a port expansion project as for the investment in a new port. In the latter case, if port customers are more waiting-time averse, it is better to install a larger new port later. One could however expect that the management of the port would have an incentive to anticipate the investment in case of port expansion projects, in order to reduce the congestion already present among its customers when the instrument of congestion pricing cannot be exploited any further in a profitable way.

Additionally, it takes a lot of time to build large infrastructure projects like port capacity investment projects (Meersman & Van de Voorde, 2014a; Vanelslander, 2014; Aguerrevere, 2003). During the project lead time, the performance of the market and hence the project's profitability could change due to the uncertainty of demand. This also has an impact on a

¹Opposed to the theoretical literature using analytical models to study port investments, more (empirical) cost-benefit analyses for port expansions are available.

port's investment decision and needs to be considered. The available RO literature on time to build is however rather limited. Although Dixit & Pindyck (1994) and Kydland & Prescott (1982) already indicated the important implications of time to build on investment decisions, Marmer & Slade (2018) more recently argued again that investment lags should be given more attention in theoretical and empirical work. In his study for non-storable commodities, Aguerrevere (2003) found an ambiguous relationship between the time to build and the size of the investment project. He moreover found that introducing time to build leads to the observation that firms anticipate their capacity expansion to moments before current capacity is fully occupied. Also Bar-Ilan & Strange (1996) found that longer construction times lead to earlier investments in a production environment. Moreover, time to build influences the effect of uncertainty on the timing of the investment. Whereas traditional RO analysis demonstrates that increased uncertainty leads to delaying the investment, increased uncertainty in combination with time to build may lead in some cases to anticipating the investment timing. This illustrates the importance of considering time to build in the model. Majd & Pindyck (1987) and Milne & Whalley (2000) quantified the option value of altering the construction lead time through the use of a different technology.

The objective of this chapter is to analyse the impact of time to build and the capacity already in place on the port capacity expansion decision under uncertainty. The following research sub-questions have been formulated: (RQ 6.1) What is the impact of time to build on the optimal capacity investment decision under uncertainty without competition? and (RQ 6.2) What is the impact of existing capacity on the optimal capacity investment decision under uncertainty without competition? The impact of the two model extensions that are introduced in this chapter, compared to Chapter 3, are analysed separately in Section 6.5. In this chapter, the port considered has the option to invest in one expansion project, which is fully deployed in one phase and which can only be used after completion. Hence, phased investment is not considered here. Since growth, uncertainty, public money involvement and congestion considerably influence the investment decision, the impact of changes in these economic characteristics on the investment decision needs to be verified in this different port setting. These additions to the literature will allow practitioners to make well-considered port investment decisions, taking into account as much relevant information as possible. However, port competition is left beyond the scope of this chapter, in order to focus on the impact of time to build on the (unrestricted) optimal expansion decision of one port. This approach moreover allows for better mathematical tractability of the model. In order to extend this chapter's model to a duopoly case, the approach of Chapter 5 can be followed.

The structure of this chapter is as follows. Section 6.1 reformulates the basic model from Chapter 3 to determine the value of an investment in a new port without time to build. This offers a basis for the model of this chapter. Section 6.2 introduces time to build and the evaluation of expansion projects in the model. Section 6.3 describes how the optimal investment decision is determined according to the real options approach in a port expansion context with time to build. The additional parameters used for the numerical simulations are given in Section 6.4. This is followed by a discussion of the results in Section 6.5 and the sensitivity of the results to previously studied economic parameters in Section 6.6. The final section contains the conclusion of this chapter.

6.1 Recapitulation of the basic model for determining the project value

The extended model in this chapter is based on the basic economic model from Chapter 3, as outlined in Section 3.1. This allows comparing the findings of this chapter with the previous chapters' findings. The profit generated in a newly built private port, operated by one single actor, was calculated by Equation (3.7). As discussed in Chapter 4, the port authority (PA) is not always privately owned. Also a public government could own shares of the PA. This was included in the PA's operational objective function, $\Pi(X, K, q)$, by considering spillover benefits for the local economy and consumer surplus partly or fully in Equation (4.6).

As indicated, throughput should always be greater than or equal to zero. As a result, q^{opt} is set to 0 if the optimal throughput level is below zero (Dangl, 1999). As opposed to the models in Chapter 3 and 4 but in line with the model of Chapter 5, a port's design capacity K is not considered as a hard constraint for the throughput level. It is however a soft constraint if the waiting cost at high occupancy rates is sufficiently high. This approach allows for the exceptional case wherein throughput exceeds capacity at a high cost.² As a result, q^{opt} is defined here in two mathematical regions R_j for $X: R_1 = [0, c - s_G \lambda)$ and $R_2 = [c - s_G \lambda, \infty)$. Applying the first-order condition to Π again leads to the optimal throughput level q^{opt} at each point in time (see also Chapter 3):

$$q^{opt}(X,K) = \begin{cases} 0, & X < c - s_G \lambda, \\ \frac{[X - (c - s_G \lambda)]K^2}{2(AX + BK^2) - s_G s_{CS} BK^2}, & X \ge c - s_G \lambda. \end{cases}$$
(6.1)

Subsequently, the optimal $\overline{\Pi}(X, K) = \Pi(X, K, q^{opt}(X, K))$ is again calculated in the same two regions as well, which then allows calculating V(X, K), the value of a greenfield project without time to build (see Section 3.3).

²It can be noted that the impact of omitting region R_3 as a result of relaxing the capacity constraint to a soft constraint only has a limited impact on the outcomes. In Table 6.3, the optimal investment strategy with K_0 , θ , s_G and s_{CS} equalling zero, is (37.71 euro per TEU, 11.19 M TEU p.a.). In Chapter 3 with region R_3 included as a result of K being a hard constraint for q, the corresponding optimum was (37.63 euro per TEU, 11.17 M TEU p.a.), encompassing a difference of only 0.2%. Including region R_3 only unnecessarily complicates the calculations of the model for expansion projects due to additional mathematical regions R_j with boundaries involving K.

6.2 Economic setting and methodology: model extensions

In order to account for (i) the amount of capacity already present in expansion projects, and (ii) the lead time between the investment decision and the project completion, two model extensions are developed in this chapter. Both extensions need to be implemented in the expression for the value of the investment project. This new expression $\Delta V'$ is to be used instead of V in the dynamic programming methodology as described in Section 3.4.

6.2.1 Accounting for existing capacity in the value of expansion projects

The value of an expansion project of size ΔK with initial (existing) capacity K_0 can be calculated as the difference in value V after and before installing the expansion project at the moment where the demand shift parameter equals X:

$$\Delta V_{j}(X, \Delta K, K_{0}) = V_{j}(X, K_{0} + \Delta K) - V_{j}(X, K_{0}),$$
(6.2)

with *j* indicating the mathematical region for *X* in which *V* is calculated and with $V_j(X, 0) = 0$, so that $\Delta V_j(X, \Delta K, 0) = V_j(X, \Delta K)$.³

6.2.2 Accounting for the impact of time to build on the project value

As noted before, it takes time to build port capacity projects, once the investment decision is made. During this lead time, the market changes in an uncertain way, as is expressed by the GBM for demand shift parameter *X*. As a result, the project value will be different from what its value would have been if the project lead time θ had equalled zero.

To account for the uncertainty impacting demand during the time to build a project, the expected evolution during this period of the uncertain parameter *X* following a GBM, defined by growth and uncertainty parameters μ and σ , needs to be included in the model. Assuming that *X* was equal to X_0 at time t = 0, the probability density function for *X* at time $t = \theta$ is given by:

$$\phi(X, X_0, \theta) = \frac{1}{\sqrt{2\pi}} \frac{1}{X\sigma\sqrt{\theta}} \exp\left(-\frac{\left(\ln X - \ln X_0 - \left(\mu - \frac{1}{2}\sigma^2\right)\theta\right)^2}{2\sigma^2\theta}\right).$$
(6.3)

Using this information, the discounted expected value $\Delta V'(X, \Delta K, K_0, \theta)$ of an expansion project with size ΔK and existing capacity K_0 , installed at the moment that the demand shift parameter equals *X* and taking θ years to complete can be calculated as follows (Aguer-

³In that case, ΔK equals *K* from Chapter 3.

revere, 2003):^{4,5}

$$\Delta V'(X, \Delta K, K_0, \theta) = e^{-r\theta} \left[\int_{0}^{c-s_G \lambda} \phi(X', X, \theta) \cdot \Delta V_1(X', \Delta K, K_0) dX' + \int_{c-s_G \lambda}^{\infty} \phi(X', X, \theta) \cdot \Delta V_2(X', \Delta K, K_0) dX' \right], \quad (6.4)$$

149

with *r* the discount rate. The integrals together calculate the weighted average project value *V* for each possible value of *X* over the different regions for *X*. The probability density function of *X* is used to determine the weights. The first factor, discount factor $e^{-r\theta}$, is required to account for the construction lead time θ during which no cash flows are generated. This time lag is situated between the moment of the investment decision and the start of port operations at the site after project completion.

6.3 Determining the optimal project investment size and timing

An analogous analysis to the one of Chapter 3 can be carried out to determine the optimal size and timing of the expansion project. To this end, Equation (3.28) here becomes

$$F(X) = \max\left\{ e^{-rdt} \mathbf{E}(F(X) + dF(X)), \max_{K} [\Delta V'(X, \Delta K, K_0, \theta) - I(\Delta K)] \right\}.$$
 (6.5)

Using Equation (6.5) and the steps described in Section 3.4, the optimal size of the expansion project $\Delta K^*(X, K_0, \theta)$ and the optimal timing, expressed as threshold $X_T^*(\Delta K, K_0, \theta)$, can be determined for an expansion project. Both are functions of existing capacity K_0 and time to build θ .

The optimal investment strategy involving both timing and size $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ is finally determined through solving the system

$$\begin{cases} X_T^{**}(K_0, \theta) = X_T^*(\Delta K^{**}(K_0, \theta), K_0, \theta), \\ \Delta K^{**}(K_0, \theta) = \Delta K^*(X_T^{**}(K_0, \theta), K_0, \theta). \end{cases}$$
(6.6)

This optimal investment strategy also depends on K_0 and θ .

The equations making up the model are summarised in Table 6.1, whereas the assumptions are in Table 6.2. Table 6.1 also contains the numerical values for the different parameters. The base case parameters determined in the previous chapters have been retained.

⁴Note that in the special case that time to build is zero, $\Delta V'(X, \Delta K, K_0, 0) = \Delta V_j(X, \Delta K, K_0)$, with *j* determined by the region R_j where *X* is located at that moment.

⁵Also note that time to build can be included in a new port project as well, where $K_0 = 0$. To this end, $\Delta V_i(X, \Delta K, K_0, \theta)$ needs to be replaced by $V_i(X, \Delta K, \theta)$, with ΔK the size of the project, i.e., *K*.

The additional parameters are determined in the next section. The sensitivity of the results to monetary scale factor *A*, growth rate μ , uncertainty σ , relative number of PA shares owned publicly s_G and relative share of total consumer surplus considered s_{CS} is discussed in Section 6.6.

6.4 Setting the additional model parameters

The expansion decision in this chapter deals again with a new container terminal with an annual capacity of about 8 to 14 million TEU per year. Due to the model extensions in Section 6.2, additional model parameters need to be determined, based on real port data. The initial capacities of a port before expansion are set to zero, five, ten or fifteen million TEU per year, representing respectively a new port, medium-sized ports such as Bremen or Valencia, large ports such as Rotterdam or Antwerp, or even larger Chinese ports (Vlaamse Havencommissie, 2016).

The time to build a new dock can amount to three years or more, as a result of various uncertain factors such as political decisions and environmental actions. This led to a final construction lead time of six years for the Deurganckdok (1999-2005), which was more than initially planned. The Maasvlakte 2 project in Rotterdam took five years to complete (2008-2013) (Port of Rotterdam, 2018), whereas the construction time of the Liverpool 2 construction project was only three years (Ship Technology, 2018). In this chapter, time to build is varied between zero and six years, to study the impact of an increase in time to build on the investment decision.

6.5 Results and discussion

The numerical model solutions in Tables 6.3 - 6.5 allow analysing the impact of a number of project and economic characteristics on the final investment decision of a port. Each table corresponds to a specific ownership structure and considers the base case, together with some scenarios involving one different economic characteristic. In the following subsections, the effect of time to build and the impact of existing capacity are analysed in greater detail.

6.5.1 The impact of time to build

Port expansion projects involving longer construction lead times, i.e., a higher θ , are installed later in most of the cases. This finding is analogous to the outcome of an increase in
Variables		
p	=	price
q	=	throughput
$K = K_0 (\in \{0, 5, 10, 15\}) + \Delta K$	=	total capacity after investment
	=	existing capacity plus capacity expansion
Inverse demand function: $p = X - B$	q-1	$4X \frac{q}{K^2}$
B(=1)	=	slope
A(=5)	=	monetary scale factor of congestion cost
Demand shift parameter X : $dX(t) =$	$\mu X($	t)d $t + \sigma X(t)$ d $Z(t)$
t (=annual)	=	time horizon
Z	=	standard Wiener process
$\mu(=0.015)$	=	drift of Z (i.e., growth)
$\sigma(=0.1)$	=	drift variability of Z (i.e., uncertainty)
Total cost $TC = c q + c_h K$		
c(=1)	=	constant marginal operational cost
$c_h(=0.5)$	=	cost to hold one unit of capacity in place
Investment cost $I = FC_I + \gamma_1 \Delta K - \gamma_2$	ΔK^2	$^{2}+\gamma_{3}\Delta K^{3}+\gamma_{4}\Delta K^{4}$
$FC_{I}(=80)$	=	fixed investment cost
$\gamma_1(=180)$	=	first-order coefficient
$\gamma_2(=19)$	=	coefficient reflecting economies of scale of
		investment cost
$\gamma_3(=0)$	=	omitted third-order coefficient
$\gamma_4 (= 0.12)$	=	coefficient posing boundary to maximum
		size of the project
Time to build the project $\theta \in \{0, 1, 2, \dots, n\}$	4,6}) years
Operational objective function $\Pi = \pi$	$\pi + s$	$_{G} \cdot \lambda q + s_{G} s_{CS} \cdot CS$
π	=	port profit = $pq - TC$
$\lambda(=0.4)$	=	spillover benefits per unit <i>q</i>
CS	=	consumer surplus, i.e., $Bq^2/2$
$s_G(\in [0;1])$	=	share of the port owned by the government
$s_{CS} (\in [0;1])$	=	share of total CS taken into account by
		the government

Table 6.1: Overview of Chapter 6's model and the selected parameters.

	Assumption(s)
Project type? Capacity constraint? Port type? Port competition? Time to build? Phased investment?	Port expansion Soft constraint Public and/or private service port* No (1 port in model) Included in the model Project installed at once: available after completion
General assumptions	Price (sum of port dues and terminal tariff) transparency \rightarrow Linear additive inverse demand function Demand uncertainty, modelled using a GBM with constant, independent growth (μ) and uncertainty (σ) parameters Growth market with $r > \mu > 0$, to ensure RO model convergence ^{**} Poisson-distributed ship arrivals at the port: delays start at about 50% and increase sharply beyond 75-80% occupancy User congestion costs depend on dock users' average value of time, which evolve with the market (i.c., increasing) Investment cost: increasing fourth-order polynomial with economies of scale and size boundary Deterministic, linear operational and capacity holding costs Flexibility of investment timing and size \rightarrow Option to postpone the investment Once-and-for-all investment decision with flexible throughput levels Infinite project life time (maintenance executed)
**: Condition r > SCOPE: Theoretical design	*: Owned and operated by one single actor. μ leads to a finite timing decision, instead of infinite investment postponement. Sensitivity tested with negative growth rates as well. i capacity for container throughput services. Other port services disregarded (sufficiently present or realised by actors other than the PA or TOC). Focus on lumpy investment in port infrastructure and irreversible elements of superstructure.

Table 6.2: Overview of Chapter 6's model assumptions.

uncertainty. If time to build were higher, the exposure to uncertainty would be higher too, because more time without profit flows passes after the investment decision. During this time, demand follows an uncertain path. This has an uncertain impact on the present value of the future profit flow. As a result, the reaction to this increased uncertainty is to postpone the investment decision, when the market is larger, in order to increase the probability of sufficiently high profits.

However in some cases, increased time to build leads to anticipating the investment decision, especially compared with cases with an investment decision threshold that is already relatively high. This can be explained by the fact that time to build also encompasses an incentive to anticipate the investment decision. The reason is that the present value of the future profit flow will be relatively lower due to this construction lag between the investment decision and project completion. Moreover, the stochastic discount factor increases more than linearly in the investment threshold (Huisman & Kort, 2015). In order to avoid higher discounting of the future profit flow and prevent a too low project value, ports may react by investing earlier. Aguerrevere (2003) in this light shows that also firms without congestion costs expand before their infrastructure is fully occupied because of the time to build.

The main impact of time to build on the investment decision can be summarised as follows:⁶

Result 6.1. Under the assumptions given in Table 6.2, the port adapts its strategy by installing less capacity and / or investing at a later moment if the construction lead time is higher, because the latter reduces the project's attractiveness.

In some example expansion projects, an opposed impact of time to build on the final project size and timing decision is observed. In Table 6.3 in the base case for example, the expansion of a port with existing capacity of 5 million TEU per year and construction lead time of one year is installed at a higher threshold, but the size of the project is smaller than if there is no construction lead time. As explained in Chapter 4, such an observation of two strong effects is uncommon in the RO literature, where the positive relationship between investment size and timing often dominates. Hence, it is more common in RO that the project size is larger when investment takes place at a moment when the market has grown more, i.e., later, or vice versa. Examples of this common RO observation are found in the same base case in Table 6.3. If the project lead time were one year, a port with existing capacity of 15 million TEU per year would install a smaller project, at a lower threshold than in the case without time to build. Oppositely, a greenfield project with one year of lead time would be larger and installed later than a greenfield project without time to build. This ambiguous impact of time to build on the investment decision is in line with the findings of Aguerrevere (2003). Hence, each port is advised to analyse its own investment options

⁶This result holds for expansion projects, as well as for the installation of new ports.

			Base case.							
			Existing	capacity						
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$					
	$\theta = 0$	(37.71, 11.19)	(31.83, 10.03)	(39.01, 10.46)	(49.46, 11.17)					
Time	$\theta = 1$	(38.57, 11.20)	(32.20, 10.02)	(39.16, 10.43)	(49.41, 11.11)					
to	$\theta = 2$	(39.47, 11.21)	(32.59, 10.02)	(39.33, 10.40)	(49.39, 11.06)					
build	$\theta = 4$	(41.40, 11.23)	(33.45, 10.00)	(39.74, 10.34)	(49.40, 10.95)					
	$\theta = 6$	(43.52, 11.25)	(34.40, 9.99)	(40.24, 10.28)	(49.48, 10.84)					
			A set to 4.							
Existing capacity $K_0 = 0$ $K_0 = 5$ $K_0 = 10$ $K_0 = 15$										
	$\theta = 0$	(34.20, 10.98)	(31.11, 10.01)	(40.00, 10.61)	(52.16, 11.44)					
Time	$\theta = 1$	(34.91, 11.00)	(31.41, 10.00)	(40.08, 10.57)	(52.01, 11.38)					
to	$\theta = 2$	(35.66, 11.01)	(31.73, 9.99)	(40.19, 10.53)	(51.89, 11.31)					
build	$\theta = 4$	(37.25, 11.04)	(32.42, 9.98)	(40.45, 10.46)	(51.72, 11.19)					
	$\theta = 6$	(39.00, 11.07)	(33.21, 9.96)	(40.80, 10.40)	(51.56, 11.07)					
			σ set to 0.15.							
			Existing	capacity						
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$					
	$\theta = 0$	(56.42, 13.47)	(43.64, 11.89)	(56.10, 12.94)	(78.61, 14.73)					
Time	$\theta = 1$	(57.48, 13.46)	(43.83, 11.84)	(55.56, 12.81)	(76.74, 14.49)					
to	$\theta = 2$	(58.54, 13.44)	(44.06, 11.79)	(55.08, 12.69)	(71.85, 13.97)					
build	$\theta = 4$	(57.16, 13.04)	(44.48, 11.69)	(52.69, 12.30)	(61.36, 12.82)					
	$\theta = 6$	(53.68, 12.42)	(43.77, 11.44)	(48.77, 11.73)	(54.44, 11.98)					
			µ set to 0.02.							
	Existing capacity									
$K_0 = 0$ $K_0 = 5$ $K_0 = 10$ $K_0 = 15$										
	$\theta = 0$	(42.75, 12.47)	(34.84, 11.16)	(44.16, 11.97)	(59.43, 13.29)					
Time	$\theta = 1$	(43.45, 12.47)	(35.00, 11.13)	(43.95, 11.90)	(58.69, 13.16)					
to	$\theta = 2$	(44.18, 12.47)	(35.19, 11.10)	(43.77, 11.83)	(58.00, 13.04)					
build	$\theta = 4$	(45.77, 12.46)	(35.62, 11.05)	(43.49, 11.69)	(56.76, 12.81)					
	$\theta = 6$	(47.49, 12.46)	(36.15, 11.00)	(43.31, 11.57)	(55.09, 12.53)					
			u sat to 0.01							
			μ set to -0.01.							
		$K_{-} = 0$	Existing $K_0 = 5$	Existing capacity $K_{1} = 5$ $K_{2} = 10$						
				$\mathbf{R}_{0} = \mathbf{I}\mathbf{U}$	$R_0 = 13$					
/E10	$\theta = 0$	$\mathbf{J} (37.74, 9.21)$	(32.46, 8.02)	(38.48, 7.98)	(46.63, 8.12)					
Tim		(39.62, 9.24)	(33.74, 8.04)	(39.72, 7.99)	(47.97, 8.12)					
t($\theta = 2$	2 (41.62, 9.26)	(35.08, 8.06)	(41.03, 8.00)	(49.37, 8.12)					
buil	$\theta = 4$	(46.00, 9.31)	(37.98, 8.10)	(43.83, 8.02)	(52.34, 8.13)					
	0=6	0 (30.96, 9.35)	(41.21, 8.14)	(40.90, 8.03)	(33.38, 8.13)					

Table 6.3: Optimal investment strategies $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ under different project lead times and initial capacities in a private port.

Base case parameter values: A = 5, B = 1, c = 1, $c_h = 0.5$, $\mu = 0.015$, $\sigma = 0.10$, r = 0.06, $FC_I = 80$, $\gamma_1 = 180$, $\gamma_2 = 19$, $\gamma_3 = 0$, $\gamma_4 = 0.12$, $s_G = 0$, $s_{CS} = 0$, $\lambda = 0.4$.

Table 6.4: Optimal investment strategies $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ under different project lead times and initial capacities in a port that is 50% publicly owned by a government taking 50% of the consumer surplus into account.

			Base case.							
		TT 0	Existing	capacity						
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$					
	$\theta = 0$	(36.07, 11.22)	(29.80, 10.02)	(36.07, 10.41)	(45.41, 11.08)					
Time	$\theta = 1$	(36.93, 11.23)	(30.18, 10.01)	(36.24, 10.38)	(45.40, 11.02)					
to	$\theta = 2$	(37.83, 11.24)	(30.57, 10.00)	(36.43, 10.34)	(45.40, 10.97)					
build	$\theta = 4$	(39.76, 11.26)	(31.43, 9.99)	(36.87, 10.29)	(45.48, 10.87)					
	$\theta = 6$	(41.87, 11.28)	(32.38, 9.98)	(37.38, 10.23)	(45.64, 11.77)					
			A set to 4							
			Existing	capacity						
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$					
	$\theta = 0$	(32.58, 11.01)	(29.03, 9.99)	(36.89, 10.54)	(47.77, 11.34)					
Time	$\theta = 1$	(33.28, 11.03)	(29.33, 9.98)	(36.99, 10.51)	(47.67, 11.28)					
to	$\theta = 2$	(34.02, 11.04)	(29.65, 9.98)	(37.11, 10.47)	(47.58, 11.22)					
build	$\theta = 4$	(35.61, 11.07)	(30.35, 9.96)	(37.41, 10.41)	(47.49, 11.10)					
	$\theta = 6$	(37.35, 11.10)	(31.14, 9.95)	(37.78, 10.35)	(47.47, 10.99)					
			σ set to 0.15							
			<u> </u>	canacity						
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$					
	$\theta = 0$	(53.72, 13.45)	i3.72, 13.45) (40.61, 11.81) (51.25, 12.							
Time	$\theta = 1$	(54.79, 13.44)	(40.84, 11.76)	(50.84, 12.64)	(69.17, 14.19)					
to	$\theta = 2$	(55.91, 13.43)	(41.11, 11.72)	(50.49, 12.53)	(66.87, 13.90)					
build	$\theta = 4$	(55.54, 13.12)	(41.70, 11.63)	(49.23, 12.24)	(58.69, 12.92)					
	$\theta = 6$	(52.54, 12.53)	(41.58, 11.44)	(46.38, 11.77)	(52.37, 12.10)					
			u set to 0.02.							
Existing capacity										
$K_0 = 0$ $K_0 = 5$ $K_0 = 10$ $K_0 = 15$										
	$\theta = 0$	(40.78, 12.48)	(32.50, 11.11)	(40.55, 11.84)	(53.92, 13.08)					
Time	$\theta = 1$	(41.49, 12.47)	(32.69, 11.08)	(40.40, 11.77)	(53.31, 12.96)					
to	$\theta = 2$	(42.24, 12.47)	(32.89, 11.05)	(40.27, 11.71)	(52.75, 12.84)					
build	$\theta = 4$	(43.85, 12.47)	(33.36, 11.00)	(40.09, 11.58)	(51.75, 12.63)					
	$\theta = 6$	(45.60, 12.47)	(33.92, 10.96)	(40.01, 11.47)	(50.74, 12.41)					
			µ set to -0.01.							
Existing capacity										
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$					
-	$\theta = 0$	0 (36.19, 9.27)	(30.51, 8.06)	(35.78, 8.00)	(43.13, 8.12)					
		(3203020)	(31.73, 8.08)	(36.96, 8.00)	(44.40, 8.12)					
Time	$\theta = 0$	(30.05, 3.23)	. , ,							
Time	$\begin{array}{c} \theta = 0 \\ \theta = 0 \\ \theta = 0 \end{array}$	$\begin{array}{c} 1 & (30.03, 9.23) \\ 2 & (39.98, 9.32) \end{array}$	(33.02, 8.10)	(38.20, 8.01)	(45.71, 8.13)					
Time to build	$\begin{array}{ccc} \theta &= 0\\ \theta &= 0\\ \theta &= 0\\ \theta &= 0 \end{array}$	2 (39.98, 9.32) 4 (44.27, 9.36)	(33.02, 8.10) (35.81, 8.14)	(38.20, 8.01) (40.86, 8.03)	(45.71, 8.13) (48.52, 8.13)					

19, $\gamma_3 = 0, \gamma_4 = 0.12, s_G = 0.5, s_{CS} = 0.5, \lambda = 0.4.$

			Base case.								
			Existing	capacity							
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$						
	$\theta = 0$	(31.58, 11.44)	(23.73, 10.06)	(26.93, 10.26)	(32.66, 10.79)						
Time	$\theta = 1$	(32.42, 11.45)	(24.10, 10.06)	(27.13, 10.24)	(32.73, 10.74)						
to	$\theta = 2$	(33.30, 11.46)	(24.49, 10.05)	(27.34, 10.21)	(32.81, 10.69)						
build	$\theta = 4$	(35.20, 11.47)	(25.32, 10.04)	(27.82, 10.16)	(33.03, 10.60)						
	$\theta = 6$	(37.28, 11.48)	(26.25, 10.03)	(28.36, 10.11)	(33.33, 10.52)						
			A set to 4								
Existing capacity											
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$						
	$\theta = 0$	(28.03, 11.26)	(22.72, 10.03)	(27.15, 10.38)	(33.94, 11.01)						
Time	$\theta = 1$	(28.71, 11.27)	(23.02, 10.02)	(27.29, 10.34)	(33.94, 10.96)						
to	$\theta = 2$	(29.43, 11.28)	(23.33, 10.01)	(27.45, 10.31)	(33.96, 10.91)						
build	$\theta = 4$	(30.96, 11.30)	(24.02, 10.00)	(27.81, 10.26)	(34.05, 10.81)						
	$\theta = 0 (3)$ Time $\theta = 1 (3)$ to $\theta = 2 (3)$ puild $\theta = 4 (3)$ $\theta = 6 (3)$ $\theta = 6 (3)$ Time $\theta = 1 (2)$ to $\theta = 2 (2)$ puild $\theta = 4 (3)$ $\theta = 6 (3)$ $\theta = 6 (3)$ $\theta = 6 (4)$ Time $\theta = 1 (4)$ to $\theta = 2 (4)$ puild $\theta = 4 (5)$ $\theta = 6 (4)$ $\theta = 6 (4)$ Time $\theta = 1 (3)$ $\theta = 6 (4)$ $\theta = 6 (4)$ $\theta = 6 (4)$ Time $\theta = 1 (3)$ $\theta = 6 (4)$ $\theta = 6 (4)$	(32.65, 11.32)	(24.77, 9.99)	(28.23, 10.21)	(34.22, 11.72)						
			<u> </u>	<u>r set to 0.15.</u>							
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$						
	$\theta = 0$	(46.60, 13.58)	(31.77, 11.64)	(36.81, 12.20)	(47.34, 13.38)						
Time	$\theta = 1$	(47.69, 13.57)	(32.08, 11.60)	(36.72, 12.11)	(46.70, 13.22)						
to	$\theta = 2$	(48.84, 13.55)	(32.43, 11.57)	(36.66, 12.03)	(46.13, 13.07)						
build	$\theta = 4$	(50.40, 13.42)	(33.22, 11.50)	(36.65, 11.87)	(44.93, 12.76)						
	$\theta = 6$	(49.07, 12.94)	(34.03, 11.42)	(36.51, 11.68)	(42.70, 12.31)						
			µ set to 0.02.								
			Existing	capacity							
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$						
	$\theta = 0$	(35.51, 12.66)	(25.59, 11.03)	(29.60, 11.47)	(37.29, 12.39)						
Time	$\theta = 1$	(36.23, 12.65)	(25.81, 11.01)	(29.58, 11.42)	(37.01, 12.30)						
to	$\theta = 2$	(36.98, 12.65)	(26.06, 10.99)	(29.59, 11.36)	(36.76, 12.21)						
build	$\theta = 4$	(38.61, 12.64)	(26.60, 10.94)	(29.66, 11.26)	(36.35, 12.03)						
	$\theta = 6$	(40.41, 12.63)	(27.22, 10.91)	(29.80, 11.16)	(36.03, 11.87)						
			u set to -0.01.								
		$K_0 = 0$	$K_0 = 5$	$K_0 = 10$	$K_0 = 15$						
	$\theta = 0$	0 (31.83, 9.53)	(24.55, 8.25)	(27.18, 8.08)	(31.83, 8.16)						
Tim	$\theta = \theta$	1 (33.55, 9.55)	(25.61, 8.26)	(28.15, 8.09)	(32.83, 8.16)						
t	$\boldsymbol{\theta} = \boldsymbol{\theta}$	2 (35.37, 9.57)	(26.72, 8.28)	(29.17, 8.10)	(33.87, 8.16)						
buil	d $\theta = 4$	4 (39.36, 9.59)	(29.15, 8.32)	(31.35, 8.12)	(36.09, 8.17)						
	$\theta = 0$	6 (43.93, 9.63)	(31.87, 8.35)	(33.77, 8.14)	(38.53, 8.18)						
	- 1	(1.32, 1.50)		15 0 10 0 0	· · · · · · · · · · · · · · · · · · ·						

Table 6.5: Optimal investment strategies $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$ under different project lead times and initial capacities in a public port.

Base case parameter values: A = 5, B = 1, c = 1, $c_h = 0.5$, $\mu = 0.015$, $\sigma = 0.10$, r = 0.06, $FC_I = 80$, $\gamma_1 = 180$, $\gamma_2 = 19$, $\gamma_3 = 0$, $\gamma_4 = 0.12$, $s_G = 1$, $s_{CS} = 1$, $\lambda = 0.4$.

using the described methodology. This allows finding the optimal point in the trade-off between higher discounting and an increased customers' willingness to pay due to investment postponement, respectively decreasing and increasing the project value.

6.5.2 The impact of existing port capacity

Next to time to build, the amount of existing capacity in the port also has an impact on the optimal investment decision. This impact of existing capacity is derived from the calculated optima in Tables 6.3-6.5 and is summarised as follows:⁷

Result 6.2. Under the assumptions given in Table 6.2, it takes more time before a port with a higher current capacity becomes too highly occupied. Hence, it is possible in such a port to postpone capacity expansion longer in order to reduce the costly congestion in the port. Because of the positive relationship between investment timing and size, the installed capacity will also be larger.

In the results, an important difference is observed between cases with existing capacity exceeding the size of an optimal new port (greenfield) project and cases with less existing capacity than this optimal greenfield project with all other economic parameters equal. If the existing capacity is below the optimal capacity that would be installed in a oneshot greenfield investment project, too much costly congestion will arise soon and expansion is required sooner than the optimal investment timing of the corresponding greenfield project. Because of the earlier investment, and because some capacity is already present, the size of the expansion will also be lower than the size of the greenfield project. However, total capacity after expansion $(K_0 + \Delta K^{**}(K_0, \theta))$ will be higher than the size of the optimal greenfield project. This is partly explained by the fact that the port will strive to adapt the investment size to benefit as much as possible from the investment size scale economies. Moreover, each expansion encompasses the installation of free, initially unused, capacity to attract the shipping lines. This is in line with the finding of Chronopoulos et al. (2017), that stepwise investment leads to the investment in more capacity than when the capacity is to be installed at once. However, if existing capacity exceeds the size of the optimal greenfield project ($K_0 > \Delta K^{**}(0, \theta)$), the timing of the expansion will be later than the timing of the optimal greenfield project. Depending on how much higher the optimal threshold for X is, the size of the project will be higher or lower than the size of the optimal greenfield project. In addition to the positive effect of a later investment on the investment size, the capacity in place has a negative effect on the size of the port expansion project. This analysis indicates the importance of accounting for the amount of existing capacity in the investment analysis. This highlights the added value of this analysis, compared with Chen & Liu (2016).

⁷This result holds for projects with and without time to build.

6.6 Sensitivity analyses

In order to verify the impact of the type of port projects studied on some of the results derived in Chapters 3 and 4, this section contains some sensitivity analyses. Congestion costs, ownership structures, growth rates and the size of the uncertainty are altered, to verify the impact of these elements on port expansion investment decisions and to compare with the impact on the decision to invest in new ports. To this end, the base cases in Tables 6.3 - 6.5, are compared with the situations in which one additional parameter is altered. These altered parameters are respectively the monetary scale factor, the uncertainty and the growth rate.

6.6.1 Sensitivity of the results to congestion costs and waiting-time aversion

For the once-and-for-all greenfield projects in the previous chapters, a positive relationship between costs of congestion and investment size and timing has been found. Such port projects for customers or goods that are more time-sensitive, resulting in a higher user cost of congestion and hence a reduced willingness to pay for the same service level, are installed later. This is a reaction to the reduced project attractiveness due to this lower willingness to pay. Additionally, more capacity is installed to reduce occupancy rates and hence the more costly congestion.⁸

Oppositely, many expansion projects with a sufficiently high existing capacity and current throughput level are installed earlier if congestion has a larger impact on the customers. If the port is already active ($K_0 > 0$), congestion has a persistent and daily impact on the customers' and the port's profit. The port has a substantial incentive to install the expansion project sooner to avoid persistent congestion in the future. The following result is derived:

Result 6.3. Under the assumptions given in Table 6.2, if the port's users are more waitingtime averse, port expansion ideally takes place earlier, i.e., when the market is smaller. Consequently, the size of the expansion project will on average be slightly smaller too.

This finding is in line with the findings of Meersman & Van de Voorde (2014a), namely that active ports have a larger incentive to expand soon enough to avoid fully occupied capacity and to dispose of sufficient free capacity in order to attract the shipping lines. Consequently, the timing decision of expanding an existing port such as the Port of Antwerp entails different incentives than the timing decision of the investment in a new port, like in Africa or Brazil.

⁸Here, phased investment is not considered. This leads to constructing a larger port at the moment of investment, which will be later. Oppositely, a first phase of a phased investment might be smaller and completed sooner, since a follow-up investment can take place later in order to expand the initial capacity (Chronopoulos et al., 2017). This reasoning is supported by the remainder of this section.

6.6.2 Sensitivity of the results to public ownership

If welfare is taken more into account in the objective function, e.g., as a result of increased public ownership or the government considering a larger share of the consumer surplus, this has a substantial impact on the investment decision. From Chapters 4, 5 and the literature, it was already known that increased public ownership leads to investing earlier in additional capacity (Asteris et al., 2012). Opposed to the common RO finding that later investment is associated with more capacity, in such a case there are some instances of greenfield projects where larger investments take place earlier.

However, as argued, disposing of additional port capacity becomes more urgent in the case of expansion projects. Existing capacity cannot accommodate the higher amount of throughput that is desired as a result of the project's increased attractiveness for a government considering social welfare more. By consequence, the following result can be formalised:

Result 6.4. Under the assumptions given in Table 6.2, the timing of the expansion will be a lot sooner in the case of higher public money involvement. Because of the sooner timing's downward pressure on size, the net impact of increased public ownership on the investment size will on average be slightly negative for expansion projects.

This result illustrates the importance of timely removing barriers to economic growth, such as a lack of port capacity. Port expansion should take place in time, to maximise the welfare effects that are generated by ports.

6.6.3 Sensitivity of the results to growth and uncertainty

The calculations here confirm the finding that increases in growth and uncertainty often lead to an increase in the size of a port capacity investment project and a later investment timing (Hagspiel et al., 2016). Bar-Ilan & Strange (1996) however show that with a long project lead time, the impact of increased uncertainty on timing can become negative. This finding is confirmed empirically by Marmer & Slade (2018) in the U.S. copper mining industry with considerable construction lead times. Since higher uncertainty may result in higher upward jumps of demand, more capacity will be needed to accommodate demand peaks. The effect of downward jumps is asymmetric, since during demand troughs it is possible to reduce throughput levels.

The results displayed in Tables 6.3 - 6.5 did not illustrate the negative impact of time to build θ on the uncertainty's effect on the investment decision. This observation of investment anticipation as a result of increased uncertainty in combination with a high time to build can however be made in this chapter's model as well. The following small example

	Uncertainty												
	$\sigma = 0.09$ $\sigma = 0.10$ $\sigma = 0.11$												
Dis-	r = 0.05	(38.27, 11.16)	(41.03, 11.64)	(44.45, 12.20)									
count	<i>r</i> = 0.06	(37.96, 10.03)	(39.74, 10.34)	(41.79, 10.67)									
rate <i>r</i> = 0.07 (39		(39.45, 9.44)	39.45, 9.44) (40.90, 9.68)										
Other parameter values:													

Table 6.6: Optimal investment strategies ($X_T^{**}(K_0 = 10, \theta = 4)$, $\Delta K^{**}(K_0 = 10, \theta = 4)$) under different values of *r* and σ in a private port.

 $A = 5, B = 1, c = 1, c_h = 0.5, \mu = 0.015, FC_I = 80, \gamma_1 = 180, \gamma_2 = 19, \gamma_3 = 0, \gamma_4 = 0.12, s_G = 0, s_{CS} = 0.$

with a substantially higher project lead time is used. In the base case from Table 6.3, expanding a private port with an existing capacity of 10 million TEU per year ($K_0 = 10$) and a project lead time θ of 15 years gives the following optima:

- $\sigma = 0.1$: (41.63 euro per TEU, 9.84 M TEU per year);
- $\sigma = 0.11$: (41.53 euro per TEU, 9.90 M TEU per year).

This example confirms the robustness of the model with respect to the finding of Bar-Ilan & Strange (1996) and Marmer & Slade (2018). Notwithstanding the anticipated timing of the investment, the size increases, in order to be sufficiently able to accommodate higher potential upward deviations in demand.

The findings from Chapter 3 with respect to negative growth rates are confirmed here. The investment threshold for *X* would become that high that it would be very unlikely to reach this value from below, because only in the exceptional case that demand were to become sufficiently high, it would be optimal to invest. This higher investment threshold is required to guarantee a sufficient project value *V*, notwithstanding the expected demand decrease during the project lead time. Hence, if the economy is expected to decline, investment will only take place at a moment that demand is sufficiently large to make up for the declining future profits. The project would also be smaller compared with cases with positive economic growth, as the negative demand growth would lead to a declining demand and lower future occupancy rates. It is also apparent that an increase in time to build in the case of negative growth results in an increase in the project size. This is due to the sharp increase in the investment threshold in such a case. Additionally, with negative growth in the port sector, it might be preferred to use public money in other sectors in order to create more welfare.

Finally, the conclusions about the joint impact of an increase in uncertainty and a resulting decrease in the discount rate from Section 3.5 can be confirmed here as well, using Table 6.6. An increase in uncertainty followed by an interest rate decrease would lead to an even larger increase in the size of the expansion than in the case of neglecting this effect on the interest rate. Again, it can be observed that the impact on the timing of the investment is ambiguous and needs to be analysed for each specific investment case. To this end, the models from this thesis and next chapter's tool can support the PA's management.

6.7 Conclusion of this chapter

In this chapter, time to build and expansion of existing capacity have been added to the analysis of one-shot port capacity investment decisions under congestion and uncertainty. These investments are to be optimised with respect to the timing and size. Considering time to build and expansion is important to more accurately model reality. Time to build reduces the attractiveness of a project and increases the uncertainty. As a result, ports tend to postpone their investment to wait for a larger market. However, time to build also urges ports to anticipate their investment, especially when the investment threshold is already high, to make sure that the capacity is ready when it is needed. The impact of time to build on size is even more ambiguous, due to the supplementary positive impact of investing later on the size, next to the negative impact of a less attractive project. One general result has been derived: due to time to build, port expansion will take place later or the expansion will be smaller. In some cases, a combination of both effects is perceived. Existing capacity has an impact on the effect of congestion costs on the investment decision as well. As the port is already operating and generating revenues, it has a larger incentive to avoid congestion as soon as possible for each unit of throughput handled. Therefore, larger waiting-time aversion of port customers leads to anticipating the port expansion decision, which will be of a slightly smaller size. This is opposed to larger greenfield projects, which are installed later under similar economic circumstances. Finally, it is observed that increased public ownership leads to earlier investment, but not necessarily more capacity in port capacity expansions. Public port ownership however facilitates the full exploitation of potential port benefits.

The findings of this chapter have some important implications for practitioners and policy makers. Time to build erodes the project value, because it reduces the present value of the profit streams. It moreover increases the exposure to uncertainty. This may among other things lead to further postponing the investment decision. As a result, large projects encompass an incentive to take actions reducing construction lead times. Governments for example may try to limit long internal discussions and arguments with external action groups. The rewards may include higher profits, sooner investment and often slightly less capacity that needs to be installed to achieve a similar level of social welfare. The resources that are saved, can be used elsewhere.

The models derived in this and the previous chapters allowed deriving new theoretical insights into the optimal investment decision in port infrastructure and superstructure under uncertainty. However, next to theoretical insights, port managers also need to be enabled to implement these models in practice. Moreover, they need to be aware of the potential benefits and costs of applying RO in practice. This is discussed in the next chapter.

Chapter 7

Moving from theoretical real options models to decision making in practice: required steps, opportunities and threats

The characteristics of the economic environment, the port type and the investment project type all have an influence on the optimal port capacity investment decision. The models developed in Chapters 3 to 6 allow deriving theoretical insights into this optimal decision. The theoretical directions of these influences provide useful information for port managers at the moment when the economic environment changes. For example, when demand starts to grow stronger, the results from Chapter 6 show the management of a single actorport that they would be better off by delaying port expansion and investing in more capacity.

The question however remains: How can a specific port implement the developed models to derive its own optimal investment decision in reality? A second question that remains is: What are the potential gains and costs of implementing RO, compared with currently used decision making methods. In practice, the NPV approach is often used to evaluate port investment projects. However, as was already indicated in Chapter 2, this approach neglects the value of flexible options in irreversible projects and uncertainty, such as the option to delay investment. Even if uncertainty is low or absent, the NPV decision rule can be (very) wrong, because the irreversible investment decision is considered as a now-ornever decision (Dixit & Pindyck, 1994). In order to apply the RO models developed in this thesis in ports, two conditions need to be fulfilled. On the one hand, the port management needs to have the information at hand to derive and estimate the values of the model parameters. On the other hand, the user needs to be able to apply advanced econometric methods in order to calculate the optimal investment timing and size, possibly supported by external econometricians (Dikos, 2008). This latter condition is less straightforward, as Marmer & Slade (2018) illustrate. They use the model of Bar-Ilan & Strange (1996) with time to build and apply it to the U.S. copper mining industry using advanced econometrics on reduced forms of a set of theoretical equations.

In this chapter, the necessary input for the RO models is discussed, as well as a possible, basic programming approach to transform the model into an optimal real-life port investment decision. The required steps that are needed to move from the complex, theoretical models, described in this thesis, to a more applicable, user-friendly implementation are discussed. These steps are gathered in a blueprint for a practical tool that can help to come one step closer to the practical implementation of the developed models, since the complex calculations are left for the algorithm. Hence, this tool should be seen as a simplified approach to yield a first approximation of the optimal investment strategy for a port expansion project, based on the RO models developed in this thesis. This tool could also support transforming the port management's input into the model parameters. Nevertheless, the quality of the input, such as the complex estimates of economic characteristics such as demand growth and uncertainty, will always be important and are left for the human user. The entered parameters are then transferred to the RO model yielding an approximation of the optimal investment size and timing in terms of observable decision variables. It is however not the objective of this chapter to explore the large number of complex estimation techniques for the parameters. These are nonetheless required to accurately implement the RO models from this thesis. More advanced econometrics and further model extensions (e.g., including phased investment) will be crucial objects of further research to increase the realism and the applicability of RO investment analysis in ports. As for every model, also for this tool holds the wisdom: "Rubbish in, rubbish out". A subsequent discussion of the advantages and disadvantages of RO models allows quantifying the potential gains and determining the costs as compared with an NPV approach.

In this chapter, the specific model of Chapter 6 has been chosen as an illustration, because the majority of port capacity investment decisions are expansion decisions (De Langen et al., 2018) and involve considerable time to build. Moreover, ports can be owned privately or publicly. Both types are accounted for in this model. Although this model does not explicitly model port competition, it is still useful to gain insight into the unrestricted optimal investment decision of a single port and to illustrate the model implementation in practice. To take into account the competition of nearby ports, the insights and different strategies of Chapter 5 need to be considered afterwards. Subsequently the aggregate optimal investment decision (from a service port model) might have to be translated into the decision of multiple actors in a port. To this end, the approach of Chapter 4 can be followed. This requires insights into the revenue and cost structures of the port authority (PA) and the terminal operating company (TOC) in order to derive their shares (the α 's of Chapter 4) in the total revenue and costs generated in the port. Moreover, Chapter 6's model is based on a number of assumptions that do not necessarily all hold in the cases that are to be studied in practice. In order to relax the most inappropriate assumptions, the RO models developed in this thesis need to be manually extended with the characteristics of the situation at hand. This can be based on Chapter 2's framework. Only after this manual extension, the new RO model can be calculated or programmed.

This chapter is structured as follows. The next section discusses the blueprint of the practical tool to implement a simplified version of the developed RO models for port managers to get a first idea about their optimal investment decision. Subsequently, Section 7.2 provides a reflection on the advantages and difficulties of using RO in a port context. These advantages are discussed and quantified for the scenario studied in this chapter. This chapter ends with a short conclusion and potential extensions of the tool, based on future research outcomes.

7.1 A real options tool blueprint for port capacity investment decisions

The next subsection provides an overview of the required inputs and how they can help to derive the parameters of Chapter 6's model. Section 7.1.2 illustrates how the tool should work, using a realistic application for an existing private port that focuses on the own unrestricted optimal investment decision. This model moreover accounts for time to build. In addition, it illustrates how a port's management could use the output of the tool to take an optimal port capacity expansion decision under demand uncertainty without neglecting the option value of waiting.

7.1.1 Input required to calculate real options model parameters

Some model parameters are relatively easy to obtain, as they can be based on information that is available for the management of the port, for example from their accounting. These include the marginal operational cost (*c*) and the holding cost to maintain the capacity in place ($c_h K_0$). Dividing the latter by the current design capacity (K_0) yields the capacity holding cost per TEU. Also the (expected) time to build (θ) can be anticipated by the port management, making abstraction of the construction lead time uncertainty. For the discount rate (*r*), the long term interest rate of state obligations is often used as a proxy for the risk-free rate of return (Dixit & Pindyck, 1994). If a government (partly) owns the port, they are expected to dispose of an estimation of the spillover benefits per TEU for the region (λ) (see e.g., Gueli et al. (2019)) and the share of total consumer surplus considered by this government (s_{CS}).¹ Finally, the relative number of shares held publicly (s_G) can be derived from the PA's ownership structure.

¹The government could ask the National Bank or another external consultant to provide these values for them.

For a number of other parameters, more complex calculations with the provided data are required. In reality, they can be executed or supported by econometricians. For the tool however, they need to be simplified and carried out automatically in order to approximate the outcome of the complex RO analysis for a PA. This yields a first idea of their optimal investment decision. Subsequent and more detailed calculations can be performed by external consultants and econometricians.

In order to approximate the demand growth rate (μ) and uncertainty (σ) in a specific port in a specific period, the regression analysis approach of Chapter 3 could be used. The growth can be estimated using the exponential model

$$\ln(TEU_t) = \alpha + \mu t + \varepsilon_t, \tag{7.1}$$

with TEU_t a port's throughput at time t, while the root-mean-square error (RMSE) of the regression can be used as a proxy for σ . This analysis can be based on a time series of the port's throughput, which is widely available for port managers (Vlaamse Havencommissie, 2016). An important condition however is that the supply of capacity is constant, to eliminate this impact on the realisations of the demand curve. Moreover, an alternative approach could be using a time series of p, the prices paid in the port by the shipping companies to have their goods handled.

An important drawback of the regression analysis to estimate the growth rate and uncertainty, is that historical data are used. In periods where structural breaks take place and where the uncertainty and the growth rate are expected to change, more forward-looking estimation techniques may be preferred. Examples of such techniques include qualitative methods on the one hand, such as expert interviews, the Delphi method and surveying the market. Quantitative examples on the other hand include forecasting demand, based on regression analysis involving other forecasted variables, such as GDP, and use this demand forecast to estimate the uncertainty and the growth rate.

The investment cost function needs to be determined as well. More specifically, the fixed cost *FC* to carry out preliminary studies and to transport machinery to the construction site determines the intercept of the investment cost function. The shape and height of the function can be determined through information about a number of investment projects of different sizes. To this end, a trend line of the most appropriate shape needs to be fitted to the situation at hand. Next to the fixed investment cost, the investment costs of port projects need to reflect three characteristics. The cost is increasing and exhibits economies of scale. Finally, there is a boundary beyond which further expansion becomes very costly, since expropriation is needed to enlarge the port perimeter. In this thesis, a situation with a fourth-order polynomial has been assumed, using the following specification:

$$I(\Delta K) = FC + \gamma_1 \Delta K - \gamma_2 \Delta K^2 + \gamma_4 \Delta K^4 + \varepsilon.$$
(7.2)

However, if there are not sufficient data of projects of different sizes, it might be needed

to assume additional data points to estimate a fourth-order polynomial. The first data point required is the fixed cost (i.e., the cost of a project of 0 TEU). Two additional data points are the cost of the smallest and the largest project considered. In order to account for larger projects being infeasible due to size restrictions, two fictional, additional points with more than linear cost increases can be assumed and added to the dataset. Also a fictional project of half the size of the smallest project, with a cost higher than $[I(\min{\{\Delta K\}}) - FC]/2$, needs to be added to account for economies of scale in the investment cost over the first part of the domain of the investment cost function. Nevertheless, it is noteworthy that if another functional shape for $I(\Delta K)$ better fitted the data, replacing it in the calculations would not impact the technicality of the calculations, as long as the function is differentiable with respect to ΔK .

The final parameter that needs to be set, is the monetary scale factor of the congestion cost, *A*, in order to translate delays into user costs. These costs depend on the port users' average value of time and reduce their willingness to pay for port services. For the developed tool, this can also be estimated using regression analysis. The formula needed, is derived as follows under the simplifying assumption of a normalised slope of the inverse demand curve (B = 1) and the assumptions in Table 6.2:

$$p = X - q - \frac{AXq}{K_{tot}^2}$$

$$(X = \rho + q)$$

$$p = \rho + q - q - \frac{A(\rho + q)q}{K_{tot}^2}$$

$$\frac{A(\rho + q)q}{K_{tot}^2} = \rho - p$$

$$(7.3)$$

with p the price corresponding to an optimal throughput level q, K_{tot} the total design capacity of the considered port and ρ the price that the port could ask at the same throughput level if there were no impact of congestion. By retrieving this information from a market analysis or survey for different throughput levels (e.g., the current level, a level slightly lower, a level slightly higher than current throughput and a level close to full occupancy), A can be fitted using OLS on the following formula:

$$\frac{(\rho - p)K^2}{(\rho + q)q} = A + \varepsilon.$$
(7.4)

It is here that the impact of competition is partly accounted for in the calculations. When less competition is present, port users have fewer alternatives. By consequence, their willingness to pay will be reduced less due to congestion and delays in the considered port. Hence, the actual port price will deviate less from the price without congestion. This results in a smaller value for *A* in a less competitive market, even when the average aversion to waiting of the port's customers is similar.

A summary of the required inputs is given in the third column of Table 7.1. Moreover, this table shows in its second column the questions that can be used in the computer tool or application to gather the data. The final column shows the calculations required to derive the model parameters. This is discussed in the next section, together with how these calculations can be used to approach a first estimate of a specific port's optimal investment decision in terms of observable decision variables.

7.1.2 An application of the investment decision tool

In order to transform the input into an optimal investment decision, an algorithm based on three steps is proposed. The first step will collect the input and transform it into the parameters that are required for the RO model, as described above. This information is passed on to the second step where the optimal investment decision is calculated. In the third step, the calculations are outputted as relevant decision variables. In this section, the three steps with their different sub-modules are described and applied using an example of a private container port. As no complete real-life data set of one port could be obtained, some real-istic assumptions have been made in the previous chapters, based on the limited amount of data that is available. These are retained here. In the considered hypothetical example, the considered private port's current annual container throughput capacity is assumed to be 9 million TEU per year. The port moreover considers investing in an expansion of 8 to 10 million TEU p.a.

The first step of the algorithm needs to collect the input from the port authority and transform it into the parameters of the RO model where necessary. First of all, the existing theoretical (design) capacity of the port K_0 at the time of using the algorithm needs to be entered into the program, together with the current price p (assumed on average 14 euro per TEU, including port dues and terminal handling charges) and annual throughput q (assumed 5.5 million TEU p.a.). The parameters c (1 euro per TEU), $c_h K_0/K_0$ (4.5/9 = 0.5 euro per TEU), θ (5 years), r (6%), λ , s_G and s_{CS} can be derived directly from the input by the algorithm.² The derivation of these parameters has been explained in Chapters 3, 4 and 6. Subsequently, a regression module can extract the parameters μ and σ from the time series of a port's container throughput, as in the example in Table 7.1. In the previous chapters, a μ of 0.015 and a σ of 0.1 have been obtained. Alternatively, the user could enter the growth and uncertainty values that result from a more forward-looking estimation method.

²The latter three are disregarded in this specific example due to the application to a private port maximising profit.

	Derived parameters	$\downarrow \qquad \downarrow \qquad$	± 2	$ \begin{array}{l} \rightarrow c_{h} (= 4.5/K = 0.5) \\ \rightarrow \theta \\ \rightarrow r \end{array} $	-	$\downarrow \downarrow \underset{S_{CS}}{\downarrow}$	310	$\begin{array}{l} \underset{i \to 1}{\overset{OLS}{\to}} \ln(TEU_i) = \alpha + (\underline{\mu}) \ t + \varepsilon \\ \underset{i \to 1}{\hookrightarrow} \mu(=0.01) \\ \underset{i \to 1}{\hookrightarrow} RMSE = \sigma(=0.1) \\ \underset{i \to 1}{\hookrightarrow} I = FC + \gamma_1 \Delta K - \gamma_2 \Delta K^2 + \gamma_4 \Delta K^4 + \varepsilon \\ \underset{i \to 2}{\overset{OLS}{\to}} I = FC + \gamma_1 \Delta K - \gamma_2 \Delta K^2 + \gamma_4 \Delta K^4 + \varepsilon \\ \underset{i \to 2}{\overset{OLS}{\to}} I = FC + \gamma_1 \Delta K - \gamma_2 \Delta K^2 + \gamma_4 \Delta K^4 + \varepsilon \end{array}$					$ \xrightarrow{OLS} \frac{OLS}{(p-p)q} = A + \mathcal{E} $ $ \xrightarrow{OA=4} 4 $					
,	Input	5.5 M TEU per year 9 M TEU per year 14 euro ner TFI1	1 euro per TEU	4.5 M euro per year 5 years 6%	NO	0% 0.4 euro per TEU 0%	Year t Throughput q (M TEU)	2009 5	: : 2015 5.5		Size ΔK Investment Cost I (M TEU p.a.) (M euro)	Fixed $cost? \rightarrow 0$ 50 Smallest project? $\rightarrow 8$ 800 Largest project? $\rightarrow 10$ 1200	Additional data points generated: 4 500	11 15000 12 2000	Annual Throughput q price p price ρ w/o (M TEU per year) (euro) congestion (euro)	$\frac{5.5}{2.5}$ (current) $\frac{14}{2.5}$ $\frac{21}{2.5}$	5 12 17 57	7.5 32 57
	Question	Current annual throughput? Current capacity? Current mice per TEH?	Current operational cost per TEU?	Current annual cost to maintain the capacity? Expected time to build the project? Discount factor to be used?	Public ownership? Yes / No? \hookrightarrow If Yes:	Relative number of government owned shares? Spillover benefits per TEU? Share of total consumer surplus considered?		Throughput of the port over the last years? (Table)				Investment costs? (Table)				Cost of congestion? (Table)		
	Line	<i>I</i>)	2)	3) 4 (4)	ତି ତି			(2				8)				(6		

Table 7.1: Step 1 of the algorithm: input.

7.1. AN RO TOOL BLUEPRINT FOR PORT CAPACITY INVESTMENT DECISIONS

169

Another regression module is required to approximate the parameters of the investment cost as a function of the new capacity that is to be installed. First, the user of the algorithm needs to specify the fixed cost of the investment, which is 50 million euro in the example. Next, the port management needs to determine the range of capacity expansion sizes considered and provide the cost of the smallest and largest project considered. In the example, a project with an annual capacity of 8 million TEU is expected to cost 800 million euro, whereas a project with a capacity of 10 million TEU per year is expected to cost 1.2 billion euro. If available, the port can also specify the cost of intermediate project sizes.

With this information at hand, the algorithm can add three more points to the dataset for the regression of the fourth-order polynomial. The first point is an additional point in the middle between size zero and the smallest project size considered. The cost to be added, is some upward rounded linear extrapolation between the fixed cost of expansion and the cost of the smallest project. For a project with a capacity of 4 million TEU per year, a cost of 500 million euro is added to the dataset, which is (800+50)/2 rounded to the higher hundredfold of million euro, to be able to model the economies of scale for the first part of the investment cost function's domain. The next two points to be added need to account for the boundary beyond which further expansion is very costly. These are costs of two fictional projects that are larger than the largest project considered by the port. In the example, these are projects with a capacity of 11 and 12 million TEU per year. The price increase compared with the project with a capacity of 10 million TEU per year needs to be larger than the increment from the second largest project to the largest project considered.³ That increment was 200 million euro per million TEU per year. Hence the algorithm adds 300 million euro to the cost of a project with a capacity of 10 million TEU per year to yield the fictional cost of a project with a capacity of 11 million TEU per year, i.c., 1.5 billion euro. For the cost of a project with a capacity of 12 million TEU per year, a more than proportionally increased increment is added again, i.e., 500 million euro per million TEU per year, yielding 2 billion euro. Using this dataset, the OLS estimation of Equation (7.2) yields the coefficients FC = 50.2, $\gamma_1 = 174$, $\gamma_2 = -17.2$ and $\gamma_4 = 0.113$. The third-order term is omitted (see Chapter 3).

A final module is required in the first step to determine the value for parameter A, which is the most difficult one to estimate. The user of the algorithm already entered the port's current capacity, throughput and price. Only the price that could have been asked under the absence of congestion still needs to be added to the first observation to estimate A more accurately. For the example, it is assumed to be 7 euro per TEU more than the current price, yielding a ρ of 21 euro per TEU. For a smaller (5 million TEU per year), and a larger annual throughput level (6 million TEU per year), the user of the program needs to provide the same information. If the port is able to provide this information for other throughput levels, it is possible to add more lines to the dataset to estimate A. In the example in Table 7.1 (line 9), information for an annual throughput of 7.5 million TEU per year is also added.

³i.e., from the 8 to the 10 million TEU project.

This final observation provides very useful information about the congestion costs, as it involves a high occupancy rate.

All modules and inputs of the program's first step are summarised in Table 7.1. The output of step 1 is the set of derived parameters in the final column. This set can be passed on to step 2, the actual RO model calculations as described in Chapter 6. The outcome of this step is an optimal investment decision $(X_T^{**}(K_0, \theta), \Delta K^{**}(K_0, \theta))$. Using the input parameters as described in this section, the outcome would be (38.42 euro per TEU, 9.47 M TEU p.a.). Some additional calculations in a third step are however still needed to translate this outcome into meaningful decision variables for the PA's management.

First, it would be interesting for PA managers to dispose of the occupancy rate at which it is optimal to invest, given that the optimal price is charged at any moment in time. As discussed in the previous chapters, the optimal throughput of a profit-maximising port as a function of *X* and K_{tot} , $q^{opt}(X, K_{tot})$, can be obtained from the first-order condition for the port profit. Inputting the optimal investment threshold for *X* returned by the RO calculations and the capacity of the port before expansion (i.c., 9 million TEU per year) in this expression for $q^{opt}(X_T^{**}, 9 \text{ M TEU p.a.})$ yields an optimal throughput level of 6.46 million TEU per year at the moment of investment. Divided by a capacity of 9 million TEU per year, it is found that investment optimally takes place at an occupancy rate of 72%. The optimal annual capacity of the expansion is 9.47 million TEU per year. After expansion, the optimal annual throughput would equal $q^{opt}(38.42 \text{ euro per TEU}, 18.47 \text{ M TEU p.a.}) = 12.89 \text{ million TEU per year. This observation illustrates that new capacity is expected to attract additional throughput, as congestion and occupancy rates will fall. The optimal price can be lowered as well, while profit will be higher. (Zhang, 2007; Wan et al., 2013).$

Another interesting output that the algorithm could present, is the time that will pass before the entire new capacity is expected to be fully utilised. According to Wilmott et al. (1993, p. 371), the following finite expected time can be calculated for a current X_0 to reach X_B , when X follows a GBM with parameters μ and σ and under the condition that $\mu > \sigma^2/2$:

$$t(X_0 \to X_B) = \frac{1}{\mu - \sigma^2/2} \ln(X_B/X_0).$$
(7.5)

Given that q^{opt} (66.93 euro per TEU, 18.47 M TEU p.a.) = 18.47 million TEU per year corresponds to full occupancy, the time for *X* to move from the investment threshold 38.42 euro per year to 66.93 euro per year can be calculated as 113 years. This implies that the utilisation of the total capacity after expansion is expected to remain below 100% for 113 years in the discussed private service port scenario, also due to the possibility to increase prices when demand and congestion increase.

The blueprint of the developed tool is summarised in Figure 7.1. The use of the tool applies to a situation where the port management wants to calculate the optimal time to





exercise the option to invest and at the same time calculate the optimal size of this investment. If the size of the expansion is fixed ex-ante, a few algorithm adaptations are required. An estimation of the investment cost function I is no longer needed and can be replaced by the exact cost of the considered expansion project. As a result, the inner maximisation of the option value maximisation in Section 6.3, equating the marginal value and cost of adding capacity, is omitted. This only leaves the outer maximisation with respect to the investment timing. The outcome, $X_T^*(K_0, \Delta K, \theta)$, can subsequently be transformed into relevant and observable decision variables for the port management in a similar way as described above. On the contrary, when the port management deems it necessary to invest right away in additional capacity, no maximisation of the option value of waiting is needed. Only the equation of marginal value and cost of adding capacity remains to determine the optimal investment size.

7.2 A critical discussion of the advantages and challenges of real options for port capacity investment decisions

Many practitioners consider the RO methodology to be a complex methodology. Calculating the RO investment option is much more difficult than calculating NPV. However, next to additional costs of this approach, there are also considerable potential gains. The critical discussion in this section is hence made up of two parts. First, the potential gains from applying RO to the investment in new port capacity are quantified for the situation elaborated in the previous section. Afterwards, the differences between RO models and traditional NPV methods are discussed and related to the additional difficulties and costs of RO models.

7.2.1 Potential gains of using real options

In order to quantify the potential gains from applying RO, the expected difference in present value between the optimal RO decision and the worst decision that would be accepted by NPV is calculated. In Figure 3.11, it was already made clear that the NPV rule would lead to installing any project with a value V exceeding the investment cost I. Hence, the worst NPV decision possible would be expanding the port at a moment and of a size leading to a net value V - I of 0 euro. However, the NPV approach does not involve optimisation of the timing or the size of the expansion. These decisions are predetermined, in an inflexible approach. Hence, two extreme scenario's exit. In reality, any situation between these two extrema is possible.

A first scenario would involve the port management applying the NPV rule at the moment that would be optimal from the RO perspective as well, i.c., when the reservation price *X* would equal 38.42 euro per TEU. In such a case, the port expansion project size could range between 0.3 and 14.31 million TEU per year to yield a positive net project value. The maximum net present value of the project is found by applying the RO rule. Installing 9.47 million TEU per year at this particular moment would yield a net present value of 1.714 billion euro. However, if 0.3 or 14.31 million TEU per year were installed, the net present value would equal zero. Hence, in the worst case, taking a decision supported by NPV could result in a net present value that is 1.714 billion euro less than what could have been achieved with RO. In the best case, the difference with RO is zero euro.

Second, a similar logic can be derived for the scenario where the size of the investment has been determined ex-ante. For instance, the port could be considering expanding the port infrastructure and superstructure with a capacity of 9.47 million TEU per year, which would be optimal under an RO decision rule. In such a case, if the NPV rule were applied at a moment that the reservation price *X* would equal 25.53 euro per TEU, the net present value of the expansion would again equal zero euro. However, compared with the optimal investment threshold $X_T^{**} = 38.42$ euro per TEU of the RO rule, a net present value of 397 million euro would be lost. This is calculated as $(25.53/38.42)^{\beta_1}(V - I)$. Only if the NPV rule was applied at the moment that the reservation price equals the optimum from an RO perspective, no loss would be encountered, compared with RO.

In conclusion, for the scenario discussed in this chapter, the potential present value gained from applying RO, compared with NPV, lies between zero and 1.714 billion euro. The actual foregone value resulting from applying the NPV rule however depends on the specific scenarios that are considered and analysed under this NPV analysis, as NPV analyses only consider a limited number of distinct scenarios with specific investments sizes and timings. In case the scenario with the optimal RO decision is included in the NPV analysis, it will result in the same present value as the RO analysis. Alternatively, some present value ranging from zero to 1.714 billion euro is foregone, depending on the specific investment sizes and timings considered in the analysed scenarios.

7.2.2 Additional complexities and costs of using real options

A number of RO model elements described in this thesis are not needed exclusively for RO analyses. For an NPV analysis, the costs of congestion are also needed. However, they can be specified more accurately under NPV, using the expected delay *times* the value of time of the goods *times* the throughput quantity. Since no optimisations are needed, the complexity of the functional expression is not bounded by potential complications of further calculations. This improves the realism and accuracy of the point estimates of expected congestion costs of the port expansion project in an NPV approach. Providing an accurate estimate of *A* in an RO approach may prove difficult and may involve hiring external econometricians at a high cost. Similarly, estimates about the other costs, such as operational and capacity holding costs, are needed under both approaches as well. These costs

are less complicated to calculate, as they can be derived from the PA's accounting by the internal accountants.

In order to be able to estimate future revenues from port activities, an estimated relationship between prices and throughput quantities will also be needed for an NPV approach. This involves estimating the most realistic shape for the demand function, and the correct parameters, e.g., for the slope, influencing price elasticities. However, because no optimisations are made under NPV, this relationship between prices and quantities is only needed over a small part of the domain of the demand function. Often, the revenues are estimated directly as the product of the expected price and throughput, which are based on a number of assumptions, but without making use of an explicit demand function. In order to accurately estimate the demand function, the PA could again rely on costly, external econometricians.

To model uncertainty in an RO approach, dynamics need to be added to the model. This is what discerns RO models the most from NPV models. In the models developed in this thesis, cost functions are assumed fixed over time. Oppositely, the demand function has been chosen to evolve over time, following an uncertain pattern. Based on the literature, the assumption has been made that this pattern can be modelled using a GBM with individual parameters for growth and uncertainty. In practice however, it is on the one hand necessary to check whether the GBM is appropriate for the studied scenario. This is complex and costly and often involves external knowledge from econometricians. On the other hand, when the GBM has been proven to hold, it is even more complex to accurately estimate the parameter values for the growth and uncertainty. Also this involves advanced econometrics or insights from costly experts. If however, the GBM hypothesis is falsified in practice, the RO model needs to be adapted as well. Also this is often too complex for the staff of the PA. As a result, the costs of hiring external experts such as scholars and econometricians will be even higher in such a case. In addition, because the growth and uncertainty parameters in the RO model are assumed fixed over time, it is necessary to re-evaluate the model if the market changes significantly before the start of the construction of the port expansion.

Another element that discerns RO models from the NPV approach is the use of an investment cost function. Whereas NPV is often used to evaluate at most a few alternative scenarios of different sizes, potentially in a binomial tree approach, continuous-time continuous-state RO models involve an optimisation of the investment size. For NPV, it is sufficient to dispose of the cost of these individual projects. For RO, a continuous function needs to be elaborated, with a correctly specified absolute value and first-order derivative to allow for an accurate optimisation. Moreover, a boundary beyond which further expansion is impossible needs to be accounted for as well. Estimating this investment cost specification in practice is time intensive and hence costly, since information and costs of more project sizes need to be collected. However, the benefit of being able to optimise the investment size over a continuum of potential port expansion projects can be very high, as was discussed in Section 7.2.1.

Finally, RO models include and attribute value to the options of flexibility on a project (i.e., a flexible investment decision) and in a project (i.e., the option to flexibly adapt operations) (T. Wang & De Neufville, 2005). In this thesis, the focus has been on options on a project. However, in practice, port expansion projects also involve options in the project. For example, the option to deploy the terminal at the new dock in phases can be considered an option in the project. Without necessarily evaluating the net present value of this approach, the port of Rotterdam successfully uses this phased investment option for the development of Maasvlakte 2. The new docks are constructed using the technique of rainbowing sand. Initially, after constructing the first dock, the terminal has only been developed partly. Paving and cranes are only installed on the area closest to the entry of the dock. The remainder can be financed and developed later, when the demand for extra capacity is expected to become present. Such an option of phased investment reduces the lumpy character of port capacity investment decisions. In this way, value can be generated (and calculated), as the time between income and expenses can be reduced. However, accounting for such options is complex and costly, as they need to be embedded in the project on the one hand, and they need to be included in the RO model calculations on the other hand. The latter implies a further complication of the model, but can be based on the RO framework from Chapter 2. At the same time, it involves an interesting approach for further research.

7.3 Conclusion of this chapter

In order to transfer the theoretical insights from the previous chapters to practical calculations for a port's optimal investment decision, this chapter has presented the description of a possible algorithm to simplify the calculations of the optimal investment decision in terms of observable decision variables. This is considered an important step to give complex RO modelling a place in practical decision making. To this end, the port management's input is transformed into RO model parameters, which are subsequently used for the RO calculations. Subsequently, the output of the RO model needs to be transformed into observable decision variables for the port management.

Using an RO approach to find the optimal investment decision potentially implies gaining a lot of money, compared with applying NPV. Depending on the scenarios and specific projects considered, these gains can largely outweigh the costs of hiring external scholars, consultants or econometricians. They could provide the necessary knowledge to further adapt the developed RO models and to estimate the parameters of the RO models. In this way, this external staff can further fine-tune the first approximation of the optimal investment decision returned by the developed tool.

However, the RO calculations are impacted by the assumptions of the RO model. One of the most challenging aspects is estimating the correct shape of the demand function, which

is assumed linear additive in this thesis. In case a linear additive curve is realistic, the external econometricians will need to empirically estimate the slope of the inverse demand curve. However, if a linear additive demand curve is not suitable and sufficiently realistic to make real-life calculations, adaptations to the RO model will be needed as well. They can be based on the RO framework from Chapter 2. It is especially important that the right elasticities of demand are included for the range of considered capacities and throughput levels. Next to the demand function, also correct representations of the specific port's congestion costs and investment costs need to be included in the model and verified by the hired staff. A similar reasoning holds for the process to model demand uncertainty, i.c., the GBM with a growth and uncertainty parameter.

Chapter 8

Conclusions, implications and future research

The conclusion of this thesis is split into three sections, each relating to a type of additions to the existing state-of-the-art in port capacity investment decisions. First, new theoretical insights from this thesis are brought together in Section 8.1. This section relates the derived theoretical knowledge and the developed models back to research sub-questions RQ 2.1 to 6.2, the theoretical part of the main research question. Subsequently, the practical implications and limitations of the methodology developed in this thesis for real-life port capacity investments are discussed in Section 8.2. These are based on the answers to research sub-questions RQ 7.1 and 7.2 from Table 1.3. Avenues for future theoretical and empirical research are discussed in Section 8.3. This further research can augment the applicability and accuracy of the RO models even more in practice.

8.1 Theoretical findings and results

Ports are crucial for an economy as a whole and for the logistics sector. They facilitate economic growth and regional development, as well as trade. Investing in ports is indispensable to provide sufficient capacity to handle the goods. This thesis focuses on the irreversible capacity investment in lumpy port infrastructure and superstructure for container throughput. Because of the uncertainty present in the port, which originates from many different sources, RO models have been identified as an accurate methodology to decide about these investments. Useful characteristics of continuous-time continuous-state RO models with demand uncertainty have been described. Already available port investment models show the importance of considering congestion costs for the users of the port capacity, i.e., the shipping lines, in the analysis. The combination of both types of models resulted in a framework that allows building new RO models and judging the applicability

of already developed RO models in a port setting.

In order to uncover the impact of project- and port-related economic characteristics on port capacity investment decisions, new RO models have been developed in this thesis. The models of Chapters 3-5 deal with the construction of new ports. These models stand out among other models in the literature, as they introduce congestion costs more accurately in port investment models considering demand uncertainty. Moreover, the different decisions of the port authority (PA) and the terminal operating company (TOC) and interport competition have explicitly been taken into account. Next to that, in Chapter 6, new insights into port expansion investments with time to build have been derived.

A crucial new finding of this thesis deals with the impact of waiting-time aversion of port users on the investment decision in a new port. The larger the delay costs for the port's customers, the later the optimal investment timing and the larger the optimal investment size. This finding has been confirmed in different models for new ports. Moreover, a frequently observed RO finding has been proven to hold in different port settings as well. When deciding about the investment in a new port, the higher the demand uncertainty, the more design capacity will be installed and the later the timing.

Moreover, it can be concluded that competition between two new ports drives investment threshold and capacity of the leader down, whereas the follower will invest later in more capacity. The reason is that competition has a negative impact on each port's option value of waiting. This impact of competition is however reduced by investment cost differences between the cost-advantaged port, who invests first, and the follower port. In a model ignoring time to build and phased investment within a landlord port, the investments of the PA and the TOC need to be geared to the same capacity and installed at the same time. This capacity and timing can be the optimum of a similar service port, which is reached in a landlord port after both actors mutually give in from their individual investment optimum. Alternatively, the PA can use its concession fee policy to force the TOC to invest in the first actor's optimum. Additionally, it has been found that ports with a larger share owned publicly invest earlier and in more capacity. This observation is uncommon in RO, where earlier (later) timing often goes hand in hand with less (more) capacity. The reason is that the profits and positive economic effects are needed soon enough, so that the growth of a region is not slowed down. To this end, a good port investment policy should facilitate sufficient capacity provision in order to avoid shortages slowing down economic growth.

In port expansion investment decisions subject to time to build, the impact of the aforementioned factors is slightly different. If congestion involves larger costs for the port users, the PA experiences a larger incentive to anticipate the expansion investment in order to avoid such costly congestion in the current operations. As a result, the optimal expansion timing will be earlier than in a case with lower congestion costs. The impact on the size of the investment is however limited. The same conclusion holds for the impact of increased public ownership, which in some cases could be crucial to beneficially invest in the port project and realise its full potential. The incentive to overcome current capacity shortages slowing down economic growth is even larger than in the case of a new port. As a result, public ownership anticipates the investment decision a lot, compared with a privately owned port. This has a reducing impact on the size of the investment. The impact of the amount of demand uncertainty remains the same in expansion projects, if time to build is neglected. However, time to build can alter the impact of uncertainty. With a lot of time to build and high uncertainty, a further increase in uncertainty may lead to an anticipation of the optimal investment, instead of postponing the investment. The installed capacity will however still be larger. The direct impact of time to build on the investment is more ambiguous. Since time to build reduces the attractiveness of a project, investment will take place later or will be smaller.

8.2 Practical implications and limitations

The previous theoretical conclusions relate to the answers to research sub-questions RQ 2.1 to 6.2 and explicate the impact of changes in the economic environment on the optimal timing and size of the capacity investment. These theoretical insights can prove useful for policy makers to create optimal circumstances for beneficial port investments. However, these results also have implications for port investment decision makers implementing these insights. These are defined in the answers to research sub-questions RQ 7.1 and 7.2. Only by combining all these answers, the main research question of this thesis can be solved and the full added value of this thesis becomes apparent.

Although in practice many port capacity investment decisions are based on the net present value (NPV) rule, this thesis confirms the statements of Dixit & Pindyck (1994). They argued already more than 20 years ago that NPV could lead to very wrong decisions when irreversible, large-scale investment decisions are studied under uncertainty. The reason is that NPV considers such an investment as a now or never investment. The findings of this thesis confirm that the flexibilities of investment size, investment timing and throughput level add value to the project. If more information and more options can be used, this leads to better decision making.

Nevertheless, these options explicitly need to be evaluated in the investment decision to optimise the project's value and take an optimal investment decision, even if uncertainty is low. Information about the uncertainty and the resulting distributions of the port's cash flows need to be incorporated in the investment decision as well. This allows determining the size of the required investment postponement wedge, protecting the port sufficiently from the negative impact of demand uncertainty. To this end, a blueprint of a tool to implement the RO model in real-life port capacity investment decisions has been proposed. This tool allows port managers in practice to approach the optimal timing and size of a real project in terms of observable decision variables, based on a number of collectable inputs and simplified calculations. This highlights the practical added value of this thesis, which is also quantified and compared to the costs of applying RO.

As the results in this thesis unveiled, the RO decision rule leads to a later investment than the NPV rule. At this later moment of investment, higher profits can be generated. If the market then experiences negative growth over a period, sufficient reserves can make up for reduced profits. Through such a protecting wedge, the project remains profitable even if the realised profit is less than the expected value. The net gains from implementing RO can become very large, depending on the specific project. Nevertheless, estimating the exact shape and the parameters of the uncertain demand function, the congestion costs or the investment cost function for RO models might involve high costs for the port management, especially if external econometricians are to be hired. This is the first of two major RO model limitations. Second, the rather theoretical RO models will always involve some simplifications of reality, as they are based on a relatively large set of assumptions. Relaxing these assumptions or considering additional options involves adapting the RO model. Such a complex task is also very costly and may require additional research.

The question is then to be posed if the advantages of RO outweigh the costs, especially compared to the costs and benefits of the NPV approach. The latter approach involves less econometric estimations of functions. In order to calculate NPV, the expected values of the different cash flows, including their expected growth, are the most important input. Using scenario analysis, the sensitivity of a specific project's profitability to economic growth or decline can be calculated too. However, the NPV rule does not enable decision makers to benefit from the project's built-in flexibilities. Using NPV, only one project can be evaluated at a time. Hence, it is impossible to determine the optimal investment size and timing using NPV. It would be too time consuming and hence practically infeasible to calculate the NPV of all investment sizes and timings of the capacity project under consideration and then invest in the most profitable option. As a result, a lot of money could be lost by installing a wedge against uncertainty that is profitable according to NPV, but that is too large or too small according to RO. By consequence, deciding on the investment through RO allows port managers to take the most beneficial decision and find an optimal point in the trade-off between costly over- and undercapacity. The high potential gains in many cases can make up for the additional costs and complexities of RO models.

Since previously developed RO models were not able to accurately account for portspecific characteristics, such as congestion costs, developing the port models in this thesis has been an important step towards the implementation of RO in real-life port capacity investment projects. Although RO models are theoretical in nature and based on a large number of assumptions, making them as realistic as possible is required to make those models applicable to real-life situations and improve port investment decision making. Even though the future is uncertain, well-developed RO models allow making the best ex-ante decision possible while taking into account the embedded project flexibility and the information available at that moment of decision. A simple analogy allows clarifying this point. Suppose there are two decision options when drunk after a party: drive yourself or call a taxi. The uncertainty is situated in the probability of having an accident. RO allows evaluating all options at once and taking the decision that is ex-ante perceived as the best. This would be to call a taxi. Even if this taxi then got involved in an accident while you were in it and you got injured, this would still have been the best decision.

8.3 Suggestions for future port-related real options research

The RO models developed in this thesis explicate the impact of economic, port and project characteristics on optimal port capacity investment decisions under uncertainty. Moreover, the added value of real options for investment decision makers has been made explicit, in addition to a potential, simplified application. Nevertheless, some limitations to the practical implications have been identified. Overcoming these limitations while further improving model realism and applicability gives rise to some future port-related RO research.

Considering the decision of one port operator (the TOC or the PA), opens up some viable ways for future research. First of all, asymmetrical information between parties may exist, or some parties might not dispose of perfect information (Delaney, 2019). This has not been included in this thesis. The impact of a relaxation of this assumption would be an interesting topic of a new analysis. Moreover, to model the difference in bargaining power of actors, Schneider et al. (2010) propose among other things a Nash bargaining game. In cases where a winner-take-all approach is not appropriate, this could be an interesting approach to define the concession fees optimising the actors' discounted project values in a specific situation. Subsequently, the impact of intra-port competition following the presence of multiple competing operators in a non-cooperative setting, as well as the impact of the actual or potential vertical integration between TOCs and other logistics chain actors, such as shipping lines or hinterland companies, would be interesting to analyse (De Borger & De Bruyne, 2011; Huisman & Kort, 2015).

In the analysis of inter-port competition, the fixed investment cost provided the basis for the cost advantage of the leader port. However, studying a different capacity holding cost per port could be an interesting approach to analyse the impact of different public private partnerships (PPP) on the different costs and the final investment decision in competing ports. Also other differences between the two ports are relevant, such as different objectives and ownership shares. Future research might prove whether the port with a higher public ownership share would indeed invest earlier in more capacity than a competitor with a lower public ownership share. The reason is that the project would be more attractive for the former because of social welfare being considered to a larger extent. Moreover, port investment decisions are influenced by the government's port development doctrine. Where the duopoly model accounts for the properties of the Anglo-Saxon and European (Continental) doctrines with individual ports, respectively private and at least partly public ports, the results may well be different under the Asian doctrine (see e.g., Bennathan & Waters (1979) and Lee & Flynn (2011)). Considering the centrally planned development of multiple ports and potential cross-subsidisation may alter the findings. Hence, this would be an interesting extension of the results found in this thesis, for example to allow for comparison with Chen et al. (2017).

In this thesis, the time to build has been assumed to be known ex-ante. It could be interesting to study the impact of uncertain time to build on the timing and the size of the investment decision. For this, the time to build could be drawn from a simple random distribution, such as the uniform distribution. If the calculations are still feasible, also a Gaussian or a right-skewed distribution such as the Chi-squared distribution could be used. Moreover, the expected value of the time to build could also be related to the size of the investment.

From an empirical viewpoint, well-developed models allow econometricians to calculate the parameters empirically and make better real-life port investment decisions, using the models at hand. However for specific applications, implementing more accurate functional representations of demand and cost functions would be a beneficial step to increase the applicability of the RO approach. This would limit some of the simplifying assumptions underlying the developed models.

Additionally, the developed models need further extensions to improve their realism and applicability to a port setting. An important extension would be incorporating the option of phased investment, e.g., starting from the port expansion model of Chapter 6. In that case, the dock or the terminal is not to be installed at once, as is assumed in the developed models of this thesis. Section 2.4.4 already discussed the modelling approach of phased investment, which is slightly different from the approach used in this thesis. Considering phased investment in future research would be beneficial, since previous research already demonstrated the significant value of this option. Stepwise investment increases the project value and the amount of capacity installed if final capacity is flexible. Moreover, the first investment often takes place earlier (Chronopoulos et al., 2017; Kort et al., 2010). However, what remains unclear is the interaction between this phased investment option and the impact of congestion costs on the investment decision(s).

Once this model is developed, the insights and other model characteristics from this thesis can be added to this phased capacity expansion model including uncertainty, congestion costs and time to build. First, a competing port could be added to the model again, as was done in Chapter 5 to study the impact of competition on the phased investment decision in a duopoly. Once the investment decision of an individual port is obtained, the approach of Chapter 4 can again be used to distinguish between the decision of the PA and

the TOC and to calculate a concession fee that satisfies the pursued objectives of the PA.

In order to add even more realism to the RO model and limit model simplifications as much as possible, it would be beneficial to investigate in a follow-up study the option of two subsequent expansions. In 1999, when the Port of Antwerp considered the investment in the Deurganckdok, it already contemplated the construction of a second tidal dock, Saeft-inghedok. Adding such a second expansion option would further reduce the myopic character of the developed RO models, which consider only one investment.

Moreover, additional sources of uncertainty could be accounted for in the models, such as a variable discount rate, ex-ante uncertainty about the time to build and cost uncertainty, next to the already included demand uncertainty. Also disaster, political, technological and legal uncertainty could be included in the analysis. However, information about the interactions between the different sources of uncertainty would then be required. In addition, different processes would be needed to model these different sources of uncertainty. It is important that these processes' accuracies in representing realistic situations are tested and validated. Nevertheless, these additions would substantially complicate the mathematical calculations of the RO model, as well as the empirical estimations needed. In any case, a good balance between model realism and (exact or approximate) solubility needs to be maintained in future research. Moreover, different port investment types might have to be combined in one decision making process model.

The newly developed models can be used in a subsequent research phase to enhance the tool proposed in Chapter 7. This will allow increasing realism and taking better and more realistic decisions in two competing ports such as Antwerp and Rotterdam in the Hamburg - Le Havre range. However, due to the increased model complexity, also the complexity of the estimation techniques might increase. Using case studies, for example on finished projects, the correctness of the developed models can be assessed and further improvements can be made.

Finally, the approaches and models developed in this thesis offer useful insights for other sectors as well. For example, the available capacity investment decision making models for airports do not explicitly account for the impact of congestion. The approach and insights presented in this thesis can be used to incorporate this characteristic, that is also present in airports, in airport capacity RO models (see e.g., Balliauw & Onghena (2018)). Of course, those characteristics distinguishing airports from ports need to be taken into account as well, before transferring any model components or findings. Next to airports, also other industries might benefit from this research. For instance, the insight of adding a fourth-order investment cost function could be useful for other investment decisions that are limited in space as well. As an example, one could think of the investment in warehouses or football stadiums. However, here again, accounting for the relevant industry characteristics is a prerequisite for finding valid and reliable optimal investment decisions under uncertainty.
Bibliography

- Abramowitz, M., & Stegun, I. A. (1972). *Handbook of mathematical functions with formulas, graphs, and mathematical tables* (10th ed.). New York, NJ: Dover Publications.
- Acciaro, M. (2014a). Real option analysis for environmental compliance: LNG and emission control areas. *Transportation Research Part D: Transport and Environment*, *28*, 41–50.
- Acciaro, M. (2014b). A real option application to investment in low-sulphur maritime transport. *International Journal of Shipping and Transport Logistics*, 6(2), 189–212.
- Aguerrevere, F. L. (2003). Equilibrium investment strategies and output price behavior: A real-options approach. *The Review of Financial Studies*, *16*(4), 1239–1272.
- Aguerrevere, F. L. (2009). Real options, product market competition, and asset returns. *The Journal of Finance*, 64(2), 957–983.
- Alderton, P. (2008). *Lloyd's practical shipping guides: Port management and operations*. London, England: Informa.
- Anda, J., Golub, A., & Strukova, E. (2009). Economics of climate change under uncertainty: Benefits of flexibility. *Energy Policy*, 37(4), 1345–1355.
- Aronietis, R., Monteiro, F., Van de Voorde, E., & Vanelslander, T. (2010, July). Concessioning in seaports: changing practices, changing market power. Paper presented at the 12th World Conference on Transport Research Society. Lisbon, Portugal.
- ASCE. (2019). Ports (2017 Infrastructure Report Card). Retrieved from https://www .infrastructurereportcard.org/cat-item/ports/
- Ashuri, B., Lu, J., & Kashani, H. (2011). A real options framework to evaluate investments in toll road projects delivered under the two-phase development strategy. *Built environment project and asset management*, *1*(1), 14–31.
- Asteris, M., Collins, A., & Jones, D. F. (2012). Container port infrastructure in north-west Europe: Policy-level modeling. *Journal of Policy Modeling*, *34*(2), 312–324.
- Azevedo, A., & Paxson, D. (2014). Developing real option game models. *European Journal* of Operational Research, 237(3), 909–920.

- Balliauw, M. (2017). An analysis of entry and exit decisions in shipping markets under uncertainty. *International Journal of Shipping and Transport Logistics*, 9(5), 527–539.
- Balliauw, M., & Onghena, E. (2018, July). *Expanding airport capacity of cities under uncertainty: Strategies to mitigate congestion*. Paper presented at the Air Transport Research Society (ATRS) Conference 2019. Amsterdam, The Netherlands.
- Bar-Ilan, A., & Strange, W. C. (1996). Investment lags. *The American Economic Review*, 86(3), 610–622.
- Bar-Ilan, A., & Strange, W. C. (1999). The timing and intensity of investment. *Journal of Macroeconomics*, *21*(1), 57–77.
- Barton, H., & Turnbull, P. (2002). Labour regulation and competitive performance in the port transport industry: The changing fortunes of three major european seaports. *European Journal of Industrial Relations*, 8(2), 133–156.
- Benacchio, M., Ferrari, C., Haralambides, H. E., & Musso, E. (2000). On the economic impact of ports: local vs. national costs and benefits. In *Forum of shipping and logistics, special interest group on maritime transport and ports international workshop*. Genoa, Italy. Retrieved from http://www.informare.it/news/forum/2000/sig2/genovade.asp
- Benacchio, M., & Musso, E. (2001). Ports and economic impact: main changes, assessment approaches and distribution disequilibrium. *European Transport / Trasporti Europei*, 47(7), 25–36.
- Bendall, H. B., & Stent, A. F. (2003). Investment strategies in market uncertainty. *Maritime Policy & Management*, 30(4), 293–303.
- Bendall, H. B., & Stent, A. F. (2005). Ship investment under uncertainty: Valuing a real option on the maximum of several strategies. *Maritime Economics & Logistics*, 7(1), 19–35.
- Bendall, H. B., & Stent, A. F. (2007). Maritime investment strategies with a portfolio of real options. *Maritime Policy & Management*, 34(5), 441–452.
- Bennathan, E., & Waters, A. A. (1979). Port pricing and investment policy for developing countries. Oxford, England: Oxford University Press.
- Bergantino, A., & Veenstra, A. (2002). Interconnection and co-ordination: An application of network theory to liner shipping. *International Journal of Maritime Economics*, *4*(3), 231–248.
- Bernanke, B. S. (1983). Irreversibility, uncertainty, and cyclical investment. *The Quarterly Journal of Economics*, 98(1), 85–106.
- Bichou, K., & Gray, R. (2005). A critical review of conventional terminology for classifying seaports. *Transportation Research Part A: Policy and Practice*, 39(1), 75–92.

- Bjerksund, P., & Ekern, S. (1995). Contingent claims evaluation of mean-reverting cash flows in shipping. In *Real options in capital investment: Models, strategies, and applications* (pp. 207–219). Westport, CT: Preager.
- Blauwens, G. (1988). *Welvaartseconomie en kosten-batenanalyse [Welfare economics and cost benefit analysis]*. Deurne, Belgium: MIM.
- Blauwens, G., De Baere, P., & Van de Voorde, E. (2016). *Transport economics* (6th ed.). Antwerp, Belgium: De Boeck.
- Blomme, J. (2014). Value creation in the port of Antwerp: an investigation of sector and cargo-related creation of value and employment. In T. Vanelslander & C. Sys (Eds.), *Port business. market challenges and management actions* (pp. 99–118). Antwerp, Belgium: University Press Antwerp.
- Blonigen, B. A., & Wilson, W. W. (2008). Port efficiency and trade flows. *Review of International Economics*, 16(1), 21–36.
- Brander, J. A., & Zhang, A. (1990). Market conduct in the airline industry: An empirical investigation. *The RAND Journal of Economics*, *21*(4), 567–583.
- Bresnahan, T. F. (1989). Empirical studies of industries with market power. In
 R. Schmalensee & R. Willig (Eds.), *Handbook of industrial organization* (Vol. 2, p. 1011-1057). Amsterdam, The Netherlands: North-Holland.
- Brooks, M. R., & Cullinane, K. (2007). Governance models defined. In M. R. Brooks & K. Cullinane (Eds.), *Devolution, port governance and port performance* (p. 405-435). London, England: Elsevier.
- Cairns, M. (2013). *Mixed ownership works for port*. Retrieved from https://www .nzherald.co.nz/business/news/article.cfm?c_id=3&objectid=10865991
- Cassimon, D., Engelen, P.-J., Meersman, H. M., & Van Wouwe, M. (2002). *Investment, uncertainty and irreversibility: evidence from Belgian accounting data* (Working Paper No. 23). Brussels, Belgium: National Bank of Belgium. Retrieved from http://ideas.repec .org/p/nbb/reswpp/200205-4.html
- Cebr. (2017). The economic contribution of the UK ports industry (Report for Maritime UK). London, UK. Retrieved from https://www.maritimeuk.org/documents/187/ Cebr_Ports_report_finalised.pdf
- Centraal Planbureau. (2001). Welvaartseffecten van Maasvlakte 2; kosten-batenanalyse van uitbreiding van de Rotterdamse haven door landaanwinning [Welfare effects of Maasvlakte 2; cost benefit analysis of an expansion of the port of Rotterdam through land reclamation] (Tech. Rep.). The Hague, The Netherlands. Retrieved from https:// www.cpb.nl/sites/default/files/publicaties/download/bijz32.pdf

- Chang, S. (1978). Production function, productivities, and capacity utilization of the Port of Mobile. *Maritime Policy & Management*, 5(4), 297–305.
- Chang, Y.-T., Tongzon, J., Luo, M., & Lee, P. T.-W. (2012). Estimation of optimal handling capacity of a container port: an economic approach. *Transport Reviews*, *32*(2), 241–258.
- Chen, H.-C., Lee, P. T.-W., Liu, S.-M., & Lee, T.-C. (2017). Governments' sequential facilities investments and ports' pricing under service differentiation and demand uncertainty. *International Journal of Shipping and Transport Logistics*, 9(4), 417–448.
- Chen, H.-C., & Liu, S.-M. (2016). Should ports expand their facilities under congestion and uncertainty? *Transportation Research Part B: Methodological*, *85*, 109–131.
- Chevalier-Roignant, B., Flath, C. M., Huchzermeier, A., & Trigeorgis, L. (2011). Strategic investment under uncertainty: A synthesis. *European Journal of Operational Research*, *215*(3), 639–650.
- Chow, J. Y. J., & Regan, A. C. (2010). Real option pricing of network design investments. *Transportation Science*, *45*(1), 50–63.
- Chronopoulos, M., Hagspiel, V., & Fleten, S.-E. (2017). Stepwise investment and capacity sizing under uncertainty. *OR Spectrum*, 39(2), 447–472.
- CONTARGO. (2018). Waiting times in the seaports of Antwerp and Rotterdam (Congestion). Retrieved February 25, 2019, from https://www.contargo.net/en/news/2018-03 -06_wartzeiten_in_dern_seeh_/
- Cooper, A., & Smith, P. (2005). *The economic catalytic effects of air transport in Europe* (EEC Technical/Scientific Report No. 2005-025). Oxford, England: Eurocontrol. Retrieved from https://www.eurocontrol.int/eec/public/standard_page/DOC_Report _2005_025.html
- Coppens, F., Lagneaux, F., Meersman, H. M., Sellekaerts, N., Van de Voorde, E., van Gastel, G., ... Verhetsel, A. (2007). *Economic impact of port activity: a disaggregate analysis - the case of Antwerp* (Working Paper No. 110). Brussels, Belgium: National Bank of Belgium. Retrieved from https://papers.ssrn.com/sol3/papers.cfm?abstract _id=1687569
- Corchón, L. C., & Marini, M. A. (2018). *Handbook of game theory and industrial organization, volume II: Applications.* Cheltenham, England: Edward Elgar Publishing.
- Cruz, C. O., & Marques, R. C. (2012). Risk-sharing in seaport terminal concessions. *Transport Reviews*, 32(4), 455–471.
- Cucchiella, F., & Gastaldi, M. (2006). Risk management in supply chain: a real option approach. *Journal of Manufacturing Technology Management*, *17*(6), 700–720.
- Cullinane, K., & Bergqvist, R. (2014). Emission control areas and their impact on maritime transport. *Transportation Research Part D: Transport and Environment, 28*, 1–5.

- Cullinane, K., Teng, Y., & Wang, T.-F. (2005). Port competition between Shanghai and Ningbo. *Maritime Policy & Management*, *32*(4), 331–346.
- Dangl, T. (1999). Investment and capacity choice under uncertain demand. *European Journal of Operational Research*, *117*(3), 415–428.
- Davis, H. C. (1983). Regional port impact studies: a critique and suggested methodology. *Transportation Journal, 23*, 61–71.
- De Borger, B., & De Bruyne, D. (2011). Port activities, hinterland congestion, and optimal government policies: The role of vertical integration in logistic operations. *Journal of Transport Economics and Policy*, 45(2), 247–275.
- De Borger, B., Proost, S., & Van Dender, K. (2008). Private port pricing and public investment in port and hinterland capacity. *Journal of Transport Economics and Policy*, *42*(3), 527–561.
- De Borger, B., & Van Dender, K. (2006). Prices, capacities and service levels in a congestible bertrand duopoly. *Journal of Urban Economics*, 60(2), 264–283.
- Degroote, G. (2015). *De kostenstructuur van een haven [The cost structure of a port]* (Master's thesis, University of Antwerp, Antwerp, Belgium). Retrieved from https://repository.uantwerpen.be/desktop/irua
- De Jong, G. (2007). Value of freight travel-time savings. In D. Hensher & K. Button (Eds.), *Handbook of transport modelling* (2nd ed., Vol. 1, pp. 649–663). Bingley, England: Emerald Group Publishing Limited.
- Dekker, S. (2005). Port investment: towards an integrated planning of port capacity (Doctoral dissertation, TU Delft, Delft, The Netherlands). Retrieved from http://resolver .tudelft.nl/uuid:7224341e-cd4d-4d4e-a5ba-f262ee089305
- Delaney, L. (2019). Symmetric equilibrium strategies in game theoretic real option models with incomplete information. *Economics Letters*, *174*, 42–47.
- De Langen, P. W. (2007). Port competition and selection in contestable hinterlands; the case of Austria. *European Journal of Transport and Infrastructure Research*, 7(1), 1–14.
- De Langen, P. W., Turró, M., Fontanet, M., & Caballé, J. (2018). The infrastructure investment needs and financing challenge of European ports (Port Investments Study 2018). Brussels, Belgium: European Seaports Organisation (ESPO). Retrieved from https://www.espo .be/media/Port%20Investment%20Study%202018_FINAL_1.pdf
- De Langen, P. W., Van Meijeren, J., & Tavasszy, L. A. (2012). Combining models and commodity chain research for making long-term projections of port throughput: an application to the Hamburg-Le Havre range. *European Journal of Transport & Infrastructure Research*, 12(3), 310–331.

- de Neufville, R. (2003). Real options: Dealing with uncertainty in systems planning and design. *Integrated Assessment*, *4*(1), 26–34.
- de Neufville, R. (2006). Real options by spreadsheet: Parking garage case example. *Journal* of *Infrastructure Systems*, *12*(2), 107–111.
- de Neufville, R. (2016). Airport systems planning and design. In L. Budd & S. Ison (Eds.), *Air transport management: An international perspective* (pp. 61–78). Abingdon, England: Routledge.
- de Weille, J., & Ray, A. (1974). The optimum port capacity. *Journal of Transport Economics and Policy*, 8(3), 244–259.
- Dikos, G. (2008). Real options econometrics for aggregate tanker investment decisions. *International Journal of Ocean Systems Management*, 1(1), 31–44.
- Dixit, A. K., & Pindyck, R. S. (1994). *Investment under uncertainty*. Princeton, NJ: Princeton University Press.
- ESCAP. (2002). Commercial development of regional ports as logistics centres (Transport and Tourism Division (TTD), ST/ESCAP/2194). Thailand: United Nations. Retrieved from https://www.unescap.org/sites/default/files/pub_2194_ch2.pdf
- Estache, A. (2001). Privatization and regulation of transport infrastructure in the 1990s. *The World Bank Research Observer, 16*(1), 85–107.
- Estache, A., & Trujillo, L. (2009). Global economic changes and the future of port authorities.
 In H. Meersman, E. Van de Voorde, & T. Vanelslander (Eds.), *Future challenges for the port* and shipping sector (pp. 69–87). London, England: Informa Law from Routledge.
- Farrell, S. (2012). The ownership and management structure of container terminal concessions. *Maritime Policy & Management*, 39(1), 7–26.
- Flemish Government. (2019). Extra container capacity in Antwerp: approach. Retrieved March 15, 2019, from https://extracontainercapaciteitantwerpen .login.kanooh.be/aanpak
- Genco, M., Sirtori, E., & Vignett, S. (2012). Ex-post evaluation of investment projects cofinanced by the European Regional Development Fund (ERDF) or Cohesion Fund (CF) in the period 1994-1999: The Port of Gioia Tauro. (Working Paper). Milan, Italy: CSIL for the European Commision. Retrieved from http://ec.europa.eu/regional_policy/ sources/docgener/evaluation/pdf/projects/port_gioia_tauro.pdf
- Gillen, D. (2011). The evolution of airport ownership and governance. *Journal of Air Transport Management*, *17*(1), 3–13.
- Giuliano, G. (2007). The changing landscape of transportation decision making. *Transportation Research Record: Journal of the Transportation Research Board*, 2036(1), 5–12.

- Grenadier, S. R. (2002). Option exercise games: An application to the equilibrium investment strategies of firms. *Review of Financial Studies*, *15*(3), 691–721.
- Gueli, E., Ringoot, P., & Van Kerckhoven, A. (2019). The economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels Report 2017 (Working Paper No. 368). Brussels, Belgium: National Bank of Belgium. Retrieved from https://www.nbb.be/doc/ts/publications/wp/wp368en.pdf
- Guthrie, G. (2012). Uncertainty and the trade-off between scale and flexibility in investment. *Journal of Economic Dynamics and Control*, 36(11), 1718–1728.
- Hagspiel, V., Huisman, K. J., & Kort, P. M. (2016). Volume flexibility and capacity investment under demand uncertainty. *International Journal of Production Economics*, *178*, 95–108.
- Haralambides, H. E. (2002). Competition, excess capacity, and the pricing of port infrastructure. *International Journal of Maritime Economics*, 4(4), 323–347.
- Harrald, J. R., Stephens, H. W., & vanDorp, J. R. (2004). A framework for sustainable port security. *Journal of Homeland Security and Emergency Management*, 1(2), Article 12.
- Hartzmark, S. M. (2016). Economic uncertainty and interest rates. *The Review of Asset Pricing Studies*, 6(2), 179–220.
- Heaver, T. (2011). Co-ordination in multi-actor logistics operations: challenges at the port interface. In P. Hall, R. J. McCalla, C. Comtois, & B. Slack (Eds.), *Integrating seaports and trade corridors* (pp. 155–170). London, England: Ashgate Publishing, Ltd.
- Heaver, T., Meersman, H., Moglia, F., & Van de Voorde, E. (2000). Do mergers and alliances influence European shipping and port competition? *Maritime Policy & Management*, *27*(4), 363–373.
- Heaver, T., Meersman, H., & Van de Voorde, E. (2001). Co-operation and competition in international container transport: strategies for ports. *Maritime Policy & Management*, 28(3), 293–305.
- Herder, P. M., de Joode, J., Ligtvoet, A., Schenk, S., & Taneja, P. (2011). Buying real options Valuing uncertainty in infrastructure planning. *Futures*, *43*(9), 961–969.
- Hu, Q., & Zhang, A. (2015). Real option analysis of aircraft acquisition: A case study. *Journal* of *Air Transport Management*, 46, 19–29.
- Huberts, N. F., Dawid, H., Huisman, K. J., & Kort, P. M. (2019). Entry deterrence by timing rather than overinvestment in a strategic real options framework. *European Journal of Operational Research*, *274*(1), 165–185.
- Huberts, N. F., Huisman, K. J., Kort, P. M., & Lavrutich, M. N. (2015). Capacity choice in (strategic) real options models: A survey. *Dynamic Games and Applications*, 5(4), 424– 439.

- Huisman, K. J., & Kort, P. M. (2015). Strategic capacity investment under uncertainty. *The RAND Journal of Economics*, 46(2), 376–408.
- Huisman, K. J., Kort, P. M., & Plasmans, J. E. (2013). Investment in high-tech industries: An example from the LCD industry. *Real options, Ambiguity, Risk and Insurance, edited by A. Bensoussan, S. Peng, and J. Sung, Studies in Probability, Optimization and Statistics Series, 5, 20–32.*
- Hull, J. (2012). Options, futures, and other derivatives (8th ed.). Boston, MA: Prentice Hall.
- Ishii, M., Lee, P. T.-W., Tezuka, K., & Chang, Y.-T. (2013). A game theoretical analysis of port competition. *Transportation Research Part E: Logistics and Transportation Review*, 49(1), 92–106.
- Jacks, D. S., & Pendakur, K. (2010). Global trade and the maritime transport revolution. *The Review of Economics and Statistics*, 92(4), 745–755.
- Jacob, A. (2013). The challenges of container terminal development (Powerpoint presentation). Long Beach, CA: Moffatt & Nichol. Retrieved from http://aapa.files.cms-plus.com/SeminarPresentations/2013Seminars/ 13TerminalTraining/Jacob%2C%20Ashebir.pdf
- Jacobs, W. (2007). Port competition between Los Angeles and Long Beach: an institutional analysis. *Tijdschrift voor economische en sociale geografie*, *98*(3), 360–372.
- Jan De Nul. (2012). Contract for capital and maintenance dredging in the port of Bahia Blanca (Argentina) awarded to Jan De Nul Group. Retrieved January 3, 2018, from http://www.jandenul.com/en/pressroom/press-releases/contract-for -capital-and-maintenance-dredging-in-the-port-of-bahia-blanca
- Jansson, J. O., & Shneerson, D. (1982). Port economics. Cambridge, MA: The MIT Press.
- Jenné, M. (2017). *Terminal finance* (Powerpoint Presentation Port Economics and Business Course at C-MAT). Antwerp, Belgium: University of Antwerp.
- Jensen, J. L. W. V. (1906). Sur les fonctions convexes et les inégalités entre les valeurs moyennes [on convex functions and the inequalities between the average values]. *Acta mathematica*, *30*(1), 175–193.
- Jia, H., Daae Lampe, O., Solteszova, V., & Strandenes, S. P. (2017). Norwegian port connectivity and its policy implications. *Maritime Policy & Management*, 44(8), 956–966.
- Jiang, C., Wan, Y., & Zhang, A. (2017). Internalization of port congestion: strategic effect behind shipping line delays and implications for terminal charges and investment. *Maritime Policy & Management*, 44(1), 112–130.
- Juan, C., Olmos, F., Pérez, J. C., & Casasus, T. (2002, July). *Optimal investment management of harbour infrastructures. A real option viewpoint.* Paper presented at the 6th Annual

Real Options Conference. Paphos, Cyprus. Retrieved from http://www.realoptions.org/papers2002/juan_realoptions.pdf

- Juhel, M., Kopicki, R., Kruk, C., & Julian, B. (2007). *Port reform toolkit* (2nd ed.). Washington, DC: World Bank & PPIAF.
- Kakimoto, R., & Seneviratne, P. (2000a). Financial risk of port infrastructure development. *Journal of Waterway, Port, Coastal, and Ocean Engineering, 126*(6), 281–287.
- Kakimoto, R., & Seneviratne, P. N. (2000b). Investment risk analysis in port infrastructure appraisal. *Journal of infrastructure systems*, 6(4), 123–129.
- Kamoto, S., & Okawa, M. (2014). Market entry, capacity choice, and product differentiation in duopolistic competition under uncertainty. *Managerial and Decision Economics*, 35(8), 503–522.
- Kapros, S. (2014). Territorial aspects of actors' cooperation strategies in maritime logistics chains and inland distribution networks. In *Port business. market challenges and management actions* (pp. 143–163). Antwerp, Belgium: ASP Publishers.
- Kauppila, J., Martinez, L., Merk, O., & Benezec, V. (2016). Capacity to grow: Transport infrastructure needs for future trade growth (Tech. Rep.). Paris, France: International Transport Forum (ITF). Retrieved from https://www.itf-oecd.org/sites/default/files/ docs/future-growth-transport-infrastructure.pdf
- Keskinen, A., Annala, I., & Miedema, P. (2017). Maintenance of automated terminals (Tech. Rep.). Helsinki, Finland: Kalmar Global. Retrieved from https://www.kalmarglobal.com/globalassets/services/kalmar-care/ kalmar-maintenance-of-automated-terminals.pdf
- Kia, M., Shayan, E., & Ghotb, F. (2000). The importance of information technology in port terminal operations. *International Journal of Physical Distribution & Logistics Management*, 30(3/4), 331–344.
- Kia, M., Shayan, E., & Ghotb, F. (2002). Investigation of port capacity under a new approach by computer simulation. *Computers & Industrial Engineering*, *42*(2), 533–540.
- Kim, H. J., Weiss, M. B., & Morel, B. (2009). Real options and technology management: Assessing technology migration options in wireless industry. *Telematics and Informatics*, 26(2), 180–192.
- Knight, F. H. (1921). *Risk, uncertainty and profit*. Boston and New York: Houghton Mifflin Company.
- Koetse, M. J., & Rietveld, P. (2012). Adaptation to climate change in the transport sector. *Transport Reviews*, *32*(3), 267–286.
- Kort, P. M., Murto, P., & Pawlina, G. (2010). Uncertainty and stepwise investment. *European Journal of Operational Research*, 202(1), 196–203.

- Kreps, D. M., & Scheinkman, J. A. (1983). Quantity precommitment and bertrand competition yield cournot outcomes. *The Bell Journal of Economics*, *14*(2), 326–337.
- Kydland, F. E., & Prescott, E. C. (1982). Time to build and aggregate fluctuations. *Econometrica: Journal of the Econometric Society*, 50(6), 1345–1370.
- Lacoste, R., & Douet, M. (2013). The adaptation of the landlord port model to France's major seaports: a critical analysis of local solutions. *Maritime Policy & Management*, 40(1), 27–47.
- Langlois, R. N., & Cosgel, M. M. (1993). Frank Knight on risk, uncertainty, and the firm: a new interpretation. *Economic Inquiry*, *31*(3), 456–465.
- Lave, L. B., & DeSalvo, J. S. (1968). Congestion, tolls, and the economic capacity of a waterway. *Journal of Political Economy*, 76(3), 375–391.
- Leachman, R. C., & Jula, P. (2011). Congestion analysis of waterborne, containerized imports from Asia to the United States. *Transportation Research Part E: Logistics and Transportation Review*, 47(6), 992–1004.
- Lee, P. T.-W., & Flynn, M. (2011). Charting a new paradigm of container hub port development policy: The Asian doctrine. *Transport Reviews*, *31*(6), 791–806.
- Li, S., & Cai, H. (2017). Government incentive impacts on private investment behaviors under demand uncertainty. *Transportation Research Part E: Logistics and Transportation Review*, *101*, 115–129.
- Lindsey, R., & De Palma, A. (2014). Cost recovery from congestion tolls with long-run uncertainty. *Economics of Transportation*, *3*(2), 119–132.
- Luck, M. (2017, June, 30). Port of Kalama awards \$1.1 million dredging contract. Retrieved January 3, 2018, from http://tdn.com/news/local/port-of-kalama -awards-million-dredging-contract/article_284b8844-4d1e-5d6b-83f3 -41f6afa86d34.html
- Luo, M., Liu, L., & Gao, F. (2010). Bertrand competition with capacity expansion. *Transportation Research Record: Journal of the Transportation Research Board*, *2166*, 74–81.
- Luo, M., Liu, L., & Gao, F. (2012). Post-entry container port capacity expansion. *Transportation Research Part B: Methodological*, 46(1), 120–138.
- Macário, R. (2014). Port capacity benchmarking: leading or lagging? In T. Vanelslander & C. Sys (Eds.), *Port business. market challenges and management actions* (pp. 61–78). Antwerp, Belgium: University Press Antwerp.
- Mackie, P., & Preston, J. (1998). Twenty-one sources of error and bias in transport project appraisal. *Transport Policy*, 5(1), 1–7.

- Majd, S., & Pindyck, R. S. (1987). Time to build, option value, and investment decisions. *Journal of Financial Economics*, *18*(1), 7–27.
- Maloni, M. J., & Jackson, E. C. (2005). North American container port capacity: A literature review. *Transportation Journal*, 44(2), 16–36.
- Marathe, R. R., & Ryan, S. M. (2005). On the validity of the geometric Brownian motion assumption. *The Engineering Economist*, *50*(2), 159–192.
- Marmer, V., & Slade, M. E. (2018). Investment and uncertainty with time to build: Evidence from entry into U.S. copper mining. *Journal of Economic Dynamics and Control*, 95, 233–254.
- Martin Associates. (2015). The 2014 national economic impact of the U.S. coastal port system (Tech. Rep.). Lancaster, PA: American Association of Port Authorities. Retrieved from http://aapa.files.cms-plus.com/US%20Coastal%20Ports%20Impact% 20Report%202014%20methodology%20-%20Martin%20Associates%204-21-2015 .doc
- Meersman, H. (2005). Port investments in an uncertain environment. *Research in Transportation Economics*, *13*, 279–298.
- Meersman, H., Steenssens, C., & Van de Voorde, E. (1997). Container throughput, port capacity and investment (SESO Working Paper No. 1997353). Antwerp, Belgium: University of Antwerp. Retrieved from http://ideas.repec.org/p/ant/wpaper/1997353 .html
- Meersman, H., Strandenes, S. P., & Van de Voorde, E. (2015). Port pricing. In C. Nash (Ed.), *Handbook of research methods and applications in transport economics and policy* (pp. 329–342). Cheltenham, England: Edward Elgar Publishing.
- Meersman, H., & Van de Voorde, E. (2002). Port management, operation and competition: a focus on North Europe. In C. T. Grammenos (Ed.), *The handbook of maritime economics and business* (pp. 765–781). London, England: Lloyd's List.
- Meersman, H., & Van de Voorde, E. (2014a). Port capacity extension A trade-off between public investment and ship owner time losses. In R. Funck & W. Rothengatter (Eds.), *Man, environment, space and time economic interactions in four dimensions* (pp. 377–394). Baden-Baden, Germany: Nomos.
- Meersman, H., & Van de Voorde, E. (2014b). Predicting the future: How to estimate future demand for port throughput and infrastructure use. In H. Meersman, E. Van de Voorde, & T. Vanelslander (Eds.), *Port infrastructure finance* (pp. 33–54). Abingdon, England: Informa Law from Routledge.
- Meersman, H., & Van de Voorde, E. (2019). Freight transport models: Ready to support transport policy of the future? *Transport Policy*. In press.

- Meersman, H., Van de Voorde, E., & Vanelslander, T. (2009). *Future challenges for the port and shipping sector* (1st ed.). London, England: Informa Law from Routledge.
- Meersman, H., Van De Voorde, E., & Vanelslander, T. (2010). Port competition revisited. *Review of Business and Economic Literature*, 55(2), 210–233.
- Memos, C. D. (2004). Port planning. In G. P. Tsinker (Ed.), *Port engineering: Planning, construction, maintenance, and security* (pp. 7–64). Hoboken, NJ: John Wiley and Sons.
- Menezes, F. M., Pracz, M., & Tyers, R. (2007). Strategic interaction amongst Australia's east coast ports. *Australian Economic Review*, 40(3), 267–278.
- Merckx, J.-P. (2018). De Vlaamse havens: Feiten, statistieken en indicatoren voor 2017 [The Flemish ports: Facts, statistics and indicators for 2017] (Annual overview). Brussels, Belgium: Vlaamse Havencommissie. Retrieved from http://www.vlaamsehavencommissie.be/sites/default/files/documenten/ VHC_Jaaroverzicht%202017%5B7%5D.pdf
- Meyer, T., & Reniers, G. (2016). *Engineering risk management*. Berlin, Germany: De Gruyter Graduate.
- Milne, A., & Whalley, A. E. (2000). 'Time to build, option value and investment decisions': a comment. *Journal of Financial economics*, *56*(2), 325–332.
- Munim, Z. H., & Schramm, H.-J. (2018). The impacts of port infrastructure and logistics performance on economic growth: the mediating role of seaborne trade. *Journal of Shipping and Trade*, *3*(1), 1-19.
- Musso, E., Ferrari, C., & Benacchio, M. (2006). Port investment: profitability, economic impact and financing. *Research in Transportation Economics*, *16*, 171–218.
- National Port Council. (2001). *De Nederlandse zeehavens in economisch perspectief* (Tech. Rep.). Den Haag, the Netherlands.
- Ng, A. K. (2006). Assessing the attractiveness of ports in the North European container transhipment market: an agenda for future research in port competition. *Maritime Economics & Logistics*, *8*(3), 234–250.
- Niu, B., & Zhang, J. (2013). Price, capacity and concession period decisions of paretoefficient bot contracts with demand uncertainty. *Transportation Research Part E: Logistics and Transportation Review*, 53, 1–14.
- Noritake, M., & Kimura, S. (1983). Optimum number and capacity of seaport berths. *Journal of Waterway, Port, Coastal, and Ocean Engineering*, 109(3), 323–339.
- Notteboom, T., Verhoeven, P., & Fontanet, M. (2012). Current practices in European ports on the awarding of seaport terminals to private operators: towards an industry good practice guide. *Maritime Policy & Management*, 39(1), 107–123.

- Notteboom, T., & Yap, W. Y. (2012). Port competition and competitiveness. In W. K. Talley (Ed.), *The Blackwell Companion to Maritime Economics* (pp. 549–570). Malden, MA: Wiley-Blackwell.
- Novaes, A. G., Scholz-Reiter, B., Silva, V. M. D., & Rosa, H. (2012). Long-term planning of a container terminal under demand uncertainty and economies of scale. *Pesquisa Operacional*, *32*(1), 55–86.
- OECD. (2014). *The competitiveness of global port-cities*. Paris, France: OECD Publishing. Retrieved from https://dx.doi.org/10.1787/9789264205277-en
- OECD. (2019). Infrastructure investment (indicator). Retrieved April 7, 2019, from https://doi.org/10.1787/b06ce3ad-en
- Onghena, E. (2013). From cost structure to strategy: The impact of the cost structure on the strategic behaviour of integrators (Doctoral dissertation, University of Antwerp, Antwerp, Belgium). Retrieved from https://repository.uantwerpen.be/desktop/irua
- Oxford Economics. (2015). The economic impact of the UK maritime services sector: Ports
 (Report for Maritime UK). Retrieved from https://www.britishports.org.uk/
 system/files/documents/ports_the_economic_impact_of_the_uk_maritime
 _services_sector.pdf
- Paflioti, P., Roumpis, E., Vitsounis, T., Tsamourgelis, I., & Bell, M. (2015, August). *Characterizing the cycle in seaports: An Australian case study.* Paper presented at IAME 2015. Kuala Lumpur, Malaysia.
- Pallis, A. A., Notteboom, T. E., & De Langen, P. W. (2008). Concession agreements and market entry in the container terminal industry. *Maritime Economics & Logistics*, 10(3), 209–228.
- Panayides, P. M. (2006). Maritime logistics and global supply chains: towards a research agenda. *Maritime Economics & Logistics*, *8*(1), 3–18.
- Pawlina, G., & Kort, P. M. (2006). Real options in an asymmetric duopoly: Who benefits from your competitive disadvantage? *Journal of Economics & Management Strategy*, 15(1), 1– 35.
- Pindyck, R. S. (1988). Irreversible investment, capacity choice, and the value of the firm. *The American Economic Review*, 78(5), 969–985.
- Port Consultants Rotterdam. (2018). *Estimation port construction costs expansion MV2*. Retrieved from http://portconsultantsrotterdam.nl/port-construction -costs-expansion-mv2/
- Port of Antwerp. (2016). *Extra capacity: why?* Retrieved April 10, 2017, from http://www.portofantwerp.com/nl/waarom

- Port of Antwerp. (2017a). Tariff regulations for sea-going vessels. Retrieved from http://www.portofantwerp.com/sites/portofantwerp/files/ def_versie_todz_2017_english.pdf
- Port of Antwerp. (2017b). Yearbook of statistics 2016. Retrieved from http:// www.portofantwerp.com/sites/portofantwerp/files/Statistisch% 20Jaarboek%202016.pdf
- Port of Rotterdam. (2018). *News*. Retrieved from https://www.maasvlakte2.com/nl/ news/index/year/2008
- Port of Vancouver. (2015). Procurement Approach | Roberts Bank Terminal 2 Project. Retrieved March 22, 2018, from http://www.robertsbankterminal2.com/business -opportunities/procurement-approach/
- Porto de Luanda. (2018). *Port website*. Retrieved June 15, 2018, from http://www .portoluanda.co.ao/
- Preston, J. (2001). Integrating transport with socio-economic activity–a research agenda for the new millennium. *Journal of Transport Geography*, 9(1), 13–24.
- Provost, F., & Fawcett, T. (2013). *Data science for business: What you need to know about data mining and data-analytic thinking.* Sebastopol, CA: O'Reilly Media, Inc.
- Quinet, E., & Vickerman, R. (2004). *Principles of transport economics*. Cheltenham, England: Edward Elgar Publishing.
- Randrianarisoa, L. M., & Zhang, A. (2019). Adaptation to climate change effects and competition between ports: Invest now or later? *Transportation Research Part B: Methodological*, *123*, 279–322.
- Rashed, Y. (2016). Container throughput modelling and forecasting: an empirical dynamic econometric time series approach (Doctoral dissertation, University of Antwerp, Antwerp, Belgium). Retrieved from https://repository.uantwerpen.be/desktop/irua
- Rau, P., & Spinler, S. (2016). Investment into container shipping capacity: A real options approach in oligopolistic competition. *Transportation Research Part E: Logistics and Transportation Review*, 93, 130–147.
- Rau, P., & Spinler, S. (2017). Alliance formation in a cooperative container shipping game: Performance of a real options investment approach. *Transportation Research Part E: Logistics and Transportation Review*, 101, 155–175.
- RebelGroup Advisory Belgium. (2015). Ontwikkelingszone Saeftinghe: een (over)capaciteitsvraagstuk? [development zone Saeftinghe: a question of (over)capacity?] (Research note). Retrieved from https://extracontainercapaciteitantwerpen .login.kanooh.be/sites/default/files/atoms/files/Oplegnota%200ZS% 2022042015_finaal_0.pdf

- Reker, R., Connell, D., & Ross, D. (1990). The development of a production function for a container terminal in the port of Melbourne. In *Proceedings of the Australiasian Transport Research Forum* (Vol. 15, pp. 209–218). Canberra, Australia.
- Rodrigue, J.-P., & Notteboom, T. (2009). The terminalization of supply chains: reassessing the role of terminals in port/hinterland logistical relationships. *Maritime Policy & Management*, *36*(2), 165–183.
- Ruiz-Aliseda, F., & Wu, J. (2012). Irreversible investment in stochastically cyclical markets. *Journal of Economics & Management Strategy*, 21(3), 801–847.
- Saeed, N., & Larsen, O. I. (2010). Container terminal concessions: A game theory application to the case of the ports of Pakistan. *Maritime Economics & Logistics*, *12*(3), 237–262.
- Schneider, G., Finke, D., & Bailer, S. (2010). Bargaining power in the European Union: An evaluation of competing game-theoretic models. *Political Studies*, *58*(1), 85–103.
- Schöne, M. (2014). *Real options valuation: The importance of stochastic process choice in commodity price modelling.* Wiesbaden, Germany: Springer Gabler.
- Ship Technology. (2018). Liverpool 2 container terminal. Retrieved November 29, 2018, from https://www.ship-technology.com/projects/liverpool-2 -container-terminal/
- Singh, N., & Vives, X. (1984). Price and quantity competition in a differentiated duopoly. *The RAND Journal of Economics*, *15*(4), 546–554.
- Slack, B., & Frémont, A. (2005). Transformation of port terminal operations: from the local to the global. *Transport Reviews*, *25*(1), 117–130.
- Smit, H. (2003). Infrastructure investment as a real options game: The case of European airport expansion. *Financial Management*, *32*(4), 27–57.
- Sødal, S. (2006). Entry and exit decisions based on a discount factor approach. *Journal of Economic Dynamics and Control, 30*(11), 1963–1986.
- Sødal, S., Koekebakker, S., & Aadland, R. (2008). Market switching in shipping A real option model applied to the valuation of combination carriers. *Review of Financial Economics*, *17*(3), 183–203.
- Stopford, M. (2009). Maritime economics (3rd ed.). Abingdon, England: Routledge.
- Strandenes, S. P. (2014). Efficiency-enhancing port pricing models. In T. Vanelslander & C. Sys (Eds.), *Port business. Market challenges and management actions* (pp. 43–60). Antwerp, Belgium: University Press Antwerp.
- Strandenes, S. P., & Marlow, P. B. (2000). Port pricing and competitiveness in short sea shipping. International Journal of Transport Economics/Rivista internazionale di economia dei trasporti, 27(3), 315–334.

- Suykens, F., & Van de Voorde, E. (1998). A quarter of a century of port management in Europe: objectives and tools. *Maritime Policy & Management, 25*(3), 251–261.
- Sys, C., Blauwens, G., Omey, E., Van De Voorde, E., & Witlox, F. (2008). In search of the link between ship size and operations. *Transportation Planning and Technology*, *31*(4), 435–463.
- Szymanski, S. (1991). The optimal timing of infrastructure investment. *Journal of Transport Economics and Policy*, 25(3), 247–258.
- Talley, W. K. (2009). Port economics. Abingdon, England: Routledge.
- Talley, W. K., Ng, M. W., & Marsillac, E. (2014). Port service chains and port performance evaluation. *Transportation Research Part E: Logistics and Transportation Review*, 69, 236–247.
- Taneja, P. (2013). The flexible port (Doctoral dissertation, TU Delft, Delft, The Netherlands). Retrieved from http://dx.doi.org/10.4233/uuid:a9f0c128-d4c3-41a2 -8790-13aec89dca63
- Thanopoulou, H., & Strandenes, S. P. (2017). A theoretical framework for analysing long-term uncertainty in shipping. *Case studies on transport policy*, *5*(2), 325–331.
- Tibben-Lembke, R. S., & Rogers, D. S. (2006). Real options: applications to logistics and transportation. *International Journal of Physical Distribution & Logistics Management*, 36(4), 252–270.
- Timperman, B. (2017, September, 17). Walter Van Mechelen: "Uitstel Saeftinghe nefast voor Antwerpen [Delaying Saeftinghe bad for Antwerp]". Retrieved from http://www.flows.be/nl/shipping/walter-van-mechelen-uitstel -saeftinghe-nefast-voor-antwerpen
- Tongzon, J. L. (1993). The Port of Melbourne Authority's pricing policy: its efficiency and distribution implications. *Maritime Policy & Management*, *20*(3), 197–205.
- Tovar, B., Jara-Díaz, S., & Trujillo, L. (2007). Econometric estimation of scale and scope economies within the Port Sector: a review. *Maritime Policy & Management*, *34*(3), 203–223.
- Trigeorgis, L. (1996). *Real options: Managerial flexibility and strategy in resource allocation*. Cambridge, MA: The MIT Press.
- Trigeorgis, L. (2005). Making use of real options simple: An overview and applications in flexible/modular decision making. *The Engineering Economist*, *50*(1), 25–53.
- Trujillo, L., & Nombela, G. (2000). *Multiservice infrastructure: Privatizing port services* (Note No. 21482). Washington, DC: World Bank Group.

- Tsamboulas, D. A., & Ballis, A. (2014). How to finance future port infrastructure investments? Potential strategies for the different actors. In H. Meersman, E. Van de Voorde, & T. Vanelslander (Eds.), *Port infrastructure finance* (pp. 211–227). Abingdon, England: Informa Law from Routledge.
- Tsamboulas, D. A., & Kapros, S. (2003). Freight village evaluation under uncertainty with public and private financing. *Transport Policy*, *10*(2), 141–156.
- Turpin, F.-M. (2013). Ppp in ports, landlord port model (Powerpoint Presentation in Tbilisi - LOGMOS: EU funded project (TRACECA)). EGIS International. Retrieved from http://www.traceca-org.org/uploads/media/04_Module_C_PPP _Francois_Marc_Turpin_new.pdf
- UNCTAD. (2015). *Review of maritime transport 2014*. Geneva, Switzerland: United Nations. Retrieved from http://unctad.org/en/PublicationsLibrary/rmt2014_en.pdf
- Van Der Horst, M. R., & De Langen, P. W. (2008). Coordination in hinterland transport chains: a major challenge for the seaport community. *Maritime Economics & Logistics, 10*(1-2), 108–129.
- van der Lugt, L., Witte, J.-J., Becker, E., & Streng, M. (2018). De economische betekenis van Nederlandse zeehavens 2002-2017 (Havenmonitor). Rotterdam, The Netherlands: Ministerie van Infrastructuur en Milieu. Retrieved from https://www.portofrotterdam .com/sites/default/files/havenmonitor-2017.pdf
- Van de Voorde, E., & Vanelslander, T. (2009). Market power and vertical and horizontal integration in the maritime shipping and port industry (OECD/ITF Joint Transport Research Centre Discussion Papers). Paris, France: Organisation for Economic Co-operation and Development. Retrieved from http://www.oecd-ilibrary.org/ content/workingpaper/227458312782
- Van de Voorde, E., & Vanelslander, T. (2014). Trends in the maritime logistics chain: vertical port cooperation strategies and relationships. In *Port business. market challenges and management actions* (pp. 119–140). Antwerp, Belgium: ASP Publishers.
- Vanelslander, T. (2005). The economics behind co-operation and competition in sea-port container handling (Doctoral dissertation, University of Antwerp, Antwerp, Belgium). Retrieved from https://repository.uantwerpen.be/desktop/irua
- Vanelslander, T. (2014). Port infrastructure: What are we talking about. In H. Meersman, E. Van de Voorde, & T. Vanelslander (Eds.), *Port infrastructure finance* (pp. 5–32). Abingdon, England: Informa Law from Routledge.
- van Gastel, G. (2016). Flemish maritime ports, Liège port complex and the port of Brussels
 report 2014 (Working Paper No. 299). Brussels, Belgium: National Bank of Belgium. Retrieved from https://www.nbb.be/doc/ts/publications/wp/wp299en.pdf

- Van Putten, A. B., & MacMillan, I. C. (2004). Making real options really work. *Harvard business review*, 82(12), 134–142.
- Van Reeven, P. (2010). The effect of competition on economic rents in seaports. *Journal of Transport Economics and Policy*, 44(1), 79–92.
- Veldman, S. J., & Bückmann, E. H. (2003). A model on container port competition: An application for the West European container hub-ports. *Maritime Economics & Logistics*, 5(1), 3–22.
- Verbruggen, A. (2008). *Economische benadering van milieu en milieubehoud [Economic approach of the environment and environmental conservation]*. Antwerp, Belgium: Garant.
- Vergauwen, B. (2010, October). *Ex-post analysis of an infrastructure project in the port of Antwerp*. Paper presented at the European Transport Conference (ETC) 2010. Glasgow, Scotland.
- Verhoeven, P. (2015). Economic assessment of management reform in European seaports (Doctoral dissertation, University of Antwerp, Antwerp, Belgium). Retrieved from https://repository.uantwerpen.be/desktop/irua
- Vilko, J. P. P., & Hallikas, J. M. (2012). Risk assessment in multimodal supply chains. *Inter*national Journal of Production Economics, 140(2), 586–595.
- Vives, X. (1999). Oligopoly pricing: Old ideas and new tools. Cambridge, MA: The MIT Press.
- Vlaamse Havencommissie. (2016). Maritime throughput in European ports [Dataset]. Retrieved November 30, 2016, from http://www.salv.be/vhc/pagina/ladingen-en -lossingen-antwerpen-overzicht-verschijningsvorm-1980
- Wan, Y., & Zhang, A. (2013). Urban road congestion and seaport competition. *Journal of Transport Economics and Policy*, 47(1), 55–70.
- Wan, Y., Zhang, A., & Yuen, A. C. (2013). Urban road congestion, capacity expansion and port competition: empirical analysis of US container ports. *Maritime Policy & Management*, 40(5), 417–438.
- Wang, G. W. Y., & Pallis, A. A. (2014). Incentive approaches to overcome moral hazard in port concession agreements. *Transportation Research Part E: Logistics and Transportation Review*, 67, 162–174.
- Wang, K., & Zhang, A. (2018). Climate change, natural disasters and adaptation investments: Inter-and intra-port competition and cooperation. *Transportation Research Part B: Methodological*, 117, 158–189.
- Wang, T., & De Neufville, R. (2005, June). *Real options "in" projects*. Paper presented at the 9th Annual Real Options Conference. Paris, France. Retrieved from http:// www.realoptions.org/papers2005/Tao_Wang_RO_in_projects.pdf

- Ward, R. (2015, July 2). Container terminals in jungle city of Manaus face growing pains. Retrieved from https://www.joc.com/port-news/south-american-ports/ container-terminals-jungle-city-manaus-face-growing-pains_20150702 .html
- Weeds, H. (2002). Strategic delay in a real options model of R&D competition. *The Review* of *Economic Studies*, 69(3), 729–747.
- Wiegmans, B., & Behdani, B. (2018). A review and analysis of the investment in, and cost structure of, intermodal rail terminals. *Transport Reviews*, *38*(1), 33–51.
- Wilmott, P., Dewynne, J., & Howison, S. (1993). *Option pricing: mathematical models and computation*. Oxford, England: Oxford Financial Press.
- Xiao, Y.-b., Fu, X., Ng, A. K., & Zhang, A. (2015). Port investments on coastal and marine disasters prevention: Economic modeling and implications. *Transportation Research Part B: Methodological*, 78, 202–221.
- Xiao, Y.-b., Fu, X., Oum, T. H., & Yan, J. (2017). Modeling airport capacity choice with real options. *Transportation Research Part B: Methodological*, *100*, 93–114.
- Xiao, Y.-b., Fu, X., & Zhang, A. (2013). Demand uncertainty and airport capacity choice. *Transportation Research Part B: Methodological*, *57*, 91–104.
- Xiao, Y.-b., Ng, A. K., Yang, H., & Fu, X. (2012). An analysis of the dynamics of ownership, capacity investments and pricing structure of ports. *Transport Reviews*, *32*(5), 629–652.
- Yang, H., & Zhang, A. (2012). Impact of regulation on transport infrastructure capacity and service quality. *Journal of Transport Economics and Policy*, 46(3), 415–430.
- Yap, W. Y., Lam, J. S., & Notteboom, T. (2006). Developments in container port competition in east asia. *Transport Reviews*, *26*(2), 167–188.
- Yuen, A., Basso, L. J., & Zhang, A. (2008). Effects of gateway congestion pricing on optimal road pricing and hinterland. *Journal of Transport Economics and Policy*, *42*(3), 495–526.
- Zachcial, M., Kramer, H., Lemper, B., & Duhme, W. (2006). *Public financing and charging practices of seaports in the EU* (EU Report). Bremen, Germany: Institute of Shipping Economics and Logistics (ISL).
- Zhang, A. (2007, May). Congestion pricing and capacity investment in a gateway-hinterland intermodal system. Paper presented at International Conference on Gateways and Corridors. Vancouver, BC. Retrieved from http://www.gateway-corridor.com/ roundconfpapers/documents/Zhang_Anming_Vancouver.pdf
- Zhang, A. (2008). The impact of hinterland access conditions on rivalry between ports (OECD/ITF Joint Transport Research Centre Discussion Papers). Paris, France: Organisation for Economic Co-operation and Development. Retrieved from http://www.oecd -ilibrary.org/content/workingpaper/235411150338

- Zhang, A., & Czerny, A. I. (2012). Airports and airlines economics and policy: An interpretive review of recent research. *Economics of Transportation*, *1*(1-2), 15–34.
- Zhang, A., & Zhang, Y. (2003). Airport charges and capacity expansion: effects of concessions and privatization. *Journal of Urban Economics*, 53(1), 54–75.
- Zhang, A., & Zhang, Y. (2006). Airport capacity and congestion when carriers have market power. *Journal of Urban Economics*, 60(2), 229–247.
- Zheng, S., & Negenborn, R. R. (2017). Terminal investment timing decisions in a competitive setting with uncertainty using a real option approach. *Maritime Policy & Management*, 44(3), 392-411.
- Zheng, T. (2015, October, 16). Bintulu Port awards \$5.3m maintenance dredging contract to Integrated Marine Works. Retrieved January 3, 2018, from https://fairplay.ihs.com/article/19580/bintulu-port-awards-5-3m -maintenance-dredging-contract-integrated-marine-works
- Zhuang, W., Luo, M., & Fu, X. (2014). A game theory analysis of port specialization implications to the Chinese port industry. *Maritime Policy & Management*, 41(3), 268– 287.
- Zuidgeest, R. (2009). Binnenvaart service centrum op Maasvlakte 2: Een haalbaarheidsstudie [inland navigation service center at Maasvlakte 2: a feasibility study] (Master's thesis, TU Delft, Delft, The Netherlands). Retrieved from http://resolver.tudelft.nl/ uuid:4e5c6ab6-645c-42fb-b523-331a6e726a91

Appendix A

Solution to differential equation (3.19), determining a new port's project value

In this Appendix, the solution to differential equation (3.19) is derived. To this end, the following equation is solved:

$$\frac{\sigma^2}{2} X^2 \frac{\partial^2 V}{\partial X^2}(X, K) + \mu X \frac{\partial V}{\partial X}(X, K) - r V(X, K) = -\overline{\pi}(X, K)$$
(A.1)

where $\overline{\pi}(X, K)$ is either a constant or rational function:

$$\pi(X, K, q^{opt}(X, K)) = \overline{\pi}(X, K) = \begin{cases} \overline{\pi}_1(X, K) = -c_h K, & X \in R_1, \\ \overline{\pi}_2(X, K) = \frac{(X-c)^2 K^2}{4(XA+BK^2)} - c_h K, & X \in R_2, \\ \overline{\pi}_3(X, K) = (K-A)X - (BK+c+c_h)K, & X \in R_3. \end{cases}$$
(A.2)

First, the homogeneous equation is considered:

$$\frac{\sigma^2}{2} X^2 \frac{\partial^2 V}{\partial X^2}(X, K) + \mu X \frac{\partial V}{\partial X}(X, K) - r V(X, K) = 0.$$
(A.3)

Note that the second-order derivative has a coefficient that is quadratic in X, and likewise for the other terms. Hence, if V is given by a power function, all terms in the differential equation will prove to be the same power functions. Thus, the two independent solutions will be given by power functions. Plugging $V = X^{\beta}$ into the equation, it can be simplified to the quadratic equation

$$\frac{\sigma^2}{2}\beta(\beta-1) + \mu\beta - r = 0, \tag{A.4}$$

which has the two distinct real roots β_1 and β_2 given in Equation (3.22). Note that the condition $r > \mu$ means that $\beta_1 > 1$ and $\beta_2 < 0$. Hence, the solution to the homogeneous equation

in the *j*-th region (j = 1, 2) is given by

$$V_{h,j}(X,K) = G_{1,j}(K)X^{\beta_1} + G_{2,j}(K)X^{\beta_2}.$$
(A.5)

In region R_1 , the right-hand side of differential equation (A.1) is a constant. A constant particular solution is then:

$$\overline{V}_1(X,K) = -\frac{c_h K}{r}.$$
(A.6)

In region R_2 , the right-hand side is more complicated:

$$\overline{\pi}_{2}(X,K) = \frac{(X-c)^{2}K^{2}}{4(AX+BK^{2})} - c_{h}K$$

$$= \frac{K^{2}}{4A}X - \left[\frac{K^{2}}{4A}\left(2c + \frac{BK^{2}}{A}\right) + c_{h}K\right] + \frac{K^{2}(c + \frac{BK^{2}}{A})^{2}}{4A}\frac{1}{X + \frac{BK^{2}}{A}}.$$
 (A.7)

This right-hand side consists of a polynomial part (of degree one) and a rational part. Finding a particular solution for the polynomial part is straightforward: if V = aX + b is entered into the differential equation, the following equation is obtained:

$$\overline{V}_{2,\text{pol}}(X,K) = \frac{K^2}{4A(r-\mu)} X - \frac{1}{r} \left[\frac{K^2}{4A} \left(2c + \frac{BK^2}{A} \right) + c_h K \right].$$
(A.8)

For the final part of the solution, it is easier to consider the monic form of the differential equation, i.e., dividing (A.7) by $\sigma^2 X^2/2$. In order to find a particular solution for the remaining rational (in *X*) part,

$$\overline{\pi}_{2,\text{rat,mon}}(X,K) = \frac{K^2 \left(c + \frac{BK^2}{A}\right)^2}{2A\sigma^2} \frac{1}{X^2 \left(X + \frac{BK^2}{A}\right)},$$
(A.9)

the method known as variation of parameters or constants is used. That is, a solution in the same form as the solution to the homogeneous equation is proposed, but now with coefficients that also depend on X:

$$\overline{V}_{2,\text{rat}}(X,K) = C_1(X,K)X^{\beta_1} + C_2(X,K)X^{\beta_2}.$$
(A.10)

As a result of this method, the coefficients are given by

$$C_1(X,K) = \int \frac{1}{W} X^{\beta_2} \overline{\pi}_{2,\text{rat}}(X,K) \, \mathrm{d}X, \qquad (A.11)$$

$$C_2(X,K) = -\int \frac{1}{W} X^{\beta_1} \overline{\pi}_{2,\text{rat}}(X,K) \, \mathrm{d}X,$$
 (A.12)

where *W* is the Wronskian, i.e., the determinant of the two independent solutions to the homogeneous equation and their derivatives:

$$W = \begin{vmatrix} X^{\beta_1} & X^{\beta_2} \\ \beta_1 X^{\beta_1 - 1} & \beta_2 X^{\beta_2 - 1} \end{vmatrix} = (\beta_2 - \beta_1) X^{\beta_1 + \beta_2 - 1}.$$
 (A.13)

Besides for integer values of β_1 and β_2 , the integrals in equations (A.11)-(A.12) cannot be evaluated using elementary functions alone. Instead, their solution can be written in terms of hypergeometric functions (Abramowitz & Stegun, 1972). The solution

$$\overline{V}_{2,\text{rat}}(X,K) = -\frac{\left(c + \frac{B}{A}K^{2}\right)^{2}}{2\sigma^{2}B(\beta_{2} - \beta_{1})}$$

$$\times \left[\frac{1}{\beta_{1}}{}_{2}F_{1}\left(1, -\beta_{1}; 1 - \beta_{1}; -\frac{AX}{BK^{2}}\right) - \frac{1}{\beta_{2}}{}_{2}F_{1}\left(1, -\beta_{2}; 1 - \beta_{2}; -\frac{AX}{BK^{2}}\right)\right]$$
(A.14)

is obtained and is valid for $X < BK^2/A$, or its analytic continuation for $X \ge BK^2/A$. The sum of the particular solutions for the polynomial and rational parts, found in (A.8) and (A.14) respectively, produces a particular solution for differential equation (A.1) in the region R_2 .

If the solution of the homogeneous equation is added to this particular solution, a particular solution is still obtained. Consequently, a linear combination of X^{β_1} and X^{β_2} can be added to the particular solution. The reason for this stems from the analysis of the asymptotic behaviour of the solution. With a large X, the first hypergeometric function in (A.14) will increase as a constant *times* X^{β_1} . However, the next term in the asymptotic expansion at infinity will merely be linear, even if $\beta_1 > 2$. This can be shown by writing down the series expansion of the hypergeometric functions at infinity. Thus, this term in X^{β_1} , calculated as

$$\frac{\Gamma(\beta_1)\Gamma(1-\beta_1)}{(BK^2)^{\beta_1}}A^{\beta_1}X^{\beta_1}$$

is subtracted, in order to obtain the particular solution omitting speculative bubbles resulting from the term in X^{β_1} (see Dixit & Pindyck (1994), Chapter 6). For a behaviour near zero omitting speculative bubbles, no further steps are needed. Since the hypergeometric series is a power series around zero, it converges in the neighbourhood of zero and no divergent terms (i.e., terms in X^{β_2}) are present. Hence, no correction in X^{β_2} is required.

Finally, in region R_3 , the right-hand side is polynomial in *X*:

$$\overline{\pi}_{3}(X,K) = (K-A)X - (BK + c + c_{h})K.$$
(A.15)

Again, by plugging V = aX + b into the differential equation,

$$\overline{V}_{3}(X,K) = \frac{K-A}{r-\mu} X - \frac{BK^{2} + cK + c_{h}K}{r}$$
(A.16)

is obtained.

All these considerations lead to the particular solution given in Equation (3.23).